# Convergence analysis in stochastic nonlinear systems: Asymptotic behaviors in extended time scales and large populations

by

Jiamin Jian

A Dissertation

Submitted to the Faculty

of the

#### WORCESTER POLYTECHNIC INSTITUTE

in partial fulfillment of the requirements for the

Degree of Doctor of Philosophy

in

Mathematical Sciences

by

APPROVED:

April 2024

Dr. Qingshuo Song, Advisor Department of Mathematical Sciences Worcester Polytechnic Institute

Dr. Gu Wang Department of Mathematical Sciences Worcester Polytechnic Institute Dr. Stephan Sturm Department of Mathematical Sciences Worcester Polytechnic Institute

Dr. Marcel Y. Blais Department of Mathematical Sciences Worcester Polytechnic Institute Dr. George Yin Department of Mathematics University of Connecticut

## Acknowledgements

Throughout my life at WPI, numerous individuals provided invaluable support, both academically and personally, each leaving a lasting impact on my research. I would like to extend my heartfelt gratitude to all who contributed to the completion of this dissertation and accompanied me on my Ph.D. journey.

First and foremost, I want to express my sincere and deepest gratitude to my advisor, Professor Qingshuo Song, for his unwavering supervision, guidance, support, and patience throughout my Ph.D. life. When I first arrived at WPI, it was Professor Song who introduced me to the field of stochastic control and mean field games. He is an outstanding mentor who provided me with significant assistance and guidance throughout my doctoral research and life at WPI. His support and encouragement empowered me to cultivate newfound confidence and tackle challenges with tremendous energy. Beyond academic matters, Professor Song extended substantial aid and mentorship in our daily interactions. I thoroughly enjoyed our time together, whether it was sharing meals, walking around our campus, or playing table tennis in Harrington Auditorium. Furthermore, Professor Song shared valuable experiences and provided insightful guidance for my future endeavors. His exemplification of a rigorous academic attitude and high standards has left a lasting impression on me. I am inspired by his profound passion for research, which significantly influences my aspirations for future research endeavors.

I extend my sincere gratitude to my dissertation committee members, Professor George Yin, Professor Gu Wang, Professor Stephan Strum, and Professor Marcel Y. Blais. Their generous allocation of time, unwavering support, guidance, and goodwill have been instrumental throughout the preparation and review of this document. It is a tremendous honor to have them in my thesis committee, and I am deeply appreciative of their invaluable comments and suggestions that have enhanced my thesis. Special appreciation goes to Professor George Yin for organizing enlightening symposiums and workshops. These academic gatherings provided me with substantial learning opportunities and enjoyable experiences, engaging with esteemed researchers. I am also thankful to Professor Gu Wang and Professor Stephan Strum for facilitating independent studies tailored to my needs. The knowledge gained from these courses about Stochastic Analysis and Stochastic Control forms the foundation of my future research. Furthermore, I express my gratitude to Professor Marcel Y. Blais for his wealth of experience in mathematical finance and valuable suggestions regarding the presentation and revision of my thesis.

I would like to extend my heartfelt gratitude to the professors and individuals who played pivotal roles in guiding me during the research projects within this dissertation. Their exemplary contributions have set a commendable standard for researchers. Collaborating and learning from Professor Peter E. Caines, Professor Daniel Ho and Professor Minyi Huang in my initial academic research project was an enriching experience. I am thankful for the opportunity to work alongside Professor Jiongmin Yong and Professor Sixian Jin, whose works and suggestions were instrumental in the completion of the turnpike property project in Chapter 2 of this dissertation. Additionally, I express sincere gratitude to Dr. Jiaxuan Ye and Ph.D. candidate Peiyao Lai for their significant contributions to the project in Chapter 3.

My appreciation extends to all faculty members and staff from the Department of Mathematical Sciences at WPI, with special mentions to Dr. Sarah Olson, Dr. Lui Roger, Dr. Min Wu, Dr. Zhongqiang Zhang, Dr. Christopher Larsen, Dr. Guanying Peng, Dr. Darko Volkov, Dr. Andrea Arnold, Mike Malone, Rhonda Podell, and Louisa Kulp. Their support, whether through imparting knowledge or providing assistance in both my life and research at WPI, has been invaluable. I am grateful for the tremendous benefit derived from their teachings and support.

A profound thank you to my friends, whose support and care have been instrumental in helping me navigate challenges, maintain a positive mindset, and stay focused on my graduate studies. Special appreciation goes to my roommates, Dr. Wenxu Han and Dr. Yanzhao Wang, for their companionship and support. We enjoyed delightful dinners and shared countless moments watching TV plays together. I express my sincere appreciation to Dr. Yuchen Dong and Dr. Feng Cheng for their assistance and support when I first arrived at WPI. Gratitude also extends to my friends, including Dr. Ruihan Zhang, Dr. Ziyang Zhang, Dr. Jiuling Yu, Dr. Hongjiang Qian, Yuying Zhao, Dr. Xiaohui Chen, Shuaichuan Feng, Dr. Zhongyuan Cao, Yipeng Jiang, Yichuan Li, Jinhong Yu and Dr. Zhenyuan Lu. A special mention goes to my badminton pals and the entire WPI badminton club. I appreciate all of you for making the graduate school experience far more enjoyable than I could have imagined, especially Yiqing Zhang, Zhuoyun Zhong, Mingzhi Hu, Xingtong Guo, Tianxing Weng, Howard Zheng, and Fangzhou Lin.

Above all, none of this would have been possible without the unconditional love and support from my family. I am profoundly indebted to my parents, grandparents and aunt, who have been a constant source of love, care, support, and strength throughout these years.

## Abstract

This thesis focuses on the convergence analysis of stochastic nonlinear systems. Specifically, it delves into the study of their asymptotic behavior, an area of paramount importance for comprehending the long-term dynamics of these systems and the systems with large populations. We concentrate on two key aspects: firstly, the examination of the turnpike property within the context of stochastic control problems, and secondly, the investigation of the convergence of N-player games towards their corresponding mean field games (MFG).

Firstly, we investigate the asymptotic behavior of the systems with long-term dynamics and the convergence is with respect to the time horizon. In the first project, we examine the limiting behavior of a specific class of linear quadratic stochastic optimal control problems and their corresponding value functions as the time horizon approaches infinity. We establish the consistency between the cell problem in weak KAM theory and the static optimization problem from the perspective of the turnpike property. Moreover, we provide the connection between the cell problem and the ergodic cost problem, and then the classical turnpike property and the turnpike property in terms of the cost function are identified.

Different from the first project, next we examine the system complexity. More precisely, we consider the convergence behavior of systems with large populations. MFG has become widely accepted as an approximation for the N-player games, especially when the number of players is large enough. A fundamental question that arises in this context concerns the convergence rate of this approximation.

In the second project, we study the convergence rate of the N-player Linear-Quadratic-Gaussian (LQG) games with a Markov chain as the common noise towards its asymptotic MFG. By postulating a Markovian structure via two auxiliary processes for the first and second moments of the MFG equilibrium and applying the fixed point condition in MFG, we first provide the characterization of the equilibrium measure in MFG with a finite-dimensional Riccati system of ODEs. Additionally, with an explicit coupling of the optimal trajectory of the N-player game driven by N dimensional Brownian motion and MFG counterpart driven by one-dimensional Brownian motion, we obtain the convergence rate  $O(N^{-1/2})$  with respect to 2-Wasserstein distance.

The number of states of the common noise considered in the above project is finite, thus it is natural to consider the case when the number of states of the common noise is infinity. In the third project, we focus on exploring the convergence properties of a generic player's trajectory and empirical measures in an N-player LQG Nash game, where Brownian motion serves as the common noise. The study establishes three distinct convergence rates concerning the representative player and empirical measure. To investigate the convergence, our methodology relies on a specific decomposition of the equilibrium path in the N-player game and utilizes the associated MFG framework.

The basic structure of standard MFG theory assumes symmetry in the connections of the agents but not necessarily in their dynamics. However, asymmetric graph connections in large population games are considered in recent studies. In the network limit, a graphon gives the communication weights. In the last project, we consider the solvability of graphon mean field games. A new type of mean field games PDE system associated with the graphon mean field games is proposed. We establish the existence of solutions via the application of Schauder's fixed point theorem and obtain the uniqueness of solution via the Lasry-Lions monotonicity assumption on the running cost.

# Contents

1 Introduction			1
	1.1	Introduction and motivation	1
		1.1.1 Turnpike properties of stochastic control problem	1
		1.1.2 Convergence of mean field games with common noise	2
		1.1.3 Graphon mean field games	3
2	Lon	ng-time behavior of stochastic LQ control problem	<b>5</b>
	2.1	Introduction	5
	2.2	Cell problem and the verification theorem	8
		2.2.1 Existence and nonuniqueness of the cell problem	8
		2.2.2 Verification theorem to probabilistic cell problem	10
		2.2.3 Consistency with the static optimization problem	15
	2.3	Turnpike property	17
		2.3.1 Main results on turnpike properties	18
		2.3.2 Proofs	19
	2.4	Example	28
	2.5	Appendix	31
3	Cor	nvergence rate of LQG mean field games with Markov chain as common noise $\sim$	35
	3.1	Introduction	35
	3.2	Problem setup and main results	37
		3.2.1 Notations	37
		3.2.2 The equilibrium of MFG	38

		3.2.3	Equilibrium of the $N$ -player game $\ldots \ldots \ldots \ldots \ldots \ldots \ldots \ldots \ldots \ldots \ldots$	40
		3.2.4	The main result with quadratic cost structures	41
		3.2.5	Remarks on the main results	44
	3.3	Main	results of MFG	44
		3.3.1	Overview	45
		3.3.2	The generic player's control with a given population measure $\ldots$	46
		3.3.3	Fixed point condition and the proof of Theorem 20	50
	3.4	The N	<i>I</i> -player game and its convergence to MFG	52
		3.4.1	Characterization of the N-player game by Riccati system $\ldots \ldots \ldots \ldots$	53
		3.4.2	Reduced Riccati form for the equilibrium	54
		3.4.3	Convergence	57
	3.5	Nume	rical results	62
		3.5.1	Simulations of Riccati system, the value function and optimal control of the	
			generic palyer	62
		3.5.2	Convergence of the <i>N</i> -player game	63
	3.6	Conclu	usion	64
	3.7	Apper	ndix	65
		3.7.1	Some explicit solutions on LQG-MFG	65
		3.7.2	Dynkin's formula for a regime-switching diffusion with a quadratic function .	67
		3.7.3	Proof of the property of G	69
		3.7.4	Proof of the existence and uniqueness of the ODE system	72
		3.7.5	Multidimensional problem	75
4	Cor	ivergei	nce rate of LQG mean field games with Brownian motion as common	
	nois	se		77
	4.1	Introd	uction	77
	4.2	Proble	em setup and main results	79
		4.2.1	The formulation of equilibrium in mean field games	79
		4.2.2	The formulation of Nash equilibrium in the $N$ -player game $\ldots \ldots \ldots$	81
		4.2.3	Main result	82
	4.3	Proof	of the main result with two propositions	85

		4.3.1	Preliminaries	. 85
		4.3.2	Empirical measures of a sequence with a common noise	. 87
		4.3.3	Generalization of the convergence to triangular arrays	. 89
		4.3.4	Proof of Theorem 44	. 91
	4.4	Propos	ition 45: Derivation of the MFG path	. 94
		4.4.1	Reformulation	. 95
		4.4.2	The generic player's control with a given population measure	. 97
		4.4.3	Fixed point condition and the proof of Proposition 45	. 100
	4.5	The $N$	-Player Game	. 103
		4.5.1	Characterization of the $N$ -player game by Riccati system $\ldots \ldots \ldots$	. 103
		4.5.2	Proof of Proposition 46: Reduced Riccati form for the equilibrium	. 105
	4.6	Further	r remark	. 107
	4.7	Append	dix	. 107
5	On	the gra	aphon mean field game equations: Individual agent affine dynamic	s
	and	mean	field dependent performance functions	111
	<b>and</b> 5.1	l <b>mean</b> : Introdu	field dependent performance functions         action	<b>111</b> . 111
	<b>and</b> 5.1 5.2	Introdu Probler	field dependent performance functions         action	<b>111</b> . 111 . 112
	and 5.1 5.2 5.3	Introdu Introdu Probler Some r	field dependent performance functions         action	<b>111</b> . 111 . 112 . 115
	and 5.1 5.2 5.3	I mean : Introdu Probler Some r 5.3.1	field dependent performance functions         action	<b>111</b> . 111 . 112 . 115 . 115
	and 5.1 5.2 5.3	Introdu Introdu Probler Some r 5.3.1 5.3.2	field dependent performance functions         action	<ol> <li>111</li> <li>111</li> <li>112</li> <li>115</li> <li>115</li> <li>121</li> </ol>
	and 5.1 5.2 5.3 5.4	Introdu Probler Some r 5.3.1 5.3.2 Exister	field dependent performance functions         action	<ol> <li>111</li> <li>111</li> <li>112</li> <li>115</li> <li>115</li> <li>121</li> <li>122</li> </ol>
	and 5.1 5.2 5.3 5.4	Introdu Probler Some r 5.3.1 5.3.2 Exister 5.4.1	field dependent performance functions         action	<ol> <li>111</li> <li>111</li> <li>112</li> <li>115</li> <li>115</li> <li>121</li> <li>122</li> <li>124</li> </ol>
	and 5.1 5.2 5.3 5.4	I mean : Introdu Problen Some r 5.3.1 5.3.2 Exister 5.4.1 5.4.2	field dependent performance functions $action \dots \dots$	<ol> <li>111</li> <li>111</li> <li>112</li> <li>115</li> <li>115</li> <li>121</li> <li>122</li> <li>124</li> <li>125</li> </ol>
	and 5.1 5.2 5.3 5.4	Introdu Problem Some r 5.3.1 5.3.2 Exister 5.4.1 5.4.2 5.4.3	field dependent performance functions $action \dots \dots$	<ol> <li>111</li> <li>111</li> <li>112</li> <li>115</li> <li>115</li> <li>121</li> <li>122</li> <li>124</li> <li>125</li> <li>127</li> </ol>
	and 5.1 5.2 5.3 5.4	Introdu Problem Some r 5.3.1 5.3.2 Exister 5.4.1 5.4.2 5.4.3 5.4.4	field dependent performance functions $action \dots \dots$	<ol> <li>111</li> <li>111</li> <li>112</li> <li>115</li> <li>115</li> <li>121</li> <li>122</li> <li>124</li> <li>125</li> <li>127</li> <li>128</li> </ol>
	and 5.1 5.2 5.3 5.4	I mean : Introdu Problen Some r 5.3.1 5.3.2 Exister 5.4.1 5.4.2 5.4.3 5.4.4 5.4.5	field dependent performance functions action	<ol> <li>111</li> <li>111</li> <li>112</li> <li>115</li> <li>115</li> <li>121</li> <li>122</li> <li>124</li> <li>125</li> <li>127</li> <li>128</li> <li>129</li> </ol>
	and 5.1 5.2 5.3 5.4	Introdu Problem Some r 5.3.1 5.3.2 Exister 5.4.1 5.4.2 5.4.3 5.4.4 5.4.5 Unique	field dependent performance functions action	<ol> <li>111</li> <li>111</li> <li>112</li> <li>115</li> <li>115</li> <li>121</li> <li>122</li> <li>124</li> <li>125</li> <li>127</li> <li>128</li> <li>129</li> <li>130</li> </ol>
	and 5.1 5.2 5.3 5.4 5.4 5.5 5.5 5.6	I mean : Introdu Problem Some r 5.3.1 5.3.2 Exister 5.4.1 5.4.2 5.4.3 5.4.4 5.4.5 Unique Conclu	field dependent performance functions action	<ol> <li>111</li> <li>111</li> <li>112</li> <li>115</li> <li>115</li> <li>121</li> <li>122</li> <li>124</li> <li>125</li> <li>127</li> <li>128</li> <li>129</li> <li>130</li> <li>131</li> </ol>

6	Con	clusion and future work	137	
	6.1	Conclusion	137	
	6.2	Future work	139	
Re	References			

# List of Notations

The following notations are used in the dissertation.

Common: for all  $i = 1, 2, \cdots, N$ ,

$W, \tilde{W}, W_i^{(N)}, B$	Brownian motion
$\mathcal{F}, \tilde{\mathcal{F}}, \mathcal{F}^{(N)}$	$\sigma$ -field
$\mathbb{F}, \tilde{\mathbb{F}}, \mathbb{F}^{(N)}$	Filtration
$\mathbb{P}, \mathbb{P}^{(N)}$	Probability measures
$\Omega, \Omega^{(N)}$	Sample space
$\mathcal{A}, \mathcal{A}_i^{(N)}$	Admissible set for controls
$L^p(\Omega, \mathbb{P})$	Space of random variable with finite $p$ -th moment
$L^p_{\mathbb{F}}([0,T]\times\Omega)$	Space of $\mathbb F\text{-}\mathrm{progressively}$ measurable random processes which are integrable on
	$[0,T]$ in $L^p$ sense
$\mathcal{P}_p(P)$	Collection of measures $m$ on Polish space $P$ having finite $p$ -th moment, i.e.,
	$\int_P  x ^p m(dx) < \infty$
$\mathbb{W}_p(\cdot, \cdot)$	<i>p</i> -Wasserstein metric
$\delta_x$	Dirac measure on the point $x$
$[m]_k$	k-th moment of the measure $m$
ho(v)	Empirical measure of the vector $v = (v_1, v_2, \ldots, v_N)$ with Dirac measure $\delta$ , i.e.,
	$\rho(v) = \frac{1}{N} \sum_{i=1}^{N} \delta_{v_i}$
$1\!\!1_N$	N-dimensional vector with all entries are 1
$e_i$	N-dimensional natural basis

# Chapter 2:

X	Controlled process
u	Control
·	The Euclidean norm of vectors in $\mathbb{R}^d$
$\ \cdot\ _2$	The spectral norm of matrices
$\mathbb{S}^{d  imes d}$	The space of all $d \times d$ symmetric matrices
$A^{ op}$	The transpose matrix of $A$
$A \ge 0 \ (A > 0)$	A is a positive semidefinite (definite) matrix
$\sqrt{A}$	The unique $B \ge 0$ satisfying $B^2 = A$ when $A \ge 0$ is a positive semidefi-
	nite matrix
$\mathbf{O}_d$	The $d \times d$ matrix in which all the entries are 0
$0_d$	The $d$ -dimensional column vector with all entries are 0
$\operatorname{diag}(\lambda_1,\lambda_2,\ldots,\lambda_d)$	The diagonal matrix with entries $\lambda_1, \lambda_2, \ldots, \lambda_d$
$\mathcal{U}_{[0,T]}$	Admissible control set defined by $L^2_{\mathbb{F}}([0,T] \times \Omega)$
U	Admissible control set such that the underlying process is square inte-
	grable for all $t \geq 0$ , i.e., $\mathcal{U} := \mathcal{U}_{[0,\infty)}$
$\mathcal{U}[\mu_{\infty}]$	Admissible control set such that for a distribution $\mu_{\infty} \in \mathcal{P}_2(\mathbb{R})$ , the
	underlying process X satisfying $\lim_{t\to\infty} \mathbb{W}_2(\mathcal{L}(X(t)), \mu_\infty) = 0$
$\langle \phi, \mu  angle$	Inner product of function $\phi$ on $\mathbb{R}$ and measure $\mu \in \mathcal{P}_2(\mathbb{R})$
$\mathcal{N}(\mu, u)$	Normal distribution with mean $\mu$ and variance (or covariance matrix) $\nu$

Chapter 3: for all  $i = 1, 2, \cdots, N$ ,

Y	Continuous-time Markov chain
${\cal Y}$	State space of Markov chain $Y$
Q	Generator of the continuous-time Markov chain
$\mathcal{L}(X Z)$	Distribution of random variable X conditional on $\sigma(Z)$
$(P, \mathcal{B}(P), d)$	Polish space
$X, X_i^{(N)}$	Controlled processes
$\alpha, \alpha_i^{(N)}$	Control

Chapter 4: for all  $i = 1, 2, \cdots, N$ ,

$X, X_i^{(N)}$	Controlled processes
$\alpha, \alpha_i^{(N)}$	Control
$C_b(\mathbb{R}^d)$	Collection of bounded and continuous functions on $\mathbb{R}^d$
$C_b^1(\mathbb{R}^d)$	Space of function on $\mathbb{R}^d$ whose first order derivative and itself are bounded and
	continuous
$\mathcal{B}(\mathbb{R}^d)$	Borel set on $\mathbb{R}^d$
$f_*m$	Pushforward measure of $m$ by function $f$
$\hat{\alpha}, \hat{\gamma}, \beta, \alpha_i, \gamma_i$	Random variables

## Chapter 5:

$\mathbb{T}^d$	<i>d</i> -dimensional torus space
g	Graphon, a symmetric measurable function from $[0,1]^2$ to $\mathbb{R}$
lpha,eta	Nodes in graphon, $\alpha, \beta \in [0, 1]$
a	Control
$(S^{1/2},\rho)$	$S^{1/2}$ is the collection of function $\mu$ : $[0,T] \times [0,1] \mapsto \mathcal{P}_1(\mathbb{T}^d)$ such that
	$ \mu _{1/2} \ = \  \mu _0 \ + \ [\mu]_{1/2} \ < \ \infty, \ \text{where} \ \  \mu _0 \ = \ \sup_{t,\alpha} \int_{\mathbb{T}^d}  x  \mu(t,\alpha,dx) \ \text{ and}$
	$[\mu]_{1/2} = \sup_{t_1 \neq t_2, \alpha} \frac{\mathbb{W}_1(\mu(t_1, \alpha), \mu(t_2, \alpha))}{ t_1 - t_2 ^{1/2}}, \text{ endowed with the metric } \rho(\mu_1, \mu_2) =$
	$\sup_{t,\alpha} \mathbb{W}_1(\mu_1(t,\alpha),\mu_2(t,\alpha))$
$B_r$	A closed convex and compact subset of $(S^{1/2},\rho)$ defined by $\{\mu\in S^{1/2}: \mu _{1/2}\leq r\}$

# List of Figures

3.1	MFG diagram.	40
3.2	Equivalent MFG diagram.	46
3.3	Simulations for $a_y, V, \alpha$ and $\nu$	63
3.4	Simulations for $b_y$ and $c_y$ .	64
3.5	Simulations for $\mu_t$ and $\nu_t$ .	64
3.6	Simulation of player 1's optimal value function $V$	65
4.1	MFG diagram 2	81
4.2	Equivalent MFG diagram 2	96

# Chapter 1

# Introduction

### **1.1** Introduction and motivation

In the realm of scientific inquiry, understanding the behavior of complex systems has been a fundamental pursuit, one that transcends disciplinary boundaries and holds profound implications for diverse fields such as biology, physics, economics, and engineering. Within this rich tapestry of complexity, stochastic nonlinear systems occupy a central role, characterized by their intricate dynamics and inherent randomness. These systems, while challenging to decipher, provide valuable insights into the intricate dance of order and chaos that governs our world.

The thesis at hand embarks on a journey into the heart of this complexity, focusing on the convergence analysis of stochastic nonlinear systems. Specifically, it delves into the study of their asymptotic behavior, an area of paramount importance for comprehending the long-term dynamics of these systems and the systems with large populations. The asymptotic behavior not only holds significance from a theoretical standpoint but also bears immense importance in practical applications. Analyzing the asymptotic behavior of systems characterized by long-term dynamics and large populations is often a challenging work. This thesis, more specifically, concentrates on two key aspects: firstly, the examination of the turnpike property within the context of stochastic control problems, and secondly, the investigation of the convergence of N-player games towards their corresponding mean field games (MFG).

#### 1.1.1 Turnpike properties of stochastic control problem

Firstly, we investigate the asymptotic behavior, i.e., turnpike properties, of the systems with longterm dynamics and the convergence is with respect to the time horizon.

The exploration of the turnpike property originated in the field of economics, serving as a tool to analyze the stationary behavior during transient periods in long-horizon control problems. Initially proposed by von Neumann [64], the terminology was later introduced by Dorfman, Samuelson, and Solow [22]. The turnpike property characterizes scenarios where the solution to an optimization problem concentrates on specific static points, evenly spaced along a defined path. Subsequently, the turnpike phenomenon has garnered significant attention in both finite and infinite-dimensional problems within deterministic discrete-time and continuous-time systems, see the book [15] as an excellent survey and a list of numerous references, including [62, 8, 57, 79, 70, 24, 81, 82, 83].

Notably, we highlight the works of [70, 71], delving into continuous-time linear quadratic (LQ) problems for ordinary differential equations (ODE), and the recent publication by Sun, Wang, and Yong [74], which addresses stochastic LQ optimal control problems. Additionally, Sun and Yong, as discussed in [75], have established the exponential, integral, and mean-square turnpike properties for optimal pairs in mean-field linear stochastic differential equations. This is contingent upon the satisfaction of the stabilizability condition for the state equation.

Chapter 2 is mainly based on the paper [49], in which we examine the limiting behavior of a specific class of linear quadratic stochastic optimal control problems and their corresponding value functions as the time horizon approaches infinity. This collaborative work involves Professor Sixian Jin from California State University San Marcos and Professor Qingshuo Song from Worcester Polytechnic Institute. Firstly, we establish the consistency between the cell problem in weak KAM theory and the static optimization problem from the perspective of the turnpike property. Secondly, we provide the connection between the cell problem and the ergodic cost problem, and then the classical turnpike property and the turnpike property in terms of the cost function are identified.

#### 1.1.2 Convergence of mean field games with common noise

Different from the first project, next we examine the system complexity. More precisely, we consider the convergence behavior of systems with large populations.

Mean field games theory was introduced by Lasry and Lions in their seminal paper [55], and by Huang, Caines, and Malhame ([45, 42, 43, 41]). It aims to provide a framework for studying the asymptotic behavior of N-player differential games being invariant under the reshuffling of the players' indices. For a comprehensive overview of recent advancements and relevant applications of MFG theory, it is recommended to refer to the two-volume book by Carmona and Delarue ([16, 17]) published in 2018 and the references provided therein.

Mean field games have been widely accepted as an approximation for the N-player games, particularly when the number of players, N, is large enough. A fundamental question that arises in this context concerns the convergence rate of this approximation. Convergence can be analyzed from different perspectives, such as convergence in value, the trajectory followed by the representative player, or the behavior of the mean field term. Each of these perspectives offers valuable insights into the behavior and characteristics of the MFG approximation. Furthermore, they raise a variety of intriguing questions within this context.

In Chapter 3, we investigate the convergence rate of the N-player game, governed by a Markov chain common noise, towards its asymptotic MFG under the linear-quadratic-Gaussian structure. To achieve this, firstly, we introduce a Markovian structure using two auxiliary processes for the first and second moments of the MFG equilibrium and employ the fixed point condition in MFG. By doing so, we characterize the equilibrium measure in MFG with a finite-dimensional Riccati system of ODEs. Consequently, we obtain the equilibrium path, equilibrium control, and the value function in MFG. Subsequently, we address the N-player game under the LQG structure, and we characterize its equilibrium path, equilibrium control, and the value function through a Riccati system of ODEs with a dimension of  $O(N^3)$ . Leveraging the N-invariant algebraic structure of this system of ODEs, we establish a dimension reduction result, facilitating a comparison between the equilibrium path in the *N*-player game and the equilibrium path in the MFG. To demonstrate the convergence between the two equilibrium paths, we embed the equilibrium path in the *N*player game to the probability space of the equilibrium path in the MFG using a distribution copy, leading to the achievement of the convergence result and the computation of the convergence rate. Lastly, some numerical examples are presented to demonstrate the convergence result. This chapter is based on the paper [50], which had been accepted by the journal *Nonlinear Analysis: Hybrid Systems* in December 2023. It is a joint work with Dr. Jiaxuan Ye, Ph.D. student Peiyao Lai and Professor Qingshuo Song from Worcester Polytechnic Institute.

Note that the number of states of the common noise considered in the Chapter 3 is finite, it is natural to consider the case when the number of states of the common noise is infinity. Thus, we consider the convergence rate of the mean field game with Brownian motion as its common noise in the next chapter.

Chapter 4 is mainly based on [51], in which we focus on a class of one-dimensional linearquadratic-Gaussian mean field games with Brownian motion as the common noise. This collaborative work involves Dr. Jiaxuan Ye and Professor Qingshuo Song from Worcester Polytechnic Institute. It is worth noting that the equilibrium path, equilibrium control, and the value function in MFG and the N-player game can be obtained by a similar methodology in Chapter 3. Our main contribution is the establishment of three different convergence rates from the N-player games to the corresponding mean field games. Firstly, we establish that the convergence rate of the p-Wasserstein metric for the distribution of the representative player in the N-player game to the distribution of the generic player in MFG is  $O(N^{-1/2})$  for  $p \in [1, 2]$ . Secondly, it demonstrates that the convergence rate of the p-Wasserstein metric for the empirical measure of the equilibrium path in the N-player game to the equilibrium measure in MFG under the  $L^p$  sense is  $O(N^{-1/(2p)})$ for  $p \in [1,2]$ . Lastly, we show that the convergence rate of the uniform p-Wasserstein metric for the empirical measure of the equilibrium path in the N-player game to the equilibrium measure in MFG under the  $L^p$  sense is  $O(N^{-1/(2p)})$  for  $p \in (1, 2]$ , and  $O(N^{-1/2} \ln(N))$  for p = 1. To investigate these convergence rates, our methodology relies on a specific decomposition of the equilibrium path in the N-player game and utilizes the associated MFG framework.

#### 1.1.3 Graphon mean field games

The basic structure of standard MFG theory assumes a symmetry in the connections of the agents but not necessarily of their dynamics. However, in the recent studies [10, 11, 12], asymmetric graph connections in large population games are considered. Large subpopulations (or clusters) of agents are placed at their particular nodes and communicate with the neighbouring subpopulations via the graph edges. The graphs are heterogeneous with the edges having not necessarily identical weights. In the network limit, a graphon gives the communication weights  $g(\alpha, \beta)$ , see for instance the introductions to each of [10, 11, 12, 34] for the graphon MFG (GMFG) framework and [58] for graphon theory. Therefore, it is interesting to investigate the large population games with asymmetric graph connections.

Along with [10, 11, 12], in Chapter 5, we consider the solvability of a type of graphon mean

field games. A new type of mean field games PDE system associated with the graphon mean field games system is proposed in this project. The graphon mean field games system consists of a collection of parameterized Hamilton-Jacobi-Bellman (HJB) equations and a collection of parameterized Fokker-Planck-Kolmogorov equations. We establish the existence of solutions via the application of Schauder's fixed point theorem and obtain the uniqueness via the Lasry-Lions monotonicity assumption on the running cost. The main difficulty is to obtain the regularity of the solution and the sensitivity of the corresponding HJB equations and Fokker-Planck-Kolmogorov equations. This chapter is mainly based on the paper [9], which is conducted in collaboration with Professor Peter E. Caines from McGill University, Professor Daniel Ho from City University of Hong Kong, Professor Minyi Huang from Carleton University, and Professor Qingshuo Song from Worcester Polytechnic Institute. It had been accepted for publication in the journal *ESAIM: Control, Optimisation and Calculus of Variations* in March 2022.

# Chapter 2

# Long-time behavior of stochastic LQ control problem

### 2.1 Introduction

In this chapter, we consider a  $\mathbb{R}^d$ -valued standard Brownian motion  $\{W(t)\}_{t\geq 0}$  defined on a complete filtered probability space  $(\Omega, \mathcal{F}, \mathbb{F} = \{\mathcal{F}_t\}_{t\geq 0}, \mathbb{P})$  which satisfies the usual conditions. Let T > 0 be a given time horizon,  $|\cdot|$  be the Euclidean norm of vectors in  $\mathbb{R}^d$ , and  $||\cdot||_2$  be the spectral norm of matrices. We use  $L^p_{\mathbb{F}}(\Omega \times [0,T])$  to denote the space of all  $\mathbb{F}$ -progressively measurable random processes  $u = \{u(t)\}_{t\in [0,T]}$  satisfying  $\mathbb{E}[\int_0^T |u(t)|^p dt] < \infty$ . Consider the following diffusion process given by a linear stochastic differential equation (SDE)

$$dX(t) = (AX(t) + u(t) + b) dt + \sigma dW(t), \quad X(0) = x, \quad t \ge 0,$$
(2.1.1)

where  $A \in \mathbb{S}^{d \times d}$  is a  $d \times d$  symmetric constant matrix,  $b, x \in \mathbb{R}^d$  are constant vectors, and  $\sigma \in \mathbb{R}^+$  is a positive constant.

The classical stochastic LQ control problem over the finite time horizon [0, T] is to find an optimal control  $u_T^*$  from the space  $\mathcal{U}_{[0,T]} := L^2_{\mathbb{F}}(\Omega \times [0,T])$  such that the quadratic cost functional

$$J_T(x; u_T) = \mathbb{E}\left[\int_0^T L(X(t), u_T(t))dt\right]$$
(2.1.2)

is minimized for a given initial state  $x \in \mathbb{R}^d$ , i.e.,

$$V_T(x) := J_T(x; u_T^*) = \inf_{u_T \in \mathcal{U}_{[0,T]}} J_T(x; u_T),$$
(2.1.3)

where

$$L(x, u) = \frac{1}{2} \left( x^{\top} Q x + |u|^2 \right) + q^{\top} x + r^{\top} u$$

with Q be a positive definite matrix in  $\mathbb{S}^{d \times d}$  and  $q, r \in \mathbb{R}^d$ . The corresponding optimal path is denoted by  $X_T^*(t)$  for  $t \in [0, T]$ . Note that, we could set Q be in  $\mathbb{R}^{d \times d}$  in general.

The turnpike property of the above finite time control problem is associated with the following static optimization problem: Determine the point  $(\hat{x}, \hat{u})$  to

$$\begin{cases} \text{minimize} \quad F(x,u) := \frac{1}{2} \left( x^{\top} Q x + |u|^2 + 2q^{\top} x + 2r^{\top} u \right) + \frac{1}{2} \sigma^2 \text{trace}(P), \\ \text{subject to} \quad Ax + u + b = \mathbf{0}_d, \end{cases}$$
(2.1.4)

where  $\mathbf{0}_d$  is the *d*-dimensional column vector with all entries are 0 and *P* is a positive definite solution to

$$P^2 - 2AP - Q = \mathbf{O}_d.$$

In the above equation,  $\mathbf{O}_d$  is the  $d \times d$  matrix in which all the entries are 0. If the underlying control problem is deterministic, i.e.,  $\sigma = 0$ , as it is shown in [70], the turnpike property refers to the following estimation: There exist some  $\lambda > 0$  and K > 0 independent of t and T such that

$$|X_T^*(t) - \hat{x}| + |u_T^*(t) - \hat{u}| \le K \left( e^{-\lambda t} + e^{-\lambda(T-t)} \right), \quad \forall t \in [0, T].$$

Indeed, this turnpike property reveals that, for a sufficiently large T, one can achieve a good approximation of the optimal trajectory during the majority time period  $[\delta T, (1 - \delta)T]$  for some  $0 < \delta << 1/2$  by simply staying at the stable point  $\hat{x}$  of the static optimization problem in the sense

$$|X_T^*(t) - \hat{x}| + |u_T^*(t) - \hat{u}| \le 2Ke^{-\lambda\delta T}, \quad \forall t \in [\delta T, (1-\delta)T], \ \delta \in (0, 1/2).$$

However, extending this turnpike property to the stochastic control problem with  $\sigma > 0$  poses challenges. This is because the presence of Brownian noise makes it impossible for any control to freeze the state unchanged at a fixed point. Recently, Sun and Yong (see Theorem 3.2 of [75]) proved the following (stochastic version) turnpike property

$$\mathbb{E}\left[|X_T^*(t) - X^*(t)|^2 + |u_T^*(t) - u^*(t)|^2\right] \le K\left(e^{-\lambda t} + e^{-\lambda(T-t)}\right), \quad \forall t \in [0, T]$$
(2.1.5)

by constructing two stochastic processes  $X^*$  and  $u^*$ , independent to T, satisfying  $\mathbb{E}[X^*(t)] = \hat{x}$ ,  $\mathbb{E}[u^*(t)] = \hat{u}$ .

In this study, we aim to reexamine the turnpike property, approaching it from a distinct perspective: the cell problem within the framework of weak Kolmogorov–Arnold–Moser (KAM) theory. Specifically, the objective of the cell problem is to seek the solution  $(v, c_0) \in C^2(\mathbb{R}^d) \times \mathbb{R}$  to the following equation

$$c_0 = H\left(x, -\nabla v(x), -D^2 v(x)\right),$$
(2.1.6)

where the Hamiltonian  $H : \mathbb{R}^d \times \mathbb{R}^d \times \mathbb{R}^{d \times d} \mapsto \mathbb{R}$  is given by

$$H(x,\bar{p},\bar{q}) := \sup_{u \in \mathbb{R}^d} \left\{ (Ax + u + b)^\top \bar{p} + \frac{1}{2} \sigma^2 \operatorname{trace}(\bar{q}) - L(x,u) \right\}.$$

Note that uniqueness does not apply to the cell problem, as  $(v, c_0)$  is a solution if and only if  $(v + k, c_0)$  is also a solution for any  $k \in \mathbb{R}$ . Therefore, when we refer to the uniqueness of the cell

problem, we specifically mean uniqueness in the value of  $c_0$ .

Initially, the weak KAM Theory was developed by Fathi [28] and Mather [61], and is linked to the theory of homogenization for Hamilton-Jacobi (HJ) equations developed by Lions, Papanicolaou, and Varadhan in [56]. It provides the connection between a type of control problem and the cell problem, offering a representation of the optimal ergodic cost. In addition to its fundamental role in the theory of homogenization, the weak KAM theory has also been used to study the long-time behavior of dynamic control problems, including the ergodic behavior of the value function and the corresponding HJ equation in the deterministic case (see [63, 7, 26, 27, 77]), as well as the Hamilton-Jacobi-Bellman (HJB) equation in the stochastic case (see [4, 47, 20]).

Compared to the above literature, this Chapter provides a distinct approach to show the turnpike properties in stochastic control theory by using the cell problem in PDE and contributions can be summarized as follows. Our first contribution lies in the formulation of a verification theorem connecting the cell problem to a specific class of infinite time horizon control problems, referred to as the probabilistic cell problem, see Lemma 5. Unlike the typical cell problem explored in the existing literature (e.g., [77]), the underlying cell problem in our context lacks uniqueness due to the non-compactness of the domain. It is the verification theorem, which establishes a tailored sufficient condition to distinguish the right solution to the probabilistic cell problem from multiple solutions of the cell problem. An immediate consequence of the verification theorem is the establishment of a link between the cell problem and the static optimization problem.

Our second contribution provides the connection between the cell problem (2.1.6) and the ergodic cost problem (refer to Remark 3.6.7 in [3]), which involves determining the constant

$$-c_* := \lim_{T \to \infty} \frac{1}{T} V_T(x) = \lim_{T \to \infty} \frac{1}{T} J_T(x; u_T^*).$$
(2.1.7)

The importance of this connection is that it unveils a new turnpike property in terms of the cost function in addition to the aforementioned turnpike property of (2.1.5) with respect to the control process and state process:

$$\lim_{T \to \infty} \frac{1}{T} J_T(x; u^*) = -c_*, \qquad (2.1.8)$$

where  $u^*$  is a *T*-independent control process obtained from the probabilistic cell problem, see Theorem 9.

The five problems, namely, the Finite Time Stochastic Control Problem (2.1.3), the Static Optimization Problem (2.1.4), the Cell Problem (2.1.6), the Ergodic Cost Problem (2.1.7), and the Probabilistic Cell Problem introduced in this Section 2.1 will be interwoven throughout the remainder of this manuscript in the following manner: In Section 2.2, we prove the verification of the cell problem and further provide the consistency of the cell problem and the static optimization problem. Additionally, we identify the turnpike properties of (2.1.5) and (2.1.8) in Section 2.3. To further elucidate the results obtained in Section 2.3, we provide an illustrative example in Section 2.4. Proofs of certain lemmas are collected in Appendix 2.5.

Throughout this chapter, we use  $\mathbb{S}^{d \times d}$  to denote the space of all  $d \times d$  symmetric matrices, and  $A \ge 0$  (A > 0) to denote a positive semidefinite (definite) matrix. If  $A \ge 0$ , then  $\sqrt{A}$  denotes the unique  $B \ge 0$  satisfying  $B^2 = A$ . We also use  $\mathbf{O}_d$  denote the  $d \times d$  matrix in which all the

entries are 0 and  $\mathbf{0}_d$  to be the *d*-dimensional column vector with all entries are 0. We also pose the following assumptions to coefficients:

(Cf)  $A \in \mathbb{S}^{d \times d}$  and Q > 0 are  $d \times d$  symmetric constant matrices,  $b, x, q, r \in \mathbb{R}^d$  are constant vectors, and  $\sigma \in \mathbb{R}^+$  is a positive constant.

## 2.2 Cell problem and the verification theorem

We commence our exploration with the solvability of the cell problem (2.1.6) within the framework of weak KAM theory. In prevailing literature, the solution of the cell problem is unique up to the constant. Interestingly, this uniqueness fails in our framework attributed to the absence of compactness in the domain. This necessitates the formulation of a meticulously crafted verification theorem: How can we discern the appropriate solution from the multitude available to accurately characterize the associated optimal control problem?

#### 2.2.1 Existence and nonuniqueness of the cell problem

We recall the cell problem: Find the solution  $(v, c_0) \in C^2(\mathbb{R}^d) \times \mathbb{R}$  to the following equation

$$c_0 = H\left(x, -\nabla v(x), -D^2 v(x)\right),$$

where the Hamiltonian  $H: \mathbb{R}^d \times \mathbb{R}^d \times \mathbb{R}^{d \times d} \mapsto \mathbb{R}$  is given by

$$H(x,\bar{p},\bar{q}) := \sup_{u \in \mathbb{R}^d} \left\{ (Ax + u + b)^\top \bar{p} + \frac{1}{2} \sigma^2 \operatorname{trace}(\bar{q}) - L(x,u) \right\} := \sup_{u \in \mathbb{R}^d} H^u(x,\bar{p},\bar{q}).$$
(2.2.1)

**Lemma 1.** The cell problem (2.1.6) can be solved by  $(v, c_0)$  in the form of

$$v(x) = \frac{1}{2}x^{\top}Z_1x + x^{\top}Z_2 + Z_3$$

and

$$-c_0 = \frac{1}{2}\sigma^2 trace(Z_1) + b^\top Z_2 - \frac{1}{2}|Z_2 + r|^2, \qquad (2.2.2)$$

where  $(Z_1, Z_2) \in \mathbb{S}^{d \times d} \times \mathbb{R}^d$  can be any solution pair to the system of equations

$$\begin{cases} Z_1^2 - 2AZ_1 - Q = \mathbf{O}_d, \\ (Z_1 - A) Z_2 - Z_1 b + Z_1 r - q = \mathbf{O}_d, \end{cases}$$
(2.2.3)

and  $Z_3 \in \mathbb{R}$  is an arbitrary constant.

Proof. Assume  $(v, c_0)$  solves the cell problem (2.1.6) with v satisfying a quadratic form  $v(x) = \frac{1}{2}x^{\top}Z_1x + x^{\top}Z_2 + Z_3$ . Note that  $Z_1$  is symmetric, then  $\nabla v(x) = Z_1x + Z_2$  and  $D^2v(x) = Z_1$ , and thus the cell problem (2.1.6) can be rewritten by  $H(x, -(Z_1x + Z_2), -Z_1) = c_0$ . We first observe

that

$$H(x,\bar{p},\bar{q}) = (Ax + \bar{u}(x,\bar{p}) + b)^{\top}\bar{p} + \frac{1}{2}\sigma^{2}\mathrm{trace}(\bar{q}) - L(x,\bar{u}(x,\bar{p})),$$

where

$$\bar{u}(x,\bar{p})=\bar{p}-r,$$

thus the cell problem can be reduced to

$$c_{0} = -(Ax - Z_{1}x + b - r - Z_{2})^{\top} (Z_{1}x + Z_{2}) - \frac{1}{2}\sigma^{2} \operatorname{trace}(Z_{1}) - \frac{1}{2}x^{\top}Qx - \frac{1}{2}|Z_{1}x + Z_{2} + r|^{2} - q^{\top}x + r^{\top}(Z_{1}x + Z_{2} + r)$$

$$(2.2.4)$$

for all  $x \in \mathbb{R}^d$ . By setting the linear and quadratic terms with respect to x on the right hand side of (2.2.4) to 0, we obtain the system of equations (2.2.3). Combining like terms for the constants in (2.2.4) provides the expression  $c_0$  of (2.2.2).

The uniqueness of  $c_0$  in the cell problem is established by Theorem 4.2 in [77] for the periodic domain. However, it is noteworthy that the uniqueness does not extend to our setting defined by equation (2.1.6). Recall that, by spectral theorem, if A > 0, it admits orthogonal decomposition  $A = QDQ^{\top}$  for some orthogonal matrix Q and diagonal matrix  $D = \text{diag}(\lambda_1, \lambda_2, \ldots, \lambda_d)$  with  $\lambda_i > 0$ . Moreover, any matrix in the form of  $B = Q\text{diag}(\pm\lambda_1, \pm\lambda_2, \ldots, \pm\lambda_d)Q^{\top}$  satisfies  $B^2 = A$ . In this below,  $\sqrt{A}$  is the unique choice of B satisfying B > 0 and  $B^2 = A$ . Also,  $\sqrt{A}$  can be represented by  $QD^{1/2}Q^{\top}$ .

**Lemma 2.** There exists multiple solution pairs  $(Z_1, Z_2)$  of (2.2.3) in  $\mathbb{S}^{d \times d} \times \mathbb{R}^d$ , hence the  $c_0$  of the cell problem (2.1.6) is not unique. Moreover, there exists unique solution pair  $(Z_1, Z_2)$  satisfying

$$D_1 = Z_1 - A > 0.$$

Moreover,  $Z_1$  and  $D_1$  are commutative, hence they share the same eigenvector matrix.

*Proof.* Taking transpose to the first equation of (2.2.3), we have  $Z_1^2 - 2Z_1A - Q = \mathbf{O}_d$  as  $Z_1, A$  and Q are symmetric matrices, which implies that  $Z_1$  and A are commutative, i.e.,  $Z_1A = AZ_1$ . Thus,  $(Z_1 - A)^2 = A^2 + Q$  holds. By spectral theorem, since  $A^2 + Q > 0$ , there exists multiple solutions to  $Z_1$  and the unique choice of  $Z_1$  to have  $D_1 = Z_1 - A > 0$  is

$$Z_1 = A + \sqrt{A^2 + Q}$$

Accordingly, from the second equation of (2.2.3),  $Z_2$  can be written in terms of  $Z_1$  as

$$Z_2 = (Z_1 - A)^{-1}(Z_1b - Z_1r + q).$$

At the end,  $D_1Z_1 = Z_1D_1 = Z_1^2 - AZ_1$  holds by commutativity of A and  $Z_1$ , thus they have the same eigenvector matrix by Page 305 of [73].

#### 2.2.2 Verification theorem to probabilistic cell problem

The lack of uniqueness in the determination of  $c_0$  as indicated by Lemma 2 introduces a compelling challenge when striving to identify the optimal value for  $c_0$  in order to substantiate the verification theorem associated with its probabilistic counterparts. In the following discussion, our objective is to delineate the connection between the cell problem (2.1.6) and the probabilistic cell problem (2.2.5)-(2.2.6), which is referred to the verification procedure in the context of the control theory.

#### 2.2.2.1 Probabilistic cell problem

We consider the following probabilistic cell problem. Let  $\mathcal{P}_2(\mathbb{R}^d)$  be the Wasserstein space of probability measures  $\mu$  on  $\mathbb{R}^d$  satisfying  $\int_{\mathbb{R}^d} |x|^2 d\mu(x) < \infty$  endowed with 2-Wasserstein metric  $\mathcal{W}_2(\cdot, \cdot)$  defined by

$$\mathcal{W}_{2}(\mu_{1},\mu_{2}) = \inf_{\pi \in \Pi(\mu_{1},\mu_{2})} \left( \int_{\mathbb{R}^{d}} \int_{\mathbb{R}^{d}} |x-y|^{2} d\pi(x,y) \right)^{\frac{1}{2}},$$

where  $\Pi(\mu_1, \mu_2)$  is the collection of all probability measures on  $\mathbb{R}^d \times \mathbb{R}^d$  with its marginals agreeing with  $\mu_1$  and  $\mu_2$ . Moreover, we denote that  $\langle \phi, \mu \rangle$  to be

$$\langle \phi, \mu \rangle = \int_{\mathbb{R}^d} \phi(x) \mu(dx)$$

for all function  $\phi$  valued on  $\mathbb{R}^d$  and all  $\mu \in \mathcal{P}_2(\mathbb{R}^d)$ . To proceed, we define  $\mathcal{U}$  as the collection of all  $\mathbb{F}$  progressively measurable processes such that

• its associated state process  $X^u = X$  given by

$$dX(t) = (AX(t) + u(t) + b) dt + \sigma dW(t), \quad X(0) = x, \quad t \ge 0,$$

is well-defined;

- $\mathbb{E}[|X(t)|^2] < \infty$  for all t > 0 and  $x \in \mathbb{R}^d$ ;
- the law of X(t) converges to some distribution  $\mu_{\infty} \in \mathcal{P}_2(\mathbb{R}^d)$  in 2-Wasserstein metric  $\mathcal{W}_2$ , i.e.,

$$\lim_{t \to \infty} \mathcal{W}_2(Law(X(t)), \mu_\infty) = 0.$$

For convenience, we also use  $\mathcal{U}[\mu_{\infty}]$  for a  $\mu_{\infty} \in \mathcal{P}_2(\mathbb{R}^d)$  to denote the collection of  $u \in \mathcal{U}$  such that  $Law(X^u(t))$  converges to  $\mu_{\infty}$ .

We define the probabilistic cell problem below: Determine (V, c) such that

$$-c = \inf_{u \in \mathcal{U}} \limsup_{T \to \infty} \frac{1}{T} \int_0^T \mathbb{E} \left[ L(X(t), u(t)) \right] dt, \qquad (2.2.5)$$

and

$$V(x) = \inf_{u \in \mathcal{U}[\mu_{\infty}^*]} \limsup_{T \to \infty} \int_0^T \mathbb{E} \left[ L(X(t), u(t)) + c \right] dt,$$
(2.2.6)

where  $\mu_{\infty}^*$  is the distribution limit of the optimal path  $X^*$  in the ergodic control problem (2.2.5).

Following the convention in stochastic control theory, we denote V(x) as the value function of the probabilistic cell problem, provided it is well-defined. However, unlike the standard control problem, the objective of the probabilistic cell problem is not solely to identify a value function V; rather, it involves determining a pair (V, c) among many solutions of the cell problem (2.1.6).

#### 2.2.2.2 Verification for a general setting

In this part, we will prove a verification for an infinite-time control problem under a general settings and all notations are independent to the rest of this chapter.

The cell problem in the general setting is to find the pair  $(v, c_0) \in C^2(\mathbb{R}^d) \times \mathbb{R}$  such that

$$c_0 = H\left(x, -\nabla v(x), -D^2 v(x)\right) := \sup_a H^a\left(x, -\nabla v(x), -D^2 v(x)\right),$$
(2.2.7)

where  $H^a: \mathbb{R}^d \times \mathbb{R}^d \times \mathbb{R}^{d \times d} \mapsto \mathbb{R}$  for any  $a \in \mathbb{R}^d$  is given by

$$H^{a}(x,\bar{p},\bar{q}) = \hat{b}(x,a) \cdot \bar{p} - \hat{L}(x,a) + \frac{1}{2}\hat{\sigma}^{2}\operatorname{trace}(\bar{q})$$

for some  $\hat{b}: \mathbb{R}^d \times \mathbb{R}^d \mapsto \mathbb{R}^d$ ,  $\hat{\sigma} \in \mathbb{R}$ , and  $\hat{L}: \mathbb{R}^d \times \mathbb{R}^d \mapsto \mathbb{R}$ .

**Assumption 3.**  $\hat{b}$  is Lipschitz continuous on  $\mathbb{R}^d \times \mathbb{R}^d$ , and  $\hat{L}$  is locally Lipschitz continuous and satisfies quadratic growth, i.e., for all  $(x, a) \in \mathbb{R}^d \times \mathbb{R}^d$ ,

$$\left|\hat{L}(x,a)\right| \leq K\left(1+|x|^2+|a|^2\right)$$
 for some  $K \in \mathbb{R}$ .

At the same time, we define its associated probabilistic cell problem: Consider  $\mathbb{R}^d$ -valued controlled process X, with a given  $\mathbb{R}^d$ -Brownian motion W(t), given by

$$dX(t) = \hat{b}(X(t), u(t)) dt + \hat{\sigma} dW(t), \quad X(0) = x \in \mathbb{R}^d, \quad t \ge 0.$$

The objective is to determine (V, c) such that

$$-c = \inf_{u \in \mathcal{U}} \limsup_{T \to \infty} \frac{1}{T} \int_0^T \mathbb{E} \left[ \hat{L}(X(t), u(t)) \right] dt, \qquad (2.2.8)$$

and

$$V(x) = \inf_{u \in \mathcal{U}[\mu_{\infty}^*]} \limsup_{T \to \infty} \int_0^T \mathbb{E} \left[ \hat{L}(X(t), u(t)) + c \right] dt,$$
(2.2.9)

where  $\mu_{\infty}^*$  is the distribution limit of the optimal path  $X^*$  in the ergodic control problem (2.2.8).

**Lemma 4.** For an arbitrary control process  $u \in \mathcal{U}[\mu_{\infty}]$  and any function  $\phi : \mathbb{R}^d \mapsto \mathbb{R}$  with a quadratic growth  $|\phi(x)| \leq K(1+|x|^2)$  for all  $x \in \mathbb{R}^d$ , we have

$$\lim_{t \to \infty} \mathbb{E}\left[\phi(X(t))\right] = \langle \phi, \mu_{\infty} \rangle.$$

*Proof.* Since  $\mu_t := Law(X(t))$  converges to some  $\mu_{\infty} \in \mathcal{P}_2(\mathbb{R}^d)$  in 2-Wasserstein distance, we have

$$\lim_{t \to \infty} \mathbb{E}[|X(t)|^2] = \int_{\mathbb{R}^d} |x|^2 \mu_{\infty}(dx).$$

By the Skorohod representation theorem, one can find another stochastic process Y(t) in a different probability space, such that  $Y(t) \to Y_{\infty}$  almost surely, as well as,  $Law(Y(t)) = \mu_t$  and  $Law(Y_{\infty}) = \mu_{\infty}$ . Hence, by the fact of

$$\phi(Y(t)) \le K\left(1+|Y(t)|^2\right), \text{ and } \mathbb{E}\left[K\left(1+|Y(t)|^2\right)\right] \to \mathbb{E}\left[K\left(1+|Y_{\infty}|^2\right)\right],$$

one can apply the dominated convergence theorem to Y(t) and obtain

$$\lim_{t \to \infty} \mathbb{E}\left[\phi(X(t))\right] = \lim_{t \to \infty} \mathbb{E}\left[\phi(Y(t))\right] = \mathbb{E}\left[\lim_{t \to \infty} \phi(Y(t))\right] = \langle \phi, \mu_{\infty} \rangle.$$

**Lemma 5.** Suppose Assumption 3 holds. Let  $(v, c_0) \in C^2(\mathbb{R}^d) \times \mathbb{R}$  solves the cell problem (2.2.7). In addition, we assume

- 1.  $\nabla v$  is Lipschitz continuous;
- 2. There exists a unique maximizer of  $H^a(x, \bar{p}, \bar{q})$  in the form of

$$\bar{u}(x,\bar{p}) = \arg\max_{a} H^{a}(x,\bar{p},\bar{q});$$

3. The distribution of the process  $X^*(t)$  controlled by  $u^*(t) = \bar{u}(X(t), -\nabla v(X(t)))$  converges to some  $\mu_{\infty}^* \in \mathcal{P}_2(\mathbb{R}^d)$  in 2-Wasserstein distance.

Then, the pair

$$(V := v - \langle v, \mu_{\infty}^* \rangle, c_0)$$

solves the probabilistic cell problem (2.2.8)-(2.2.9).

*Proof.* Since  $(v, c_0) \in C^2(\mathbb{R}^d) \times \mathbb{R}$  solves the cell problem (2.2.7), for a control  $u \in \mathcal{U}$  and its associated state process  $X^u = X$ , by Itô's formula, we obtain

$$\begin{aligned} v(X(t)) &= v(x) + \int_0^t \left( \hat{b}(X(s), u(s)) \cdot \nabla v(X(s)) + \frac{1}{2} \hat{\sigma}^2 \Delta v(X(s)) \right) ds \\ &+ \int_0^t \hat{\sigma} \nabla v(X(s)) \cdot dW(s). \end{aligned}$$

Fixing t > 0 and taking expectation on both sides and note that

$$\mathbb{E}\left[\int_0^t \hat{\sigma}^2 \left|\nabla v(X(s))\right|^2 ds\right] \le \hat{\sigma}^2 K^2 \mathbb{E}\left[\int_0^t \left(1 + |X(s)|^2\right) ds\right] \le \hat{\sigma}^2 K^2 \left(t + \mathbb{E}\left[\int_0^t |X(s)|^2 ds\right]\right)$$

is finite, we have

$$\mathbb{E}\left[v(X(t))\right] = v(x) + \mathbb{E}\left[\int_0^t \left(\hat{b}(X(s), u(s)) \cdot \nabla v(X(s)) + \frac{1}{2}\hat{\sigma}^2 \Delta v(X(s))\right) ds\right].$$

The cell problem (2.2.7) implies that, for all  $x, a \in \mathbb{R}^d$ ,

$$-\hat{b}(x,a)\cdot\nabla v(x) - \hat{L}(x,a) - \frac{1}{2}\hat{\sigma}^2\Delta v(x) - c_0 \le 0,$$

hence

$$\hat{b}(X(s), u(s)) \cdot \nabla v(X(s)) + \frac{1}{2}\hat{\sigma}^2 \Delta v(X(s)) \ge -\hat{L}(X(s), u(s)) - c_0$$

for all  $s \in [0, t]$ . Thus

$$v(x) \le \mathbb{E}\left[v(X(t))\right] + \mathbb{E}\left[\int_0^t \left(\hat{L}(X(s), u(s)) + c_0\right) ds\right].$$
(2.2.10)

The inequality (2.2.10) holds for all  $u \in \mathcal{U}$  and equality holds if  $u = u^*$ .

Moreover, due to Lipschitz continuity of  $\nabla v$ , the value function v satisfies a quadratic growth condition, i.e.,  $|v(x)| \leq K(1+|x|^2)$  for all  $x \in \mathbb{R}^d$ . By Lemma 4, we have

$$\lim_{t \to \infty} \mathbb{E}\left[v(X(t))\right] = \langle v, \mu_{\infty} \rangle,$$

where  $\mu_{\infty}$  is the distribution limit of  $X_t$ .

Hence, by taking  $\limsup_{t\to\infty} \frac{1}{t}$  on both sides of (2.2.10), we have

$$-c_0 \leq \limsup_{T \to \infty} \frac{1}{T} \int_0^T \mathbb{E} \left[ \hat{L}(X(t), u(t)) \right] dt.$$

The above inequality holds for all  $u \in \mathcal{U}$  and equality holds if  $u = u^*$ . Thus, we conclude the first identity of the probabilistic cell problem (2.2.8)-(2.2.9).

Moreover, by taking  $\limsup_{t\to\infty}$  on both sides of (2.2.10), we have

$$v(x) \le \langle v, \mu_{\infty} \rangle + \limsup_{t \to \infty} \int_{0}^{t} \mathbb{E} \left[ \hat{L}(X(s), u(s)) + c_{0} \right] ds$$

since  $\mathbb{E}[|X(t)|^2] < \infty$  for all t > 0 and  $\hat{L}(x, a)$  is quadratic growth with respect to x and a from Assumption 3. The above inequality holds for arbitrary  $u \in \mathcal{U}[\mu_{\infty}^*]$ , and an equality holds for  $u^*$ , i.e.,

$$v(x) = \langle v, \mu_{\infty}^* \rangle + \limsup_{t \to \infty} \int_0^t \mathbb{E} \left[ \hat{L}(X^*(s), u^*(s)) + c_0 \right] ds.$$

Therefore, we conclude that

$$v(x) - \langle v, \mu_{\infty}^* \rangle = \inf_{u \in \mathcal{U}[\mu_{\infty}^*]} \limsup_{t \to \infty} \int_0^t \mathbb{E} \left[ \hat{L}(X(s), u(s)) + c_0 \right] ds.$$

#### 2.2.2.3 Verification for LQG setting

In the following, we provide a complete characterization of the probabilistic cell problem (2.2.5)-(2.2.6) in the LQ setting by the help of cell problem given by Lemma 1.

**Theorem 6.** Let  $(v, c_0) \in C^2(\mathbb{R}^d) \times \mathbb{R}$  be the solution to the cell problem (2.1.6) in the form of

$$v(x) = \frac{1}{2}x^{\top}Z_1x + x^{\top}Z_2$$

associated to the

$$Z_1 = A + \sqrt{A^2 + Q}, \ Z_2 = (Z_1 - A)^{-1} (Z_1 b - Z_1 r + q).$$
 (2.2.11)

Then, the pair

$$(V := v - \langle v, \mu_{\infty}^* \rangle, c_0)$$

with

$$\mu_{\infty}^{*} = \mathcal{N}\left(D_{1}^{-1}D_{2}, \frac{1}{2}\sigma^{2}D_{1}^{-1}\right), \ D_{1} = Z_{1} - A, \ D_{2} = b - r - Z_{2}$$
(2.2.12)

solves the probabilistic cell problem (2.2.5)-(2.2.6). Moreover, the optimal path  $X^*$  of the probabilistic cell problem (2.2.5)-(2.2.6) is an OU process given by

$$dX^*(t) = (-D_1X^*(t) + D_2) dt + \sigma dW(t), \quad X^*(0) = x,$$
(2.2.13)

whose distribution converges to  $\mu_{\infty}^*$  in 2-Wasserstein metric and the optimal control process  $u^*$  admits a feedback form of

$$u^{*}(t) = \bar{u} \left( X^{*}(t), -\nabla v \left( X^{*}(t) \right) \right) = -Z_{1} X^{*}(t) - Z_{2} - r, \qquad (2.2.14)$$

where

$$\bar{u}(x,\bar{p})=\bar{p}-r.$$

*Proof.* From Lemma 1, we know that the cell problem (2.1.6) admits solutions  $(v, c_0)$ , where v is in the form of

$$v(x) = \frac{1}{2}x^{\top}Z_1x + x^{\top}Z_2 + Z_3$$

Note that  $v - \langle v, \mu_{\infty}^* \rangle$  is independent to  $Z_3$ , hence it's enough to show the results with  $Z_3 = 0$ . By Lemma 2, there exists unique choice of  $(Z_1, Z_2)$  of (2.2.3) such that  $D_1$  is positive definite.

Therefore, it's enough to check all of the assumptions in Lemma 5. First of all, v is a quadratic function and thus its first order derivative  $\nabla v$  is Lipschitz continuous. Moreover, by the first order condition, the maximizer of  $H^a$  of (2.2.1) uniquely exists in the form of (2.2.14).

For the third assumption, we shall check the convergence of the process associated to the optimal control. The explicit solution to (2.2.13) can be written by

$$X^{*}(t) = \Phi_{t}x + \int_{0}^{t} \Phi_{t-s} ds D_{2} + \sigma \int_{0}^{t} \Phi_{t-s} dW(s),$$

where  $\Phi_t$  is a  $d \times d$  fundamental matrix satisfying  $\Phi_0 = I_d$  and the homogeneous matrix ODE

$$d\Phi_t = -D_1 \Phi_t dt.$$

It is clear that  $X^*$  is an Ornstein-Uhlenbeck (OU) process and

$$X^*(t) \sim \mathcal{N}(m_t, \nu_t),$$

where

$$m_t = \Phi_t x + \int_0^t \Phi_{t-s} ds D_2, \ \nu_t = \sigma^2 \int_0^t \Phi_{t-s}^2 ds$$

By Lemma 2, there exists unique choice of the positive definite  $D_1$ , and one can write its orthogonal nal diagonalization  $D_1 = \tilde{Q}\Lambda\tilde{Q}^{\top}$ , where  $\tilde{Q}$  is an orthogonal matrix and  $\Lambda = \text{diag}(\lambda_1, \lambda_2, \ldots, \lambda_d)$  is a diagonal matrix with all  $\lambda_i > 0$ . This implies that  $D_1$  has its inverse in the form of  $D_1^{-1} = \tilde{Q}\Lambda^{-1}\tilde{Q}^{\top}$  and  $\Phi_t$  can be factored into  $\Phi_t = e^{-D_1 t} = \tilde{Q}e^{-\Lambda t}\tilde{Q}^{\top}$ . Moreover, we have

$$\int_0^t \Phi_{t-s} ds = (I_d - \Phi_t) D_1^{-1}, \ \int_0^t \Phi_{t-s}^2 ds = \frac{1}{2} (I_d - \Phi_t^2) D_1^{-1}.$$

Therefore, the mean and variance of  $X_t^*$  can be rewritten as

$$m_t = \Phi_t x + (I_d - \Phi_t) D_1^{-1} D_2, \ \nu_t = \frac{1}{2} \sigma^2 (I_d - \Phi_t^2) D_1^{-1}.$$

In addition, due to the positive definiteness  $D_1$ ,  $\Phi_t \to 0$  as  $t \to \infty$  and there exist

$$m_{\infty} = D_1^{-1} D_2 \in \mathbb{R}^d$$
 and  $\nu_{\infty} = \frac{1}{2} \sigma^2 D_1^{-1} \in \mathbb{S}^{d \times d}$ 

such that  $m_t$  converges to  $m_\infty$  and  $\nu_t$  converges to  $\nu_\infty$ . Hence, we obtain the desired result that  $X^*(t) \xrightarrow{d} \bar{X}$  as  $t \to \infty$ , where

$$\bar{X} \sim \mathcal{N}\left(m_{\infty}, \nu_{\infty}\right) := \mu_{\infty}^{*} \tag{2.2.15}$$

is a normal random variable. The proof can be concluded from Lemma 5.

#### 2.2.3 Consistency with the static optimization problem

The static optimization problem (2.1.4) plays an important role in the study of the turnpike properties of the control problem in the deterministic type [70, 79] and the stochastic type [74]. In this subsection, we provide a result to show the equality between the constant  $-c_0$  from the cell problem (2.1.6) and the value  $F(\hat{x}, \hat{u})$  from the static optimization problem (2.1.4). Recall that  $L(x, u) = \frac{1}{2}(x^{\top}Qx + |u|^2 + 2q^{\top}x + 2r^{\top}u)$  and  $F(x, u) = L(x, u) + \frac{1}{2}\sigma^2 \operatorname{trace}(P)$ , where P is a positive definite solution to  $P^2 - 2AP - Q = \mathbf{O}_d$ . The solution  $(\hat{x}, \hat{u})$  to the static optimization problem (2.1.4) is the one that solves

$$\begin{cases} \text{Minimize} \quad L(x,u), \\ \text{subject to} \quad u = -(Ax+b), \end{cases}$$

since  $\frac{1}{2}\sigma^2 \operatorname{trace}(P)$  is a constant and it is independent with (x, u).

**Lemma 7.** The optimal static value to the static optimization problem (2.1.4) is

$$F(\hat{x},\hat{u}) = -\frac{1}{2}(Ab+q-Ar)^{\top} \left(Q+A^2\right)^{-1} \left(Ab+q-Ar\right) + \frac{1}{2}b^{\top}b - r^{\top}b + \frac{1}{2}\sigma^2 trace(P), \quad (2.2.16)$$

where P is a positive definite solution to

$$P^2 - 2AP - Q = \mathbf{O}_d$$

and the optimal solution  $(\hat{x}, \hat{u})$  is given by

$$\hat{x} = -(Q + A^2)^{-1} (Ab + q - Ar), 
\hat{u} = A (Q + A^2)^{-1} (Ab + q - Ar) - b.$$

*Proof.* Plugging u = -(Ax + b), we know that L(x, u) is a quadratic function with respect to x and it is given by

$$L(x,u) = \frac{1}{2}x^{\top} (Q + A^2) x + (Ab + q - Ar)^{\top} x + \frac{1}{2}b^{\top}b - r^{\top}b.$$

It is straightforward to obtain the desired result by minimizing the quadratic function under the condition that  $Q + A^2 > 0$ .

Now, we are ready to show the consistency between  $-c_0$  and  $F(\hat{x}, \hat{u})$ .

**Lemma 8.** The constant  $-c_0$  of (2.1.6) is identical to the value  $F(\hat{x}, \hat{u})$  of the static optimization problem (2.1.4).

*Proof.* It's equivalent to verify the equality between the representation of in (2.2.2) and the optimal static value in (2.2.16). Choosing  $Z_1$  be the solution to  $Z_1^2 - 2AZ_1 - Q = \mathbf{O}_d$  such that  $D_1 = Z_1 - A$  is positive definite, i.e.,  $Z_1 = A + \sqrt{A^2 + Q}$ , we obtain that  $Z_1 = P$  and  $Z_2 = (Z_1 - A)^{-1}(Z_1 b + q - Z_1 r)$ . To verify that  $-c_0 = F(\hat{x}, \hat{u})$ , we only need to check that

$$b^{\top} Z_2 - \frac{1}{2} |Z_2 + r|^2 = -\frac{1}{2} (Ab + q - Ar)^{\top} (Q + A^2)^{-1} (Ab + q - Ar) + \frac{1}{2} b^{\top} b - r^{\top} b.$$

By calculation and applying the representation of  $Z_2$ , we have

$$b^{\top} Z_{2} - \frac{1}{2} |Z_{2} + r|^{2}$$

$$= -\frac{1}{2} \left( (Z_{1} - A)^{-1} (Z_{1}b + q - Z_{1}r) + r \right)^{\top} \left( (Z_{1} - A)^{-1} (Z_{1}b + q - Z_{1}r) + r \right)$$

$$+ b^{\top} (Z_{1} - A)^{-1} (Z_{1}b + q - Z_{1}r)$$

$$= -\frac{1}{2} (Z_{1}b + q - Z_{1}r)^{\top} \left( (Z_{1} - A)^{-1} \right)^{2} (Z_{1}b + q - Z_{1}r) - \frac{1}{2}r^{\top}r$$

$$+ (Z_{1}b + q - Z_{1}r)^{\top} \left( (Z_{1} - A)^{-1} \right)^{2} (Z_{1}b - Ab - Z_{1}r + Ar)$$

$$= \frac{1}{2} (Z_{1}b + q - Z_{1}r)^{\top} \left( (Z_{1} - A)^{-1} \right)^{2} (Z_{1}b - Z_{1}r - q - 2Ab + 2Ar) - \frac{1}{2}r^{\top}r.$$

Note that  $Z_1 - A = \sqrt{Q + A^2}$ , then  $((Z_1 - A)^{-1})^2 = (Q + A^2)^{-1}$ . Substituting  $Z_1$  by  $A + \sqrt{Q + A^2}$ , we could obtain

$$b^{\top} Z_{2} - \frac{1}{2} |Z_{2} + r|^{2}$$

$$= \frac{1}{2} (Z_{1}b + q - Z_{1}r)^{\top} (Q + A^{2})^{-1} (Z_{1}b - Z_{1}r - q - 2Ab + 2Ar) - \frac{1}{2}r^{\top}r$$

$$= \frac{1}{2} \left( \left( A + \sqrt{Q + A^{2}} \right) (b - r) + q \right)^{\top} (Q + A^{2})^{-1} \left( \left( A + \sqrt{Q + A^{2}} \right) (b - r) - q - 2Ab + 2Ar \right) - \frac{1}{2}r^{\top}r$$

$$= -\frac{1}{2} (Ab + q - Ar)^{\top} (Q + A^{2})^{-1} (Ab + q - Ar) + \frac{1}{2} (b - r)^{\top} (b - r) - \frac{1}{2}r^{\top}r$$

$$= -\frac{1}{2} (Ab + q - Ar)^{\top} (Q + A^{2})^{-1} (Ab + q - Ar) + \frac{1}{2}b^{\top}b - r^{\top}b,$$

which yields the desired result.

### 2.3 Turnpike property

In this section, we uncover the turnpike property applicable to the optimal trajectory  $X^*$  and optimal control  $u^*$ , both stemming from the probabilistic cell problem as outlined in (2.2.5)-(2.2.6). Moreover, distinct from the aforementioned turnpike property elucidated by (2.1.5), we also establish the turnpike behavior concerning the cost function:

$$\lim_{T \to \infty} \frac{1}{T} J_T(x; u^*) = -c_*,$$

where  $-c_* := \lim_{T\to\infty} \frac{1}{T} J_T(x; u_T^*)$  is the ergodic cost defined in (2.1.7). In other words, achieving a near optimality in terms of the average cost over an extended period doesn't require calculating the optimal control  $u_T^*$  for every T. Instead, one only needs to compute the optimal control  $u^*$  for the probabilistic cell problem.

To accomplish this objective, it becomes imperative to establish a relation between the cost function  $J_T(x; u^*)$  with the  $u^*$  from the probabilistic cell problem (2.2.5)-(2.2.6), and the corresponding function  $V_T(x) = J_T(x, u_T^*)$  originating from the finite time stochastic control problem

(2.1.3).

#### 2.3.1 Main results on turnpike properties

To distinguish the finite time control problem (2.1.3) from the probabilistic cell problem (2.2.5)-(2.2.6), we denote by  $\{X_T(t)\}_{0 \le t \le T}$  as the underlying process controlled by finite time control  $\{u_T(t)\}_{0 \le t \le T}$  in the finite time stochastic control problem, while denote by  $\{X(t)\}_{t \ge 0}$  as the underlying process controlled by infinite time control  $\{u(t)\}_{t \ge 0}$  in the probabilistic cell problem. In other words,  $\{X_T(t)\}_{0 \le t \le T}$  follows

$$dX_T(t) = (AX_T(t) + u_T(t) + b) dt + \sigma dW(t), \quad t \in [0, T]$$

and  $\{X(t)\}_{t\geq 0}$  follows

$$dX(t) = (AX(t) + u(t) + b) dt + \sigma dW(t), \quad t \ge 0.$$

We also extend the cost functional for the finite time control problem (2.1.3) to an initial condition  $X_T(t) = x$ , that is

$$J_T(t,x;u_T) := \mathbb{E}\left[\int_t^T L(X_T(s),u_T(s))ds \,\middle|\, X_T(t) = x\right],$$

where  $u_T \in \mathcal{U}_{[t,T]}$  is an admissible control between t and T and the value function is

$$V_T(t,x) := J_T(t,x;u_T^*) = \inf_{u_T \in \mathcal{U}_{[t,T]}} J_T(t,x;u_T)$$

with its optimal control and optimal path denoted by  $u_T^*$  and  $X_T^*$ .

We recall that the value function V(x) of the probabilistic cell problem is defined in (2.2.5)-(2.2.6) as

$$-c_0 = \inf_{u \in \mathcal{U}} \limsup_{T \to \infty} \frac{1}{T} \int_0^T \mathbb{E} \left[ L(X(t), u(t)) \right] dt$$

and

$$V(x) := \inf_{u \in \mathcal{U}[\mu_{\infty}^*]} \limsup_{T \to \infty} \int_0^T \mathbb{E} \left[ L\left(X_t, u_t\right) + c_0 \right] dt,$$

where  $\mu_{\infty}^*$  is the distribution limit of the optimal path  $X^*$  in the ergodic control problem (2.2.5). The optimal control and optimal path of the probabilistic cell problem is denoted by  $u^*$  and  $X^*$  respectively, which are characterized by Theorem 6. Note that, the constant c in the above definition is replaced by  $c_0$  of the cell problem corresponding to  $Z_1$  such that  $D_1$  is positive definite according to Theorem 6. Next, we present our main results.

**Theorem 9.** Let  $J_T(0, x; u^*)$  be the cost functional evaluated along the optimal control of the
probabilistic cell problem  $u^*$  on the finite time horizon [0, T],

$$J_T(0, x; u^*) := \mathbb{E}\left[\int_0^T L(X_T(s), u^*(s)) \, ds\right].$$

For all  $x \in \mathbb{R}^d$ , the following estimation holds:

$$0 \le J_T(0, x; u^*) - V_T(0, x) = O(1).$$

Moreover, the constant  $c_0$  of the cell problem (2.1.6) is the ergodic cost defined via (2.1.7), i.e.,

$$c_0 = \lim_{T \to \infty} \frac{1}{T} V_T(0, x), \quad \forall x \in \mathbb{R}^d.$$

From the results of Theorem 9, we could establish the turnpike behavior in terms of the average cost function straightforwardly. Note that  $V_T(0, x) \leq J_T(0, x; u^*) = V_T(0, x) + O(1)$ , it follows that

$$\lim_{T \to \infty} \frac{1}{T} J_T(0, x; u^*) = \lim_{T \to \infty} \frac{1}{T} V_T(0, x) = -c_*.$$
(2.3.1)

**Theorem 10.** For all  $x \in \mathbb{R}^d$ , there exist some  $\lambda > 0$  and K > 0 independent of t and T such that

$$\mathbb{E}\left[\left|X_{T}^{*}(t) - X^{*}(t)\right|^{2} + \left|u_{T}^{*}(t) - u^{*}(t)\right|^{2}\right] \le K\left(e^{-\lambda t} + e^{-\lambda(T-t)}\right), \quad \forall t \in [0,T].$$

## 2.3.2 Proofs

The proofs of these two theorems rely on the analytical expressions of the value functions for both the finite time control problem and probabilistic cell problem, and some comparison results between solutions to the Riccati system of ODEs. We first present some preliminary lemmas with their proof posted in Appendix 2.5. The first lemma gives an analytical expression for the value function of the finite time control problem.

**Lemma 11.** The value function  $V_T(t, x)$  of the finite time control problem has the following form

$$V_T(t,x) = \frac{1}{2}x^{\top}\tilde{Z}_1(t)x + x^{\top}\tilde{Z}_2(t) + \tilde{Z}_3(t),$$

where  $\{\tilde{Z}_1(t), \tilde{Z}_2(t), \tilde{Z}_3(t) : t \in [0, T]\}$  solves the Riccati system of ODEs

$$\begin{cases} \tilde{Z}_{1}(t) + 2A\tilde{Z}_{1}(t) - \tilde{Z}_{1}^{2}(t) + Q = \mathbf{O}_{d}; \\ \dot{\tilde{Z}}_{2}(t) + A\tilde{Z}_{2}(t) - \tilde{Z}_{1}(t)\tilde{Z}_{2}(t) + \tilde{Z}_{1}(t)(b-r) + q = \mathbf{0}_{d}; \\ \dot{\tilde{Z}}_{3}(t) + \frac{1}{2}\sigma^{2}trace(\tilde{Z}_{1}(t)) + b^{\top}\tilde{Z}_{2}(t) - \frac{1}{2}|\tilde{Z}_{2}(t) + r|^{2} = 0, \end{cases}$$

$$(2.3.2)$$

with terminal conditions  $\tilde{Z}_1(T) = \mathbf{O}_d$ ,  $\tilde{Z}_2(T) = \mathbf{O}_d$ ,  $\tilde{Z}_3(T) = 0$ . Moreover, the optimal feedback

control of the finite time control problem is given by

$$u_T^*(t) = -\tilde{Z}_1(t)X_T^*(t) - \tilde{Z}_2(t) - r.$$

Next lemma gives a similar analytical structure for  $J_T(t, x; u^*)$ , which is the cost functional of the finite time control problem evaluated along the optimal control  $u^*$  from the probabilistic cell problem. Recall that  $(Z_1, Z_2)$  is the solution to the system of algebraic equations (2.2.3) in which  $Z_1$  is uniquely chosen such that  $D_1$  to be positive definite, see Lemma 2.

**Lemma 12.** The cost functional  $J_T(t, x; u^*)$  of the finite time control problem evaluated along the optimal control  $u^*$  from the probabilistic cell problem has the form

$$J_T(t, x; u^*) = \frac{1}{2} x^\top f_1(t) x + x^\top f_2(t) + f_3(t),$$

where  $\{f_1(t), f_2(t), f_3(t) : t \in [0, T]\}$  solves the Riccati system of ODEs

$$\begin{cases} \dot{f}_1(t) - 2(Z_1 - A)f_1(t) + Q + Z_1^2 = \mathbf{O}_d; \\ \dot{f}_2(t) - (Z_1 - A)f_2(t) + f_1(t)(b - r - Z_2) + Z_1Z_2 + q = \mathbf{O}_d; \\ \dot{f}_3(t) + \frac{1}{2}\sigma^2 trace(f_1(t)) + (b - r - Z_2)^\top f_2(t) + \frac{1}{2}(|Z_2|^2 - |r|^2) = 0, \end{cases}$$
(2.3.3)

with terminal conditions  $f_1(T) = \mathbf{O}_d$ ,  $f_2(T) = \mathbf{O}_d$ ,  $f_3(T) = 0$ .

From the above lemmas, we can observe that

$$J_T(0, x; u^*) - V_T(0, x) = \frac{1}{2} x^\top \Gamma_1(0) x + x^\top \Gamma_2(0) + \Gamma_3(0), \qquad (2.3.4)$$

where, for i = 1, 2, 3,

$$\Gamma_i(t) = f_i(t) - \tilde{Z}_i(t), \ i = 1, 2, 3$$

If we introduce  $\gamma_i$  by

$$\gamma_i(t) = Z_i - Z_i(t), \ i = 1, 2,$$

 $\{\Gamma_i : i = 1, 2, 3\}$  satisfies the following system of ODE

$$\begin{cases} \dot{\Gamma}_{1}(t) - 2(Z_{1} - A)\Gamma_{1}(t) + \gamma_{1}^{2}(t) = \mathbf{O}_{d}; \\ \dot{\Gamma}_{2}(t) - (Z_{1} - A)\Gamma_{2}(t) + \Gamma_{1}(t)(b - r - Z_{2}) + \gamma_{1}(t)\gamma_{2}(t) = \mathbf{0}_{d}; \\ \dot{\Gamma}_{3}(t) + \frac{1}{2}\sigma^{2}\mathrm{trace}(\Gamma_{1}(t)) + (b - r - Z_{2})^{\top}\Gamma_{2}(t) + \frac{1}{2}|\gamma_{2}(t)|^{2} = 0, \end{cases}$$

$$(2.3.5)$$

with the terminal conditions  $\Gamma_1(T) = \mathbf{O}_d$ ,  $\Gamma_2(T) = \mathbf{O}_d$ ,  $\Gamma_3(T) = 0$ . Moreover,  $\{\gamma_1, \gamma_2\}$  is the solution

to the system of ODEs

$$\begin{cases} \dot{\gamma}_1(t) - 2(Z_1 - A)\gamma_1(t) + \gamma_1^2(t) = \mathbf{O}_d; \\ \dot{\gamma}_2(t) - (Z_1 - A)\gamma_2(t) + \gamma_1(t)\gamma_2(t) + \gamma_1(t)(b - r - Z_2) = \mathbf{0}_d, \end{cases}$$
(2.3.6)

with the terminal conditions  $\gamma_1(T) = Z_1$  and  $\gamma_2(T) = Z_2$ .

It is enough to give a proper estimations for the  $\gamma_1, \gamma_2$  in (2.3.6) and  $\Gamma_1, \Gamma_2, \Gamma_3$  in (2.3.5) to obtain the bound for the difference between  $J_T(0, x; u^*)$  and  $V_T(0, x)$ .

To proceed, we recall that

$$D_1 = Z_1 - A > 0, \quad D_2 = b - r - Z_2.$$

**Lemma 13.** Let  $\lambda_1 \geq \lambda_2 \geq \cdots \geq \lambda_d > 0$  be eigenvalues of  $D_1$  in a descending order. Then, there exists a unique solution  $\gamma_1, \gamma_2 \in C^1([0,T])$  of the system of ODEs (2.3.6) satisfying, for some constant k > 0,

$$\|\gamma_1(t)\|_2 \le ke^{-2\lambda_d(T-t)}, \ |\gamma_2(t)| \le ke^{-\lambda_d(T-t)}, \ \forall t \in [0,T].$$
 (2.3.7)

Moreover,  $\gamma_1(t)$  is positive definite for all  $t \in [0, T]$  and  $\gamma_1$  is an increasing function with respect to Loewner order on [0, T].

*Proof.* ODE satisfied by  $\gamma_1$  can be rewritten by

$$\dot{\gamma}_1(t) - 2D_1\gamma_1(t) + \gamma_1^2(t) = \mathbf{O}_d, \quad \gamma_1(T) = Z_1.$$

Let  $\tau = T - t$  and denote  $\bar{\gamma}_1(\tau) = \gamma_1(t)$ ,  $\bar{\gamma}_1$  is the solution to the ODE

$$\dot{\bar{\gamma}}_1(t) = -2D_1\bar{\gamma}_1(t) + \bar{\gamma}_1^2(t), \quad \bar{\gamma}_1(0) = Z_1.$$

Recall that  $D_1 = \tilde{Q}\Lambda \tilde{Q}^{\top}$ , where  $\tilde{Q}$  is an orthogonal matrix and  $\Lambda$  is a diagonal matrix with descending diagonal entries  $\lambda_i$ . Since  $Z_1$  and  $\tilde{Z}_1(t)$  are symmetric matrices,  $\bar{\gamma}_1(t)$  is also symmetric as their difference. Taking transpose to the equation satisfied by  $\bar{\gamma}_1$ , we have  $\dot{\bar{\gamma}}_1(t) = -2\bar{\gamma}_1(t)D_1 + \bar{\gamma}_1^2(t)$ , which implies that  $D_1$  and  $\bar{\gamma}_1(t)$  are commute, i.e.,  $D_1\bar{\gamma}_1(t) = \bar{\gamma}_1(t)D_1$ . Thus,  $\bar{\gamma}_1$  and  $D_1$ share the same eigenvector matrix  $\tilde{Q}$  by the results in Page 305 of [73].

Hence, we can write  $\bar{\gamma}_1(t) = \tilde{Q}\Sigma(t)\tilde{Q}^{\top}$  for some  $\Sigma(t) = \text{diag}(\sigma_1(t), \sigma_2(t), \dots, \sigma_d(t))$  for  $t \ge 0$ . It follows that

$$\dot{\Sigma}(t) = -2\Lambda\Sigma(t) + \Sigma^2(t), \quad \forall t > 0$$
(2.3.8)

with initial condition  $\bar{\gamma}_1(0) = \tilde{Q}\Sigma(0)\tilde{Q}^{\top} = Z_1$ . It is equivalent to write

$$\dot{\sigma}_i(t) = -2\lambda_i \sigma_i(t) + \sigma_i^2(t), \quad t > 0, \ i = 1, 2, \dots, d_i$$

Note that  $2D_1 - Z_1 = Z_1 - 2A = \sqrt{A^2 + Q} - A > 0$  and  $2D_1 - Z_1 = \tilde{Q}(2\Lambda - \Sigma(0))\tilde{Q}^{\top}$ , thus  $2\Lambda - \Sigma(0) > 0$ , i.e.,  $\sigma_i(0) < 2\lambda_i$  for i = 1, 2, ..., d. Therefore, there exists unique solution  $\{\sigma_i : i = 1, 2, ..., d\}$ .

 $1, 2, \ldots, d$  in the form of

$$0 < \sigma_i(t) = \frac{2\lambda_i}{1 + \left(\frac{2\lambda_i}{\sigma_i(0)} - 1\right)e^{2\lambda_i t}} < \max_{i=1,2,\dots,d} \left(\frac{2\lambda_i \sigma_i(0)}{2\lambda_i - \sigma_i(0)}\right)e^{-2\lambda_i t} := a_1 e^{-2\lambda_i t}, \quad \forall t \ge 0.$$

Clearly,  $\sigma_i$  is strictly decreasing on  $[0, \infty)$ , which implies that  $\bar{\gamma}_1$  is a strictly decreasing function on  $[0, \infty)$  with respect to Loewner order. Moreover, since  $\lambda_i > 0$  for all  $i = 1, 2, \ldots, d, \bar{\gamma}_1(t)$  is positive definite for all  $t \ge 0$ . Thus, we have

$$\|\bar{\gamma}_1(t)\|_2 = \|\Sigma(t)\|_2 \le a_1 e^{-2\lambda_d t}, \quad \forall t \ge 0,$$
(2.3.9)

which yields the desired result that

$$\|\gamma_1(t)\|_2 \le a_1 e^{-2\lambda_d(T-t)}, \quad t \in [0,T].$$
 (2.3.10)

Moreover,  $\gamma_1(t)$  is positive definite for all  $t \in [0, T]$  and  $\gamma_1$  is an increasing function with respect to Loewner order on [0, T].

By (2.3.6),  $\gamma_2$  satisfies the ODE

$$\dot{\gamma}_2(t) + (\gamma_1(t) - D_1)\gamma_2(t) + \gamma_1(t)D_2 = \mathbf{0}_d, \quad \gamma_2(T) = Z_2.$$

Let  $\tau = T - t$  and denote  $\bar{\gamma}_2(\tau) = \gamma_2(t)$ ,  $\bar{\gamma}_2$  is the solution to the ODE

$$\dot{\bar{\gamma}}_2(t) = (\bar{\gamma}_1(t) - D_1)\bar{\gamma}_2(t) + \bar{\gamma}_1(t)D_2, \quad \bar{\gamma}_2(0) = Z_2.$$

Denote that  $A_1(t) = \bar{\gamma}_1(t) - D_1$  for  $t \ge 0$ , the explicit form of  $\bar{\gamma}_2$  is given by

$$\bar{\gamma}_2(t) = e^{\int_0^t A_1(s)ds} Z_2 + \int_0^t e^{\int_s^t A_1(r)dr} \bar{\gamma}_1(s) D_2 ds$$
(2.3.11)

for all  $t \ge 0$ , which implies that  $\gamma_2 \in C^1([0,T])$ .

To proceed, we first observe from (2.3.9) that

$$\left\|\int_0^t \Sigma(s) ds\right\|_2 \le \int_0^t \|\Sigma(s)\|_2 ds \le \frac{a_1}{2\lambda_d} (1 - e^{-2\lambda_d t}) \le \frac{a_1}{2\lambda_d}$$

From the fact that  $||e^D||_2 = e^{||D||_2}$  for any  $D \ge 0$ , it implies that

$$\left\| e^{\int_0^t \Sigma(s) ds} \right\|_2 \le a_2 := \exp\left\{\frac{a_1}{2\lambda_d}\right\}.$$

Hence, we have

$$\left\| e^{\int_0^t A_1(s)ds} \right\|_2 = \left\| e^{\int_0^t (\Sigma(s) - \Lambda)ds} \right\|_2 \le \left\| e^{\int_0^t \Sigma(s)ds} \right\|_2 \left\| e^{-\Lambda t} \right\|_2 \le a_2 e^{-\lambda_d t}.$$
 (2.3.12)

Therefore, the estimate of (2.3.11) is

$$\begin{aligned} |\bar{\gamma}_{2}(t)| &\leq \left\| e^{\int_{0}^{t} A_{1}(s)ds} \right\|_{2} |Z_{2}| + \int_{0}^{t} \left\| e^{\int_{s}^{t} A_{1}(r)dr} \right\|_{2} \|\bar{\gamma}_{1}(s)\|_{2} |D_{2}|ds \\ &\leq a_{2}e^{-\lambda_{d}t}|Z_{2}| + \int_{0}^{t} a_{2}e^{-\lambda_{d}(t-s)}a_{1}e^{-2\lambda_{d}s}ds|D_{2}| \\ &\leq \left(a_{2}|Z_{2}| + \frac{a_{1}a_{2}|D_{2}|}{\lambda_{d}}\right)e^{-\lambda_{d}t} := a_{3}e^{-\lambda_{d}t} \end{aligned}$$

for all  $t \ge 0$ . Thus, we obtain that

$$|\gamma_2(t)| \le a_3 e^{-\lambda_d(T-t)}, \quad \forall t \in [0,T].$$

$$(2.3.13)$$

**Lemma 14.** With eigenvalues of  $D_1$  denoted by Lemma 13, The system of ODEs (2.3.5) has a unique solution  $\Gamma_1, \Gamma_2$  and  $\Gamma_3 \in C^1([0,T])$ , and satisfies the following properties for some constant k > 0:

$$\|\Gamma_1(t)\|_2 \le ke^{-2\lambda_d(T-t)}, \quad |\Gamma_2(t)| \le ke^{-\lambda_d(T-t)}, \quad |\Gamma_3(t)| \le k$$
 (2.3.14)

for all  $t \in [0, T]$ .

*Proof.* We recall that,  $\Gamma_1$  satisfies the ODE

$$\dot{\Gamma}_1(t) - 2D_1\Gamma_1(t) + \gamma_1^2(t) = \mathbf{O}_d, \quad \Gamma_1(T) = \mathbf{O}_d,$$

which can be written in terms of  $\gamma_1$  in the form of

$$\Gamma_1(t) = \int_t^T e^{2D_1(t-s)} \gamma_1^2(s) ds, \quad \forall t \in [0,T].$$

From (2.3.10), the estimation of  $\Gamma_1$  is

$$\|\Gamma_1(t)\|_2 \le \int_t^T e^{2\lambda_d(t-s)} \|\gamma_1(s)\|_2^2 ds \le \frac{a_1^2}{2\lambda_d} e^{-2\lambda_d(T-t)}$$
(2.3.15)

for all  $t \in [0, T]$ . Moreover, from the explicit form of  $\Gamma_1$ , it is clear that  $\Gamma_1(t)$  is positive semi-definite for all  $t \in [0, T]$ . Similarly, the ODE for  $\Gamma_2$  is

$$\dot{\Gamma}_2(t) - D_1\Gamma_2(t) + \Gamma_1(t)D_2 + \gamma_1(t)\gamma_2(t) = \mathbf{0}_d, \quad \Gamma_2(T) = \mathbf{0}_d,$$

which yields an expression in terms of  $\gamma_1, \gamma_2$  and  $\Gamma_1$ :

$$\Gamma_2(t) = \int_t^T e^{D_1(t-s)} \left( \Gamma_1(s) D_2 + \gamma_1(s) \gamma_2(s) \right) ds, \quad \forall t \in [0,T].$$

By the estimation (2.3.10), (2.3.13), and (2.3.15), for all  $t \in [0, T]$ ,

$$\begin{aligned} |\Gamma_{2}(t)| &\leq \int_{t}^{T} \left\| e^{D_{1}(t-s)} \right\|_{2} \left( \|\Gamma_{1}(s)\|_{2} |D_{2}| + \|\gamma_{1}(s)\|_{2} |\gamma_{2}(s)| \right) ds \\ &\leq \int_{t}^{T} e^{\lambda_{d}(t-s)} \left( \frac{a_{1}^{2}}{2\lambda_{d}} e^{-2\lambda_{d}(T-s)} |D_{2}| + a_{1}a_{3}e^{-3\lambda_{d}(T-s)} \right) ds \\ &\leq \frac{a_{1}^{2} |D_{2}| + a_{1}a_{3}\lambda_{d}}{2\lambda_{d}^{2}} e^{-\lambda_{d}(T-t)} := a_{4}e^{-\lambda_{d}(T-t)}. \end{aligned}$$

$$(2.3.16)$$

Next, the term  $\Gamma_3$  satisfies the ODE

$$\dot{\Gamma}_3(t) + \frac{1}{2}\sigma^2 \operatorname{trace}(\Gamma_1(t)) + D_2^{\top}\Gamma_2(t) + \frac{1}{2}|\gamma_2(t)|^2 = 0, \quad \Gamma_3(T) = 0,$$

which can be rewritten with the above  $\gamma_2, \Gamma_1$  and  $\Gamma_2$ :

$$\Gamma_{3}(t) = \int_{t}^{T} \left( \frac{1}{2} \sigma^{2} \operatorname{trace}(\Gamma_{1}(s)) + D_{2}^{\top} \Gamma_{2}(s) + \frac{1}{2} |\gamma_{2}(s)|^{2} \right) ds, \quad \forall t \in [0, T].$$

Thus, we obtain the estimation for  $\Gamma_3$  with the help of (2.3.13), (2.3.15), and (2.3.16)

$$|\Gamma_3(t)| \le \int_t^T \left(\frac{1}{2} d\sigma^2 \|\Gamma_1(s)\|_2 + |D_2| |\Gamma_2(s)| + \frac{1}{2} |\gamma_2(s)|^2\right) ds \le a_5$$
(2.3.17)

for some constant  $a_5 > 0$ .

Now, we are well prepared to prove Theorem 9.

Proof of Theorem 9. From (2.3.14) in Lemma 14, we have

$$|J_T(0,x;u^*) - V_T(0,x)| = \left|\frac{1}{2}x^{\top}\Gamma_1(0)x + x^{\top}\Gamma_2(0) + \Gamma_3(0)\right|$$
  
$$\leq \frac{1}{2}ke^{-2\lambda_d T}|x|^2 + ke^{-\lambda_d T}|x| + k.$$

Hence, for all  $x \in \mathbb{R}^d$ , we obtain that  $J_T(0, x; u^*) - V_T(0, x) = O(1)$ . Since  $u_T^*$  is the optimal control of the finite time control problem, by the definition of  $V_T(t, x)$  and  $J_T(t, x; u)$ , we have  $0 \leq J_T(0, x; u^*) - V_T(0, x)$  for all  $x \in \mathbb{R}^d$ . Therefore, we obtain the desired result.

Next, the estimations of  $\gamma_1$  and  $\gamma_2$  in (2.3.7) can help us to verify that the identity  $V_T(0, x) + c_0T = o(T)$  holds, where  $c_0$  is given by (2.2.2) and can also be obtained from the solution to the cell problem. Note that  $V_T(0, x) = \frac{1}{2}x^{\top}\tilde{Z}_1(0)x + x^{\top}\tilde{Z}_2(0) + \tilde{Z}_3(0)$ , where  $\{\tilde{Z}_1(t), \tilde{Z}_2(t), \tilde{Z}_3(t) : t \in [0, T]\}$  is the solution to the system of Riccati equation (2.3.2). Then, we only need to show that for all  $x \in \mathbb{R}^d$ 

$$\lim_{T \to \infty} \frac{1}{T} \left| \frac{1}{2} x^{\top} \tilde{Z}_1(0) x + x^{\top} \tilde{Z}_2(0) + \tilde{Z}_3(0) + c_0 T \right| = 0.$$

From (2.3.7) in Lemma 13, we have the following inequalities

$$\left\|\tilde{Z}_{1}(0)\right\|_{2} = \|Z_{1} - \gamma_{1}(0)\|_{2} \le \|Z_{1}\|_{2} + \|\gamma_{1}(0)\|_{2} \le \|Z_{1}\|_{2} + ke^{-2\lambda_{d}T}$$

and

$$\left|\tilde{Z}_{2}(0)\right| = |Z_{2} - \gamma_{2}(0)| \le |Z_{2}| + |\gamma_{2}(0)| \le |Z_{2}| + ke^{-\lambda_{d}T},$$

which implies that

$$\lim_{T \to \infty} \frac{1}{T} \left| \frac{1}{2} x^\top \tilde{Z}_1(0) x + x^\top \tilde{Z}_2(0) \right| = 0, \quad \forall x \in \mathbb{R}^d.$$

Moreover, from the ODE satisfied by  $Z_3(t)$  in (2.3.2),

$$\tilde{Z}_{3}(0) = \int_{0}^{T} \left( \frac{1}{2} \sigma^{2} \operatorname{trace} \left( \tilde{Z}_{1}(s) \right) + b^{\top} \tilde{Z}_{2}(s) - \frac{1}{2} \left| \tilde{Z}_{2}(s) + r \right|^{2} \right) ds$$
$$= \int_{0}^{T} \left( -\frac{1}{2} \sigma^{2} \operatorname{trace} \left( \gamma_{1}(s) \right) - D_{2}^{\top} \gamma_{2}(s) - \frac{1}{2} \left| \gamma_{2}(s) \right|^{2} \right) ds - c_{0} T,$$

where  $c_0 = -\frac{1}{2}\sigma^2 \operatorname{trace}(Z_1) - b^\top Z_2 + \frac{1}{2}|Z_2 + r|^2$ . Then, it follows that

$$\lim_{T \to \infty} \frac{1}{T} \left| \tilde{Z}_3(0) + c_0 T \right| = \lim_{T \to \infty} \frac{1}{T} \left| \int_0^T \left( -\frac{1}{2} \sigma^2 \operatorname{trace}\left(\gamma_1(s)\right) - D_2^\top \gamma_2(s) - \frac{1}{2} \left| \gamma_2(s) \right|^2 \right) ds \right|.$$

Using the triangle inequality and the estimations (2.3.7), we have

$$\begin{split} \lim_{T \to \infty} \frac{1}{T} \left| \int_0^T \left( -\frac{1}{2} \sigma^2 \operatorname{trace} \left( \gamma_1(s) \right) - D_2^\top \gamma_2(s) - \frac{1}{2} |\gamma_2(s)|^2 \right) ds \right| \\ &\leq \lim_{T \to \infty} \frac{1}{T} \int_0^T \left( \frac{1}{2} d\sigma^2 ||\gamma_1(s)||_2 + |D_2| |\gamma_2(s)| + \frac{1}{2} |\gamma_2(s)|^2 \right) ds \\ &\leq \lim_{T \to \infty} \frac{1}{T} \frac{dk \sigma^2 + 4 |D_2| k + k^2}{4\lambda_d} = 0, \end{split}$$

which implies that  $\lim_{T\to\infty} \frac{1}{T} |\tilde{Z}_3(0) + c_0 T| = 0$ . Therefore, we obtain the desired result that

$$\lim_{T \to \infty} \frac{1}{T} |V_T(0, x) + c_0 T| = 0, \quad \forall x \in \mathbb{R}^d,$$

i.e.,  $V_T(0, x) + c_0 T = o(T)$  for all  $x \in \mathbb{R}^d$  with  $c_0$  given by (2.2.2). The identical result between  $c_0$  and  $c_*$  yields from Definition (2.1.7).

Recall Theorem 6: From (2.2.14) and (2.2.13), the optimal control  $u^*$  of the probabilistic cell problem is

$$u^*(t) = -Z_1 X^*(t) - Z_2 - r,$$

and the optimal path  $X^*$  of the probabilistic cell problem is given by

$$dX^*(t) = -D_1X^*(t)dt + D_2dt + \sigma dW(t),$$

with  $X^*(0) = x$ , where  $D_1$  and  $D_2$  are given by (2.2.12). Moreover, the optimal path  $X^*(t)$  in the probabilistic cell problem (2.2.5)-(2.2.6) converges in distribution to a normal random variable  $\bar{X} \sim \mathcal{N}(m_{\infty}, \nu_{\infty})$  as  $t \to \infty$ , where  $m_{\infty} = D_1^{-1}D_2$  and  $\nu_{\infty} = \frac{1}{2}\sigma^2 D_1^{-1}$ . Next, we prove the classical turnpike property that is described in Theorem 10.

Proof of Theorem 10. Here we assume the optimal path  $X^*$  in (2.2.13) of the probabilistic cell problem has the initial point  $X^*(0) = \overline{X} \sim \mathcal{N}(m_{\infty}, \nu_{\infty})$ , a normal random variable independent to the Brownian motion W, instead of a real-valued vector  $x \in \mathbb{R}^d$ . To calculate  $\mathbb{E}[|X_T^*(t) - X^*(t)|^2]$ , we first observe that the optimal control of the finite time control problem is given by

$$u_T^*(t) = -\tilde{Z}_1(t)X_T^*(t) - \tilde{Z}_2(t) - r$$

from the results in Lemma 11. Thus the optimal path for the finite time control problem satisfies

$$dX_T^*(t) = \left(A - \tilde{Z}_1(t)\right) X_T^*(t) dt + \left(b - r - \tilde{Z}_2(t)\right) dt + \sigma dW(t).$$
(2.3.18)

Denote that  $\delta_T(t) = X_T^*(t) - X^*(t)$ , by (2.2.13) and (2.3.18), we have

$$d\delta_T(t) = \left(A - \tilde{Z}_1(t)\right)\delta_T(t)dt + \gamma_1(t)X^*(t)dt + \gamma_2(t)dt$$

with the initial value  $\delta_T(0) = X_T^*(0) - X^*(0) = x - \bar{X} \sim \mathcal{N}(x - m_\infty, \nu_\infty).$ 

Let  $\bar{A}_1(t) = A - \tilde{Z}_1(t) = \gamma_1(t) - D_1$  for all  $t \in [0, T]$ , then we have  $\bar{A}_1(t) = A_1(T - t)$ . By the similar method as the estimation in (2.3.12), for  $0 \le s \le t \le T$ , we have

$$\left\| e^{\int_0^t \bar{A}_1(r)dr} \right\|_2 \le a_2 e^{-\lambda_d t} \text{ and } \left\| e^{\int_s^t \bar{A}_1(r)dr} \right\|_2 \le a_2 e^{-\lambda_d(t-s)}.$$

Applying the integrating factor method, we obtain the explicit form of  $\delta_T(t)$  as

$$\delta_T(t) = e^{\int_0^t \bar{A}_1(r)dr} \delta_T(0) + \int_0^t e^{\int_s^t \bar{A}_1(r)dr} \gamma_1(s) X^*(s) ds + \int_0^t e^{\int_s^t \bar{A}_1(r)dr} \gamma_2(s) ds.$$

Therefore, we have the following estimation:

$$\mathbb{E}\left[|\delta_{T}(t)|^{2}\right] \leq 3\left\|e^{\int_{0}^{t}\bar{A}_{1}(r)dr}\right\|_{2}^{2}\mathbb{E}\left[|\delta_{T}(0)|^{2}\right] + 3\mathbb{E}\left[\left|\int_{0}^{t}e^{\int_{s}^{t}\bar{A}_{1}(r)dr}\gamma_{1}(s)X^{*}(s)ds\right|^{2}\right] + 3\left|\int_{0}^{t}e^{\int_{s}^{t}\bar{A}_{1}(r)dr}\gamma_{2}(s)ds\right|^{2}.$$

Firstly, by calculation,

$$3 \left\| e^{\int_0^t \bar{A}_1(r)dr} \right\|_2^2 \mathbb{E} \left[ |\delta_T(0)|^2 \right] \le 3a_2^2 \mathbb{E} \left[ \left| x - \bar{X} \right|^2 \right] e^{-2\lambda_d t} \le K_1 e^{-2\lambda_d t}$$

for some constant  $K_1 \ge 3a_2^2 \mathbb{E}\left[|x - \bar{X}|^2\right]$ . Next, using the estimation for  $\gamma_2$  in (2.3.7) from Lemma

13 and by Hölder's inequality, we get

for some constant  $K_2 \geq \frac{3a_2^2k^2}{4\lambda_d^2}$ . Lastly, using the Hölder's inequality again, we have

$$3\mathbb{E}\left[\left|\int_{0}^{t} e^{\int_{s}^{t} \bar{A}_{1}(r)dr} \gamma_{1}(s)X^{*}(s)ds\right|^{2}\right] \leq 3\int_{0}^{t} \left\|e^{\int_{s}^{t} \bar{A}_{1}(r)dr}\right\|_{2}^{2} ds \ \mathbb{E}\left[\int_{0}^{t} \|\gamma_{1}(s)\|_{2}^{2} |X^{*}(s)|^{2} ds\right].$$

Similar with the calculation of expectation and variance of  $X^*$  in Subsection 2.2.2.3, we know the expectation and the variance of  $X^*(s)$  in (2.2.13) with the initial  $X^*(0) = \overline{X}$  is  $\mathbb{E}[X^*(s)] = D_1^{-1}D_2$  and  $\mathbb{V}ar(X^*(s)) = \frac{1}{2}\sigma^2 D_1^{-1}$  respectively, which implies  $\mathbb{E}[|X^*(s)|^2] = a_6$  for all s > 0 for some positive constant  $a_6$ . Thus, from the estimation of  $\gamma_1$  in Lemma 13, we obtain the estimation as follows

$$3\mathbb{E}\left[\left|\int_{0}^{t} e^{\int_{s}^{t} \bar{A}_{1}(r)dr} \gamma_{1}(s)X^{*}(s)ds\right|^{2}\right] \leq 3k^{2}a_{6}a_{2}^{2}\int_{0}^{t} e^{-2\lambda_{d}(t-s)}ds\int_{0}^{t} e^{-4\lambda_{d}(T-s)}ds \leq K_{3}e^{-4\lambda_{d}(T-t)}ds$$

for some constant  $K_3 \ge \frac{3k^2 a_6 a_2^2}{8\lambda_d^2}$ .

To summarize from the above estimation results, we have

$$\mathbb{E}\left[\left|\delta_{T}(t)\right|^{2}\right] \leq K_{1}e^{-2\lambda_{d}t} + K_{2}e^{-2\lambda_{d}(T-t)} + K_{3}e^{-4\lambda_{d}(T-t)} \\ \leq K_{4}\left(e^{-2\lambda_{d}t} + e^{-2\lambda_{d}(T-t)}\right)$$
(2.3.19)

for all  $t \in [0, T]$  for some positive constant  $K_4$ .

Similarly, from the optimal control of the probabilistic cell problem and the optimal control of the finite time control problem, it is clear that

$$u_T^*(t) - u^*(t) = \gamma_1(t)X^*(t) - \tilde{Z}_1(t)\delta_T(t) + \gamma_2(t),$$

which yields the inequality

$$\mathbb{E}\left[\left|u_{T}^{*}(t)-u^{*}(t)\right|^{2}\right] \leq 3\left(\left\|\gamma_{1}(t)\right\|_{2}^{2}\mathbb{E}\left[\left|X^{*}(t)\right|^{2}\right]+\left\|\tilde{Z}_{1}(t)\right\|_{2}^{2}\mathbb{E}\left[\left|\delta_{T}(t)\right|^{2}\right]+\left|\gamma_{2}(t)\right|^{2}\right)\right.$$

Due to the facts that  $\gamma_1(t) = Z_1 - \tilde{Z}_1(t)$  and  $\|\gamma_1(t)\|_2 \leq ke^{-2\lambda_d(T-t)}$ , we have  $\|\tilde{Z}_1(t)\|_2 \leq \|Z_1\|_2 + ke^{-2\lambda_d(T-t)} \leq \|Z_1\|_2 + k$  for all  $t \in [0, T]$ . Applying the estimations (2.3.7) in Lemma 13 and the inequality (2.3.19), we get the estimation

$$\mathbb{E}\left[\left|u_{T}^{*}(t)-u^{*}(t)\right|^{2}\right] \leq K_{5}\left(e^{-2\lambda_{d}t}+e^{-2\lambda_{d}(T-t)}\right)$$

for all  $t \in [0,T]$  for some  $K_5 > 0$  independent of t and T. Therefore, let  $\lambda = 2\lambda_d$ , the classical turnpike property

$$\mathbb{E}\left[|X_T^*(t) - X^*(t)|^2 + |u_T^*(t) - u^*(t)|^2\right] \le K\left(e^{-\lambda t} + e^{-\lambda(T-t)}\right), \quad \forall t \in [0,T]$$
  
ed for some  $K > K_4 + K_5$ .

is obtained for some  $K \ge K_4 + K_5$ .

**Remark 15.** In the above proof, we show the turnpike property between  $X_T^*(t)$  and the optimal path  $X^*(t)$  in (2.2.13) for the probabilistic cell problem taking an initial  $\overline{X} \sim \mathcal{N}(m_{\infty}, \nu_{\infty})$  instead of a real-valued vector  $x \in \mathbb{R}^d$ . The reason for taking this normal random variable as the initial is that the optimal path  $X^*(t)$  starting with any constant initial  $x \in \mathbb{R}^d$  converges in distribution to a normal random variable  $\bar{X} \sim \mathcal{N}(m_{\infty}, \nu_{\infty})$  as  $t \to \infty$  from (2.2.15). We refer the asymptomatic  $\bar{X} \sim \mathcal{N}(m_{\infty}, \nu_{\infty})$  to be the equilibrium point of  $X^*(t)$  in (2.2.13). The proof of Theorem 10 when  $X^*$  taking an initial value  $x \in \mathbb{R}^d$  follows a similar approach.

#### $\mathbf{2.4}$ Example

In this section, we give an example to illustrate the results in Theorem 9 and Theorem 10. In this example, we consider the case when d = 1.

Let A = 0, b = 0, Q = 1, q = 0, r = -1. Then, the underlying process (2.1.1) is reduced to

$$dX(t) = u(t)dt + \sigma dW(t), \quad X_0 = x,$$

and the cost functional (2.1.2) can be simplified as following

$$J_T(0,x;u) = \mathbb{E}\left[\frac{1}{2}\int_0^T \left(X^2(t) + u^2(t) - 2u(t)\right)dt\right].$$

Firstly, we verify the result in Theorem 9. It is clear that  $Z_1 = 1$  and  $Z_2 = 1$  when we take  $Z_1 > 0$  from the results in Lemma 1, and  $\{\tilde{Z}_1(t), \tilde{Z}_2(t), \tilde{Z}_3(t) : t \in [0,T]\}$  is the solution to the system of ODEs

$$\begin{cases} \tilde{Z}_{1}(t) - \tilde{Z}_{1}^{2}(t) + 1 = 0; \\ \dot{\tilde{Z}}_{2}(t) - \tilde{Z}_{1}(t)\tilde{Z}_{2}(t) + \tilde{Z}_{1}(t) = 0; \\ \dot{\tilde{Z}}_{3}(t) + \tilde{Z}_{2}(t) - \frac{1}{2}\tilde{Z}_{2}^{2}(t) + \frac{1}{2}\sigma^{2}\tilde{Z}_{1}(t) - \frac{1}{2} = 0; \\ \tilde{Z}_{1}(T) = \tilde{Z}_{2}(T) = \tilde{Z}_{3}(T) = 0. \end{cases}$$

By calculation, we obtain the solution to the above system of ODEs as follows

$$\tilde{Z}_1(t) = \frac{1 - e^{2t - 2T}}{1 + e^{2t - 2T}}, \quad \tilde{Z}_2(t) = 1 - e^{-\int_t^T \tilde{Z}_1(s)ds} = 1 - \frac{2e^{t - T}}{1 + e^{2t - 2T}},$$

and

$$\tilde{Z}_3(t) = \frac{1}{2}\sigma^2(T-t) - \frac{1}{2}\sigma^2 \ln\left(\frac{2}{1+e^{2t-2T}}\right) + \frac{1}{2} - \frac{1}{1+e^{2t-2T}}.$$

Similarly, we know that  $f_1, f_2, f_3$  is the solution to

$$\begin{cases} \dot{f}_1(t) - 2f_1(t) + 2 = 0; \\ \dot{f}_2(t) - f_2(t) + 1 = 0; \\ \dot{f}_3(t) + \frac{1}{2}\sigma^2 f_1(t) = 0; \\ f_1(T) = f_2(T) = f_3(T) = 0, \end{cases}$$

which gives that

$$f_1(t) = 1 - e^{2t - 2T}, \ f_2(t) = 1 - e^{t - T}, \ f_3(t) = \frac{1}{2}\sigma^2(T - t) - \frac{1}{4}\sigma^2 + \frac{1}{4}\sigma^2 e^{2t - 2T}.$$

Thus, we could obtain the difference between  $f_i(t)$  and  $\tilde{Z}_i(t)$  for i = 1, 2, 3 by

$$\Gamma_1(0) = e^{-2T} \frac{1 - e^{-2T}}{1 + e^{-2T}}, \quad \Gamma_2(0) = e^{-T} \frac{1 - e^{-2T}}{1 + e^{-2T}},$$

and

$$\Gamma_3(0) = \frac{1}{4}\sigma^2 \left( e^{-2T} - 1 + 2\ln\left(\frac{2}{1 + e^{-2T}}\right) \right) + \frac{1}{1 + e^{-2T}} - \frac{1}{2}.$$

Note that for  $x \in \mathbb{R}$  and T large enough,

$$J_T(0,x;u^*) - V_T(0,x) = \frac{1}{2}\Gamma_1(0)x^2 + \Gamma_2(0)x + \Gamma_3(0) \ge \Gamma_3(0) - \frac{(\Gamma_2(0))^2}{2\Gamma_1(0)} \ge 0.$$

On the other hand side, we have

$$J_T(0,x;u^*) - V_T(0,x) = \frac{e^{-2T}(1-e^{-2T})}{2(1+e^{-2T})}x^2 + \frac{e^{-T}(1-e^{-2T})}{1+e^{-2T}}x + \frac{\sigma^2}{4}\left(e^{-2T}-1+2\ln\left(\frac{2}{1+e^{-2T}}\right)\right) + \frac{1}{1+e^{-2T}} - \frac{1}{2}$$

Let  $T \to \infty$ , we find that

$$\lim_{T \to \infty} \left( J_T(0, x; u^*) - V_T(0, x) \right) = \frac{1}{4} \sigma^2 (\ln 4 - 1) + \frac{1}{2},$$

which yields that  $J_T(0, x; u^*) - V_T(0, x) = O(1)$  for all  $x \in \mathbb{R}$ . Thus, we obtain the desired result in Theorem 9 under this example.

Next, we check the result in Theorem 10. It is easy to get that  $D_1 = 1$  and  $D_2 = 0$  from (2.2.12). Denote  $\delta_T(t) = X_T^*(t) - X^*(t)$ , we have

$$d\delta_T(t) = -\tilde{Z}_1(t)\delta_T(t)dt + \left(1 - \tilde{Z}_2(t)\right)dt + \left(1 - \tilde{Z}_1(t)\right)X^*(t)dt$$

with the initial value  $\delta_T(0) = x - \bar{X} \sim \mathcal{N}(x, \frac{\sigma^2}{2})$ . Applying the integrating factor method, we have

the explicit form of  $\delta_T(t)$  as following

$$\delta_T(t) = \delta_T(0) e^{-\int_0^t \tilde{Z}_1(s)ds} + \int_0^t e^{-\int_s^t \tilde{Z}_1(r)dr} \left(1 - \tilde{Z}_2(s)\right) ds + \int_0^t e^{-\int_s^t \tilde{Z}_1(r)dr} \left(1 - \tilde{Z}_1(s)\right) X^*(s)ds.$$

Then, we get the estimation

$$\mathbb{E}\left[|\delta_{T}(t)|^{2}\right] \leq 3e^{-\int_{0}^{t} 2\tilde{Z}_{1}(s)ds} \mathbb{E}\left[|\delta_{T}(0)|^{2}\right] + 3\left(\int_{0}^{t} e^{-\int_{s}^{t} \tilde{Z}_{1}(r)dr} \left(1 - \tilde{Z}_{2}(s)\right)ds\right)^{2} + 3\mathbb{E}\left[\left(\int_{0}^{t} e^{-\int_{s}^{t} \tilde{Z}_{1}(r)dr} \left(1 - \tilde{Z}_{1}(s)\right)X^{*}(s)ds\right)^{2}\right].$$

Firstly, since  $\delta_T(0) = x - \bar{X} \sim \mathcal{N}(x, \frac{\sigma^2}{2})$ , we know that

$$3e^{-\int_0^t 2\tilde{Z}_1(s)ds} \mathbb{E}\left[\left|\delta_T(0)\right|^2\right] = 3e^{-2t} \left(\frac{1+e^{2t-2T}}{1+e^{-2T}}\right)^2 \left(x^2 + \frac{\sigma^2}{2}\right) \le 12e^{-2t} \left(x^2 + \frac{\sigma^2}{2}\right).$$

Next, by the Hölder's inequality and some simplifications, we can estimate the second term as

$$3\left(\int_0^t e^{-\int_s^t \tilde{Z}_1(r)dr} \left(1 - \tilde{Z}_2(s)\right) ds\right)^2 \le 12e^{-2(T-t)}.$$

Lastly, using the Hölder's inequality again, we have

$$3\mathbb{E}\left[\left(\int_{0}^{t} e^{-\int_{s}^{t} \tilde{Z}_{1}(r)dr} \left(1 - \tilde{Z}_{1}(s)\right) X^{*}(s)ds\right)^{2}\right] \le 24e^{-4T} \int_{0}^{t} e^{4s}\mathbb{E}\left[(X^{*}(s))^{2}\right]ds.$$

Similar with the calculation of expectation and variance of  $X^*$  in Subsection 2.2.2.3, we know that

$$\mathbb{E}[X^*(s)] = \mathbb{E}[\bar{X}] e^{-s} = 0, \quad \mathbb{V}ar(X^*(s)) = \frac{\sigma^2}{2} (1 - e^{-2s}) + \mathbb{V}ar(\bar{X}) e^{-2s} = \frac{\sigma^2}{2},$$

as  $\bar{X} \sim \mathcal{N}(0, \frac{\sigma^2}{2})$ , which gives  $\mathbb{E}[(X^*(s))^2] = \frac{\sigma^2}{2}$ . Then

$$3\mathbb{E}\left[\left(\int_0^t e^{-\int_s^t \tilde{Z}_1(r)dr} \left(1-\tilde{Z}_1(s)\right) X^*(s)ds\right)^2\right] \le 3\sigma^2 e^{-4(T-t)}.$$

Thus, we obtain the desired inequality

$$\mathbb{E}\left[\left|\delta_{T}(t)\right|^{2}\right] \leq 12\left(x + \frac{\sigma^{2}}{2}\right)e^{-2t} + 12e^{-2(T-t)} + 3\sigma^{2}e^{-4(T-t)}$$
$$\leq K_{6}\left(e^{-2t} + e^{-2(T-t)}\right)$$

for all  $t \in [0,T]$ , where  $K_6 = \max\{12(x + \frac{\sigma^2}{2}), 12 + 3\sigma^2\}$ . Similarly, since

$$u_T^*(t) - u^*(t) = 1 - \tilde{Z}_2(t) + \left(1 - \tilde{Z}_1(t)\right) X^*(t) - \tilde{Z}_1(t)\delta_T(t),$$

by some analytical calculations and simplifications, we have

$$\mathbb{E}\left[|u_T^*(t) - u^*(t)|^2\right] \le K_7 \left(e^{-2t} + e^{-2(T-t)}\right)$$

for all  $t \in [0, T]$ , where  $K_7 \ge 12 + 6\sigma^2 + 3K_6$ . Therefore, the desired turnpike property is obtained.

# 2.5 Appendix

*Proof of Lemma 11.* From the standard dynamic programming principle, we obtain the HJB equation

$$\begin{cases} -\partial_t V_T(t,x) + H\left(x, -\nabla_x V_T(t,x), -D_x^2 V_T(t,x)\right) = 0, \\ V_T(T,x) = 0, \end{cases}$$
(2.5.1)

where

$$L(x,u) = \frac{1}{2} \left( x^{\top} Q x + |u|^2 + 2q^{\top} x + 2r^{\top} u \right),$$
  
$$H(x,\bar{p},\bar{q}) = \sup_{u \in \mathbb{R}} \left\{ (Ax + u + b)^{\top} \bar{p} + \frac{1}{2} \sigma^2 \operatorname{trace}(\bar{q}) - L(x,u) \right\}.$$

Taking derivative to the terms in the supermum with respect to u, and letting it be zero, we have  $u + r + \nabla_x V_T(t, x) = 0$ . Thus, the optimal feedback control is given by

$$u_T^*(t) = -(r + \nabla_x V_T(t, X_T^*(t))),$$

and the value function  $V_T(t, x)$  satisfies

$$0 = \partial_t V_T(t, x) + x^\top A \nabla_x V_T(t, x) + (b - r)^\top \nabla_x V_T(t, x) - \frac{1}{2} |\nabla_x V_T(t, x)|^2 + \frac{1}{2} \sigma^2 \Delta_x V_T(t, x) + \frac{1}{2} x^\top Q x + q^\top x - \frac{1}{2} |r|^2.$$
(2.5.2)

Next, we give the semi-explicit solution to the HJB equation (2.5.1). Suppose the solution  $V_T$  to the HJB equation (2.5.1) is in  $C^{1,2}([0,T] \times \mathbb{R}^d)$ , we assume the value function has the form

$$V_T(t,x) = \frac{1}{2}x^{\top}\tilde{Z}_1(t)x + x^{\top}\tilde{Z}_2(t) + \tilde{Z}_3(t),$$

where  $\tilde{Z}_1 : [0,T] \mapsto \mathbb{S}^{d \times d}, \ \tilde{Z}_2 : [0,T] \mapsto \mathbb{R}^d$  and  $\tilde{Z}_3 : [0,T] \mapsto \mathbb{R}$  are real-valued functions in

 $C^{1}([0,T])$ . Then, it is straightforward to get that

$$\partial_t V_T(t,x) = \frac{1}{2} x^\top \dot{\tilde{Z}}_1(t) x + x^\top \dot{\tilde{Z}}_2(t) + \dot{\tilde{Z}}_3(t), \nabla_x V_T(t,x) = \tilde{Z}_1(t) x + \tilde{Z}_2(t), D_x^2 V_T(t,x) = \tilde{Z}_1(t).$$

Plugging the above terms into equation (2.5.2), we have

$$0 = \frac{1}{2}x^{\top} \left(\dot{\tilde{Z}}_{1}(t) + 2A\tilde{Z}_{1}(t) - \tilde{Z}_{1}^{2}(t) + Q\right)x$$
  
+  $x^{\top} \left(\dot{\tilde{Z}}_{2}(t) + A\tilde{Z}_{2}(t) - \tilde{Z}_{1}(t)\tilde{Z}_{2}(t) + \tilde{Z}_{1}(t)(b-r) + q\right)$   
+  $\dot{\tilde{Z}}_{3}(t) + \frac{1}{2}\sigma^{2}\operatorname{trace}(\tilde{Z}_{1}(t)) + b^{\top}\tilde{Z}_{2}(t) - \frac{1}{2}|\tilde{Z}_{2}(t) + r|^{2}$ 

for all  $x \in \mathbb{R}^d$  with  $\tilde{Z}_1(T) = \mathbf{O}_d$ ,  $\tilde{Z}_2(T) = \mathbf{O}_d$ , and  $\tilde{Z}_3(T) = 0$ . Setting up the coefficients of linear and quadratic terms with respect to x and constant to 0, we obtain the Riccati system of ODEs (2.3.2). Moreover, from the explicit form of  $V_T(t, x)$ , the optimal feedback control of the finite time control problem is given by  $u_T^*(t) = -(\tilde{Z}_1(t)X_T^*(t) + \tilde{Z}_2(t) + r)$  for all  $t \in [0, T]$ .

*Proof of Lemma 12.* From (2.2.14), the optimal control for the probabilistic cell problem is

$$u^*(t) = -(Z_1 X^*(t) + Z_2 + r)$$

which is a feedback control. If we take this control in the finite time control problem, then the underlying process becomes

$$dX_T(t) = (AX_T(t) - (Z_1X_T(t) + Z_2 + r) + b) dt + \sigma dW(t)$$
  
=  $-D_1X_T(t)dt + D_2dt + \sigma dW(t)$ 

with  $X_T(0) = x$ , where  $D_1$  and  $D_2$  are constants given in (2.2.12). By inserting the optimal control of the probabilistic cell problem into the cost functional of the finite time control problem, we obtain  $J_T(t, x; u^*)$  equals to

$$\mathbb{E}\left[\int_{t}^{T} \left(\frac{1}{2}X_{T}^{\top}(s)\left(Q+Z_{1}^{2}\right)X_{T}(s)+X_{T}^{\top}(s)(Z_{1}Z_{2}+q)+\frac{1}{2}\left(|Z_{2}|^{2}-|r|^{2}\right)\right)ds\right].$$

From Feynman-Kac's formula, if  $J_T(\cdot, \cdot; u^*) \in C^{1,2}([0, T] \times \mathbb{R}^d)$ , it is the solution to the following PDE

$$\begin{cases} \partial_t J_T(t,x;u^*) + (-D_1 x + D_2)^\top \nabla_x J_T(t,x;u^*) + \frac{1}{2} \sigma^2 \Delta_x J_T(t,x;u^*) \\ + \frac{1}{2} x^\top \left( Q + Z_1^2 \right) x + x^\top \left( Z_1 Z_2 + q \right) + \frac{1}{2} \left( |Z_2|^2 - |r|^2 \right) = 0, \qquad (2.5.3)\\ J_T(T,x;u^*) = 0. \end{cases}$$

We consider that the solution to (2.5.3) has the form

$$J_T(t, x; u^*) = \frac{1}{2} x^\top f_1(t) x + x^\top f_2(t) + f_3(t),$$

where  $f_1: [0,T] \mapsto \mathbb{S}^{d \times d}$ ,  $f_2: [0,T] \mapsto \mathbb{R}^d$  and  $f_3: [0,T] \mapsto \mathbb{R}$  are real-valued functions in  $C^1([0,T])$ . Then, it is clear that

$$\partial_t J_T(t, x; u^*) = \frac{1}{2} x^\top \dot{f}_1(t) x + x^\top \dot{f}_2(t) + \dot{f}_3(t)$$
$$\nabla_x J_T(t, x; u^*) = f_1(t) x + f_2(t)$$
$$D_x^2 J_T(t, x; u^*) = f_1(t).$$

Plugging the above terms into the PDE (2.5.3) satisfied by  $J_T(t, x; u^*)$ , we obtain

$$\frac{1}{2}x^{\top}\dot{f}_{1}(t)x + x^{\top}\dot{f}_{2}(t) + \dot{f}_{3}(t) + (-D_{1}x + D_{2})^{\top}(f_{1}(t)x + f_{2}(t)) + \frac{1}{2}\sigma^{2}\operatorname{trace}(f_{1}(t)) \\ + \frac{1}{2}x^{\top}\left(Q + Z_{1}^{2}\right)x + x^{\top}\left(Z_{1}Z_{2} + q\right) + \frac{1}{2}\left(|Z_{2}|^{2} - |r|^{2}\right) = 0$$

for all  $x \in \mathbb{R}^d$  with  $f_1(T) = \mathbf{O}_d$ ,  $f_2(T) = \mathbf{O}_d$ , and  $f_3(T) = 0$ . Setting up the coefficients of linear and quadratic terms with respect to x and constant to 0, and plugging  $D_1, D_2$  in (2.2.12) back, we obtain (2.3.3).

# Chapter 3

# Convergence rate of LQG mean field games with Markov chain as common noise

# 3.1 Introduction

In this chapter, we study the convergence rate of equilibrium measures of N-player differential game in the context of Linear-Quadratic (LQ) structure with a common noise to its limiting MFG system. Different from the works mentioned above, the common noise in this chapter is a continuous-time Markov chain (CTMC) instead of Brownian motion, which often models the real-world control problems associated with hybrid systems. Markov chains are widely used to model systems that exhibit randomness and transition between different states. In various real-world scenarios, especially in economics (see [78]), finance (see [85]), biology (see [86]), and engineering (see [84]), the dynamics of systems can be effectively represented as discrete states with probabilistic transitions between them. By using CTMC, the applications aim to model less frequently changing common noises, such as government policies implemented by two different regimes.

LQ control problems have been widely recognized in the stochastic control theory due to their broad applications. More importantly, LQ structure leads to solvability in a closed form, namely the Ricatti system, and this usually sheds light on many fundamental properties of the control theory. For this reason, LQ structure has also been studied in MFG with or without common noises for its importance. The related literature include major and minor Linear-Quadratic-Gaussian (LQG) Mean Field Games system ([39, 65, 30]); social optimal in LQG Mean Field Games ([44, 29]); the LQG Mean Field Games with different model settings ([5, 37, 6, 38]); and LQG Graphon Mean Field Games ([35]). Recently, LQ Mean Field Games with a Brownian motion as the common noise have also been studied in [1, 76] with restrictions of the dependence of measure on its mean alone. Moreover, some literature considers various topics of Mean Field control and game problems with Markov chain common noise, see [59, 66, 67].

A fundamental question in this regard is the convergence rate of the N-player game to the desired MFG system. A well-known result is about the convergence rate of value functions of the

generic player, which can be shown  $O(N^{-1})$ , see for instance [13, 14, 16, 17, 46]. In particular, [46] establishes the convergence rate of value functions in the sense of

$$J_1^N(\hat{\alpha}_1, \hat{\alpha}_{-1}) \le J_1^N(\alpha_1, \hat{\alpha}_{-1}) + O(N^{-1}).$$

where  $J_1^N$  is the value of the first player in the *N*-player game and  $\hat{\alpha}$  is the Nash equilibrium decentralized control process for the Mean Field Game problem.

In contrast, the convergence rate of equilibrium measures is another challenging question due to the complication of the correlation structures among N players. To be more concrete, we examine the behavior of the  $\hat{X}_{it}^{(N)}$ , who represents the equilibrium state of the *i*-th player at time *t* in the Nplayer game defined within the probability space  $(\Omega^{(N)}, \mathcal{F}^{(N)}, \mathbb{P}^{(N)}, \mathbb{P}^{(N)})$ . Additionally, we denote  $\hat{X}_t$  as the equilibrium path at time *t* derived from the associated MFG defined in the probability space  $(\Omega, \mathcal{F}, \mathbb{F}, \mathbb{P})$ . The question pertains to the convergence of  $\hat{X}_{1t}^{(N)}$  as follows:

(Q) The  $\mathbb{W}_p$ -convergence rate of the representative equilibrium path,

$$\mathbb{W}_p\left(\mathcal{L}\left(\hat{X}_{1t}^{(N)}\right), \mathcal{L}\left(\hat{X}_t\right)\right) = O\left(N^{-?}\right).$$

Here,  $\mathbb{W}_p$  denotes the *p*-Wasserstein metric.

The existing literature extensively explores the convergence rate in this context. For (Q), Theorem 2.4.9 of the monograph [14] establishes a convergence rate of  $O(N^{-1/2})$  using the  $W_1$ metric. More recently, [48] addresses (Q) by introducing displacement monotonicity and controlled common noise, and Theorem 2.23 applies the maximum principle of forward-backward propagation of chaos to achieve the same convergence rate. It is important to note that these results are not applicable to the LQG framework, primarily due to the assumption concerning the linear growth of the cost functional.

The main result of this chapter establishes that the equilibrium measures exhibit a convergence rate of 1/2 concerning the 2-Wasserstein distance. The precise statement of this result can be found in Theorem 21. In comparison to the aforementioned literature, two primary distinctions emerge. Firstly, within the framework of Mean Field Games, the common noise is modeled as a Continuous-Time Markov Chain. Secondly, a significant difference lies in the cost function's behavior, as it does not possess linear growth within the context of the LQG framework.

To obtain the desired convergence rate in this chapter, the first building block is the characterization of the equilibrium measure of the limiting MFG by a finite-dimensional ODE system. The key step leading us to a desired finite-dimensional system is that, instead of searching for the infinite-dimensional function directly, we postulate a Markovian structure via two auxiliary processes (3.3.1) governed by its finite-dimensional coefficient functions, which exhibits the distinct feature of Markov chain common noise relatives to the Brownian motion counterpart.

The next stage towards the convergence rate is to compare the limiting MFG system to an N-player game. In contrast to the characterization of the MFG system, it is relatively routine to solve the N-player game due to its LQ structure. Therefore, the convergence rate problem can be recasted to the following question about a coupling of the two following processes: For two

equilibrium processes  $\hat{X}$  of MFG in  $\Omega$  and  $\hat{X}_{1}^{(N)}$  of the *N*-player game in  $\Omega^{(N)}$ , finding a random process  $Z^{N}$  in  $\Omega$  whose distribution is identical to  $\hat{X}_{1}^{(N)}$  satisfying the estimate in the form of  $\mathbb{E}[|\hat{X}_{t} - Z_{t}^{N}|^{2}] = O(N^{-?})$ . For this purpose, we first show an *N*-invariant algebraic structure of the seemingly intractable  $\kappa N^{3}$  dimensional ODE system (3.4.1), which originated from [46, Huang and Yang] as a dimensional reduction in the system with Brownian common noise. Thanks to this *N*-invariant structure, the complex ODE system (3.4.1) can be reduced to the ODE system (3.4.5) whose dimension agrees with the ODE (3.2.12) of MFG system. Moreover,  $\hat{X}_{1}^{(N)}$  can be represented as a stochastic flow driven by two Brownian motions  $W_{1}^{(N)}$  and  $W_{-1}^{(N)} := \frac{1}{\sqrt{N-1}} \sum_{i=2}^{N} W_{i}^{(N)}$ , which enables us to embed the equilibrium process  $\hat{X}_{1}^{(N)}$  to any probability space having only two Brownian motions.

The rest of this chapter is outlined as follows: Section 3.2 presents a precise formulation of the problem and two main results. Section 3.3 is devoted to the derivation of our first result: the equilibrium of MFG. In Section 3.4, we show in detail the convergence of the *N*-player game to MFG, which yields our second main result. Section 3.5 demonstrates the convergence by some numerical examples. The conclusion and some possible future works are summarised in Section 3.6. Section 3.7 is an appendix that collects some related facts to support our main theme.

# 3.2 Problem setup and main results

First, we collect common notations used in this chapter in Subsection 3.2.1. Then, we set up problems on MFG and the *N*-player game separately in Subsections 3.2.2 and 3.2.3. The main results are presented in Subsection 3.2.4 and some interpretations of our main results are added in Subsection 3.2.5.

#### 3.2.1 Notations

Let T > 0 be a fixed terminal time and  $(\Omega, \mathcal{F}_T, \mathbb{F} = \{\mathcal{F}_t : 0 \le t \le T\}, \mathbb{P})$  be a completed filtered probability space satisfying the usual conditions, on which W and B are two independent standard Brownian motions, and Y is a continuous time Markov chain (CTMC) independent of (W, B) taking values in a finite state space  $\mathcal{Y} = \{1, 2, \ldots, \kappa\}$  with a generator

$$Q = (q_{i,j})_{i,j \in \mathcal{Y}} \tag{3.2.1}$$

satisfying  $q_{i,j} \geq 0$  for all  $i \neq j \in \mathcal{Y}$  and  $\sum_{i\neq j} q_{i,j} + q_{i,i} = 0$  for each  $i \in \mathcal{Y}$ . In the above, the Brownian motion *B* does not play any role in MFG problem formulation until the convergence proof of the *N*-player game to MFG.

By  $L^p := L^p(\Omega, \mathbb{P})$ , we denote the space of random variables X on  $(\Omega, \mathcal{F}_T, \mathbb{P})$  with finite p-th moment with norm  $||X||_p = (\mathbb{E}[|X|^p])^{1/p}$ . We also denote by  $L^p_{\mathbb{F}} := L^p_{\mathbb{F}}([0,T] \times \Omega)$  the space of all  $\mathbb{F}$ -progressively measurable random processes  $\alpha = (\alpha_t)_{0 \le t \le T}$  satisfying

$$\mathbb{E}\left[\int_0^T |\alpha_t|^p dt\right] < \infty.$$

For any polish (complete separable metric) space  $(P, \mathcal{B}(P), d)$ , we use  $\delta_x$  to denote the Dirac measure on the point  $x \in P$ . Then, the collection of all probabilities m on  $(P, \mathcal{B}(P), d)$  having finite k-th moment is denoted by  $\mathcal{P}_k(P)$ , i.e.

$$[m]_k := \int x^k m(dx) < \infty, \quad \forall m \in \mathcal{P}_k(P).$$

The equilibrium of MFG with the common noise yields the conditional distribution. For realvalued random variables X and Z in  $(\Omega, \mathcal{F}_T, \mathbb{P})$ , we denote the distribution of X conditional on  $\sigma(Z)$  by  $\mathcal{L}(X|Z)$ , or equivalently

$$\mathcal{L}(X|Z)(A) = \mathbb{E}[I_A(X)|Z], \quad \forall A \in \mathcal{F}_T.$$

Note that  $\mathcal{L}(X|Z)(A) : \Omega \mapsto \mathbb{R}$  is a  $\sigma(Z)$ -measurable random variable, therefore,  $\mathcal{L}(X|Z)$  is  $\sigma(Z)$ -measurable random probability distribution with k-th moment  $[\mathcal{L}(X|Z)]_k = \mathbb{E}[X^k|Z]$ , if it exists. We refer to more details on the conditional distribution in Volume II of [17]. The next proposition provides an embedding approach to prove a convergence in distribution, which will be used later in the convergence of the N-player game to MFG.

**Proposition 16.** Suppose  $(\Omega^{(N)}, \mathcal{F}_T^{(N)}, \mathbb{P}^{(N)})$  is a complete probability space. Let  $X^{(N)}$  and X be random variables of  $\Omega^{(N)} \mapsto P$  and  $\Omega \mapsto P$ , respectively. Then,  $X^{(N)}$  is convergent in distribution to X, denoted by  $X^{(N)} \Rightarrow X$ , if there exists  $Z^N : \Omega \mapsto P$  satisfying  $\mathcal{L}(Z^N) = \mathcal{L}(X^{(N)})$ , such that  $Z^N \to X$  holds almost surely, i.e.

$$\lim_{N \to \infty} d(Z^N, X) = 0, \text{ almost surely in } \mathbb{P},$$

where d represents the metric assigned to the space P.

In this chapter, we formulate the N-player game in the completed filtered probability space

$$(\Omega^{(N)}, \mathcal{F}_T^{(N)}, \mathbb{F}^{(N)} := \{\mathcal{F}_t^{(N)} : 0 \le t \le T\}, \mathbb{P}^{(N)}),\$$

and  $Y^{(N)}$  is the continuous time Markov chain in  $\Omega^{(N)}$  with the same generator given by (3.2.1) and  $W^{(N)} = (W_i^{(N)} : i = 1, 2, ..., N)$  is an N-dimensional standard Brownian motion. We assume  $Y^{(N)}$  and  $W^{(N)}$  are independent of each other.

For better clarity, we use the superscript (N) for a random variable to emphasize the probability space  $\Omega^{(N)}$  it belongs to. For example, Proposition 16 denotes a random variable in  $\Omega^{(N)}$  by  $X^{(N)}$ , while its distribution copy in  $\Omega$  by  $Z^N$ , but not by  $Z^{(N)}$ .

#### 3.2.2 The equilibrium of MFG

In this section, we define the equilibrium of MFG associated with a generic player's stochastic control problem in the probability setting  $\Omega$ , see Section 3.2.1.

Given a random measure flow  $m: (0,T] \times \Omega \mapsto \mathcal{P}_2(\mathbb{R})$ , consider a generic player who wants to

minimize her expected accumulated cost on [0, T]:

$$J(y,x,\alpha) = \mathbb{E}\left[\int_0^T \left(\frac{1}{2}\alpha_s^2 + F(Y_s, X_s, m_s)\right) ds + G(Y_T, X_T, m_T) \Big| Y_0 = y, X_0 = x\right]$$
(3.2.2)

with some given cost functions  $F, G : \mathcal{Y} \times \mathbb{R} \times \mathcal{P}_2(\mathbb{R}) \mapsto \mathbb{R}$  and underlying random processes  $(Y, X) : [0, T] \times \Omega \mapsto \mathcal{Y} \times \mathbb{R}$ . Among three processes (Y, X, m), the generic player can control the process X via  $\alpha$  in the form of

$$X_t = X_0 + \int_0^t \left( \tilde{b}_1(Y_s, s) X_s + \tilde{b}_2(Y_s, s) \alpha_s \right) ds + W_t, \quad \forall t \in [0, T],$$
(3.2.3)

where  $\tilde{b}_1(\cdot, \cdot)$  and  $\tilde{b}_2(\cdot, \cdot)$  are two deterministic functions. We assume that the initial state  $X_0$  is independent of Y. The Brownian motion W is the individual noise of the generic player, the process Y of (3.2.1) represents the common noise, and  $m = (m_t)_{0 \le t \le T}$  is a given random density flow normalized up to total mass one.

The objective of the control problem for the generic player is to find its optimal control  $\hat{\alpha} \in \mathcal{A} := L^4_{\mathbb{F}}$  to minimize the total cost, i.e.

$$V[m](y,x) = J[m](y,x,\hat{\alpha}) \le J[m](y,x,\alpha), \quad \forall \alpha \in \mathcal{A}.$$
(3.2.4)

Associated with the optimal control  $\hat{\alpha}$ , we denote the optimal path by  $\hat{X} = (\hat{X}_t)_{0 \leq t \leq T}$ . To introduce MFG Nash equilibrium, it is often convenient to highlight the dependence of the optimal path and optimal control of the generic player and its associated value on the underlying density flow m, which are denoted by

$$\hat{X}_t[m], \hat{\alpha}_t[m], \text{ and } V[m],$$

respectively. Now, we present the definition of the equilibrium below, see also Volume II-P127 of [17] for a general setup with a common noise.

**Definition 17.** Given an initial distribution  $\mathcal{L}(X_0) = m_0 \in \mathcal{P}_2(\mathbb{R})$ , a random measure flow  $\hat{m} = \hat{m}(m_0)$  is said to be an MFG equilibrium measure if it satisfies the fixed point condition

$$\hat{m}_t = \mathcal{L}(\hat{X}_t[\hat{m}]|Y), \ \forall 0 < t \le T, \ almost \ surely \ in \ \mathbb{P}.$$
 (3.2.5)

The path  $\hat{X}$  and the control  $\hat{\alpha}$  associated to  $\hat{m}$  is called the MFG equilibrium path and equilibrium control, respectively. The value function of the control problem associated with the equilibrium measure  $\hat{m}$  is called as MFG value function, denoted by

$$U(m_0, y, x) = V[\hat{m}](y, x).$$
(3.2.6)

The flowchart of MFG diagram is given in Figure 3.1. It is noted from the optimality condition (3.2.4) and the fixed point condition (3.2.5) that

$$J[\hat{m}](y, x, \hat{\alpha}) \le J[\hat{m}](y, x, \alpha), \quad \forall \alpha$$



Figure 3.1: MFG diagram.

holds for the equilibrium measure  $\hat{m}$  and its associated equilibrium control  $\hat{\alpha}$ , while it is not

$$J[\hat{m}](y, x, \hat{\alpha}) \le J[m](y, x, \alpha), \quad \forall \alpha, m.$$

Otherwise, this problem turns into a McKean-Vlasov control problem discussed in [66]. Furthermore, it's important to note that the Continuous-Time Markov Chain Y serves a role as common noise. This is due to the fact that the mean field term is conditioned on the distribution of Y.

#### 3.2.3 Equilibrium of the *N*-player game

The discrete counterpart of MFG is an N-player game, which is formulated below in the probability space  $\Omega^{(N)}$ , see Section 3.2.1 for more details on the probability setup.

Recall that,  $W_{it}^{(N)}$  and  $W_{jt}^{(N)}$  are independent Brownian motions for  $j \neq i$  and they are called individual noises in the N-player game. The common noise  $Y^{(N)}$  is the continuous time Markov chain in  $\Omega^{(N)}$  with the generator given by (3.2.1). Let the player *i* follow the dynamic, for  $i = 1, 2, \ldots, N$ ,

$$dX_{it}^{(N)} = \left(\tilde{b}_1(Y_t^{(N)}, t)X_{it}^{(N)} + \tilde{b}_2(Y_t^{(N)}, t)\alpha_{it}^{(N)}\right)dt + dW_{it}^{(N)}, \quad X_{i0}^{(N)} = x_i^{(N)}.$$
(3.2.7)

The cost function for player *i* associated to the control  $\alpha^{(N)} = (\alpha_i^{(N)} : i = 1, 2, ..., N)$  is

$$J_{i}^{N}(y, x^{(N)}, \alpha^{(N)}) = \mathbb{E}\left[\int_{0}^{T} \left(\frac{1}{2} |\alpha_{it}^{(N)}|^{2} + F(Y_{t}^{(N)}, X_{it}^{(N)}, \rho(X_{t}^{(N)}))\right) dt + G(Y_{T}^{(N)}, X_{iT}^{(N)}, \rho(X_{T}^{(N)})) \Big| X_{0}^{(N)} = x^{(N)}, Y_{0}^{(N)} = y \right],$$
(3.2.8)

where  $x^{(N)} = (x_1^{(N)}, x_2^{(N)}, \dots, x_N^{(N)})$  is an  $\mathbb{R}^N$ -valued random vector in  $\Omega^{(N)}$  to denote the initial

state for N player,  $\alpha_i^{(N)} \in \mathcal{A}^{(N)} := L^4_{\mathbb{F}^{(N)}},$  and

$$\rho(x^{(N)}) = \frac{1}{N} \sum_{i=1}^{N} \delta_{x_i^{(N)}}$$

is the empirical measure of a vector  $x^{(N)}$  with Dirac measure  $\delta$ . We use the notation for the control  $\alpha^{(N)} = (\alpha_i^{(N)}, \alpha_{-i}^{(N)}) = (\alpha_1^{(N)}, \alpha_2^{(N)}, \dots, \alpha_N^{(N)}).$ 

**Definition 18.** 1. The value function of player i for i = 1, 2, ..., N of the Nash game is defined by  $V^N = (V_i^N : i = 1, 2, ..., N)$  satisfying the equilibrium condition

$$V_i^N(y, x^{(N)}) = J_i^N(y, x^{(N)}, \hat{\alpha}_i^{(N)}, \hat{\alpha}_{-i}^{(N)}) \le J_i^N(y, x^{(N)}, \alpha_i^{(N)}, \hat{\alpha}_{-i}^{(N)}),$$
(3.2.9)

for all  $\alpha_i^{(N)} \in \mathcal{A}^{(N)}$ .

2. The equilibrium path of the N-player game is the random path  $\hat{X}_t^{(N)} = (\hat{X}_{1t}^{(N)}, X_{2t}^{(N)}, \dots, \hat{X}_{Nt}^{(N)})$ driven by (3.2.7) associated to the control  $\hat{\alpha}_t^{(N)}$  satisfying the equilibrium condition of (3.2.9).

## 3.2.4 The main result with quadratic cost structures

We consider the following two functions  $F, G: \mathcal{Y} \times \mathbb{R} \times \mathcal{P}_2(\mathbb{R}) \mapsto \mathbb{R}$  in the cost functional (3.2.2):

$$F(y, x, m) = h(y) \int_{\mathbb{R}} (x - z)^2 m(dz), \qquad (3.2.10)$$

and

$$G(y, x, m) = g(y) \int_{\mathbb{R}} (x - z)^2 m(dz), \qquad (3.2.11)$$

for some  $h, g: \mathcal{Y} \mapsto \mathbb{R}^+$ . In this case, the F and G terms in (3.2.8) of the N-player game can be written by

$$F(Y_t^{(N)}, X_{it}^{(N)}, \rho(X_t^{(N)})) = \frac{h(Y_t^{(N)})}{N} \sum_{j=1}^N (X_{it}^{(N)} - X_{jt}^{(N)})^2,$$

and

$$G(Y_T^{(N)}, X_{iT}^{(N)}, \rho(X_T^{(N)})) = \frac{g(Y_T^{(N)})}{N} \sum_{j=1}^N (X_{iT}^{(N)} - X_{jT}^{(N)})^2,$$

respectively.

**Remark 19.** First, we note that F and G possess the quadratic structures in x. Secondly, the coefficients h(y) and g(y) provide the sensitivity to the mean field effects, which depend on the current CTMC state. For another remark, let us consider the scenario where the number of states is 2 and sensitivities are invariant, say

$$h(0) = h(1) = h, g(0) = g(1) = 0.$$

Then the cost function and hence the entire problem is free from the common noise. Interestingly, as shown in the Appendix 3.7.1, there is no global solution for MFG when h < 0, while there is a global solution when h > 0.

Moreover, the uniqueness of Mean Field Game can be achieved under the displacement monotonicity condition. It is easy to check that (3.2.10)-(3.2.11) satisfy the displacement monotonicity condition. Note that

$$F_x(y,x,m) = 2h(y)(x - [m]_1), \quad G_x(y,x,m) = 2g(y)(x - [m]_1),$$

which gives that

$$\mathbb{E}\left[\left(F_x(y, X_1, m_{X_1}) - F_x(y, X_2, m_{X_2})\right)(X_1 - X_2)\right] = 2h(y)\left(\mathbb{E}\left[\left(X_1 - X_2\right)^2\right] - \left(\mathbb{E}[X_1] - \mathbb{E}[X_2]\right)^2\right) \ge 0$$

for all  $y \in \mathcal{Y}$  if h > 0 on  $\mathcal{Y}$ , where  $m_{X_1}$  and  $m_{X_2}$  is the law of  $X_1$  and  $X_2$  respectively. Similarly, we can obtain that

$$\mathbb{E}\left[\left(G_x(y, X_1, m_{X_1}) - G_x(y, X_2, m_{X_2})\right)(X_1 - X_2)\right] \ge 0$$

for all  $y \in \mathcal{Y}$  if g > 0 on  $\mathcal{Y}$ . Therefore, we require positive values for all sensitivities for simplicity. It is of course an interesting problem to investigate the explosion when some sensitivities are negative.

Wrapping up the above discussions, we impose the following assumptions:

- (A0)  $\tilde{b}_1(y,\cdot), \tilde{b}_2(y,\cdot): [0,T] \mapsto \mathbb{R}$  are continuous functions for all  $y \in \mathcal{Y}$ .
- (A1) The cost functions are given by (3.2.10)-(3.2.11) with h, g > 0; The initial  $X_0$  of MFG satisfies  $\mathbb{E}[X_0^2] < \infty$ .
- (A2) In addition to (A1), the initial  $x^{(N)} = (x_1^{(N)}, x_2^{(N)}, \dots, x_N^{(N)})$  of the *N*-player game is a vector of i.i.d. random variables in  $\Omega^{(N)}$  with the same distribution as the initial  $\mathcal{L}(X_0)$  of MFG.

Our objective for this chapter is to understand the Nash equilibrium of MFG and its connection to the N-player game equilibrium:

(P1) With Assumptions (A0), (A1), and (A2), obtain the convergence rate of  $(\hat{X}_{1t}^{(N)}, Y^{(N)})$  from the N-player game of Definition 18 to  $(\hat{X}_t, Y)$  from MFG of Definition 17 in distribution.

To answer (P1), it is critical to have a solid understanding of the joint distribution  $(\hat{X}_t, Y)$  for the underlying MFG, which yields another question:

(P2) With Assumptions (A0) and (A1), characterize the MFG equilibrium path  $\hat{X}$ , as well as associated equilibrium measure  $\hat{m}$  along the Definition 17;

For our first main result, we first answer (P2) via the following Riccati system for unknowns

 $(a_y, b_y, c_y, k_y : y \in \mathcal{Y})$ :

$$\begin{cases} a'_{y} + 2\tilde{b}_{1y}a_{y} - 2\tilde{b}_{2y}^{2}a_{y}^{2} + \sum_{i=1}^{\kappa} q_{y,i}a_{i} + h_{y} = 0, \\ b'_{y} + \left(2\tilde{b}_{1y} - 4\tilde{b}_{2y}^{2}a_{y}\right)b_{y} + \sum_{i=1}^{\kappa} q_{y,i}b_{i} + h_{y} = 0, \\ c'_{y} + a_{y} + b_{y} + \sum_{i=1}^{\kappa} q_{y,i}c_{i} = 0, \\ c'_{y} - 2\tilde{b}_{2y}^{2}a_{y}^{2} + 4\tilde{b}_{2y}^{2}a_{y}b_{y} + 2\tilde{b}_{1y}k_{y} + \sum_{i=1}^{\kappa} q_{y,i}k_{i} = 0, \\ a_{y}(T) = b_{y}(T) = g_{y}, c_{y}(T) = k_{y}(T) = 0, \end{cases}$$

$$(3.2.12)$$

where  $h_y = h(y)$ ,  $g_y = g(y)$  for  $y \in \mathcal{Y}$ . Next, we present our first main result about the equilibrium path, the equilibrium control, and the value function in MFG.

**Theorem 20** (MFG). Under (A0)-(A1), there exists a unique solution  $(a_y, b_y, c_y, k_y : y \in \mathcal{Y})$  for the Riccati system (3.2.12). With these solutions, the MFG equilibrium path  $\hat{X} = \hat{X}[\hat{m}]$  is given by

$$d\hat{X}_{t} = \left(\tilde{b}_{1}(Y_{t}, t)\hat{X}_{t} - 2\tilde{b}_{2}^{2}(Y_{t}, t)a_{Y_{t}}(t)\left(\hat{X}_{t} - \hat{\mu}_{t}\right)\right)dt + dW_{t}, \quad \hat{X}_{0} = X_{0}, \quad (3.2.13)$$

with equilibrium control

$$\hat{\alpha}_t = -2\tilde{b}_2(Y_t, t)a_{Y_t}(t)\left(\hat{X}_t - \hat{\mu}_t\right), \qquad (3.2.14)$$

where

$$d\hat{\mu}_t = \tilde{b}_1(Y_t, t)\hat{\mu}_t dt, \quad \hat{\mu}_0 = \mathbb{E}[X_0].$$

Moreover, the value function U is

$$U(m_0, y, x) = a_y(0)x^2 - 2a_y(0)x[m_0]_1 + k_y(0)[m_0]_1^2 + b_y(0)[m_0]_2 + c_y(0), \quad y \in \mathcal{Y}.$$

The proof of theorem 20 is based on the Markovian structure of the equilibrium and the fixed point condition of the MFG problem, and it is provided in Subsection 3.3.3. The next theorem establishes the convergence result and answers the problem (P1) with the convergence rate  $\frac{1}{2}$ .

**Theorem 21** (Convergence rate). Under Assumptions (A0)-(A1)-(A2), the joint law  $(\hat{X}_{1t}^{(N)}, Y_t^{(N)})$  of the N-player game converges in distribution to that of the MFG equilibrium  $(\hat{X}_t, Y_t)$  for any  $t \in (0,T]$  at the convergence rate

$$\mathbb{W}_2\left(\mathcal{L}(\hat{X}_{1t}^{(N)}, Y_t^{(N)}), \mathcal{L}(\hat{X}_t, Y_t)\right) = O\left(N^{-\frac{1}{2}}\right), \quad as \ N \to \infty.$$

The proof of Theorem 21 is given in Subsection 3.4.3 since it needs the comparison between the equilibrium path  $\hat{X}_{1t}^{(N)}$  in the N-player game and the equilibrium path  $\hat{X}_t$  in MFG.

#### 3.2.5 Remarks on the main results

One can interpret the main results in plain words: For the N-player game with dynamic (3.2.7) and cost structure (3.2.8) for large N, the equilibrium control of the generic player can be effectively approximated by steering itself toward the population center  $\hat{\mu}_t$  depending only on the function  $\tilde{b}_1(\cdot)$  and the entire past of the common noise, whose velocity is dependent on only the function  $\tilde{b}_2(\cdot)$ and the entire past of the common noise. The effectiveness can be quantified by the convergence rate of 1/2 for the one-dimensional MFG under LQ structure and CMTC common noise. A natural question is whether the convergence rate can be generalized to more general settings.

We may utilize the above outcome to depict a simplified scenario of traffic flow control, where individual vehicles strive to minimize their own costs while taking into account their impact on overall traffic dynamics. We could denote the dynamics of vehicle *i* within a city by  $X_i^{(N)}$ , and employ the common noise  $Y^{(N)}$  to represent traffic conditions across various districts or intersections. Each vehicle's behavior is influenced not only by its driving habits and preferences but also by the traffic situation in different districts or intersections of the city. Additionally, vehicles must factor in their proximity to traffic centers and endeavor to find more efficient routes to their destinations.

This chapter focuses on the one-dimensional problem to avoid unnecessary symbol complexity. The main convergence rate 1/2 still holds for multidimensional problems using the same coupling procedure. For convenience to check, we summarize the computation involved in multidimensional problems in Appendix 3.7.5.

The current coupling procedure can also be adapted with suitable modifications to the LQ Mean Field Game problems with Brownian common noise, see [51]. In particular, the reduction of the  $O(N^3)$ -dimensional ODE can be conducted similarly and the convergence rate is still maintained as 1/2. However, the dependence of the mean and variance process on the common noise and subsequent calculations are significantly different from the current chapter, see Definition 4 of [51].

Indeed, choosing the CTMC common noise instead of Brownian motion does not simplify the underlying problem, since it preserves the path-dependence feature of the equilibrium measure. On the contrary, the advantage of CTMC common noise is that the applications aim to model less frequently changing environment settings, such as government policies implemented by multiple different regimes. Due to its realistic applications, stochastic control theory perturbed by CTMC is extensively studied in the context of hybrid control problems, see books [60, 80] and the references therein.

We close this section with a remark on the uniqueness. The uniqueness of Mean Field Game can be achieved under Lasry-Lions monotonicity [55] or displacement monotonicity [33] and our setting in Section 3.2.2 satisfies the displacement monotonicity. Thus, the convergence of Theorem 21 implies that the unique equilibrium path of the N-player game converges to the unique equilibrium paths of the limiting MFG, which is characterized by Theorem 20.

# 3.3 Main results of MFG

This section is devoted to the proof of the first main result Theorem 20 on the MFG solution. First, we outline the scheme based on the Markovian structure of the equilibrium by reformulating the

MFG problem in Subsection 3.3.1. Next, we solve the underlying control problem in Subsection 3.3.2 and provide the corresponding Riccati system. Finally, Subsection 3.3.3 proves Theorem 20 by checking the fixed point condition of MFG problem.

#### 3.3.1 Overview

By Definition 18, to solve for the equilibrium measure, one shall search the infinite dimensional space of the random measure flows  $m : (0,T] \times \Omega \mapsto \mathcal{P}_2(\mathbb{R})$ , until a measure flow satisfies the fixed point condition  $m_t = \mathcal{L}(\hat{X}_t|Y), \forall t \in (0,T]$ , see Figure 3.1, which requires to check the following infinitely many conditions:

$$[m_t]_k = \mathbb{E}[\hat{X}_t^k | Y], \quad \forall k = 1, 2, \dots,$$

if they exist.

The first observation is that the cost functions F and G in (3.2.10)-(3.2.11) are dependent on the measure m only via the first two moments:

$$F(y, x, m) = h(y)(x^2 - 2x[m]_1 + [m]_2),$$
  

$$G(y, x, m) = g(y)(x^2 - 2x[m]_1 + [m]_2).$$

Therefore, the underlying stochastic control problem for MFG can be entirely determined by the input given by  $\mathbb{R}^2$  valued random process  $\mu_t = [m_t]_1$  and  $\nu_t = [m_t]_2$ , which implies that the fixed point condition can be effectively reduced to check two conditions only:

$$\mu_t = \mathbb{E}[\hat{X}_t|Y], \ \nu_t = \mathbb{E}[\hat{X}_t^2|Y]$$

This observation effectively reduces our search from the space of random measure-valued processes  $m: (0,T] \times \Omega \mapsto \mathcal{P}_2(\mathbb{R})$  to the space of  $\mathbb{R}^2$ -valued random processes  $(\mu,\nu): (0,T] \times \Omega \mapsto \mathbb{R}^2$ .

Note that, if underlying MFG have no common noise Y, then  $(\mu, \nu)$  is a deterministic mapping  $[0, T] \mapsto \mathbb{R}^2$  and the above observation is enough to reduce the original infinite-dimensional MFG into a finite-dimensional system. However, the following example shows that this is not the case for MFG with a common noise and it becomes the main drawback to characterizing MFG via a finite-dimensional system.

**Example 1.** To illustrate, we consider the following uncontrolled mean field dynamics: Let the mean field term  $\mu_t := \mathbb{E}[\hat{X}_t|Y]$ , where the underlying dynamic is given by

$$d\hat{X}_t = -\mu_t Y_t dt + dW_t.$$

•  $\mu_t$  is path dependent on Y, i.e.,

$$\mu_t = \mu_0 \exp\Big\{-\int_0^t Y_s ds\Big\}.$$

This implies that no finite dimensional system is possible to characterize the process  $\mu_t$ , since the  $(t, Y) \mapsto \mu_t$  is a function on an infinite dimensional domain.

•  $\mu_t$  is Markovian, i.e.,

$$d\mu_t = -Y_t \mu_t dt.$$

It might be possible to characterize  $\mu_t$  via a function  $(t, Y_t, \mu_t) \mapsto \frac{d\mu_t}{dt}$  on a finite dimensional domain.

To solidify the above idea, we need to postulate the Markovian structure for the first and second moments of the MFG equilibrium. More precisely, our search for the equilibrium will be confined to the space  $\mathcal{M}$  of measure flows whose first and second moment exhibits Markovian structure.

**Definition 22.** The space  $\mathcal{M}$  is the collection of all  $\mathcal{F}_t^Y$ -adapted measure flows  $m : [0,T] \times \Omega \mapsto \mathcal{P}_2(\mathbb{R})$ , whose first moment  $[m_t]_1 := \mu_t$  and second moment  $[m_t]_2 := \nu_t$  satisfy

$$\mu_t = \mu_0 + \int_0^t \left( w_0(Y_s, s) \mu_s + w_1(Y_s, s) \right) ds,$$
  

$$\nu_t = \nu_0 + \int_0^t \left( w_2(Y_s, s) \mu_s + w_3(Y_s, s) \nu_s + w_4(Y_s, s) \mu_s^2 + w_5(Y_s, s) \right) ds,$$
(3.3.1)

for all  $t \in [0,T]$  and for some smooth deterministic functions  $(w_i : i = 0, 1, ..., 5)$ .



Figure 3.2: Equivalent MFG diagram.

The flowchart for our equilibrium is depicted in Figure 3.2. Subsection 3.3.2 covers the derivation of the Riccati system for the LQG system with a given population measure flow  $m \in \mathcal{M}$ , which provides the key building block to MFG. In Subsection 3.3.3, we check the fixed point condition and provide a finite-dimensional characterization of MFG, which gives the first main result Theorem 20.

#### 3.3.2 The generic player's control with a given population measure

The advantage of the generic player's control problem associated with  $m \in \mathcal{M}$  is that its optimal path can be characterized via the following classical stochastic control problem:

• (P3) Given smooth functions  $w = (w_i : i = 0, 1, ..., 5)$ , find the optimal value  $\overline{V} = \overline{V}[w]$ 

$$\bar{V}(y, x, t, \bar{\mu}, \bar{v}) = \inf_{\alpha \in \mathcal{A}} \mathbb{E} \left[ \int_t^T \left( \frac{1}{2} \alpha_s^2 + \bar{F}(Y_s, X_s, \mu_s, \nu_s) \right) ds + \bar{G}(Y_T, X_T, \mu_T, \nu_T) \right| Y_t = y, X_t = x, \mu_t = \bar{\mu}, \nu_t = \bar{\nu}$$

underlying  $\mathbb{R}^4$ -valued processes  $(Y, X, \mu, \nu)$  defined through (3.2.1)-(3.2.3)-(3.3.1) with the finite dimensional cost functions:  $\overline{F}, \overline{G} : \mathbb{R}^4 \to \mathbb{R}$  given by

$$\bar{F}(y, x, \bar{\mu}, \bar{\nu}) = h(y)(x^2 - 2x\bar{\mu} + \bar{\nu}),$$
$$\bar{G}(y, x, \bar{\mu}, \bar{\nu}) = g(y)(x^2 - 2x\bar{\mu} + \bar{\nu}),$$

where  $\bar{\mu}, \bar{\nu}$  are scalars, while  $\mu, \nu$  are used as processes.

**Lemma 23.** Given  $m \in \mathcal{M}$  associated with  $w = (w_i : i = 0, 1, ..., 5)$ , the player's value (3.2.4) under assumption (A1) is

$$U[m_0](y,x) = V(y,x,0,[m_0]_1,[m_0]_2),$$

and the optimal control has a feedback form

$$\hat{\alpha}_t = \bar{\alpha}(Y_t, X_t, t, \mu_t, \nu_t)$$

underlying the processes  $(Y, X, \mu, \nu)$  defined through (3.2.1)-(3.2.3)-(3.3.1), whenever there exists a feedback optimal control  $\bar{\alpha}$  for the problem (P3).

*Proof.* Due to the quadratic cost structure in (3.2.10)-(3.2.11), we have enough regularity to all concerned value functions and the details are omitted.

Next, we turn to the solution to the control problem (P3).

#### 3.3.2.1 HJB equation

For the simplicity of notations, for each  $i \in \{0, 1, 2, 3, 4, 5\}$  and  $y \in \mathcal{Y}$ , denote the function  $(x, t, \bar{\mu}, \bar{\nu}) \mapsto v(y, x, t, \bar{\mu}, \bar{\nu})$  as  $v_y$ , and denote  $t \mapsto w_i(y, t)$  as  $w_{iy}$ . We apply similar notations for other functions whenever they have a variable  $y \in \mathcal{Y}$ . Formally, under enough regularity conditions, the value function  $\bar{V}$  defined in (P3) is the solution v of the following coupled HJBs

$$\begin{cases} \partial_t v_y + \tilde{b}_{1y} x \partial_x v_y - \frac{1}{2} \left( \tilde{b}_{2y} \partial_x v_y \right)^2 + \frac{1}{2} \partial_{xx} v_y + \partial_\mu v_y \left( w_{0y} \bar{\mu} + w_{1y} \right) + \\ \partial_\nu v_y \left( w_{2y} \bar{\mu} + w_{3y} \bar{\nu} + w_{4y} \bar{\mu}^2 + w_{5y} \right) + \sum_{i=1}^{\kappa} q_{y,i} v_i + \bar{F}_y = 0, \qquad (3.3.2) \\ v_y (x, T, \mu_T, \nu_T) = \bar{G}_y (x, \mu_T, \nu_T), \ y \in \mathcal{Y}. \end{cases}$$

Furthermore, the optimal control has to admit the feedback form of

$$\hat{\alpha}(t) = -\tilde{b}_2(Y_t, t)\partial_x v(Y_t, \hat{X}_t, t, \mu_t, \nu_t).$$
(3.3.3)

Next, we identify what conditions are needed for equating the control problem (P3) and HJB equation. Denote

$$\mathbb{S} = \left\{ v \in C^{\infty} : \begin{array}{c} (1+|x|^2)^{-1}(|v|+|\partial_t v|) + \\ v \in C^{\infty} : & (1+|x|)^{-1}(|\partial_x v|+|\partial_\mu v|+|\partial_\nu v|) + |\partial_{xx} v| < K, \\ \forall (y, x, t, \mu, \nu), \text{ for some } K \end{array} \right\}$$

**Lemma 24.** (Verification theorem) Consider the control problem (P3) with some given smooth w. Suppose there exists a solution  $v \in \mathbb{S}$  of (3.3.2). Then,  $v_y(x, t, \bar{\mu}, \bar{\nu}) = \bar{V}(y, x, t, \bar{\mu}, \bar{\nu})$  holds, and an optimal control is provided by (3.3.3).

*Proof.* We first prove the verification theorem. Since  $v \in S$ , for any admissible  $\alpha \in L^4_{\mathbb{F}}$ , the process  $X^{\alpha}$  is well defined and one can use Dynkin's formula given by Lemma 34 to write

$$\mathbb{E}\left[v(Y_T, X_T, T, \mu_T, \nu_T)\right] = v(y, x, t, \bar{\mu}, \bar{\nu}) + \mathbb{E}\left[\int_t^T \mathcal{G}^{\alpha(s)} v(Y_s, X_s, s, \mu_s, \nu_s) ds\right],$$

where

$$\mathcal{G}^{a}f(y,x,s,\bar{\mu},\bar{\nu}) = \left(\partial_{t} + \left(\tilde{b}_{1y}x + \tilde{b}_{2y}a\right)\partial_{x} + \frac{1}{2}\partial_{xx} + \mathcal{Q} + \left(w_{0y}\bar{\mu} + w_{1y}\right)\partial_{\bar{\mu}} + \left(w_{2y}\bar{\mu} + w_{3y}\bar{\nu} + w_{4y}\bar{\mu}^{2} + w_{5y}\right)\partial_{\bar{\nu}}\right)f(y,x,s,\bar{\mu},\bar{\nu}).$$

Note that HJB actually implies that

$$\inf_{a} \left\{ \mathcal{G}^a v + \frac{1}{2}a^2 \right\} = -\bar{F},$$

which again implies

$$-\mathcal{G}^a v \le \frac{1}{2}a^2 + \bar{F}.$$

Hence, we obtain that for all  $\alpha \in L^4_{\mathbb{F}}$ ,

$$\begin{aligned} v(y, x, t, \bar{\mu}, \bar{\nu}) \\ &= \mathbb{E}\left[\int_{t}^{T} -\mathcal{G}^{\alpha(s)}v(Y_{s}, X_{s}, s, \mu_{s}, \nu_{s})ds\right] + \mathbb{E}\left[v(Y_{T}, X_{T}, T, \mu_{T}, \nu_{T})\right] \\ &\leq \mathbb{E}\left[\int_{t}^{T}\left(\frac{1}{2}\alpha^{2}(s) + \bar{F}(Y_{s}, X_{s}, \mu_{s}, \nu_{s})\right)ds\right] + \mathbb{E}\left[\bar{G}(Y_{T}, X_{T}, \mu_{T}, \nu_{T})\right] \\ &= J(y, x, t, \alpha, \bar{\mu}, \bar{\nu}). \end{aligned}$$

In the above, if  $\alpha$  is replaced by  $\hat{\alpha}$  given by the feedback form (3.3.3), then since  $\partial_x v$  is Lipschitz continuous in x, there exists corresponding optimal path  $\hat{X} \in L^4_{\mathbb{F}}$ . Thus,  $\hat{\alpha}$  is also in  $L^4_{\mathbb{F}}$ . One can repeat all above steps by replacing X and  $\alpha$  by  $\hat{X}$  and  $\hat{\alpha}$ , and  $\leq$  sign by = sign to conclude that v is indeed the optimal value.

#### 3.3.2.2 LQG solution

Note that, the costs  $\bar{F}$  and  $\bar{G}$  of (P3) are quadratic functions in  $(x, \bar{\mu}, \bar{\nu})$ , while the drift function of the process  $\nu$  of (3.3.1) is not linear in  $(x, \bar{\mu}, \bar{\nu})$ . Therefore, the control problem (P3) does not fall into the standard LQG control framework. Nevertheless, similar to the LQG solution, we guess the value function as a quadratic function in the form of

$$v_y(x,t,\bar{\mu},\bar{\nu}) = a_y(t)x^2 + d_y(t)x + e_y(t)\bar{\mu} + f_y(t)x\bar{\mu} + k_y(t)\bar{\mu}^2 + b_y(t)\bar{\nu} + c_y(t), \quad y \in \mathcal{Y}.$$
 (3.3.4)

With the above setup, for  $t \in [0, T]$ , the optimal control is

$$\hat{\alpha}_t = -\tilde{b}_2(Y_t, t)\partial_x v(Y_t, \hat{X}_t, t, \mu_t, \nu_t) = -\tilde{b}_2(Y_t, t)\left(2a_{Y_t}(t)\hat{X}_t + d_{Y_t}(t) + f_{Y_t}(t)\mu_t\right),$$
(3.3.5)

and the optimal path  $\hat{X}$  is

$$d\hat{X}_{t} = \left(\tilde{b}_{1}(Y_{t}, t)\hat{X}_{t} - \tilde{b}_{2}^{2}(Y_{t}, t)\left(2a_{Y_{t}}(t)\hat{X}_{t} + d_{Y_{t}}(t) + f_{Y_{t}}(t)\mu_{t}\right)\right)dt + dW_{t}.$$
(3.3.6)

Denote the following ODE systems for  $y \in \mathcal{Y}$ ,

$$\begin{cases} a'_{y} + 2\tilde{b}_{1y}a_{y} - 2\tilde{b}_{2y}^{2}a_{y}^{2} + \sum_{i=1}^{\kappa} q_{y,i}a_{i} + h_{y} = 0, \\ d'_{y} + \tilde{b}_{1y}d_{y} - 2\tilde{b}_{2y}^{2}a_{y}d_{y} + f_{y}w_{1y} + \sum_{i=1}^{\kappa} q_{y,i}d_{i} = 0, \\ e'_{y} - \tilde{b}_{2y}^{2}d_{y}f_{y} + 2k_{y}w_{1y} + e_{y}w_{0y} + b_{y}w_{2y} + \sum_{i=1}^{\kappa} q_{y,i}e_{i} = 0, \\ f'_{y} + \tilde{b}_{1y}f_{y} - 2\tilde{b}_{2y}^{2}a_{y}f_{y} + f_{y}w_{0y} + \sum_{i=1}^{\kappa} q_{y,i}f_{i} - 2h_{y} = 0, \\ k'_{y} - \frac{1}{2}\tilde{b}_{2y}^{2}f_{y}^{2} + 2k_{y}w_{0y} + b_{y}w_{4y} + \sum_{i=1}^{\kappa} q_{y,i}k_{i} = 0, \\ b'_{y} + b_{y}w_{3y} + \sum_{i=1}^{\kappa} q_{y,i}b_{i} + h_{y} = 0, \\ c'_{y} + a_{y} - \frac{1}{2}\tilde{b}_{2y}^{2}d_{y}^{2} + e_{y}w_{1y} + b_{y}w_{5y} + \sum_{i=1}^{\kappa} q_{y,i}c_{i} = 0, \end{cases}$$
(3.3.7)

with terminal conditions

$$a_y(T) = g_y, \ b_y(T) = g_y, \ c_y(T) = 0, \ d_y(T) = 0,$$
  

$$e_y(T) = 0, \ f_y(T) = -2g_y, \ k_y(T) = 0.$$
(3.3.8)

**Lemma 25.** Suppose there exists a unique solution  $(a_y, b_y, c_y, d_y, e_y, f_y, k_y : y \in \mathcal{Y})$  to the ODE

system (3.3.7)-(3.3.8) on [0,T]. Then the value function of (P3) is

$$V(y, x, t, \bar{\mu}, \bar{\nu}) = v_y(x, t, \bar{\mu}, \bar{\nu})$$
  
=  $a_y(t)x^2 + d_y(t)x + e_y(t)\bar{\mu} + f_y(t)x\bar{\mu} + k_y(t)\bar{\mu}^2 + b_y(t)\bar{\nu} + c_y(t)$  (3.3.9)

for  $y \in \mathcal{Y}$  and the optimal control and optimal path are given by (3.3.5) and (3.3.6), respectively.

*Proof.* With the form of value function  $v_y$  given in (3.3.4) and the first and second moment of the conditional population density given in (3.3.1), we have

$$\begin{aligned} \partial_t v_y &= a'_y(t)x^2 + d'_y(t)x + e'_y(t)\bar{\mu} + f'_y(t)x\bar{\mu} + k'_y(t)\bar{\mu}^2 + b'_y(t)\bar{\nu} + c'_y(t), \\ \partial_x v_y &= 2xa_y(t) + d_y(t) + f_y(t)\bar{\mu}, \\ \partial_{xx}v_y &= 2a_y(t), \\ \partial_{\bar{\mu}}v_y &= e_y(t) + f_y(t)x + 2k_y(t)\bar{\mu}, \\ \partial_{\bar{\nu}}v_y &= b_y(t), \end{aligned}$$

for  $y \in \mathcal{Y}$ . Plugging them back to the coupled HJBs in (3.3.2), we get a system of ODEs in (3.3.7) by equating  $x, \bar{\mu}, \bar{\nu}$ -like terms in each equation.

Therefore, any solution  $(a_y, b_y, c_y, d_y, e_y, f_y, k_y : y \in \mathcal{Y})$  of ODE system (3.3.7) leads to the solution of HJB (3.3.2) in the form of the quadratic function given by (3.3.9). Since the  $(a_y, b_y, c_y, d_y, e_y, f_y, k_y : y \in \mathcal{Y})$  are differentiable functions on the closed set [0, T], they are also bounded, and the function v meets regularity conditions required by Lemma 24 to conclude the desired result.

#### 3.3.3 Fixed point condition and the proof of Theorem 20

Going back to the ODE system (3.3.7), there are  $7\kappa$  equations, while we have total  $13\kappa$  deterministic functions of  $[0, T] \times \mathbb{R}$  to be determined to characterize MFG. Those are

$$(a_y, b_y, c_y, d_y, e_y, f_y, k_y : y \in \mathcal{Y})$$
 and  $(w_{iy} : i = 0, 1, \dots, 5, y \in \mathcal{Y}).$ 

In the following, we identify the missing  $6\kappa$  equations by checking the fixed point condition:

$$\mu_s = \mathbb{E}\left[\left.\hat{X}_s\right|Y\right], \ \nu_s = \mathbb{E}\left[\left.\hat{X}_s^2\right|Y\right], \quad \forall s \in [0,T],$$
(3.3.10)

where  $\mu$  and  $\nu$  are two auxiliary processes  $(\mu, \nu)[w]$  defined in (3.3.1), see Figure 3.2. This leads to a complete characterization of the equilibrium for the MFG posed by (P2).

Note that based on the dynamic of the optimal  $\hat{X}$  defined in (3.3.6), the fixed point condition (3.3.10) implies that the first moment  $\hat{\mu}_s := \mathbb{E}[\hat{X}_s | Y]$  and the second moment  $\hat{\nu}_s := \mathbb{E}[\hat{X}_s^2 | Y]$  of the optimal path conditioned on Y satisfy

$$\begin{cases} \hat{\mu}_s = \bar{\mu} + \int_t^s \left( \left( \tilde{b}_1(Y_r, r) - \tilde{b}_2^2(Y_r, r) \left( 2a_{Y_r}(r) + f_{Y_r}(r) \right) \right) \hat{\mu}_r - \tilde{b}_2^2(Y_r, r) d_{Y_r}(r) \right) dr, \\ \hat{\nu}_s = \bar{\nu} + \int_t^s \left( 1 + 2\tilde{b}_1(Y_r, r) \hat{\nu}_r - \tilde{b}_2^2(Y_r, r) \left( 4a_{Y_r}(r) \hat{\nu}_r + 2d_{Y_r}(r) \hat{\mu}_r + 2f_{Y_r}(r) \hat{\mu}_r^2 \right) \right) dr, \end{cases}$$
(3.3.11)

for  $s \ge t$ . Note that under the optimal control in (3.3.5), comparing the terms in (3.3.1) and (3.3.11), we obtain another  $6\kappa$  equations:

$$\begin{split} w_{0y} &= \tilde{b}_{1y} - 2\tilde{b}_{2y}^2 a_y - \tilde{b}_{2y}^2 f_y, \ w_{1y} = -\tilde{b}_{2y}^2 d_y, \ w_{2y} = -2\tilde{b}_{2y}^2 d_y, \\ w_{3y} &= -4\tilde{b}_{2y}^2 a_y + 2\tilde{b}_{1y}, \ w_{4y} = -2\tilde{b}_{2y}^2 f_y, \ w_{5y} = 1, \end{split}$$
(3.3.12)

for  $y \in \mathcal{Y}$ . Using further algebraic structures, one can reduce the ODE system of  $13\kappa$  equations composed by (3.3.7) and (3.3.12) into a system of  $4\kappa$  equations of the form (3.2.12) for the MFG characterization in Theorem 20.

**Proof of Theorem 20.** Since  $a_y$  ( $y \in \mathcal{Y}$ ) has the same expressions as (3.2.12), its existence, uniqueness and boundedness are shown in Lemma 38. Given  $a_y$  ( $y \in \mathcal{Y}$ ) and smooth bounded w's,

$$(b_y, d_y, e_y, f_y : y \in \mathcal{Y})$$

is a coupled linear system, and their existence, uniqueness and boundedness is shown by Theorem 12.1 in [2]. Similarly, given  $(b_y, d_y, f_y : y \in \mathcal{Y})$ ,  $(k_y, c_y : y \in \mathcal{Y})$  is a linear system, and their existence and uniqueness is also guaranteed by Theorem 12.1 in [2].

The ODE system (3.3.7) can be rewritten by

$$\begin{cases} a'_{y} + 2\tilde{b}_{1y}a_{y} - 2\tilde{b}_{2y}^{2}a_{y}^{2} + \sum_{i=1}^{\kappa} q_{y,i}a_{i} + h_{y} = 0, \\ d'_{y} + \tilde{b}_{1y}d_{y} - 2\tilde{b}_{2y}^{2}a_{y}d_{y} - \tilde{b}_{2y}^{2}f_{y}d_{y} + \sum_{i=1}^{\kappa} q_{y,i}d_{i} = 0, \\ e'_{y} - \tilde{b}_{2y}^{2}d_{y}f_{y} - 2\tilde{b}_{2y}^{2}k_{y}d_{y} + e_{y}\left(\tilde{b}_{1y} - 2\tilde{b}_{2y}^{2}a_{y} - \tilde{b}_{2y}^{2}f_{y}\right) - 2\tilde{b}_{2y}^{2}b_{y}d_{y} + \sum_{i=1}^{\kappa} q_{y,i}e_{i} = 0, \\ f'_{y} + \tilde{b}_{1y}f_{y} - 2\tilde{b}_{2y}^{2}a_{y}f_{y} + f_{y}\left(\tilde{b}_{1y} - 2\tilde{b}_{2y}^{2}a_{y} - \tilde{b}_{2y}^{2}f_{y}\right) + \sum_{i=1}^{\kappa} q_{y,i}f_{i} - 2h_{y} = 0, \\ k'_{y} - \frac{1}{2}\tilde{b}_{2y}^{2}f_{y}^{2} + 2k_{y}\left(\tilde{b}_{1y} - 2\tilde{b}_{2y}^{2}a_{y} - \tilde{b}_{2y}^{2}f_{y}\right) - 2\tilde{b}_{2y}^{2}b_{y}f_{y} + \sum_{i=1}^{\kappa} q_{y,i}k_{i} = 0, \\ b'_{y} + b_{y}\left(-4\tilde{b}_{2y}^{2}a_{y} + 2\tilde{b}_{1y}\right) + \sum_{i=1}^{\kappa} q_{y,i}b_{i} + h_{y} = 0, \\ c'_{y} + a_{y} - \frac{1}{2}\tilde{b}_{2y}^{2}d_{y}^{2} - 2\tilde{b}_{2y}^{2}d_{y}e_{y} + b_{y} + \sum_{i=1}^{\kappa} q_{y,i}c_{i} = 0, \end{cases}$$

with the terminal conditions

$$a_y(T) = g_y, \ b_y(T) = g_y, \ c_y(T) = 0, \ d_y(T) = 0, \ e_y(T) = 0, \ f_y(T) = -2g_y, \ k_y(T) = 0.$$

Since  $a_y, b_y$   $(y \in \mathcal{Y})$  has the same expressions as (3.2.12), its existence, uniqueness and boundedness are shown in Lemma 38. Meanwhile, with the given  $(a_y, b_y : y \in \mathcal{Y})$ , we denote  $l_y = 2a_y + f_y$ , and then

$$l'_{y} + 2\tilde{b}_{1y}l_{y} - \tilde{b}_{2y}^{2}l_{y}^{2} + \sum_{i=1}^{\kappa} q_{y,i}l_{i} = 0 , \ l_{y}(T) = 0.$$

By Lemma 36 and Lemma 37 in Appendix, there exists a unique solution for  $l_y$   $(y \in \mathcal{Y})$ , which is  $l_y = 0, y \in \mathcal{Y}$ . This gives  $f_y = -2a_y$  and  $d'_y + \tilde{b}_{1y}d_y + \sum_{i=1}^{\kappa} q_{y,i}d_i = 0$ , which implies  $d_y = 0, y \in \mathcal{Y}$ . Then, the equation for  $e_y$  can be simplified as  $e'_y + \tilde{b}_{1y}e_y + \sum_{i=1}^{\kappa} q_{y,i}e_i = 0$ , which indicates that  $e_y = 0, y \in \mathcal{Y}$ . For  $k_y, c_y$ , with the given of  $(a_y, b_y : y \in \mathcal{Y})$ , we have

$$k'_{y} + 2\tilde{b}_{1y}k_{y} - 2\tilde{b}_{2y}^{2}a_{y}^{2} + 4\tilde{b}_{2y}^{2}a_{y}b_{y} + \sum_{i=1}^{\kappa} q_{y,i}k_{i} = 0 , \ k_{y}(T) = 0,$$
  
$$c'_{y} + a_{y} + b_{y} + \sum_{i=1}^{\kappa} q_{y,i}c_{i} = 0 , \ c_{y}(T) = 0.$$

The existence and uniqueness of the solution for  $k_y, c_y \ (y \in \mathcal{Y})$  are yielded by Theorem 12.1 in [2].

Note that in this case, since  $2a_y + f_y = 0$  and  $d_y = 0$  for  $y \in \mathcal{Y}$ , from (3.3.11) we have

$$\hat{\mu}_s = \bar{\mu} + \int_t^s \tilde{b}_1(Y_r, r)\hat{\mu}_r \ dr$$

for all  $s \in [t, T]$ . Then

$$\hat{\nu}_s = \bar{\nu} + \int_t^s \left( 1 + 2\tilde{b}_1(Y_r, r)\hat{\nu}_r - 4\tilde{b}_2^2(Y_r, r)a_{Y_r}(r)\hat{\nu}_r + 4\tilde{b}_2^2(Y_r, r)a_{Y_r}(r)\hat{\mu}_r^2 \right) dr.$$

Plugging  $d_y = 0$  for  $y \in \mathcal{Y}$  back to (3.3.5), we obtain the optimal control by

$$\hat{\alpha}_s = -2\tilde{b}_2^2(Y_s, s)a_{Y_s}(s)\left(\hat{X}_s - \hat{\mu}_s\right)$$

Since we have  $d_y = 0$  for  $y \in \mathcal{Y}$ , the value function can be simplified from (3.3.4) to

$$v_y(x,t,\bar{\mu},\bar{\nu}) = a_y(t)x^2 - 2a_y(t)x\bar{\mu} + k_y(t)\bar{\mu}^2 + b_y(t)\bar{\nu} + c_y(t)\bar{\nu}$$

By the equivalence Lemma 23, it yields the value function U of Theorem 20. Moreover, since  $f_y = -2a_y$  and  $k_y \neq 0$ , the ODE system (3.3.7) together with (3.3.12) can be reduced into (3.2.12). From the Lemma 38, the existence and uniqueness of  $(a_y, b_y, c_y, k_y : y \in \mathcal{Y})$  in (3.2.12) is guaranteed.

# 3.4 The *N*-player game and its convergence to MFG

In this section, we show the convergence of the N-player game to MFG. To simplify the presentation, we may omit the superscript (N) for the processes in the probability space  $\Omega^{(N)}$ , whenever there is no confusion. First, we solve the N-player game in Subsection 3.4.1, which provides a Riccati system consisting of  $O(N^3)$  equations. Subsection 3.4.2 reduces the corresponding Riccati system

into an ODE system whose dimension is independent of N. This becomes the key building block of the convergence rate obtained in Subsection 3.4.3. To obtain the convergence rate, Subsection 3.4.3 provides an explicit embedding of some processes in  $\Omega^{(N)}$  into the probability space  $\Omega$ . Note that,  $\Omega^{(N)}$  is much richer than  $\Omega$  since  $\Omega^{(N)}$  contains N Brownian motions while  $\Omega$  has only two Brownian motions. Therefore, careful treatment has to be carried out to some processes of our interest, otherwise, such an embedding is in general implausible.

#### 3.4.1 Characterization of the *N*-player game by Riccati system

The N-player game is indeed an N-coupled stochastic LQG problem by its very own definition, see Subsection 3.2.3. Therefore, the solution can be derived via Riccati system with the existing LQG theory given below: For  $i = 1, 2, ..., N, y \in \mathcal{Y}$ ,

$$\begin{cases} A'_{iy} + 2\tilde{b}_{1y}e_{i}e_{i}^{\top}A_{iy} - 2\tilde{b}_{2y}^{2}A_{iy}^{\top}e_{i}e_{i}^{\top}A_{iy} + \sum_{j\neq i}^{N} \left(2\tilde{b}_{1y}e_{j}e_{j}^{\top}A_{iy} - 4\tilde{b}_{2y}^{2}A_{jy}^{\top}e_{j}e_{j}^{\top}A_{iy}\right) \\ + \sum_{j=1}^{\kappa} q_{y,j}A_{ij} + \frac{h_{y}}{N}\sum_{j\neq i}^{N} \left(e_{i} - e_{j}\right)\left(e_{i} - e_{j}\right)^{\top} = 0, \\ B'_{iy} + \sum_{j\neq i}^{N} \left(\tilde{b}_{1y}e_{j}e_{j}^{\top}B_{iy} - 2\tilde{b}_{2y}^{2}A_{iy}^{\top}e_{j}e_{j}^{\top}B_{jy} - 2\tilde{b}_{2y}^{2}A_{jy}^{\top}e_{j}e_{j}^{\top}B_{iy}\right) \\ + \tilde{b}_{1y}e_{i}e_{i}^{\top}B_{iy} - 2\tilde{b}_{2y}^{2}A_{iy}^{\top}e_{i}e_{i}^{\top}B_{iy} + \sum_{j=1}^{\kappa} q_{y,j}B_{ij} = 0, \\ C'_{iy} - \frac{1}{2}\tilde{b}_{2y}^{2}B_{iy}^{\top}e_{i}e_{i}^{\top}B_{iy} - \sum_{j\neq i}^{N}\tilde{b}_{2y}^{2}B_{jy}^{\top}e_{j}e_{j}^{\top}B_{iy} + \sum_{j=1}^{N} tr(A_{jy}) + \sum_{j=1}^{\kappa} q_{y,j}C_{ij} = 0, \\ A_{iy}(T) = \frac{g_{y}}{N}\Lambda_{i}, \ B_{iy}(T) = 0 \cdot \mathbbm{N}, \ C_{iy}(T) = 0, \end{cases}$$

$$(3.4.1)$$

where the solutions consist of  $N \times N$  symmetric matrices  $A_{iy}$ 's, N-dimensional vectors  $B_{iy}$ 's, and  $C_{iy} \in \mathbb{R}$ . In the above,  $\mathbb{1}_N$  is the N-dimensional vector with all entries are 1,  $\Lambda_i$ 's are  $N \times N$  matrices with diagonal 1 except  $(\Lambda_i)_{ii} = N - 1$ ,  $(\Lambda_i)_{ij} = (\Lambda_i)_{ji} = -1$  for any  $j \neq i$  and the rest entries as 0, and  $e_i$ 's are the N-dimensional natural basis.

**Lemma 26.** Suppose  $(A_{iy}, B_{iy}, C_{iy} : i = 1, 2, ..., N, y \in \mathcal{Y})$  is the solution of (3.4.1). Then, the value functions of the N-player game defined by (3.2.9) are

$$V_i(y, x^{(N)}) = (x^{(N)})^\top A_{iy}(0) x^{(N)} + (x^{(N)})^\top B_{iy}(0) + C_{iy}(0), \quad i = 1, 2, \dots, N$$

Moreover, the path and the control under the equilibrium are

$$d\hat{X}_{it} = \left(\tilde{b}_1(Y_t, t)\hat{X}_{it} - \tilde{b}_2^2(Y_t, t)\left(2(A_{iY_t})_i^{\top}\hat{X}_t + (B_{iY_t})_i\right)\right)dt + dW_{it}, \quad i = 1, 2, \dots, N,$$
(3.4.2)

and

$$\hat{\alpha}_{it} = -\tilde{b}_2(Y_t, t) \left( 2(A_{iY_t})_i^\top \hat{X}_t + (B_{iY_t})_i \right),$$

where  $(A)_i$  denotes the *i*-th column of matrix A,  $(B)_i$  denotes the *i*-th entry of vector B and  $\hat{X}_t = [\hat{X}_{1t}, \hat{X}_{2t}, \dots, \hat{X}_{Nt}]^\top$ .

*Proof.* It is standard that, under enough regularities, the value function  $V(y, x^{(N)}) = (V_1, V_2, \ldots, V_N)$  $(y, x^{(N)})$  of the N-player game can be lifted to the solution  $v_{iy}(x^{(N)}, t)$  of the following system of HJB equations, for  $i = 1, 2, \ldots, N$  and  $y \in \mathcal{Y}$ ,

$$\begin{cases} \partial_t v_{iy} + \tilde{b}_{1y} x_i \partial_i v_{iy} - \frac{1}{2} \left( \tilde{b}_{2y} \partial_i v_{iy} \right)^2 + \sum_{j \neq i}^N \left( \tilde{b}_{1y} x_j - \tilde{b}_{2y}^2 \partial_j v_{jy} \right) \partial_j v_{iy} \\ + \frac{1}{2} \Delta v_{iy} + \sum_{j=1}^\kappa q_{y,j} v_{ij} + \frac{h_y}{N} \sum_{j \neq i}^N \left( (e_i - e_j)^\top x^{(N)} \right)^2 = 0, \qquad (3.4.3) \\ v_{iy} (x^{(N)}, T) = \frac{g_y}{N} \sum_{j \neq i}^N \left( (e_i - e_j)^\top x^{(N)} \right)^2. \end{cases}$$

Then, the value functions V of the N-player game defined by (3.2.9) is  $V_i(y, x^{(N)}) = v_{iy}(x^{(N)}, 0)$ for all i = 1, 2, ..., N. Moreover, the path and the control under the equilibrium are

$$d\hat{X}_{it} = \left(\tilde{b}_1(Y_t, t)\hat{X}_{it} - \tilde{b}_2^2(Y_t, t)\partial_i v_{iY_t}(\hat{X}_t, t)\right)dt + dW_{it}, \quad i = 1, 2, \dots, N,$$

and

$$\hat{\alpha}_{it} = -\tilde{b}_2(Y_t, t)\partial_i v_{iY_t}(\hat{X}_t, t).$$

The proof is the application of Dynkin's formula and the details are omitted here. Due to its LQG structure, the value function leads to a quadratic function of the form

$$v_{iy}(x^{(N)},t) = (x^{(N)})^{\top} A_{iy}(t) x^{(N)} + (x^{(N)})^{\top} B_{iy}(t) + C_{iy}(t).$$

For each i = 1, 2, ..., N, after plugging  $V_{iy}$  into (3.4.3), and matching the coefficient of variables, we get the desired results.

#### 3.4.2 Reduced Riccati form for the equilibrium

So far, the N-player game and MFG have been characterized by Lemma 26 and Theorem 20, respectively. One of our main objectives is to investigate the convergence of the generic optimal path  $\hat{X}_{1t}^{(N)}$  of the N-player game generated (3.4.1)-(3.4.2) to the optimal path  $\hat{X}_t$  of MFG generated by (3.2.12)-(3.2.13).

Note that  $\hat{X}_t$  relies only on  $\kappa$  functions  $(a_y : y \in \mathcal{Y})$  from the simple ODE system (3.2.12) while  $\rho(\hat{X}_t^{(N)})$  depends on  $O(N^3)$  functions from  $(A_{iy} : i = 1, 2, ..., N, y \in \mathcal{Y})$  solved from a huge Riccati system (3.4.1). Therefore, it is almost a hopeless task for a meaningful comparison between these two processes without gaining further insight into the complex structure of the Riccati system (3.4.1).

To proceed, let us first observe some hidden patterns from a numerical result for the solution of Riccati (3.4.1). The following matrix shows  $A_{20}$  at t = 1 for N = 5 with the same parameters
as in Figure 3.3 and Figure 3.4 in Section 3.5.1:

$$A_{20}(1) = \begin{bmatrix} 0.1319 & -0.1924 & 0.0202 & 0.0202 & 0.0202 \\ -0.1924 & 0.7696 & -0.1924 & -0.1924 & -0.1924 \\ 0.0202 & -0.1924 & 0.1319 & 0.0202 & 0.0202 \\ 0.0202 & -0.1924 & 0.0202 & 0.1319 & 0.0202 \\ 0.0202 & -0.1924 & 0.0202 & 0.0202 & 0.1319 \end{bmatrix}.$$

Interestingly enough, we observe that the entire 25 entries of  $A_{20}(1)$  indeed consists of 4 distinct values. Moreover, similar computation with different values of N only yields a larger table depending on N, but always consists of 4 values. Inspired by this accidental discovery from the above numerical example, we may want to believe and prove a pattern of the matrix  $A_{iy}$  in the following form:

$$(A_{iy})_{pq} = \begin{cases} a_{1y}(t), & \text{if } p = q = i, \\ a_{2y}(t), & \text{if } p = q \neq i, \\ a_{3y}(t), & \text{if } p \neq q, p = i \text{ or } q = i, \\ a_{4y}(t), & \text{otherwise}, \end{cases}$$
(3.4.4)

for  $y \in \mathcal{Y}$ . The next result justifies the above pattern: the  $N^2$  entries of the matrix  $A_{iy}$  can be embedded to a  $2\kappa$ -dimensional vector space no matter how big N is.

**Lemma 27.** There exists a unique solution  $(a_{1y}^N, a_{2y}^N)$  from the ODE system (3.4.5)

$$\begin{cases} a'_{1y} + 2\tilde{b}_{1y}a_{1y} - \frac{2(N+1)}{N-1}\tilde{b}_{2y}^2a_{1y}^2 + \sum_{j=1}^{\kappa}q_{y,j}a_{1j} + \frac{N-1}{N}h_y = 0, \\ a'_{2y} + 2\tilde{b}_{1y}a_{2y} + \frac{2}{(N-1)^2}\tilde{b}_{2y}^2a_{1y}^2 - \frac{4N}{N-1}\tilde{b}_{2y}^2a_{1y}a_{2y} + \sum_{j=1}^{\kappa}q_{y,j}a_{2j} + \frac{h_y}{N} = 0, \\ a_{1y}(T) = \frac{N-1}{N}g_y, \ a_{2y}(T) = \frac{g_y}{N}, \end{cases}$$
(3.4.5)

for  $y \in \mathcal{Y}$ . Moreover, the path and the control of player i under the equilibrium are

$$d\hat{X}_{it}^{(N)} = \left(\tilde{b}_1(Y_t^{(N)}, t)\hat{X}_{it}^{(N)} - 2\tilde{b}_2^2(Y_t^{(N)}, t)a_{1Y_t^{(N)}}^N(t)\left(\hat{X}_{it}^{(N)} - \frac{1}{N-1}\sum_{j\neq i}^N\hat{X}_{jt}^{(N)}\right)\right)dt + dW_{it}^{(N)},$$
(3.4.6)

and

$$\hat{\alpha}_{it}^{(N)} = -2\tilde{b}_2(Y_t^{(N)}, t)a_{1Y_t^{(N)}}^N(t) \left(\hat{X}_{it}^{(N)} - \frac{1}{N-1}\sum_{j\neq i}^N \hat{X}_{jt}^{(N)}\right)$$

for i = 1, 2, ..., N.

*Proof.* It is obvious to see that in the Riccati system (3.4.1),  $B_{iy} = 0$  for all i = 1, 2, ..., N and

 $y \in \mathcal{Y}$ . Note that in this case, for  $i = 1, 2, \ldots, N$ , the optimal control is given by

$$\hat{\alpha}_{i}^{(N)} = -2\tilde{b}_{2}(Y_{t}^{(N)}, t) \sum_{j=1}^{N} (A_{iY_{t}^{(N)}})_{ij} \hat{X}_{jt}^{(N)} = -2\tilde{b}_{2}(Y_{t}^{(N)}, t) \left(A_{iY_{t}^{(N)}}\right)_{i}^{\top} \hat{X}_{t}^{(N)}.$$

Plugging the pattern (3.4.4) into the differential equation of  $A_{iy}$ , we have

$$\begin{aligned} a_{1y}' + 2\tilde{b}_{1y}a_{1y} - 2\tilde{b}_{2y}^2a_{1y}^2 - 4(N-1)\tilde{b}_{2y}^2a_{3y}^2 + \sum_{j=1}^{\kappa} q_{y,j}a_{1j} + \frac{N-1}{N}h_y &= 0, \\ a_{2y}' + 2\tilde{b}_{1y}a_{2y} - 2\tilde{b}_{2y}^2a_{3y}^2 - 4\tilde{b}_{2y}^2\left(a_{1y}a_{2y} + (N-2)a_{3y}a_{4y}\right) + \sum_{j=1}^{\kappa} q_{y,j}a_{2j} + \frac{h_y}{N} &= 0, \\ a_{3y}' + 2\tilde{b}_{1y}a_{3y} - 2\tilde{b}_{2y}^2a_{1y}a_{3y} - 4\tilde{b}_{2y}^2\left(a_{1y}a_{3y} + (N-2)a_{3y}^2\right) + \sum_{j=1}^{\kappa} q_{y,j}a_{3j} - \frac{h_y}{N} &= 0, \\ a_{3y}' + 2\tilde{b}_{1y}a_{3y} - 2\tilde{b}_{2y}^2a_{1y}a_{3y} - 4\tilde{b}_{2y}^2\left(a_{2y}a_{3y} + (N-2)a_{3y}a_{4y}\right) + \sum_{j=1}^{\kappa} q_{y,j}a_{3j} - \frac{h_y}{N} &= 0, \\ a_{4y}' + 2\tilde{b}_{1y}a_{4y} - 2\tilde{b}_{2y}^2a_{3y}^2 - 4\tilde{b}_{2y}^2\left(a_{2y}a_{3y} + a_{1y}a_{4y} + (N-3)a_{3y}a_{4y}\right) + \sum_{j=1}^{\kappa} q_{y,j}a_{4j} &= 0, \end{aligned}$$

which gives  $a_{1y} + (N-2)a_{3y} = a_{2y} + (N-2)a_{4y}$  since two expressions for  $a_{3y}$  should be identical. This implies that  $(a_{1y} + (N-2)a_{3y})' = (a_{2y} + (N-2)a_{4y})'$  or

$$\begin{split} &-2\tilde{b}_{1y}a_{1y}+2\tilde{b}_{2y}^2a_{1y}^2+4(N-1)\tilde{b}_{2y}^2a_{3y}^2-\frac{N-1}{N}h_y-\sum_{j=1}^{\kappa}q_{y,j}a_{1j}\\ &+(N-2)\Big(-2\tilde{b}_{1y}a_{3y}+2\tilde{b}_{2y}^2a_{1y}a_{3y}+4\tilde{b}_{2y}^2\left(a_{2y}a_{3y}+(N-2)a_{3y}a_{4y}\right)-\sum_{j=1}^{\kappa}q_{y,j}a_{3j}+\frac{h_y}{N}\Big)\\ &=-2\tilde{b}_{1y}a_{2y}+2\tilde{b}_{2y}^2a_{3y}^2+4\tilde{b}_{2y}^2\left(a_{1y}a_{2y}+(N-2)a_{3y}a_{4y}\right)-\sum_{j=1}^{\kappa}q_{y,j}a_{2j}-\frac{h_y}{N}\\ &+(N-2)\Big(-2\tilde{b}_{1y}a_{4y}+2\tilde{b}_{2y}^2a_{3y}^2+4\tilde{b}_{2y}^2\left(a_{1y}a_{4y}+a_{2y}a_{3y}+(N-3)a_{3y}a_{4y}\right)-\sum_{j=1}^{\kappa}q_{y,j}a_{4j}\Big). \end{split}$$

After combining terms and substituting  $a_{2y} + (N-2)a_{4y}$  with  $a_{1y} + (N-2)a_{3y}$ , we get  $a_{1y}^2 + (N-2)a_{1y}a_{3y} - (N-1)a_{3y}^2 = 0$ , which yields  $a_{3y} = a_{1y}$  or  $a_{3y} = -\frac{1}{N-1}a_{1y}$ . Note that  $a_{3y} \neq a_{1y}$  due to their differential equations. Hence, we can conclude that  $a_{3y} = -\frac{1}{N-1}a_{1y}$ . In conclusion,

for i = 1, 2, ..., N,  $A_{iy}$   $(y \in \mathcal{Y})$  has the following expressions:

$$(A_{iy})_{pq} = \begin{cases} a_{1y}(t), & \text{if } p = q = i, \\ a_{2y}(t), & \text{if } p = q \neq i, \\ -\frac{1}{N-1}a_{1y}(t), & \text{if } p \neq q, p = i \text{ or } q = i \\ \frac{1}{(N-1)(N-2)}a_{1y}(t) - \frac{1}{N-2}a_{2y}(t), & \text{otherwise.} \end{cases}$$

The existence and uniqueness of (3.4.1) is equivalent to the existence and uniqueness of (3.4.5). For  $a_{1y}$ , the existence and uniqueness can be deduced from Lemma 36 and 37. Given  $a_{1y}$ 's,  $a_{2y}$ 's are linear equations, thus their existence and uniqueness are guaranteed by Theorem 12.1 in [2]. Together with previous discussions, we conclude the results.

#### 3.4.3 Convergence

Based on the current progress, let us reiterate our goal (P1) for the convergence. Our objective is the convergence of the joint distribution  $\mathcal{L}(\hat{X}_{1t}^{(N)}, Y_t^{(N)})$  of the *N*-player game generated (3.4.5)-(3.4.6) in the probability space  $\Omega^{(N)}$  to the distribution  $\mathcal{L}(\hat{X}_t, Y_t)$  of MFG generated by (3.2.12)-(3.2.13) in the probability space  $\Omega$ . More precisely, we want to find a number  $\eta > 0$  satisfying

$$\mathbb{W}_2\left(\mathcal{L}(\hat{X}_{1t}^{(N)}, Y_t^{(N)}), \mathcal{L}(\hat{X}_t, Y_t)\right) = O\left(N^{-\eta}\right),\tag{3.4.7}$$

where  $\mathbb{W}_2$  is the 2-Wasserstein metric. This procedure is given in the following two steps:

1. We will construct a process  $Z^N$  in the probability space  $\Omega$ , who provides exact copy of the joint distribution in the sense of

$$\mathcal{L}(\hat{X}_{1t}^{(N)}, Y_t^{(N)}) = \mathcal{L}(Z_t^N, Y_t), \quad \forall t \in [0, T].$$

Note that, the (3.4.6) shows that  $\hat{X}_{1t}^{(N)}$  correlates to N many Brownian motions  $\{W_i^{(N)} : i = 1, 2, \ldots, N\}$  from a much richer space  $\Omega^{(N)}$  while  $\Omega$  is a much smaller space having only two Brownian motions W and B. Therefore, such an embedding essentially requires to represent  $\hat{X}_{1t}^{(N)}$  by two independent Brownian motions and is in general not possible. However, due to the symmetric structure of MFG (or the nature of the mean field effect), the embedding is possible and the details are provided in Lemma 28.

2. By Proposition 16, we can use distribution copy  $(Z^N, Y)$  in  $\Omega$  to write

$$\mathbb{W}_2^2\left(\mathcal{L}(\hat{X}_{1t}^{(N)}, Y_t^{(N)}), \mathcal{L}(\hat{X}_t, Y_t)\right) \le \mathbb{E}\left[\left|Z_t^N - \hat{X}_t\right|^2\right].$$
(3.4.8)

To obtain the estimate of the above right hand side, we shall compare the (3.4.9) of  $Z^N$  and (3.2.13) of  $\hat{X}$ , and it becomes essential to obtain the convergence rate of the ODE system (3.4.5) towards the ODE system (3.2.12). The details are provided in Lemma 29.

**Lemma 28.** Let  $\{X_0^i : i \in \mathbb{N}\}$  be *i.i.d.* random variables in  $\Omega$  independent to (W, B, Y) with  $X_0^1 = X_0$ . Let  $Z^N$  be the solution of

$$Z_t^N = X_0 + \int_0^t \tilde{b}_1(Y_s, s) Z_s^N ds - \int_0^t 2\tilde{b}_2^2(Y_s, s) \hat{a}_{1Y_s}^N(s) \left(Z_s^N - \bar{X}_s^N\right) ds + W_t,$$
(3.4.9)

where

$$d\bar{X}_{t}^{N} = \tilde{b}_{1}(Y_{t}, t)\bar{X}_{t}^{N}dt + \frac{\sqrt{N-1}}{N}dB_{t} + \frac{1}{N}dW_{t}, \quad \bar{X}_{0}^{N} = \frac{1}{N}\sum_{i=1}^{N}X_{0}^{i},$$

and

$$\hat{a}_{1y}^N = \frac{N}{N-1} a_{1y}^N$$

In the above,  $a_{1y}^N$  is from the ODE system(3.4.5). Then,  $(Z_t^N, Y_t)$  in  $(\Omega, \mathcal{F}_T, \mathbb{P})$  has the same distribution as  $(\hat{X}_{1t}^{(N)}, Y_t^{(N)})$  in  $(\Omega^{(N)}, \mathcal{F}_T^{(N)}, \mathbb{P}^{(N)})$ .

Proof. Continued from the Lemma 27, player i's path in the N-player game follows

$$\begin{split} \hat{X}_{it}^{(N)} &= x_i^{(N)} + \int_0^t \tilde{b}_1(Y_s^{(N)}, s) \hat{X}_{is}^{(N)} ds \\ &- \int_0^t 2 \tilde{b}_2^2(Y_s^{(N)}, s) a_{1Y_s^{(N)}}^N(s) \left( \hat{X}_{is}^{(N)} - \frac{1}{N-1} \sum_{j \neq i}^N \hat{X}_{js}^{(N)} \right) ds + W_{it}^{(N)}. \end{split}$$

With the notation

$$\bar{X}_{s}^{(N)} = \frac{1}{N} \sum_{i=1}^{N} \hat{X}_{is}^{(N)},$$

one can rewrite the path by

$$\hat{X}_{it}^{(N)} = x_i^{(N)} + \int_0^t \tilde{b}_1(Y_s^{(N)}, s) \hat{X}_{is}^{(N)} ds 
- \int_0^t 2\tilde{b}_2^2(Y_s^{(N)}, s) \hat{a}_{1Y_s^{(N)}}^N(s) \left(\hat{X}_{is}^{(N)} - \bar{X}_s^{(N)}\right) ds + W_{it}^{(N)}.$$
(3.4.10)

By adding up the above equations (3.4.10) indexed by i = 1 to N, one can have

$$\bar{X}_{t}^{(N)} = \bar{x}^{(N)} + \int_{0}^{t} \tilde{b}_{1}(Y_{s}^{(N)}, s) \bar{X}_{s}^{(N)} ds + \frac{1}{N} \sum_{i=1}^{N} W_{it}^{(N)}$$

$$= \bar{x}^{(N)} + \int_{0}^{t} \tilde{b}_{1}(Y_{s}^{(N)}, s) \bar{X}_{s}^{(N)} ds + \frac{\sqrt{N-1}}{N} \left(\sqrt{N-1}\bar{W}_{-it}^{(N)}\right) + \frac{1}{N} W_{it}^{(N)},$$
(3.4.11)

where  $\bar{W}_{-it}^{(N)} := \frac{1}{N-1} \sum_{j \neq i} W_{jt}^{(N)}$ .

Next, we define solution maps of (3.4.10) and (3.4.11):

$$\bar{G}_t(x,\phi,W_1,W_2) = \mathcal{E}_t(\phi) \left( x + \int_0^t \mathcal{E}_s(-\phi)d(W_{1s} + W_{2s}) \right)$$
(3.4.12)

and

$$G_t(x,\phi_1,\phi_2,\phi_3,W) = x\mathcal{E}_t(\phi_1 - \phi_2) + \mathcal{E}_t(\phi_1 - \phi_2) \int_0^t \mathcal{E}_s(-\phi_1 + \phi_2) \left(\phi_2(s)\phi_3(s)ds + dW_s\right), \quad (3.4.13)$$

where

$$\mathcal{E}_t(\phi) = \exp\left\{\int_0^t \phi_s ds\right\}.$$

Now, we can rewrite  $\bar{X}_t^{(N)}$  of (3.4.11) and  $\hat{X}_{1t}^{(N)}$  of (3.4.10) as

$$\bar{X}_{t}^{(N)} = \bar{G}_{t} \left( \frac{1}{N} \sum_{i=1}^{N} x_{i}^{(N)}, \tilde{b}_{1}(Y_{\cdot}^{(N)}, \cdot), \frac{\sqrt{N-1}}{N} \left( \sqrt{N-1} \bar{W}_{-1}^{(N)} \right), \frac{1}{N} W_{1}^{(N)} \right),$$

and

$$\hat{X}_{1t}^{(N)} = G_t \left( x_1^{(N)}, \tilde{b}_1(Y_{\cdot}^{(N)}, \cdot), 2\tilde{b}_2(Y_{\cdot}^{(N)}, \cdot) \hat{a}_1^N(Y_{\cdot}^{(N)}, \cdot), \bar{X}^{(N)}(\cdot), W_1^{(N)} \right)$$

Meanwhile,  $(Z^N, \overline{X}^N)$  of (3.4.9) can also be written in the form of

$$\bar{X}_{t}^{N} = \bar{G}_{t} \left( \frac{1}{N} \sum_{i=1}^{N} X_{0}^{i}, \tilde{b}_{1}(Y_{\cdot}, \cdot), \frac{\sqrt{N-1}}{N} B, \frac{1}{N} W \right),$$

and

$$Z_t^N = G_t\left(X_0, \tilde{b}_1(Y_{\cdot}, \cdot), 2\tilde{b}_2(Y_{\cdot}, \cdot)\hat{a}_1^N(Y_{\cdot}, \cdot), \bar{X}^N(\cdot), W\right)$$
(3.4.14)

Finally, the fact that the distribution of  $(Z^N, Y)$  in the space  $\Omega$  is identical distribution to  $(\hat{X}_1^{(N)}, Y^{(N)})$  in  $\Omega^{(N)}$  comes from the followings:

- $\tilde{b}_1, \tilde{b}_2, \hat{a}_1^N$  are deterministic functions.
- The random processes  $(\sqrt{N-1}\overline{W}_{-1}^{(N)}, W_1^{(N)}, Y^{(N)})$  are independent mutually in  $\Omega^{(N)}$ , while the random elements (B, W, Y) are also independent triples. Moreover, two random triples have identical joint distributions.
- Initial states are generated from identical joint distributions  $\{x_i^{(N)} : i = 1, 2, ..., N\}$  and  $\{X_0^i : i = 1, 2, ..., N\}$ .

Therefore,  $(Z^N, Y)$  and  $(\hat{X}_1^{(N)}, Y^{(N)})$  have the same distributions. This completes the proof.  $\Box$ 

In view of (3.4.8), we shall estimate the second moment  $\mathbb{E}[|Z_t^N - \hat{X}_t|^2]$ . First, we can rewrite  $\hat{X}$  of (3.2.13) using above representations via  $G_t$ :

$$\hat{X}_t = G_t\left(X_0, \tilde{b}_1(Y_{\cdot}, \cdot), 2\tilde{b}_2(Y_{\cdot}, \cdot)a(Y_{\cdot}, \cdot), \hat{\mu}(\cdot), W\right)$$

which leads to a better comparison with  $Z^N$  in the form of (3.4.14). To proceed, a Lipschitz properties of  $G_t$  are useful for the estimate of the second moment, whose proof is relegated to the Appendix 3.7.3. Throughout the proof of the next lemma, we will use K in various places as a generic constant which varies from line to line.

**Lemma 29.** The convergence rate under the Wasserstein metric  $\mathbb{W}_2(\cdot, \cdot)$  is

$$\mathbb{W}_2\left(\mathcal{L}(\hat{X}_{1t}^{(N)}, Y_t^{(N)}), \mathcal{L}(\hat{X}_t, Y_t)\right) = O\left(N^{-\frac{1}{2}}\right).$$

*Proof.* In view of (3.4.8), we start with

$$\begin{split} \mathbb{W}_{2}^{2} \left( \mathcal{L}(\hat{X}_{1t}^{(N)}, Y_{t}^{(N)}), \mathcal{L}(\hat{X}_{t}, Y_{t}) \right) &\leq \mathbb{E} \left[ \left| Z_{t}^{N} - \hat{X}_{t} \right|^{2} \right] \\ &= \mathbb{E} \left[ \left| G_{t} \left( X_{0}, \tilde{b}_{1}(Y_{\cdot}, \cdot), 2\tilde{b}_{2}(Y_{\cdot}, \cdot) \hat{a}_{1}^{N}(Y_{\cdot}, \cdot), \bar{X}^{N}(\cdot), W \right) - G_{t} \left( X_{0}, \tilde{b}_{1}(Y_{\cdot}, \cdot), 2\tilde{b}_{2}(Y_{\cdot}, \cdot) a(Y_{\cdot}, \cdot), \hat{\mu}(\cdot), W \right) \right|^{2} \right] \\ &:= \mathbb{E} \left[ \left| I_{1}(t) - I_{2}(t) \right|^{2} \right]. \end{split}$$

Applying the Lipschitz continuity of  $(\phi_2, \phi_3) \mapsto G_t(x, \phi_1, \phi_2, \phi_3, W)$  by Appendix 3.7.3 on the conditional expectation  $\mathbb{E}\left[ |I_1(t) - I_2(t)|^2 |Y \right]$ , we have

$$\begin{split} \mathbb{E}\left[|Z_{t}^{N}-\hat{X}_{t}|^{2}\right] &\leq K\mathbb{E}\left[\sup_{0\leq t\leq T}\left(2\tilde{b}_{2}(Y_{t},t)\hat{a}_{1Y_{t}}^{N}(t)-2\tilde{b}_{2}(Y_{t},t)a_{Y_{t}}(t)\right)^{2}+\sup_{0\leq t\leq T}\left(\bar{X}^{N}(t)-\hat{\mu}(t)\right)^{2}\right] \\ &\leq K\mathbb{E}\left[\sup_{0\leq t\leq T}\left|\tilde{b}_{2}(Y_{t},t)\right|^{2}\sup_{0\leq t\leq T}\left|\hat{a}_{1Y_{t}}^{N}(t)-a_{Y_{t}}(t)\right|^{2}+\sup_{0\leq t\leq T}\left|\bar{X}^{N}(t)-\hat{\mu}(t)\right|^{2}\right] \\ &\leq K\mathbb{E}\left[\sup_{0\leq t\leq T}\left|\hat{a}_{1Y_{t}}^{N}(t)-a_{Y_{t}}(t)\right|^{2}+\sup_{0\leq t\leq T}\left|\bar{X}^{N}(t)-\hat{\mu}(t)\right|^{2}\right]. \end{split}$$

From the dynamic of  $\bar{X}^N$  and  $\hat{\mu}$ ,

$$\begin{cases} d\left(\bar{X}_{t}^{N}-\hat{\mu}_{t}\right)=\tilde{b}_{1}(Y_{t},t)\left(\bar{X}_{t}^{N}-\hat{\mu}_{t}\right)dt+\frac{\sqrt{N-1}}{N}dB_{t}+\frac{1}{N}dW_{t},\\ \bar{X}_{0}^{N}-\hat{\mu}_{0}=\frac{1}{N}\sum_{i=1}^{N}X_{0}^{i}-\hat{\mu}_{0}, \end{cases}$$

which can be written in terms of  $\bar{G}_t$  of (3.4.12):

$$\bar{X}^{N}(t) - \hat{\mu}(t) = \bar{G}_{t} \left( \frac{1}{N} \sum_{i=1}^{N} X_{0}^{i} - \hat{\mu}_{0}, \tilde{b}_{1}(Y_{\cdot}, \cdot), \frac{\sqrt{N-1}}{N} B, \frac{1}{N} W \right).$$

Using the fact of  $\left|\tilde{b}_{1y}\right|_{\infty}<\infty$  and Ito's isometry, this yields the following estimation:

$$\mathbb{E}\left[\sup_{0\leq t\leq T}\left|\bar{X}^{N}(t)-\hat{\mu}(t)\right|^{2}\right]\leq K\left(\frac{1}{N}+\mathbb{E}\left[\left|\frac{1}{N}\sum_{i=1}^{N}X_{0}^{i}-\hat{\mu}_{0}\right|^{2}\right]\right).$$

Note that, by central limit theorem, we have

$$N\mathbb{E}\left[\left|\frac{1}{N}\sum_{i=1}^{N}X_{0}^{i}-\hat{\mu}_{0}\right|^{2}\right] = \mathbb{E}\left[\left|\frac{\sum_{i=1}^{N}(X_{0}^{i}-\hat{\mu}_{0})}{\sqrt{N}}\right|^{2}\right] \rightarrow Var(X_{0}^{1}) < \infty, \quad N \rightarrow \infty,$$

and we conclude that

$$\mathbb{E}\left[\sup_{0\leq t\leq T}\left|\bar{X}^{N}(t)-\hat{\mu}(t)\right|^{2}\right]=O(N^{-1}).$$
(3.4.15)

Next we investigate the boundness of

$$\sup_{0 \le t \le T} \left| \hat{a}_{1Y_t}^N(t) - a_{Y_t}(t) \right|^2.$$

From (3.4.5) and  $\hat{a}_{1y}^N = \frac{N}{N-1} a_{1y}^N$ , we have

$$\begin{cases} \left(\hat{a}_{1y}^{N}\right)' + 2\tilde{b}_{1y}\hat{a}_{1y}^{N} - \frac{2(N+1)}{N}\tilde{b}_{2y}^{2}\left(\hat{a}_{1y}^{N}\right)^{2} + \sum_{i=1}^{\kappa} q_{y,i}\hat{a}_{1i}^{N} + h_{y} = 0\\ \hat{a}_{1y}^{N}(T) = g_{y}. \end{cases}$$

Define  $u_y = a_y - \hat{a}_{1y}^N$ , let  $\tau = T - t$  and denote  $u_y(\tau) := u_y(T - t)$ , we have

$$\begin{cases} u_{y}'(\tau) = 2\tilde{b}_{1y}(\tau)u_{y}(\tau) - 2\tilde{b}_{2y}^{2}(\tau)\left(a_{y}(\tau) + \hat{a}_{1y}^{N}(\tau)\right)u_{y}(\tau) \\ + \frac{2}{N}\tilde{b}_{2y}^{2}(\tau)\left(\hat{a}_{1y}^{N}(\tau)\right)^{2} + \sum_{i=1}^{\kappa}q_{y,i}u_{i}(\tau) \\ u_{y}(0) = 0, \end{cases}$$
(3.4.16)

which gives that

$$u_y(\tau) = \int_0^\tau \left( 2\tilde{b}_{1y}(s)u_y(s) - 2\tilde{b}_{2y}^2(s) \left( a_y(s) + \hat{a}_{1y}^N(s) \right) u_y(s) + \frac{2}{N} \tilde{b}_{2y}^2(s) \left( \hat{a}_{1y}^N(s) \right)^2 + \sum_{i=1}^\kappa q_{y,i} u_i(s) \right) ds.$$

Thus for  $\tau \in [0, T]$ ,

$$\begin{aligned} |u_y(\tau)| &\leq \int_0^\tau \left( 2\left| \tilde{b}_{1y} \right|_\infty |u_y(s)| + 2\left| \tilde{b}_{2y} \right|_\infty^2 \left( |a_y|_\infty + \left| \hat{a}_{1y}^N \right|_\infty \right) |u_y(s)| \\ &+ \frac{2}{N} \left| \tilde{b}_{2y} \right|_\infty^2 \left| \hat{a}_{1y}^N \right|_\infty^2 + \sum_{i=1}^\kappa |q_{y,i}| |u_i(s)| \right) ds. \end{aligned}$$

Let 
$$\left(\left|\tilde{b}_{1y}\right|_{\infty}, \left|\tilde{b}_{2y}\right|_{\infty}, |a_y|_{\infty}, |\hat{a}_{1y}^N|_{\infty}, \sup_{i \in \mathcal{Y}} |q_{y,i}|\right) \le K_1$$
, then  
 $|u_y(\tau)| \le \frac{2}{N} K_1^4 T + \int_0^{\tau} \left(\left(2K_1 + 4K_1^3\right) |u_y(s)| + K_1 \sum_{i=1}^{\kappa} |u_i(s)|\right) ds.$ 

By adding up the above equation indexed by y = 1 to  $\kappa$ , one can have

$$\sum_{y=1}^{\kappa} |u_y(\tau)| \le \frac{2\kappa K_1^4 T}{N} + \left(2K_1 + 4K_1^3 + \kappa K_1\right) \int_0^{\tau} \sum_{y=1}^{\kappa} |u_y(s)| ds.$$

Let  $K_2 = 2\kappa K_1^4 T$  and  $K_3 = 2K_1 + 4K_1^3 + \kappa K_1$ , by the Grönwall's inequality,

$$\sum_{y=1}^{\kappa} |u_y(\tau)| \le \frac{K_2}{N} e^{K_3 \tau} \le \frac{K_2}{N} e^{K_3 T}, \quad \forall \tau \in [0, T],$$

which implies that

$$\sum_{y=1}^{\kappa} |u_y(\tau)| \le \frac{K}{N}, \quad \forall \tau \in [0, T].$$

Thus, we have

$$\sup_{0 \le t \le T} \left| \hat{a}_{1Y_t}^N(t) - a_{Y_t}(t) \right|^2 \le \frac{K}{N^2}, \quad \text{almsot surely.}$$
(3.4.17)

Therefore, the convergence is obtained from (3.4.15) and (3.4.17):

$$\mathbb{W}_{2}^{2}\left(\mathcal{L}(Z_{t}^{N}),\mathcal{L}(\hat{X}_{t})\right) \leq K\mathbb{E}\left[\sup_{0\leq t\leq T}\left|\hat{a}_{1Y_{t}}^{N}(t)-a_{Y_{t}}(t)\right|^{2}+\sup_{0\leq t\leq T}\left|\bar{X}^{N}(t)-\hat{\mu}(t)\right|^{2}\right]=O(N^{-1}).$$

# 3.5 Numerical results

# 3.5.1 Simulations of Riccati system, the value function and optimal control of the generic palyer

We have derived a  $4\kappa$  dimensional Riccati ODE system (3.2.12) to determine the parameter functions

$$(a_y, b_y, c_y, k_y : y \in \mathcal{Y})$$

needed for the characterization of the equilibrium and the value function. Meanwhile, we also show the solvability of the Riccati ODE system in Section 3.3.

As mentioned earlier, different from the MFG characterization with the common noise, the derived Riccati system is essentially finite-dimensional. In this subsection, we present a numerical experiment and show some numerical results for solving the Riccati system to demonstrate its computational advantages.

For the illustration purpose, assume the finite time horizon is given with T = 5 and the coefficients of the dynamic equation are listed below:

$$\begin{aligned} \mathcal{Y} &= \{0, 1\}, \\ Q &= \begin{bmatrix} -0.5 & 0.5 \\ 0.6 & -0.6 \end{bmatrix}, \\ \tilde{b}_1(\cdot, \cdot) &= 0, \ \tilde{b}_2(\cdot, \cdot) = 1, \\ h_0 &= 2, \ h_1 = 5, \ g_0 = 3, \ g_1 = 1, \\ \mu_0 &= 0, \nu_0 = 2. \end{aligned}$$

Firstly, using the forward Euler's method with the step size  $\delta = 10^{-2}$ , we can obtain trajectories of  $(a_y, b_y, c_y : y \in \mathcal{Y})$ , which is the solution of ODE system (3.2.12). Next, using the trajectories of the parameter functions and Markov chain  $Y_t$ , we can achieve the simulations for  $\hat{\alpha}_t$  and  $\hat{X}_t$ . The Matlab code can be found at https://github.com/JiaminJIAN/Regime\_switching\_MFG.



-----

Figure 3.3: Simulations for  $a_y, V, \alpha$  and  $\nu$ .

As shown in Figure 3.3, people tend to centralize since the conditional second moment of the population density  $\nu_t$  is always decreasing.

#### 3.5.2 Convergence of the *N*-player game

In Section 3.4, we showed that the generic player's path for the N-player game is convergent to the generic player's path for MFG. In this subsection, we demonstrate the convergence of the conditional first moment, conditional second moment, and the value functions of the N-player game to the corresponding terms of the generic player in the Mean Field Game setup by using some numerical examples.



Figure 3.4: Simulations for  $b_y$  and  $c_y$ .

The following figures show the value functions,  $\mu^{(N)}$  and  $\nu^{(N)}$  under  $N \in \{10, 20, 50, 100\}$  with the same parameters' settings as in Figure 3.3 and Figure 3.4 in Section 3.5.1. We can clearly see the convergence to the solution of the generic player in Figure 3.5 and Figure 3.6.



(a)  $\mu_t$ : conditional mean of the population density

(b)  $\nu_t$ : conditional 2nd moment of the population density

Figure 3.5: Simulations for  $\mu_t$  and  $\nu_t$ .

# 3.6 Conclusion

This chapter investigates the convergence rate of the *N*-player game, governed by a Markov chain common noise, towards its asymptotic MFG under the LQG structure. To achieve this, firstly, we introduce a Markovian structure using two auxiliary processes for the first and second moments of the MFG equilibrium and employ the fixed point condition in MFG. By doing so, we characterize the equilibrium measure in MFG with a finite-dimensional Riccati system of ODEs. Consequently, we obtain the equilibrium path, equilibrium control, and the value function in MFG.



Figure 3.6: Simulation of player 1's optimal value function V.

Subsequently, we address the N-player game under the LQG structure, and we characterize its equilibrium path, equilibrium control, and the value function through a Riccati system of ODEs with a dimension of  $O(N^3)$ . Leveraging the N-invariant algebraic structure of this system of ODEs, we establish a dimension reduction result, facilitating a comparison between the equilibrium path  $\hat{X}_1^{(N)}$  in the N-player game and the equilibrium path  $\hat{X}$  in the MFG.

To demonstrate the convergence between the two equilibrium paths, we embed  $\hat{X}_1^{(N)}$  from  $\Omega^{(N)}$  to  $\Omega$  using a distribution copy  $Z^N \in \Omega$ , leading to the achievement of the convergence result and the computation of the convergence rate. Lastly, some numerical examples are presented to demonstrate the convergence result.

In the future, firstly, we can consider the MFG in more general settings, such as with time delays and Poisson jumps. Next, except for considering the LQG structure, we could consider the convergence of MFG with common noise under more general structures. Furthermore, in this chapter, we require positive values for all sensitivities in the cost functional. We find that there is no global solution for MFG when the coefficient of the cost functional is negative, while there is a global solution when the coefficient is positive. So, it is also an interesting problem to investigate the explosion when some sensitivities are negative.

# 3.7 Appendix

#### 3.7.1 Some explicit solutions on LQG-MFG

In this part, we only provide explicit solutions to some LQG-MFG without the common noise. The methodology could be the utilization of the standard Stochastic Maximum Principle or Dynamic Programming approach, and all proofs will be omitted.

Suppose the position of a generic player  $X_t$  follows

$$dX_t = \alpha_t dt + \sigma dW_t, \ X_0 \sim \mathcal{N}(0, 1).$$

The goal of the generic player is to minimize the running cost

$$\inf_{\alpha \in \mathcal{A}} \mathbb{E}\left[\int_0^T \left(\frac{1}{2}\alpha_t^2 + h \int_{\mathbb{R}} (X_t - y)^2 m(t, dy)\right) dt\right],$$

subject to

$$m_t = \mathcal{L}aw(X_t), \quad \forall t \in [0, T],$$

where  $h \in \mathbb{R}$  is a constant.

Denote

$$V(x,t) = \inf_{\alpha} \mathbb{E}\left[\int_{t}^{T} \left(\frac{1}{2}\alpha_{s}^{2} + h \int_{\mathbb{R}} (X_{s} - y)^{2} m(s, dy)\right) ds \,\middle| \, X_{t} = x\right].$$

Note that the model can be characterized by Hamilton-Jacobian-Bellman equation coupled by Fokker-Planck-Kolmogorov equation:

$$\begin{cases} \partial_t V + \frac{1}{2}\sigma^2 \partial_{xx} V - \frac{1}{2}(\partial_x V)^2 + F(x,m) = 0, & (t,x) \in [0,T] \times \mathbb{R}, \\ \partial_t m - \frac{1}{2}\sigma^2 \partial_{xx} m - \partial_x (m\partial_x V) = 0, & (t,x) \in [0,T] \times \mathbb{R}, \\ m_0 \sim \mathcal{N}(0,1), V(x,T) = 0, & x \in \mathbb{R}, \end{cases}$$

where  $F(x,m) = h \int_{\mathbb{R}} (x-y)^2 m(dy)$ .

The monotonicity condition on the source term F in the variable m plays a crucial role in the uniqueness of the MFG system. A monotone function  $f : \mathbb{R} \to \mathbb{R}$  is said to be increasing if it satisfies  $(f(x_1) - f(x_2))(x_1 - x_2) \ge 0$ , and decreasing if -f is increasing. This definition can be generalized to an infinite dimensional function F(x, m).

**Definition 30.** The real function F on  $\mathbb{R} \times \mathcal{P}_2(\mathbb{R})$  is said to be monotone, if, for all  $m \in \mathcal{P}_2(\mathbb{R})$ , the mapping  $\mathbb{R} \ni x \mapsto F(x,m)$  is at most of quadratic growth, and for all  $m_1, m_2$  it satisfies

$$\int_{\mathbb{R}} \left( F(x, m_1) - F(x, m_2) \right) d(m_1 - m_2)(x) \ge 0.$$

F is said to be anti-monotone, if (-F) is monotone.

According to [13], if F is monotone, then MFG have at most one solution. Interestingly, the monotonicity of F is dependent on the sign of h.

**Lemma 31.**  $F(x,m) = h \int_{\mathbb{R}} (x-y)^2 m(dy)$  is monotone if h < 0, and anti-monotone if h > 0.

A natural question is how the MFG system behaves differently to the monotonicity of F?

#### **3.7.1.1** Case I: h > 0

**Lemma 32.** For h > 0, there exists a solution (may not be unique) to the MFG system in the form of  $V(x,t) = f_1(t)x^2 + f_3(t)$  and  $m(t) \sim \mathcal{N}(0,\gamma(t))$ , where

$$f_{1}(t) = \sqrt{\frac{h}{2}} \frac{1 - e^{-2\sqrt{2h}(T-t)}}{1 + e^{-2\sqrt{2h}(T-t)}},$$
  

$$\gamma(t) = e^{-\int_{0}^{t} 4f_{1}(s)ds} \left(1 + \int_{0}^{t} \sigma^{2} e^{\int_{0}^{s} 4f_{1}(u)du} ds\right),$$
  

$$f_{3}(t) = \int_{t}^{T} (\sigma^{2} f_{1}(s) + h\gamma(s))ds.$$

#### **3.7.1.2** Case II: h < 0

**Lemma 33.** For h < 0, there exists a unique solution in  $(t_0, T]$  to the MFG system in the form of  $V(x,t) = g_1(t)x^2 + g_3(t)$  and  $m(t) \sim \mathcal{N}(0, \lambda(t))$ , where

$$g_{1}(t) = -\sqrt{-\frac{h}{2}} \tan\left(\sqrt{-2h}(T-t)\right),$$
  

$$\lambda(t) = e^{-\int_{0}^{t} 4g_{1}(s)ds} \left(1 + \int_{0}^{t} \sigma^{2} e^{\int_{0}^{s} 4g_{1}(u)du}ds\right),$$
  

$$g_{3}(t) = \int_{t}^{T} (\sigma^{2}g_{1}(s) + h\lambda(s))ds,$$
  

$$t_{0} = max\left(0, T - \frac{1}{\sqrt{-2h}}\frac{\pi}{2}\right).$$

#### 3.7.1.3 Remark

When h > 0, the cost is anti-monotone, and there exists at least one global solution. When h < 0, the cost is monotone, and there exists at most one solution. Unfortunately, this solution lives in a short period. Lemma 33 coincides with the notes in Section 3.8 of [16] saying that due to the opposite time evolution of the system of HJB-FPK, the existence of the solution may exist for only a short period.

# 3.7.2 Dynkin's formula for a regime-switching diffusion with a quadratic function

Since the running cost (3.2.10) has a quadratic growth in the state variable, the value function  $V[\hat{m}](y, x, t)$  is expected to possess similar growth. Next, we present a version of Dynkin's formula for the functions of quadratic growth, which is sufficient for our purpose. Throughout this subsection, we will use K in various places as a generic constant that varies from line to line. The notions of this subsection are independent of other parts of the chapter.

**Lemma 34.** Let X be the  $\mathbb{R}^d$ -valued process satisfying

$$X_t = X_0 + \int_0^t \left( \tilde{b}_1(Y_s, s) X_s + \tilde{b}_2(Y_s, s) \alpha_s \right) ds + \int_0^t \sigma(s) dW_s$$

where Y is CTMC with a generator

$$Y \sim Q = (q_{ij})_{i,j=1,2,\dots,\kappa},$$

Suppose  $\sigma(\cdot)$ ,  $\tilde{b}_1(y, \cdot)$  and  $\tilde{b}_2(y, \cdot)$  are continuous functions on [0, T] for every  $y \in \mathcal{Y} := \{1, 2, ..., \kappa\}$ . If  $X_0 \in L^4$ ,  $\alpha \in L^4_{\mathbb{F}}$  and  $f: \mathcal{Y} \times \mathbb{R}^d \times \mathbb{R} \mapsto \mathbb{R}$  satisfies, for some large K

 $\sup_{y \in \mathcal{Y}, t \in [0,T]} \{ |f(y,x,t)| + (1+|x|) |\nabla f(y,x,t)| + (1+|x|)^2 |\Delta f(y,x,t)| + |\partial_t f(y,x,t)| \} \le K(|x|^2 + 1),$ 

then the following identity holds for all  $t \in [0, T]$ :

$$\mathbb{E}\left[f(Y_t, X_t, t)\right] = \mathbb{E}\left[f(Y_0, X_0, 0)\right] + \mathbb{E}\left[\int_0^t (\partial_t + \mathcal{L}^{\alpha_s} + \mathcal{Q})f(Y_s, X_s, s)ds\right],$$

where

$$\mathcal{L}^{a}f(y,x,s) = \left(\frac{1}{2}\operatorname{Tr}\left(\sigma_{s}\sigma_{s}^{\top}\Delta\right) + \left(\tilde{b}_{1y}x + \tilde{b}_{2y}a\right) \cdot \nabla_{x}\right)f(y,x,s)$$

and

$$\mathcal{Q}f(y,x,s) = \sum_{i=1}^{n} q_{y,i}f(i,x,s).$$

*Proof.* It's enough to show that the local martingale defined by Itô's formula

$$M_t^f = f(Y_t, X_t, t) - f(Y_0, X_0, 0) - \int_0^t (\partial_t + \mathcal{L}^{\alpha_s} + \mathcal{Q}) f(Y_s, X_s, s) ds$$
(3.7.1)

is uniformly integrable, hence is a true martingale.

First, note that from the assumptions on  $X_0$  and  $\alpha$ , we have

$$\mathbb{E}\left[\|X_{t}\|^{4}\right] \leq K\mathbb{E}\left[\|X_{0}\|^{4} + \int_{0}^{t} \|\tilde{b}_{1}(Y_{s},s)X_{s} + \tilde{b}_{2}(Y_{s},s)\alpha_{s}\|^{4}ds + \int_{0}^{t} \|\sigma_{s}W_{s}\|^{4}ds\right]$$
$$\leq K\mathbb{E}\left[\|X_{0}\|^{4} + \int_{0}^{t} \|X_{s}\|^{4}ds + \int_{0}^{t} \|\alpha_{s}\|^{4}ds + \int_{0}^{t} \|\sigma_{s}W_{s}\|^{4}ds\right]$$
$$\leq K + K\int_{0}^{t} \mathbb{E}\left[\|X_{s}\|^{4}\right]ds,$$

where K is a generic constant that varies from line to line. Then, by the Grönwall's inequality,

$$\mathbb{E}\left[\|X_t\|^4\right] \le K e^{Kt} \le K,$$

which implies that  $\{X_t : 0 \le t \le T\}$  is  $L^4$  bounded uniformly in t.

On the other hand, since  $x \mapsto f(y, x, t)$  is at most quadratic growth uniformly in (y, t), we conclude that  $f(Y_t, X_t, t)$  is uniformly  $L^2$  bounded from the fact

$$\sup_{t \in [0,T]} \mathbb{E}\left[f^2(Y_t, X_t, t)\right] \le K \sup_{t \in [0,T]} \mathbb{E}\left[\|X_t\|^4\right] + K \le K.$$

The uniform  $L^2$ -boundedness of  $\int_0^t \partial_t f(Y_s, X_s, s) ds$  follows from our assumption on  $\partial_t f$ . Similarly, since  $\mathcal{Q}f$  has a quadratic growth uniformly in y and t, and

$$\left\{\int_0^t \mathcal{Q}f(Y_s, X_s, s)ds : 0 \le t \le T\right\}$$

is  $L^2$  bounded. At last, we have

$$\begin{split} & \mathbb{E}\left[\left(\int_{0}^{t} \mathcal{L}^{\alpha_{s}} f(Y_{s}, X_{s}, s) ds\right)^{2}\right] \\ &\leq K \mathbb{E}\left[\int_{0}^{t} \left(\left(\tilde{b}_{1}(Y_{s}, s) X_{s} + \tilde{b}_{2}(Y_{s}, s) \alpha_{s}\right) \cdot \nabla f + \frac{1}{2} \mathrm{Tr}\left(\sigma_{s} \sigma_{s}^{\top} \Delta f\right)\right)^{2} (Y_{s}, X_{s}, s) ds\right] \\ &\leq K \mathbb{E}\left[\int_{0}^{t} \|\tilde{b}_{1}(Y_{s}, s) X_{s} + \tilde{b}_{2}(Y_{s}, s) \alpha_{s}\|^{2} \|\nabla f\|^{2} (Y_{s}, X_{s}, s) ds\right] \\ &\quad + K \mathbb{E}\left[\int_{0}^{t} \frac{1}{4} \|\mathrm{Tr}\left(\sigma_{s} \sigma_{s}^{\top} \Delta f\right)\|^{2} (Y_{s}, X_{s}, s) ds\right] \\ &\leq K \mathbb{E}\left[\int_{0}^{t} \|\alpha_{s}\|^{4} ds\right] + K \mathbb{E}\left[\int_{0}^{t} \|X_{s}\|^{4} ds\right] + K \mathbb{E}\left[\int_{0}^{t} |\nabla f|^{4} (Y_{s}, X_{s}, s) ds\right] \\ &\quad + K \mathbb{E}\left[\int_{0}^{t} \frac{1}{4} \|\mathrm{Tr}\Delta f\|^{2} (Y_{s}, X_{s}, s) ds\right]. \end{split}$$

Since  $\nabla f$  is linear growth in x, the second term  $\sup_{t \in [0,T]} \mathbb{E} \left[ \int_0^t \|\nabla f\|^4 (Y_s, X_s, s) ds \right]$  is finite. Together with assumptions on  $\Delta f$  and  $\alpha$ , we have uniform  $L^2$ -boundedness of  $\int_0^t \mathcal{L}^{\alpha_s} f(Y_s, X_s, s) ds$ .

As a result, each term of the right-hand side of (3.7.1) is uniform  $L^2$ -bounded in t, and thus  $M_t^f$  belongs to  $L_{\mathbb{F}}^2$  and this implies the uniform integrability.

### 3.7.3 Proof of the property of G

Lemma 35. Define

$$\mathcal{E}_t(\phi) = \exp\left\{\int_0^t \phi_s ds\right\},$$

and

$$G_t(x,\phi_1,\phi_2,\phi_3,W) = x\mathcal{E}_t(\phi_1 - \phi_2) + \mathcal{E}_t(\phi_1 - \phi_2) \int_0^t \mathcal{E}_s(-\phi_1 + \phi_2) \left(\phi_2(s)\phi_3(s)ds + dW_s\right),$$

where x is a given constant,  $\phi_1, \phi_2, \phi_3$  are RCLL functions on [0, T]. Then

$$\mathbb{E}\left[\left|G_t(x^1,\phi_1,\phi_1^1,\phi_3^1,W) - G_t(x^2,\phi_1,\phi_2^2,\phi_3^2,W)\right|^2\right]$$
  
$$\leq K\left(|x^1 - x^2|^2 + \sup_{0 \le t \le T} |\phi_2^1(t) - \phi_2^2(t)|^2 + \sup_{0 \le t \le T} |\phi_3^1(t) - \phi_3^2(t)|^2\right).$$

*Proof.* Firstly, it can be shown that  $G(\cdot, \phi_1, \phi_2, \phi_3, W)$  is Lipschitz continuous with respect to x

$$\mathbb{E}\left[\left|G_{t}(x^{1},\phi_{1},\phi_{2},\phi_{3},W)-G(x^{2},\phi_{1},\phi_{2},\phi_{3},W)\right|\right] \leq \left|x^{1}\mathcal{E}_{t}(\phi_{1}-\phi_{2})-x^{2}\mathcal{E}_{t}(\phi_{1}-\phi_{2})\right|$$
$$\leq \mathcal{E}_{t}(\phi_{1}-\phi_{2})|x^{1}-x^{2}|$$
$$\leq K(|\phi_{1}|_{\infty},|\phi_{2}|_{\infty},T)|x^{1}-x^{2}|.$$

Next, we have

$$\begin{split} & \mathbb{E}\left[\left|G_{t}(x,\phi_{1},\phi_{2},\phi_{3}^{1},W)-G(x,\phi_{1},\phi_{2},\phi_{3}^{2},W)\right|^{2}\right] \\ &=\left|\mathcal{E}_{t}(\phi_{1}-\phi_{2})\int_{0}^{t}\mathcal{E}_{s}(\phi_{1}-\phi_{2})\phi_{2}(s)(\phi_{3}^{1}(s)-\phi_{3}^{2}(s))ds\right|^{2} \\ &\leq \mathcal{E}_{t}(2\phi_{1}-2\phi_{2})\left(\int_{0}^{t}\mathcal{E}_{s}(\phi_{1}-\phi_{2})|\phi_{2}(s)||(\phi_{3}^{1}(s)-\phi_{3}^{2}(s))|ds\right)^{2} \\ &\leq K(|\phi_{1}|_{\infty},|\phi_{2}|_{\infty},T)\left(\int_{0}^{T}|\phi_{3}^{1}(s)-\phi_{3}^{2}(s)|ds\right)^{2} \\ &\leq K(|\phi_{1}|_{\infty},|\phi_{2}|_{\infty},T)\sup_{0\leq t\leq T}\left|\phi_{3}^{1}(t)-\phi_{3}^{2}(t)\right|^{2}. \end{split}$$

Similarly, for  $\phi_2^1(\cdot), \phi_2^2(\cdot) \in C([0,T]),$ 

$$\begin{split} & \mathbb{E}\left[\left|G_{t}(x,\phi_{1},\phi_{2}^{1},\phi_{3},W)-G(x,\phi_{1},\phi_{2}^{2},\phi_{3},W)\right|^{2}\right] \\ & \leq K\left|x\mathcal{E}_{t}(\phi_{1}-\phi_{2}^{1})-x\mathcal{E}_{t}(\phi_{1}-\phi_{2}^{2})\right|^{2} \\ & +K\left|\mathcal{E}_{t}(\phi_{1}-\phi_{2}^{1})\int_{0}^{t}\mathcal{E}_{s}(-\phi_{1}+\phi_{2}^{1})\phi_{2}^{1}(s)\phi_{3}(s)ds-\mathcal{E}_{t}(\phi_{1}-\phi_{2}^{2})\int_{0}^{t}\mathcal{E}_{s}(-\phi_{1}+\phi_{2}^{2})\phi_{2}^{2}(s)\phi_{3}(s)ds\right|^{2} \\ & +K\mathbb{E}\left[\left|\mathcal{E}_{t}(\phi_{1}-\phi_{2}^{1})\int_{0}^{t}\mathcal{E}_{s}(-\phi_{1}+\phi_{2}^{1})dW_{s}-\mathcal{E}_{t}(\phi_{1}-\phi_{2}^{2})\int_{0}^{t}\mathcal{E}_{s}(-\phi_{1}+\phi_{2}^{2})dW_{s}\right|^{2}\right] \\ & :=K(J_{1}+J_{2}+J_{3}). \end{split}$$

Note that by the mean-value theorem and the continuity of  $\phi_1, \phi_2^1$  and  $\phi_2^2$  on [0, T], we can get

$$J_{1} = \left| x \mathcal{E}_{t}(\phi_{1} - \phi_{2}^{1}) - x \mathcal{E}_{t}(\phi_{1} - \phi_{2}^{2}) \right|^{2}$$
  
$$= x^{2} \left( e^{\int_{0}^{t} (\phi_{1}(s) - \phi_{2}^{1}(s)) ds} - e^{\int_{0}^{t} (\phi_{1}(s) - \phi_{2}^{2}(s)) ds} \right)^{2}$$
  
$$\leq K \left( x, \left| \phi_{2}^{1} \right|_{\infty}, \left| \phi_{2}^{2} \right|_{\infty}, T \right) e^{\int_{0}^{t} 2\phi_{1}(s) ds} \left| \phi_{2}^{1} - \phi_{2}^{2} \right|_{\infty}^{2}$$
  
$$\leq K \left( x, \left| \phi_{1} \right|_{\infty}, \left| \phi_{2}^{1} \right|_{\infty}, \left| \phi_{2}^{2} \right|_{\infty}, T \right) \left| \phi_{2}^{1} - \phi_{2}^{2} \right|_{\infty}^{2},$$

and

$$\begin{split} J_{3} &= \mathbb{E}\left[\left|\mathcal{E}_{t}(\phi_{1}-\phi_{2}^{1})\int_{0}^{t}\mathcal{E}_{s}(-\phi_{1}+\phi_{2}^{1})dW_{s}-\mathcal{E}_{t}(\phi_{1}-\phi_{2}^{2})\int_{0}^{t}\mathcal{E}_{s}(-\phi_{1}+\phi_{2}^{2})dW_{s}\right|^{2}\right]\\ &= \mathbb{E}\left[\left|\mathcal{E}_{t}(\phi_{1}-\phi_{2}^{1})\int_{0}^{t}\mathcal{E}_{s}(-\phi_{1}+\phi_{2}^{1})dW_{s}-\mathcal{E}_{t}(\phi_{1}-\phi_{2}^{1})\int_{0}^{t}\mathcal{E}_{s}(-\phi_{1}+\phi_{2}^{2})dW_{s}\right.\\ &+\left.\mathcal{E}_{t}(\phi_{1}-\phi_{2}^{1})\int_{0}^{t}\mathcal{E}_{s}(-\phi_{1}+\phi_{2}^{2})dW_{s}-\mathcal{E}_{t}(\phi_{1}-\phi_{2}^{2})\int_{0}^{t}\mathcal{E}_{s}(-\phi_{1}+\phi_{2}^{2})dW_{s}\right|^{2}\right]\\ &\leq 2\mathcal{E}_{t}(2\phi_{1}-2\phi_{2}^{1})\int_{0}^{t}\left(\mathcal{E}_{s}(-\phi_{1}+\phi_{2}^{1})-\mathcal{E}_{s}(-\phi_{1}+\phi_{2}^{2})\right)^{2}ds\\ &+2\left(\mathcal{E}_{t}(\phi_{1}-\phi_{2}^{1})-\mathcal{E}_{t}(\phi_{1}-\phi_{2}^{2})\right)^{2}\int_{0}^{t}\mathcal{E}_{s}(-2\phi_{1}+2\phi_{2}^{2})ds\\ &\leq K\left(|\phi_{1}|_{\infty},|\phi_{2}^{1}|_{\infty},|\phi_{2}^{2}|_{\infty},T\right)|\phi_{1}^{1}-\phi_{2}^{2}|_{\infty}^{2}. \end{split}$$

Lastly, using the similar argument, we have

$$\begin{split} J_{2} &= \left| \mathcal{E}_{t}(\phi_{1} - \phi_{2}^{1}) \int_{0}^{t} \mathcal{E}_{s}(-\phi_{1} + \phi_{2}^{1}) \phi_{2}^{1}(s) \phi_{3}(s) ds - \mathcal{E}_{t}(\phi_{1} - \phi_{2}^{2}) \int_{0}^{t} \mathcal{E}_{s}(-\phi_{1} + \phi_{2}^{2}) \phi_{2}^{2}(s) \phi_{3}(s) ds \right|^{2} \\ &= \left| \mathcal{E}_{t}(\phi_{1} - \phi_{2}^{1}) \int_{0}^{t} \mathcal{E}_{s}(-\phi_{1} + \phi_{2}^{1}) \phi_{2}^{1}(s) \phi_{3}(s) ds - \mathcal{E}_{t}(\phi_{1} - \phi_{2}^{2}) \int_{0}^{t} \mathcal{E}_{s}(-\phi_{1} + \phi_{2}^{1}) \phi_{2}^{1}(s) \phi_{3}(s) ds \\ &+ \mathcal{E}_{t}(\phi_{1} - \phi_{2}^{2}) \int_{0}^{t} \mathcal{E}_{s}(-\phi_{1} + \phi_{2}^{1}) \phi_{2}^{1}(s) \phi_{3}(s) ds - \mathcal{E}_{t}(\phi_{1} - \phi_{2}^{2}) \int_{0}^{t} \mathcal{E}_{s}(-\phi_{1} + \phi_{2}^{2}) \phi_{2}^{2}(s) \phi_{3}(s) ds \right|^{2} \\ &\leq 2 \left| \left( \mathcal{E}_{t}(\phi_{1} - \phi_{2}^{1}) - \mathcal{E}_{t}(\phi_{1} - \phi_{2}^{2}) \right) \int_{0}^{t} \mathcal{E}_{s}(-\phi_{1} + \phi_{2}^{1}) \phi_{2}^{1}(s) \phi_{3}(s) ds - \int_{0}^{t} \mathcal{E}_{s}(-\phi_{1} + \phi_{2}^{2}) \phi_{2}^{2}(s) \phi_{3}(s) ds \right|^{2} \\ &+ 2 \left| \mathcal{E}_{t}(\phi_{1} - \phi_{2}^{2}) \left( \int_{0}^{t} \mathcal{E}_{s}(-\phi_{1} + \phi_{2}^{1}) \phi_{2}^{1}(s) \phi_{3}(s) ds - \int_{0}^{t} \mathcal{E}_{s}(-\phi_{1} + \phi_{2}^{2}) \phi_{2}^{2}(s) \phi_{3}(s) ds \right) \right|^{2} \\ &\leq K \left( |\phi_{1}|_{\infty}, |\phi_{2}^{1}|_{\infty}, |\phi_{2}^{2}|_{\infty}, |\phi_{3}|_{\infty}, T) |\phi_{2}^{1} - \phi_{2}^{2}|_{\infty}^{2} \\ &+ 2 \left| \mathcal{E}_{t}(\phi_{1} - \phi_{2}^{2}) \left( \int_{0}^{t} \mathcal{E}_{s}(-\phi_{1} + \phi_{2}^{1}) \phi_{2}^{1}(s) \phi_{3}(s) ds - \int_{0}^{t} \mathcal{E}_{s}(-\phi_{1} + \phi_{2}^{2}) \phi_{2}^{1}(s) \phi_{3}(s) ds \right) \\ &+ \mathcal{E}_{t}(\phi_{1} - \phi_{2}^{2}) \left( \int_{0}^{t} \mathcal{E}_{s}(-\phi_{1} + \phi_{2}^{2}) \phi_{2}^{1}(s) \phi_{3}(s) ds - \int_{0}^{t} \mathcal{E}_{s}(-\phi_{1} + \phi_{2}^{2}) \phi_{2}^{2}(s) \phi_{3}(s) ds \right) \right|^{2} \\ &\leq K \left( |\phi_{1}|_{\infty}, |\phi_{2}^{1}|_{\infty}, |\phi_{2}^{1}|_{\infty}, |\phi_{3}|_{\infty}, T) |\phi_{2}^{1} - \phi_{2}^{2}|_{\infty}^{2} \right]. \end{split}$$

Sum up the above inequalities for  $J_1, J_2$  and  $J_3$ , then  $\mathbb{E} \left[ \int_{-\infty}^{1} G_1(x_1 - y_2) + \int_{-\infty}^{1} G_2(x_1 - y_2) + \int_{-\infty}^{1} G_2$ 

$$\mathbb{E}\left[\left|G_{t}(x,\phi_{1},\phi_{2}^{1},\phi_{3},W)-G(x,\phi_{1},\phi_{2}^{2},\phi_{3},W)\right|^{2}\right] \le K\left(x,|\phi_{1}|_{\infty},|\phi_{2}^{1}|_{\infty},|\phi_{2}^{2}|_{\infty},|\phi_{3}|_{\infty},T\right)\left|\phi_{2}^{1}-\phi_{2}^{2}\right|_{\infty}^{2}.$$

Thus, we can obtain the desired result.

## 3.7.4 Proof of the existence and uniqueness of the ODE system

Consider the following ODE system

$$\begin{cases} a'_y + C_1 \tilde{b}_{1y} a_y - C_2 \tilde{b}_{2y}^2 a_y^2 + \sum_{i=1}^{\kappa} q_{y,i} a_i + h_y = 0, \\ a_y(T) = g_y, \end{cases}$$
(3.7.2)

for  $y \in \mathcal{Y} = \{1, 2, \dots, \kappa\}$ , where  $C_1, C_2, h_y, g_y$  are in  $\mathbb{R}^+$ . We need to show the existence and uniqueness of the solution to (3.7.2). Define  $T_y^{(N)}$  as

$$T_{y}^{(N)}[a](t) = \left[ \left( g_{y} + \int_{t}^{T} \left( h_{y} + C_{1}\tilde{b}_{1y}(s)a_{y}(s) - C_{2}\tilde{b}_{2y}^{2}(s)a_{y}^{2}(s) + \sum_{i=1}^{\kappa} q_{y,i}a_{i}(s) \right) ds \right) \wedge N \right] \vee 0,$$

where  $a = [a_1, a_2, \dots, a_{\kappa}]^{\top}$ . Let  $D = \{f \in C([0, T]) : 0 \leq \sup_{t \in [0, T]} f(t) \leq N\}$ . Note that  $T_y^{(N)}(y \in \mathcal{Y})$  maps  $D^{\kappa}$  to  $D^{\kappa}$ .

**Lemma 36.** For fixed N, there exists a unique solution in C([0,T]) to

$$a = T_y^{(N)}[a]. (3.7.3)$$

*Proof.* Denote the norm  $||f||_k = ||e^{kt} \max_{y \in \mathcal{Y}} |f_y|||_{\infty}$ , where k needs to be determined later and f is a  $\kappa$  dimensional vector with entry of  $f_y, y \in \mathcal{Y}$ , which is equivalent to the infinite norm. Define the iteration rule  $a_y^{(n+1)} = T_y^{(N)} \left[a_y^{(n)}\right]$  for  $y \in \mathcal{Y}$ . Note that

$$\begin{split} \left\| e^{kt} \left( a_{y}^{(n+1)}(t) - a_{y}^{(n)}(t) \right) \right\|_{\infty} \\ &\leq \sup_{t \in [0,T]} e^{kt} \int_{t}^{T} \left( C_{1} \left| \tilde{b}_{1y} \right|_{\infty} \left| a_{y}^{(n)}(s) - a_{y}^{(n-1)}(s) \right| + C_{2} \left| \tilde{b}_{2y} \right|_{\infty}^{2} \left| \left( a_{y}^{(n)}(s) \right)^{2} - \left( a_{y}^{(n-1)}(s) \right)^{2} \right. \\ &+ \sum_{i=1}^{\kappa} q_{y,i} \left| a_{i}^{(n)}(s) - a_{i}^{(n-1)}(s) \right| \right) ds \\ &\leq \sup_{t \in [0,T]} e^{kt} \int_{t}^{T} \left( C_{1} \left| \tilde{b}_{1y} \right|_{\infty} \left| a_{y}^{(n)}(s) - a_{y}^{(n-1)}(s) \right| + 2NC_{2} \left| \tilde{b}_{2y} \right|_{\infty}^{2} \left| a_{y}^{(n)}(s) - a_{y}^{(n-1)}(s) \right| \right. \\ &+ \sum_{i=1}^{\kappa} q_{y,i} \left| a_{i}^{(n)}(s) - a_{i}^{(n-1)}(s) \right| \right) ds \\ &\leq \sup_{t \in [0,T]} e^{kt} \int_{t}^{T} e^{-ks} \left( C_{1} \left| \tilde{b}_{1y} \right|_{\infty} + 2NC_{2} \left| \tilde{b}_{2y} \right|_{\infty}^{2} + \kappa \max_{i \in \mathcal{Y}} \left| q_{y,i} \right| \right) \left\| a^{(n)} - a^{(n-1)} \right\|_{k} ds \\ &\leq \frac{C_{1} \left| \tilde{b}_{1y} \right|_{\infty} + 2NC_{2} \left| \tilde{b}_{2y} \right|_{\infty}^{2} + \kappa \max_{i \in \mathcal{Y}} \left| q_{y,i} \right| }{k} \left\| a^{(n)} - a^{(n-1)} \right\|_{k}. \end{split}$$

Choose  $k > C_1 \left| \tilde{b}_{1y} \right|_{\infty} + 2NC_2 \left| \tilde{b}_{2y} \right|_{\infty}^2 + \kappa \max_{i \in \mathcal{Y}} |q_{y,i}|$ , then

$$\left\| a^{(n+1)} - a^{(n)} \right\|_{k} \leq \frac{C_{1} \left| \tilde{b}_{1y} \right|_{\infty} + 2NC_{2} \left| \tilde{b}_{2y} \right|_{\infty}^{2} + \kappa \max_{i \in \mathcal{Y}} |q_{y,i}|}{k} \left\| a^{(n)} - a^{(n-1)} \right\|_{k}$$

which gives us a contraction mapping from  $D^{\kappa}$  to  $D^{\kappa}$ . Hence, by the Banach fixed point theorem, there exists a unique solution to (3.7.3).

Next, we want to show that for large enough N, the solution to (3.7.3) is also the solution to (3.7.2).

Lemma 37. For

$$N \ge e^{KT} \left( \sum_{y=1}^{\kappa} g_y + T \sum_{y=1}^{\kappa} h_y \right),$$

where  $K := C_1 \max_{y \in \mathcal{Y}} \left| \tilde{b}_{1y} \right|_{\infty} + \max_{i \in \mathcal{Y}} \sum_{y=1}^{\kappa} |q_{y,i}|$ , the solution  $a^{(N)}$  to (3.7.3) satisfies the inequalities

$$0 \le g_y + \int_t^T \left( h_y + C_1 \tilde{b}_{1y}(s) a_y^{(N)}(s) - C_2 \tilde{b}_{2y}^2(s) \left( a_y^{(N)}(s) \right)^2 + \sum_{i=1}^\kappa q_{y,i} a_i^{(N)}(s) \right) ds \le N$$
(3.7.4)

for all  $t \in [0, T]$ , where  $y \in \mathcal{Y}$ .

*Proof.* For simplicity of notations,  $a_y$  is used instead of  $a_y^{(N)}$  for  $y \in \mathcal{Y}$  if there is no confusion.

First, for  $y \in \mathcal{Y}$ , we prove the positiveness of  $a_y$  by contradiction. Suppose  $a_y$   $(y \in \mathcal{Y})$  are not positive functions on [0, T]. Since  $a_1$  is continuous and  $a_1(T) = g_1 > 0$ , there exists some  $\tau_1 \in [0, T]$ as the closest time to T such that  $a_1(\tau_1) = 0$ . Note that finding such a  $\tau_1$  is possible. Let  $t_n \in [0, T]$ be a non-decreasing sequence such that  $a_1(t_n) = 0$ , there exists some  $\tau_1$  such that  $t_n \to \tau_1 < T$  as  $n \to \infty$  since  $a_1$  is continuous and  $a_1(T) = g_1 > 0$ . By the continuity of  $a_1$ , we have  $a_1(\tau_1) = 0$ , which gives the desirable point  $\tau_1$ . Then for all  $t \in (\tau_1, T]$ ,  $a_1(t) > 0$  and it implies that  $a'_1(\tau_1) > 0$ . In this case, plugging  $t = \tau_1$  to (3.7.2), we have

$$a_1'(\tau_1) = -h_1 - \sum_{i \neq 1}^{\kappa} q_{1,i} a_i(\tau_1) > 0,$$

which implies there is some  $y \in \mathcal{Y}$  and  $y \neq 1$  such that  $a_y(\tau_1) < 0$ . Without loss of generality, we let  $a_2(\tau_1) < 0$ . Since  $a_2$  is continuous on [0,T] and  $a_2(T) = g_2 > 0$ , from the intermediate value theorem, there exists some  $\tau_2 \in (\tau_1, T)$  such that  $a_2(\tau_2) = 0$  and  $a'_2(\tau_2) > 0$ . This indicates that  $a'_2(\tau_2) = -h_2 - \sum_{i\neq 2}^{\kappa} q_{2,i}a_i(\tau_2) > 0$  by plugging  $t = \tau_2$  back to (3.7.2), and it implies that there is some  $y \in \mathcal{Y}$  and  $y \neq 1, 2$  such that  $a_y(\tau_2) < 0$  since we already know  $a_1(\tau_2) > 0$ . Without loss of generality, we can let  $a_3(\tau_2) < 0$ . By induction with the same argument, there is a  $\tau_{\kappa} \in (\tau_{\kappa-1}, T)$ 

such that  $a_{\kappa}(\tau_{\kappa}) = 0$  and  $a'_{\kappa}(\tau_{\kappa}) > 0$ , which gives

$$a'_{\kappa}(\tau_{\kappa}) + h_{\kappa} + \sum_{i \neq \kappa}^{\kappa} q_{\kappa,i} a_i(\tau_{\kappa}) = 0.$$

But it contradicts with the fact that

$$a'_{\kappa}(\tau_{\kappa}) > 0, \ h_{\kappa} > 0, \ q_{\kappa,i} > 0, \ a_i(\tau_{\kappa}) > 0$$

for  $i \in \{1, 2, \ldots, \kappa - 1\}$ . Thus the positiveness of  $a_y$  on [0, T] for all  $y \in \mathcal{Y}$  is obtained.

Next, we prove the upper boundness for the integral in (3.7.4). Note that for all  $t \in [0, T]$  and  $y \in \mathcal{Y}$ , let  $\tau = T - t$ , we have

$$a'_{y}(\tau) = h_{y} + C_{1}\tilde{b}_{1y}(\tau)a_{y}(\tau) - C_{2}\tilde{b}^{2}_{2y}(\tau)a^{2}_{y}(\tau) + \sum_{i=1}^{\kappa} q_{y,i}a_{i}(\tau),$$

and thus

$$\begin{split} \sum_{y=1}^{\kappa} a_{y}'(\tau) &= \sum_{y=1}^{\kappa} h_{y} + C_{1} \sum_{y=1}^{\kappa} \tilde{b}_{1y}(\tau) a_{y}(\tau) - C_{2} \sum_{y=1}^{\kappa} \tilde{b}_{2y}^{2}(\tau) a_{y}^{2}(\tau) + \sum_{y=1}^{\kappa} \sum_{i=1}^{\kappa} q_{y,i} a_{i}(\tau) \\ &\leq \sum_{y=1}^{\kappa} h_{y} + C_{1} \max_{y \in \mathcal{Y}} \left| \tilde{b}_{1y} \right|_{\infty} \sum_{y=1}^{\kappa} a_{y}(\tau) + \sum_{y=1}^{\kappa} \sum_{i=1}^{\kappa} |q_{y,i}| a_{i}(\tau) \\ &\leq \sum_{y=1}^{\kappa} h_{y} + \sum_{i=1}^{\kappa} \left( C_{1} \max_{y \in \mathcal{Y}} \left| \tilde{b}_{1y} \right|_{\infty} + \sum_{y=1}^{\kappa} |q_{y,i}| \right) a_{i}(\tau) \\ &\leq \sum_{y=1}^{\kappa} h_{y} + K \sum_{i=1}^{\kappa} a_{i}(\tau), \end{split}$$

where

$$K := C_1 \max_{y \in \mathcal{Y}} \left| \tilde{b}_{1y} \right|_{\infty} + \max_{i \in \mathcal{Y}} \sum_{y=1}^{\kappa} |q_{y,i}|$$

with  $\sum_{y=1}^{\kappa} a_y(T) = \sum_{y=1}^{\kappa} g_y$ . By Grönwall's inequality, for all  $\tau \in [0, T]$ ,

$$\sum_{y=1}^{\kappa} a_y(\tau) \le e^{KT} \left( \sum_{y=1}^{\kappa} g_y + T \sum_{y=1}^{\kappa} h_y \right).$$

Hence  $a_y(t) \le e^{KT} \left( \sum_{y=1}^{\kappa} g_y + T \sum_{y=1}^{\kappa} h_y \right)$  for all  $t \in [0,T]$  and  $y \in \mathcal{Y}$ . Hence, when

$$e^{KT}\left(\sum_{y=1}^{\kappa}g_y+T\sum_{y=1}^{\kappa}h_y\right)\leq N,$$

**Lemma 38.** With the given of  $h_y, g_y \in \mathbb{R}^+$ ,  $y \in \mathcal{Y}$ , there exists a unique solution to the Riccati system (3.2.12).

Proof. The existence, uniqueness and boundedness of the solution to  $a_y \ (y \in \mathcal{Y})$  are shown in Lemma 36 and Lemma 37. Given  $(a_y : y \in \mathcal{Y})$ , the coefficient functions  $b_y \ (y \in \mathcal{Y})$  form a linear ordinary differential equation system. Their existence and uniqueness are guaranteed by Theorem 12.1 in [2]. Similarly, with the given of  $(a_y, b_y : y \in \mathcal{Y})$ , the coefficient functions  $c_y, k_y \ (y \in \mathcal{Y})$ also form a linear ordinary differential equation system. Applying the Theorem 12.1 in [2], we can obtain the existence and uniqueness of  $c_y, k_y \ (y \in \mathcal{Y})$ .

#### 3.7.5 Multidimensional problem

In this subsection, we consider the multidimensional problem, which is a straightforward extension of the previous one-dimensional setup. The same type of Ricatti system to characterize the equilibrium and the value function is obtained, and we have a similar result as the Theorem 20.

Suppose that  $X_t$ ,  $W_t$  and  $\alpha_t$  take values in  $\mathbb{R}^d$ , and all components of  $W_t$  are independent. Suppose that the dynamic of the generic player is given by

$$X_t = X_0 + \int_0^t \left( \tilde{b}_1(Y_s, s) X_s + \tilde{b}_2(Y_s, s) \alpha_s \right) ds + W_t.$$

Consider the cost function

$$\begin{split} J[m](y, x, t, \bar{\mu}, \bar{\nu}) \\ &= \mathbb{E} \left[ \int_{t}^{T} \left( \frac{1}{2} \| \alpha_{s} \|_{2}^{2} + h(Y_{s}) \int_{\mathbb{R}^{d}} \| X_{s} - z \|_{2}^{2} m(dz) \right) ds \\ &\quad + g(Y_{T}) \int_{\mathbb{R}^{d}} \| X_{T} - z \|_{2}^{2} m(dz) \Big| X_{t} = x, Y_{t} = y, \mu_{t} = \bar{\mu}, \nu_{t} = \bar{\nu} \right] \\ &= \mathbb{E} \left[ \int_{t}^{T} \left( \frac{1}{2} \alpha_{s}^{\top} \alpha_{s} + h(Y_{s}) \left( X_{s}^{\top} X_{s} - 2 \mu_{s}^{\top} X_{s} + \nu_{s} \cdot \mathbb{1}_{d} \right) \right) ds \\ &\quad + g(Y_{T}) \left( X_{T}^{\top} X_{T} - 2 \mu_{T}^{\top} X_{T} + \nu_{T} \cdot \mathbb{1}_{d} \right) \Big| X_{t} = x, Y_{t} = y, \mu_{t} = \bar{\mu}, \nu_{t} = \bar{\nu} \right], \end{split}$$

where m is the joint density function in  $\mathbb{R}^d$ , and  $\mu, \nu$  take value in  $\mathbb{R}^d$ . For  $y \in \mathcal{Y}$ , define the Riccati

 $\operatorname{system}$ 

$$\begin{cases} a'_{y} + 2\tilde{b}_{1y}a_{y} - 2\tilde{b}_{2y}^{2}a_{y}^{2} + \sum_{i=1}^{\kappa} q_{y,i}a_{i} + h_{y} = 0, \\ b'_{y} + \left(2\tilde{b}_{1y} - 4\tilde{b}_{2y}^{2}a_{y}\right)b_{y} + \sum_{i=1}^{\kappa} q_{y,i}b_{i} + h_{y} = 0, \\ c'_{y} + da_{y} + db_{y} + \sum_{i=1}^{\kappa} q_{y,i}c_{i} = 0, \\ k'_{y} - 2\tilde{b}_{2y}^{2}a_{y}^{2} + 4\tilde{b}_{2y}^{2}a_{y}b_{y} + 2\tilde{b}_{1y}k_{y} + \sum_{i=1}^{\kappa} q_{y,i}k_{i} = 0, \\ a_{y}(T) = b_{y}(T) = g_{y}, c_{y}(T) = k_{y}(T) = 0. \end{cases}$$

$$(3.7.5)$$

**Theorem 39** (Verification theorem for MFG). There exists a unique solution  $(a_y, b_y, c_y, k_y : y \in \mathcal{Y})$ for the Riccati system (3.7.5). With these solutions, for  $t \in [0, T]$ , the MFG equilibrium path follows  $\hat{X} = \hat{X}[\hat{m}]$  is given by

$$d\hat{X}_{t} = \left(\tilde{b}_{1}(Y_{t}, t)\hat{X}_{t} - 2\tilde{b}_{2}^{2}(Y_{t}, t)a_{Y_{t}}(t)\left(\hat{X}_{t} - \hat{\mu}_{t}\right)\right)dt + dW_{t}, \quad \hat{X}_{0} = X_{0},$$

with equilibrium control  $\hat{\alpha}_t = -2\tilde{b}_2(Y_t, t)a_{Y_t}(t)\left(\hat{X}_t - \hat{\mu}_t\right)$ , where

$$d\hat{\mu}_t = \tilde{b}_1(Y_t, t)\hat{\mu}_t dt, \quad \hat{\mu}_0 = \mathbb{E}[X_0].$$

Moreover, the value function U is

$$U(m_0, y, x) = a_y(0)x^{\top}x - 2a_y(0)x^{\top}[m_0]_1 + k_y(0)[m_0]_1^{\top}[m_0]_1 + b_y(0)[m_0]_2^{\top}\mathbb{1}_d + c_y(0)$$

for  $y \in \mathcal{Y}$ .

The proof is similar to the one-dimensional problem, and we don't show the details here.

# Chapter 4

# Convergence rate of LQG mean field games with Brownian motion as common noise

# 4.1 Introduction

In this chapter, we examine the behavior of the triangular array  $\hat{X}_{t}^{(N)} = (\hat{X}_{it}^{(N)} : 1 \leq i \leq N)$  as  $N \to \infty$ , where  $\hat{X}_{it}^{(N)}$  represents the equilibrium state of the *i*-th player at time *t* in the *N*-player game, defined within the probability space  $(\Omega^{(N)}, \mathcal{F}^{(N)}, \mathbb{P}^{(N)}, \mathbb{P}^{(N)})$ . Additionally, we denote  $\hat{X}_{t}$  as the equilibrium path at time *t* derived from the associated MFG, defined in the probability space  $(\Omega, \mathcal{F}, \mathbb{F}, \mathbb{P})$ .

Considering the identical but not independent distribution  $\mathcal{L}(\hat{X}_{it}^{(N)})$ , the first question pertains to the convergence of  $\hat{X}_{1t}^{(N)}$ , which represents the generic path. It can be framed as follows:

(Q1) The  $\mathbb{W}_p$ -convergence rate of the representative equilibrium path,

$$\mathbb{W}_p\left(\mathcal{L}\left(\hat{X}_{1t}^{(N)}\right), \mathcal{L}\left(\hat{X}_t\right)\right) = O\left(N^{-?}\right).$$

Here,  $\mathbb{W}_p$  denotes the *p*-Wasserstein metric.

The existing literature extensively explores the convergence rate in this context. For (Q1), Theorem 2.4.9 of the monograph [14] establishes a convergence rate of  $O(N^{-1/2})$  using the  $W_1$ metric. More recently, [48] addresses (Q1) by introducing displacement monotonicity and controlled common noise, and Theorem 2.23 applies the maximum principle of forward-backward propagation of chaos to achieve the same convergence rate. Within the LQG framework, [50] also provides a convergence rate of 1/2 for the representative player.

The second question pertains to the convergence of the mean-field term, which is equivalent to the convergence of the empirical measure  $\rho(\hat{X}_t^{(N)}) = \frac{1}{N} \sum_{i=1}^N \delta_{\hat{X}_{it}^{(N)}}$  of N players. Given the Brownian motion, denoted as  $\tilde{W}_t$ , to be the common noise, the problem lies in determining the rate of convergence of the empirical measures to the MFG equilibrium measure

$$\hat{m}_t = \mathcal{L}\left(\hat{X}_t \middle| \mathcal{F}_t^{\tilde{W}}\right), \quad \forall t \in (0, T].$$

Thus, the second question can be stated as follows:

(Q2) The  $\mathbb{W}_p$ -convergence rate of empirical measures in  $L^p$  sense,

$$\left(\mathbb{E}\left[\mathbb{W}_{p}^{p}\left(\rho\left(\hat{X}_{t}^{\left(N\right)}\right),\mathcal{L}\left(\hat{X}_{t}\middle|\mathcal{F}_{t}^{\tilde{W}}\right)\right)\right]\right)^{\frac{1}{p}}=O\left(N^{-?}\right).$$

As for (Q2), Theorem 3.1 of [21] provides an answer, stating that the empirical measures exhibit a convergence rate of  $O(N^{-1/(2p)})$  in the  $\mathbb{W}_p$  distance for  $p \in [1, 2]$ . In [21], they also explore a related question that is both similar and more intriguing, which concerns the uniform  $\mathbb{W}_p$ -convergence rate:

(Q3) The t-uniform  $\mathbb{W}_p$ -convergence rate of empirical measures in  $L^p$  sense,

$$\left(\mathbb{E}\left[\sup_{t\in[0,T]}\mathbb{W}_{p}^{p}\left(\rho\left(\hat{X}_{t}^{(N)}\right),\mathcal{L}\left(\hat{X}_{t}\middle|\mathcal{F}_{t}^{\tilde{W}}\right)\right)\right]\right)^{\frac{1}{p}}=O\left(N^{-?}\right)$$

The answer provided by Theorem 3.1 in [21] reveals that the uniform convergence rate, as formulated in (Q3), is considerably slower compared to the convergence rate mentioned in (Q2). Specifically, the convergence rate for (Q3) is  $O(N^{-1/(d+8)})$  when p = 2, where d represents the dimension of the state space.

In this chapter, we specifically focus on a class of one-dimensional Linear-Quadratic-Gaussian (LQG) Mean Field Nash Games with Brownian motion as the common noise. It is important to note that the assumptions made in the aforementioned papers except [50] only account for linear growth in the state and control elements for the running cost, thus excluding the consideration of LQG. It is also noted that differences between [50] and the current chapter lie in various aspects: (1) The problem setting in this chapter considers Brownian motion as the common noise, whereas [50] employs a Markov chain. This discrepancy leads to significant differences in the subsequent analysis; (2) The work in [50] does not address the questions posed in (Q2) and (Q3).

Our main contribution is the establishment of the convergence rate of all three questions in the above in LQG framework. Firstly, this chapter establishes that the convergence rate of the *p*-Wasserstein metric for the distribution of the representative player is  $O(N^{-1/2})$  for  $p \in [1, 2]$ . Secondly, it demonstrates that the convergence rate of the *p*-Wasserstein metric for the empirical measure in the  $L^p$  sense is  $O(N^{-1/(2p)})$  for  $p \in [1, 2]$ . Lastly, this chapter shows that the convergence rate of the uniform *p*-Wasserstein metric for the empirical measure in the  $L^p$  sense is  $O(N^{-1/(2p)})$ for  $p \in [1, 2]$ , and  $O(N^{-1/2} \ln(N))$  for p = 1.

It is worth noting that the convergence rates obtained for (Q1) and (Q2) in the LQG framework align with the results found in existing literature, albeit under different conditions. Additionally, it is revealed that the uniform convergence rate of (Q3) may be slower than that of (Q2), which is consistent with the observations made by [21] from a similar perspective. Interestingly, when considering the specific case where p = 2 and d = 1, the uniform convergence rate of (Q3) is established as  $O(N^{-1/9})$  according to [21], while it is determined to be  $O(N^{-1/4})$  within our framework that incorporates the LQG structure.

Regarding (Q2), if the states  $(\hat{X}_{it}^{(N)} : 1 \leq i \leq N)$  were independent, the convergence rate could be determined as 1/(2p) based on Theorem 1 of [32] and Theorem 5.8 of [16], which provide convergence rates for empirical measures of independent and identically distributed sequences. However, in the mean-field game, the states  $\hat{X}_{it}^{(N)}$  are not independent of each other, despite having identical distributions. The correlation is introduced mainly by two factors: One is the system coupling arising from the mean-field term and the other is the common noise. Consequently, determining the convergence rate requires understanding the contributions of these two factors to the correlation among players.

In our proof, we rely on a specific decomposition (refer to Lemma 53 and the proof of the main theorem) of the underlying states. This decomposition reveals that the states can be expressed as a sum of a weakly correlated triangular array and a common noise. By analyzing the behavior of these components, we can address the correlation and establish the convergence rate.

Additionally, it is worth mentioning that a similar technique of dimension reduction in N-player LQG games have been previously utilized in [46] and related papers to establish decentralized Nash equilibria and the convergence rate in terms of value functions.

The remainder of the chapter is organized as follows: Section 4.2 outlines the problem setup and presents the main result. The proof of the main result, which relies on two propositions, is provided in Section 4.3. We establish the proof for these two propositions in Section 4.4 and Section 4.5. Some lemmas are given in the Appendix.

### 4.2 Problem setup and main results

#### 4.2.1 The formulation of equilibrium in mean field games

In this section, we present the formulation of the MFG in the sample space  $\Omega$ .

Let T > 0 be a given time horizon. We assume that  $W = \{W_t\}_{t\geq 0}$  is a standard Brownian motion constructed on the probability space  $(\bar{\Omega}, \bar{\mathcal{F}} = \bar{\mathcal{F}}_T, \bar{\mathbb{P}}, \bar{\mathbb{F}} = \{\bar{\mathcal{F}}_t\}_{t\geq 0})$ . Similarly, the process  $\tilde{W} = \{\tilde{W}_t\}_{t\geq 0}$  is a standard Brownian motion constructed on the probability space  $(\tilde{\Omega}, \tilde{\mathcal{F}} = \tilde{\mathcal{F}}_T, \tilde{\mathbb{P}}, \tilde{\mathbb{F}} = \{\tilde{\mathcal{F}}_t\}_{t\geq 0})$ . We define the product structure as follows:

$$\Omega = \bar{\Omega} \times \bar{\Omega}, \quad \mathcal{F}, \quad \mathbb{F} = \{\mathcal{F}_t\}_{t \ge 0}, \quad \mathbb{P},$$

where  $(\mathcal{F}, \mathbb{P})$  is the completion of  $(\overline{\mathcal{F}} \otimes \tilde{\mathcal{F}}, \overline{\mathbb{P}} \otimes \widetilde{\mathbb{P}})$  and  $\mathbb{F}$  is the complete and right continuous augmentation of  $\{\overline{\mathcal{F}}_t \otimes \tilde{\mathcal{F}}_t\}_{t \geq 0}$ .

Note that, W and  $\tilde{W}$  are two Brownian motions from separate sample spaces  $\bar{\Omega}$  and  $\tilde{\Omega}$ , they are independent of each other in their product space  $\Omega$ . In our manuscript, W is called individual or idiosyncratic noise, and  $\tilde{W}$  is called common noise, see their different roles in the problem formulation later defined via fixed point condition (4.2.4). To proceed, we denote by  $L^p := L^p(\Omega, \mathbb{P})$  the set of random variables X on  $(\Omega, \mathcal{F}, \mathbb{P})$  with finite *p*-th moment with norm  $||X||_p = (\mathbb{E}[|X|^p])^{1/p}$ and by  $L^p_{\mathbb{F}} := L^p_{\mathbb{F}}(\Omega \times [0,T])$  the space of all  $\mathbb{R}$  valued  $\mathbb{F}$ -progressively measurable random processes  $\alpha$  such that

$$\mathbb{E}\left[\int_0^T |\alpha_t|^p dt\right] < \infty.$$

Let  $\mathcal{P}_p(\mathbb{R})$  denote the Wasserstein space of probability measures  $\mu$  on  $\mathbb{R}$  satisfying  $\int_{\mathbb{R}} |x|^p d\mu(x) < \infty$  endowed with *p*-Wasserstein metric  $\mathbb{W}_p(\cdot, \cdot)$  defined by

$$\mathbb{W}_p(\mu,\nu) = \inf_{\pi \in \Pi(\mu,\nu)} \left( \int_{\mathbb{R} \times \mathbb{R}} |x-y|^p d\pi(x,y) \right)^{\frac{1}{p}},$$

where  $\Pi(\mu, \nu)$  is the collection of all probability measures on  $\mathbb{R} \times \mathbb{R}$  with its marginals agreeing with  $\mu$  and  $\nu$ .

Let  $X_0 \in L^2$  be a random variable that is independent with W and  $\tilde{W}$ . For any control  $\alpha \in L^2_{\mathbb{F}}$ , consider the state  $X = \{X_t\}_{t\geq 0}$  of the generic player governed by a stochastic differential equation (SDE)

$$dX_t = \alpha_t dt + dW_t + d\tilde{W}_t \tag{4.2.1}$$

with the initial value  $X_0$ , where the underlying process  $X : [0, T] \times \Omega \mapsto \mathbb{R}$ . Given a random measure flow  $m : (0, T] \times \Omega \mapsto \mathcal{P}_2(\mathbb{R})$ , the generic player wants to minimize the expected accumulated cost on [0, T]:

$$J(x,\alpha) = \mathbb{E}\left[\int_0^T \left(\frac{1}{2}\alpha_s^2 + F(X_s, m_s)\right) ds \middle| X_0 = x\right]$$
(4.2.2)

with some given cost function  $F : \mathbb{R} \times \mathcal{P}_2(\mathbb{R}) \mapsto \mathbb{R}$ .

The objective of the control problem for the generic player is to find its optimal control  $\hat{\alpha} \in \mathcal{A} := L^4_{\mathbb{F}}$  to minimize the total cost, i.e.,

$$V[m](x) = J[m](x, \hat{\alpha}) \le J[m](x, \alpha), \quad \forall \alpha \in \mathcal{A}.$$
(4.2.3)

Associated to the optimal control  $\hat{\alpha}$ , we denote the optimal path by  $\hat{X} = {\{\hat{X}_t\}_{t \ge 0}}$ .

Next, to introduce the MFG Nash equilibrium, it is useful to emphasize the dependence of the optimal path and optimal control of the generic player, as well as its associated value, on the underlying measure flow m. These quantities are denoted as  $\hat{X}_t[m]$ ,  $\hat{\alpha}_t[m]$ , J[m], and V[m], respectively.

We now present the definitions of the equilibrium measure, equilibrium path, and equilibrium control. Please also refer to page 127 of [17] for a general setup with a common noise.

**Definition 40.** Given an initial distribution  $\mathcal{L}(X_0) = m_0 \in \mathcal{P}_2(\mathbb{R})$ , a random measure flow  $\hat{m} = \hat{m}(m_0)$  is said to be an MFG equilibrium measure if it satisfies the fixed point condition

$$\hat{m}_t = \mathcal{L}\left(\left. \hat{X}_t[\hat{m}] \right| \tilde{\mathcal{F}}_t \right), \ \forall 0 < t \le T, \quad almost \ surely \ in \ \mathbb{P}.$$

$$(4.2.4)$$

The path  $\hat{X}$  and the control  $\hat{\alpha}$  associated with  $\hat{m}$  are called the MFG equilibrium path and equilibrium

control, respectively.



Figure 4.1: MFG diagram 2.

The flowchart of the MFG diagram is given in Figure 4.1. It is noted from the optimality condition (4.2.3) and the fixed point condition (4.2.4) that

$$J[\hat{m}](x,\hat{\alpha}) \le J[\hat{m}](x,\alpha), \quad \forall \alpha$$

holds for the equilibrium measure  $\hat{m}$  and its associated equilibrium control  $\hat{\alpha}$ , while it is not

$$J[\hat{m}](x,\hat{\alpha}) \leq J[m](x,\alpha), \quad \forall \alpha,m.$$

Otherwise, this problem turns into a McKean-Vlasov control problem, which is essentially different from the current Mean Field Games setup. Readers refer to [19, 18] to see the analysis of this different model as well as some discussion of the differences between these two problems.

#### 4.2.2 The formulation of Nash equilibrium in the *N*-player game

In this subsection, we set up the N-player game and define the Nash equilibrium of the N-player game in the sample space  $\Omega^{(N)}$ . Firstly, let  $W^{(N)} = (W_i^{(N)} : i = 1, 2, ..., N)$  be an N-dimensional standard Brownian motion constructed on the space  $(\bar{\Omega}^{(N)}, \bar{\mathcal{F}}^{(N)}, \bar{\mathbb{P}}^{(N)}, \bar{\mathbb{F}}^{(N)} = \{\bar{\mathcal{F}}_t^{(N)}\}_{t\geq 0})$  and  $\tilde{W} = \{\tilde{W}_t\}_{t\geq 0}$  be the common noise in MFG defined in Section 4.2.1 on  $(\tilde{\Omega}, \tilde{\mathcal{F}}, \tilde{\mathbb{P}})$ . The probability space for the N-player game is  $(\Omega^{(N)}, \mathcal{F}^{(N)}, \mathbb{P}^{(N)}, \mathbb{P}^{(N)})$ , which is constructed via the product structure with

$$\Omega^{(N)} = \bar{\Omega}^{(N)} \times \tilde{\Omega}, \quad \mathcal{F}^{(N)}, \quad \mathbb{F}^{(N)} = \left\{ \mathcal{F}_t^{(N)} \right\}_{t \ge 0}, \quad \mathbb{P}^{(N)}$$

where  $(\mathcal{F}^{(N)}, \mathbb{P}^{(N)})$  is the completion of  $(\bar{\mathcal{F}}^{(N)} \otimes \tilde{\mathcal{F}}, \bar{\mathbb{P}}^{(N)} \otimes \tilde{\mathbb{P}})$  and  $\mathbb{F}^{(N)}$  is the complete and right continuous augmentation of  $\{\bar{\mathcal{F}}_t^{(N)} \otimes \tilde{\mathcal{F}}_t\}_{t \geq 0}$ .

Consider a stochastic dynamic game with N players, where each player  $i \in \{1, 2, ..., N\}$  controls a state process  $X_i^{(N)} = \{X_{it}^{(N)}\}_{t \ge 0}$  in  $\mathbb{R}$  given by

$$dX_{it}^{(N)} = \alpha_{it}^{(N)}dt + dW_{it}^{(N)} + d\tilde{W}_t, \quad X_{i0}^{(N)} = x_i^{(N)}$$
(4.2.5)

with a control  $\alpha_i^{(N)}$  in an admissible set  $\mathcal{A}^{(N)} := L^4_{\mathbb{F}^{(N)}}$  and random initial state  $x_i^{(N)}$ . Given the strategies  $\alpha_{-i}^{(N)} = (\alpha_1^{(N)}, \dots, \alpha_{i-1}^{(N)}, \alpha_{i+1}^{(N)}, \dots, \alpha_N^{(N)})$  from other players, the objective of player *i* is to select a control  $\alpha_i^{(N)} \in \mathcal{A}^{(N)}$  to minimize her expected total cost given by

$$J_{i}^{N}\left(x^{(N)},\alpha_{i}^{(N)};\alpha_{-i}^{(N)}\right) = \mathbb{E}\left[\int_{0}^{T}\left(\frac{1}{2}\left(\alpha_{it}^{(N)}\right)^{2} + F\left(X_{it}^{(N)},\rho\left(X_{t}^{(N)}\right)\right)\right)dt \middle| X_{0}^{(N)} = x^{(N)}\right],\tag{4.2.6}$$

where  $x^{(N)} = (x_1^{(N)}, x_2^{(N)}, \dots, x_N^{(N)})$  is a  $\mathbb{R}^N$ -valued random vector in  $\Omega^{(N)}$  to denote the initial state for N players, and

$$\rho\left(x^{(N)}\right) = \frac{1}{N} \sum_{i=1}^{N} \delta_{x_i^{(N)}}$$

is the empirical measure of the vector  $x^{(N)}$  with Dirac measure  $\delta$ . We use the notation  $\alpha^{(N)} := (\alpha_i^{(N)}, \alpha_{-i}^{(N)}) = (\alpha_1^{(N)}, \alpha_2^{(N)}, \dots, \alpha_N^{(N)})$  to denote the control from N players as a whole. Next, we give the equilibrium value function and equilibrium path in the sense of the Nash game.

1. The value function of player i for i = 1, 2, ..., N of the Nash game is defined Definition 41. by  $V^N = (V_i^N : i = 1, 2, ..., N)$  satisfying the equilibrium condition

$$V_i^N\left(x^{(N)}\right) := J_i^N\left(x^{(N)}, \hat{\alpha}_i^{(N)}; \hat{\alpha}_{-i}^{(N)}\right) \le J_i^N\left(x^{(N)}, \alpha_i^{(N)}; \hat{\alpha}_{-i}^{(N)}\right), \tag{4.2.7}$$

for all  $\alpha_i^{(N)} \in \mathcal{A}^{(N)}$ .

2. The equilibrium path of the N-player game is the N-dimensional random path  $\hat{X}_t^{(N)} = (\hat{X}_{1t}^{(N)}, \hat{X}_{2t}^{(N)}, \dots, \hat{X}_{Nt}^{(N)})$  driven by (4.2.5) associated to the control  $\hat{\alpha}_t^{(N)}$  satisfying the equilibrium condition of (4.2.7).

#### 4.2.3Main result

We consider three convergence questions on the N-player game defined in  $\Omega^{(N)}$ : The first one is the convergence of the representative path  $\hat{X}_{it}^{(N)}$ , the second one is the convergence of the empirical measure  $\rho(\hat{X}_t^{(N)})$ , while the last one is the *t*-uniform convergence of the empirical measure  $\rho(\hat{X}_t^{(N)})$ . To be precise, we shall assume the following throughout this chapter:

•  $\mathbb{E}[|X_0|^q] < \infty$  for some q > 4. Assumption 42.

• The initials  $X_{i0}^{(N)}$  of the N-player game are i.i.d. random variables in  $\Omega^{(N)}$  with the same distribution as  $\mathcal{L}(X_0)$  in the MFG.

Note that the equilibrium path  $\hat{X}_{t}^{(N)} = (\hat{X}_{it}^{(N)} : i = 1, 2, ..., N)$  is a vector-valued stochastic process. Due to the Assumption 42, the game is invariant to index reshuffling of N players and the elements in  $(\hat{X}_{it}^{(N)} : i = 1, 2, ..., N)$  have identical distributions, but they are not independent of each other.

So, the first question on the representative path is indeed about  $\hat{X}_{1t}^{(N)}$  in  $\Omega^{(N)}$  and we are interested in how fast it converges to  $\hat{X}_t$  in  $\Omega$  in distribution:

(Q1) The  $\mathbb{W}_p$ -convergence rate of the representative equilibrium path,

$$\mathbb{W}_p\left(\mathcal{L}\left(\hat{X}_{1t}^{(N)}\right), \mathcal{L}\left(\hat{X}_t\right)\right) = O\left(N^{-?}\right).$$

The second question is about the convergence of the empirical measure  $\rho(\hat{X}_t^{(N)})$  of the N-player game defined by

$$\rho\left(\hat{X}_{t}^{(N)}\right) = \frac{1}{N} \sum_{i=1}^{N} \delta_{\hat{X}_{it}^{(N)}}.$$

We are interested in how fast this converges to the MFG equilibrium measure given by

$$\hat{m}_t = \mathcal{L}\left(\hat{X}_t \middle| \tilde{\mathcal{F}}_t\right), \quad \forall t \in (0,T].$$

(Q2') The  $\mathbb{W}_p$ -convergence rate of empirical measures,

$$\mathbb{W}_p\left(\rho\left(\hat{X}_t^{(N)}\right), \mathcal{L}\left(\hat{X}_t\middle| \tilde{\mathcal{F}}_t\right)\right) = O\left(N^{-?}\right).$$

Note that the left-hand side of the above equality is a random quantity and one shall be more precise about what the Big O notation means in this context. Indeed, by the definition of the empirical measure,  $\rho(\hat{X}_t^{(N)})$  is a random distribution measurable by  $\sigma$ -algebra generated by the random vector  $\hat{X}_t^{(N)}$ . On the other hand,  $\mathcal{L}(\hat{X}_t|\tilde{\mathcal{F}}_t)$  is a random distribution measurable by the  $\sigma$ -algebra  $\tilde{\mathcal{F}}_t$ . Therefore, from the construction of the product probability space  $\Omega^{(N)}$  in Section 4.2.2, both random distributions  $\rho(\hat{X}_t^{(N)})$  and  $\mathcal{L}(\hat{X}_t|\tilde{\mathcal{F}}_t)$  are measurable with respect to  $\mathcal{F}_t^{(N)} = \bar{\mathcal{F}}_t^{(N)} \otimes \tilde{\mathcal{F}}_t$ . Consequently,  $\mathbb{W}_p(\rho(\hat{X}_t^{(N)}), \mathcal{L}(\hat{X}_t|\tilde{\mathcal{F}}_t))$  is a random variable in the probability space  $(\Omega^{(N)}, \mathcal{F}^{(N)}, \mathbb{P}^{(N)})$  and we will focus on a version of (Q2') in the  $L^p$  sense:

(Q2) The  $\mathbb{W}_p$ -convergence rate of empirical measures in  $L^p$  sense for each  $t \in [0, T]$ ,

$$\left(\mathbb{E}\left[\mathbb{W}_{p}^{p}\left(\rho\left(\hat{X}_{t}^{\left(N\right)}\right),\mathcal{L}\left(\left.\hat{X}_{t}\right|\right.\tilde{\mathcal{F}}_{t}\right)\right)\right]\right)^{\frac{1}{p}}=O\left(N^{-?}\right).$$

In addition, we also study the following related question:

(Q3) The *t*-uniform  $\mathbb{W}_p$ -convergence rate of empirical measures in  $L^p$  sense,

$$\left(\mathbb{E}\left[\sup_{0\leq t\leq T}\mathbb{W}_{p}^{p}\left(\rho\left(\hat{X}_{t}^{\left(N\right)}\right),\mathcal{L}\left(\hat{X}_{t}\middle|\,\tilde{\mathcal{F}}_{t}\right)\right)\right]\right)^{\frac{1}{p}}=O\left(N^{-?}\right).$$

In this chapter, we will study the above three questions (Q1), (Q2), and (Q3) in the framework of LQG structure with Brownian motion as a common noise with the following function F in the cost functional (4.2.2).

**Assumption 43.** Let the function  $F : \mathbb{R} \times \mathcal{P}_2(\mathbb{R}) \mapsto \mathbb{R}$  be given in the form of

$$F(x,m) = k \int_{\mathbb{R}} (x-z)^2 m(dz) = k(x^2 - 2x[m]_1 + [m]_2)$$
(4.2.8)

for some k > 0, where  $[m]_1, [m]_2$  are the first and second moment of the measure m.

The main result of this chapter is presented below. Let us recall that q denotes the parameter defined in Assumption 42.

**Theorem 44.** Under Assumptions 42-43, for any  $p \in [1, 2]$ , we have

1. The  $\mathbb{W}_p$ -convergence rate of the representative equilibrium path is 1/2, i.e.,

$$\mathbb{W}_p\left(\mathcal{L}\left(\hat{X}_{1t}^{(N)}\right), \mathcal{L}\left(\hat{X}_t\right)\right) = O\left(N^{-\frac{1}{2}}\right).$$

2. The  $\mathbb{W}_p$ -convergence rate of empirical measures in  $L^p$  sense is

$$\mathbb{E}\left[\mathbb{W}_{p}^{p}\left(\rho\left(\hat{X}_{t}^{\left(N\right)}\right),\mathcal{L}\left(\hat{X}_{t}\middle|\,\tilde{\mathcal{F}}_{t}\right)\right)\right]=O\left(N^{-\frac{1}{2}}\right).$$

3. The uniform  $\mathbb{W}_p$ -convergence rate of empirical measures in  $L^p$  sense is

$$\mathbb{E}\left[\sup_{0 \le t \le T} \mathbb{W}_p^p\left(\rho\left(\hat{X}_t^{(N)}\right), \mathcal{L}\left(\hat{X}_t\middle| \tilde{\mathcal{F}}_t\right)\right)\right] = \begin{cases} O\left(N^{-\frac{1}{2}}\ln(N)\right), & \text{if } p = 1, \\ O\left(N^{-\frac{1}{2}}\right), & \text{if } 1$$

We would like to provide some additional remarks on our main result. Firstly, the cost function F defined in (4.2.6) applies to the running cost for the *i*-th player in the *N*-player game, and it takes the form:

$$F\left(X_{it}^{(N)}, \rho\left(X_{t}^{(N)}\right)\right) = \frac{k}{N} \sum_{j=1}^{N} \left(X_{it}^{(N)} - X_{jt}^{(N)}\right)^{2}.$$
(4.2.9)

Interestingly, if k < 0, although F does satisfy the Lasry-Lions monotonicity ([13]) as demonstrated in Appendix 6.1 of [50], there is no global solution for MFG due to the concavity in x. On the contrary, when k > 0, F satisfies the displacement monotonicity proposed in [33] as shown by the following derivation:

$$\mathbb{E}\left[\left(F_x(X_1,\mathcal{L}(X_1)) - F_x(X_2,\mathcal{L}(X_2))\right)(X_1 - X_2)\right] = 2k\left(\mathbb{E}\left[\left(X_1 - X_2\right)^2\right] - \left(\mathbb{E}[X_1 - X_2]\right)^2\right) \ge 0.$$

# 4.3 Proof of the main result with two propositions

Our objective is to investigate the relations between  $(\hat{X}_{1t}^{(N)}, \hat{X}_{2t}^{(N)}, \dots, \hat{X}_{Nt}^{(N)})$  and  $\hat{X}_t$  described in (Q1), (Q2), and (Q3). In this part, we will give the proof of Theorem 44 based on two propositions whose proofs will be given later.

**Proposition 45.** Under Assumptions 42-43, the MFG equilibrium path  $\hat{X} = \hat{X}[\hat{m}]$  is given by

$$d\hat{X}_{t} = -2a(t)\left(\hat{X}_{t} - \hat{\mu}_{t}\right)dt + dW_{t} + d\tilde{W}_{t}, \quad \hat{X}_{0} = X_{0},$$
(4.3.1)

where a is the solution of

$$a'(t) - 2a^{2}(t) + k = 0, \quad a(T) = 0,$$
 (4.3.2)

and  $\hat{\mu}$  is

$$\hat{\mu}_t := \mathbb{E}\left[\left.\hat{X}_t\right| \tilde{\mathcal{F}}_t\right] = \mathbb{E}[X_0] + \tilde{W}_t.$$

Moreover, the equilibrium control follows

$$\hat{\alpha}_t = -2a(t)\left(\hat{X}_t - \hat{\mu}_t\right). \tag{4.3.3}$$

**Proposition 46.** Suppose Assumptions 42-43 hold. For the N-player game, the path and the control of player i under the equilibrium are given by

$$d\hat{X}_{it}^{(N)} = -2a^N(t) \left( \hat{X}_{it}^{(N)} - \frac{1}{N-1} \sum_{j \neq i}^N \hat{X}_{jt}^{(N)} \right) dt + dW_{it}^{(N)} + d\tilde{W}_t,$$
(4.3.4)

and

$$\hat{\alpha}_{it}^{(N)} = -2a^N(t) \left( \hat{X}_{it}^{(N)} - \frac{1}{N-1} \sum_{j \neq i}^N \hat{X}_{jt}^{(N)} \right)$$

respectively for i = 1, 2, ..., N, where  $a^N$  is the solution of

$$a' - \frac{2(N+1)}{N-1}a^2 + \frac{N-1}{N}k = 0, \quad a(T) = 0.$$
(4.3.5)

#### 4.3.1 Preliminaries

We first recall the convergence rate of empirical measures of i.i.d. sequence provided in Theorem 1 of [32] and Theorem 5.8 of [16].

**Lemma 47.** Let d = 1 or 2. Suppose  $\{X_i : i \in \mathbb{N}\}$  is a sequence of d dimensional i.i.d. random variables with  $\mathbb{E}[|X_1|^q] < \infty$  for some q > 4. Then, the empirical measure

$$\rho^N(X) = \frac{1}{N} \sum_{i=1}^N \delta_{X_i}$$

satisfies

$$\mathbb{E}\left[\mathbb{W}_{p}^{p}\left(\rho^{N}(X),\mathcal{L}(X_{1})\right)\right] = \begin{cases} O\left(N^{-1/2}\right), & \text{if } p \in (1,2], \\ O\left(N^{-1/2}\right), & \text{if } p = 1, d = 1, \\ O\left(N^{-1/2}\ln N\right), & \text{if } p = 1, d = 2. \end{cases}$$

Next, we give the definition of some notations that will be used in the following part. Denote  $C_b(\mathbb{R}^d)$  to be the collection of bounded and continuous functions on  $\mathbb{R}^d$ , and let  $C_b^1(\mathbb{R}^d) \subset C_b(\mathbb{R}^d)$  be the space of functions on  $\mathbb{R}^d$  whose first order derivative is also bounded and continuous.

**Lemma 48.** Suppose  $m_1, m_2$  are two probability measures on  $\mathcal{B}(\mathbb{R}^d)$  and  $f \in C_b^1(\mathbb{R}^d, \mathbb{R})$ , where  $\mathcal{B}(\mathbb{R}^d)$  is the Borel set on  $\mathbb{R}^d$ . Then,

$$\mathbb{W}_p(f_*m_1, f_*m_2) \le |Df|_0 \mathbb{W}_p(m_1, m_2)$$

where  $f_*m_j$  is the pushforward measure for j = 1, 2, and  $|Df|_0 = \sup_{x \in \mathbb{R}^d} \max\{|\partial_{x_i}f(x)| : i = 1, 2, \dots, d\}.$ 

Proof. We define a function  $F(x, y) = (f(x), f(y)) : \mathbb{R}^{2d} \mapsto \mathbb{R}^2$ . Note that, for any  $\pi \in \Pi(m_1, m_2)$ ,  $F_*\pi \in \Pi(f_*m_1, f_*m_2)$ , i.e.,

$$F_*\Pi(m_1, m_2) \subset \Pi(f_*m_1, f_*m_2).$$

Therefore, we have the following inequalities:

$$\begin{split} \mathbb{W}_{p}^{p}(f_{*}m_{1},f_{*}m_{2}) &= \inf_{\pi'\in\Pi(f_{*}m_{1},f_{*}m_{2})} \int_{\mathbb{R}^{2}} |x-y|^{p}\pi'(dx,dy) \\ &\leq \inf_{\pi'\in F_{*}\Pi(m_{1},m_{2})} \int_{\mathbb{R}^{2}} |x-y|^{p}\pi'(dx,dy) \\ &= \inf_{\pi\in\Pi(m_{1},m_{2})} \int_{\mathbb{R}^{2d}} |f(x)-f(y)|^{p}\pi(dx,dy) \\ &\leq |Df|_{0}^{p} \inf_{\pi\in\Pi(m_{1},m_{2})} \int_{\mathbb{R}^{2d}} |x-y|^{p}\pi(dx,dy) \\ &= |Df|_{0}^{p}\mathbb{W}_{p}^{p}(m_{1},m_{2}). \end{split}$$

**Lemma 49.** Let  $\{X_i : i \in \mathbb{N}\}$  be a sequence of d dimensional random variables in  $(\Omega, \mathcal{F}, \mathbb{P})$ . Let  $f \in C_b^1(\mathbb{R}^d)$ . We also denote by f(X) the sequence  $\{f(X_i) : i \in \mathbb{N}\}$ . Then

$$\mathbb{W}_p\left(\rho^N(f(X)), \mathcal{L}(f(X_1))\right) \le |Df|_0 \mathbb{W}_p\left(\rho^N(X), \mathcal{L}(X_1)\right), \quad almost \ surely$$

where  $|Df|_0 = \sup_{x \in \mathbb{R}^d} \max\{|\partial_{x_i} f(x)| : i = 1, 2, ..., d\}.$ 

*Proof.* For any sequence  $\{c_i : i \in \mathbb{N}\}$  in  $\mathbb{R}^d$ , the empirical measure  $\rho^N(c) := \frac{1}{N} \sum_{i=1}^N \delta_{c_i}$  satisfies

$$\rho^N(f(c)) = f_*\rho^N(c),$$

since

$$\langle \phi, \rho^N(f(c)) \rangle = \frac{1}{N} \sum_{i=1}^N \phi(f(c_i)) = \langle \phi \circ f, \rho^N(c) \rangle, \quad \forall \phi \in C_b(\mathbb{R}^d).$$

This implies that

$$\rho^N(f(X)) = f_*\rho^N(X)$$
, almost surely.

On the other hand, we also have

$$\mathcal{L}(f(X_1))(A) = \mathbb{P}(f(X_1) \in A) = \mathbb{P}(X_1 \in f^{-1}(A)) = f_*\mathcal{L}(X_1)(A), \quad \forall A \in \mathcal{B}(\mathbb{R}^d).$$

Therefore, the conclusion follows by applying Lemma 48.

#### 4.3.2 Empirical measures of a sequence with a common noise

We are going to apply lemmas from the previous subsection to study the convergence of empirical measures of a sequence with a common noise in the following sense.

**Definition 50.** We say a sequence of random variables  $X = \{X_i : i \in \mathbb{N}\}$  is a sequence with a common noise, if there exists a random variable  $\beta$  such that

- $X \beta = \{X_i \beta : i \in \mathbb{N}\}$  is a sequence of *i.i.d.* random variables,
- $\beta$  is independent to  $X \beta$ .

By this definition, a sequence with a common noise is i.i.d. if and only if  $\beta$  is a deterministic constant.

**Example 2.** Let q > 4 be a given constant and  $X = \{X_i : i \in \mathbb{N}\}$  be a 1-dimensional sequence of  $L^q$  random variables with a common noise term  $\beta$ , where

$$X_i - \beta = \gamma_i + \lambda \alpha_i.$$

In above,  $\{(\alpha_i, \gamma_i) : i \in \mathbb{N}\}$  is a sequence of 2-dimensional i.i.d. random variables independent to  $\beta$ , and  $\lambda$  is a given non-negative constant. Let  $\rho^N(X)$  be the empirical measure defined by

$$\rho^N(X) = \frac{1}{N} \sum_{i=1}^N \delta_{X_i}.$$

The first question is

(Qa) In Example 2, where does  $\rho^N(X)$  converge to?

For any test function  $\phi \in C_b(\mathbb{R})$ ,

$$\langle \phi, \rho^N(X) \rangle = \frac{1}{N} \sum_{i=1}^N \phi(X_i) = \frac{1}{N} \sum_{i=1}^N \phi(\gamma_i + \lambda \alpha_i + \beta).$$

Since  $\beta$  is independent to  $(\alpha_i, \gamma_i)$ , by Example 4.1.5 of [23] together with the Law of Large Numbers, we have

$$\frac{1}{N}\sum_{i=1}^{N}\phi(\gamma_i+\lambda\alpha_i+c)\to\mathbb{E}[\phi(\gamma_1+\lambda\alpha_1+c)]=\mathbb{E}[\phi(\gamma_1+\lambda\alpha_1+\beta)|\beta=c],\quad\forall c\in\mathbb{R}.$$

Therefore, we conclude that

$$\langle \phi, \rho^N(X) \rangle \to \mathbb{E}[\phi(\gamma_1 + \lambda \alpha_1 + \beta)|\beta], \quad \beta - a.s.$$
  
=  $\langle \phi, \mathcal{L}(\gamma_1 + \lambda \alpha_1 + \beta|\beta) \rangle, \quad \beta - a.s.$ 

Hence, the answer for the (Qa) is

•  $\rho^N(X) \Rightarrow \mathcal{L}(X_1|\beta)$ ,  $\beta$ -a.s. More precisely, since all random variables are square-integrable, the weak convergence implies, for all  $p \in [1, 2]$ ,

$$\mathbb{W}_p\left(\rho^N(X), \mathcal{L}(X_1|\beta)\right) \to 0, \quad \beta - a.s.$$

The next question is

(Qb) In Example 2, what's the convergence rate in the sense  $\mathbb{E}\left[\mathbb{W}_{p}^{p}\left(\rho^{N}(X),\mathcal{L}\left(X_{1}|\beta\right)\right)\right]$ ?

Since  $\beta$  is independent to  $\gamma_1 + \lambda \alpha_1$ , by Example 4.1.5 of [23], we have

$$\mathbb{E}[\phi(\gamma_1 + \lambda \alpha_1 + \beta) | \beta = c] = \mathbb{E}[\phi(\gamma_1 + \lambda \alpha_1 + c)], \quad \forall \phi \in C_b(\mathbb{R}), c \in \mathbb{R},$$

or equivalently, if one takes  $c = \beta(\omega)$ ,

$$\mathcal{L}(X_1|\beta)(\omega) = \mathcal{L}(\gamma_1 + \lambda\alpha_1 + \beta|\beta)(\omega) = \mathcal{L}(\gamma_1 + \lambda\alpha_1 + c)$$

On the other hand, with  $c = \beta(\omega)$ ,

$$\rho^{N}(X)(\omega) = \rho^{N}(X(\omega)) = \frac{1}{N} \sum_{i=1}^{N} \delta_{\gamma_{i}(\omega) + \lambda \alpha_{i}(\omega) + c}$$

From the above two identities, with  $c = \beta(\omega)$ , we can write

$$\mathbb{W}_p\left(\rho^N(X)(\omega), \mathcal{L}(X_1|\beta=c)(\omega)\right) = \mathbb{W}_p\left(\frac{1}{N}\sum_{i=1}^N \delta_{\gamma_i(\omega)+\lambda\alpha_i(\omega)+c}, \mathcal{L}(\gamma_1+\lambda\alpha_1+c)\right).$$
(4.3.6)

Now we can conclude (Qb) in the next lemma.

**Lemma 51.** Let  $p \in [1,2]$  be a given constant. For a sequence  $X = \{X_i : i \in \mathbb{N}\}$  with a common noise  $\beta$  as of Example 2, we have

$$\mathbb{E}\left[\mathbb{W}_p^p\left(\rho^N(X),\mathcal{L}(X_1|\beta)\right)\right] = O\left(N^{-\frac{1}{2}}\right).$$

*Proof.* Originally,  $X_i = \gamma_i + \lambda \alpha_i + \beta$  of Example 2 are dependent due to the common term  $\beta$ . We apply (4.7.2) in Lemma 60 in Appendix to (4.3.6) and obtain

$$\mathbb{W}_p\left(\rho^N(X)(\omega), \mathcal{L}(X_1|\beta)(\omega)\right) = \mathbb{W}_p\left(\frac{1}{N}\sum_{i=1}^N \delta_{\gamma_i(\omega)+\lambda\alpha_i(\omega)+\beta(\omega)}, \mathcal{L}(\gamma_1+\lambda\alpha_1+\beta(\omega))\right)$$
$$= \mathbb{W}_p\left(\rho^N(\gamma(\omega)+\lambda\alpha(\omega)), \mathcal{L}(\gamma_1+\lambda\alpha_1)\right).$$

Now, the convergence of empirical measures is equivalent to the ones of i.i.d. sequence  $\{\gamma_i + \lambda \alpha_i : i \in \mathbb{N}\}$ . The conclusion follows from Lemma 47.

Next, we present the uniform convergence rate by combining Lemma 49.

**Lemma 52.** In Example 2, we use  $X(\lambda)$  to denote X to emphasize its dependence on  $\lambda$ . Then,

$$\mathbb{E}\left[\sup_{\lambda \in [0,1]} \mathbb{W}_p^p\left(\rho^N(X(\lambda)), \mathcal{L}\left(X_1(\lambda)|\beta\right)\right)\right] = \begin{cases} O\left(N^{-\frac{1}{2}}\ln(N)\right), & \text{if } p = 1, \\ O\left(N^{-\frac{1}{2}}\right), & \text{if } 1$$

*Proof.* Note that, by (4.7.2) in Lemma 60 in Appendix,

$$\mathbb{W}_{p}^{p}\left(\rho^{N}(X(\lambda)),\mathcal{L}\left(X_{1}(\lambda)|\beta\right)\right)=\mathbb{W}_{p}^{p}\left(\rho^{N}(\gamma+\lambda\alpha),\mathcal{L}\left(\gamma_{1}+\lambda\alpha_{1}\right)\right)$$

Next, applying Lemma 49 with  $f(x, y) = x + \lambda y$ , we obtain

$$\sup_{\lambda \in [0,1]} \mathbb{W}_p^p \left( \rho^N(\gamma + \lambda \alpha), \mathcal{L} \left( \gamma_1 + \lambda \alpha_1 \right) \right) \leq \sup_{\lambda \in [0,1]} \max\{1, \lambda^p\} \mathbb{W}_p^p \left( \rho^N((\gamma, \alpha)), \mathcal{L} \left( (\gamma_1, \alpha_1) \right) \right) \\ = \mathbb{W}_p^p \left( \rho^N((\gamma, \alpha)), \mathcal{L} \left( (\gamma_1, \alpha_1) \right) \right).$$

At last, using Lemma 47 for the 2-dimensional i.i.d. sequence  $\{(\gamma_i, \alpha_i) : i \in \mathbb{N}\}$ , we obtain the desired conclusion.

#### 4.3.3 Generalization of the convergence to triangular arrays

Unfortunately,  $(\hat{X}_{1t}^{(N)}, \hat{X}_{2t}^{(N)}, \dots, \hat{X}_{Nt}^{(N)})$  of the *N*-player's game does not have a clean structure with a common noise term  $\beta$  given in Example 2. Therefore, we need a generalization of the convergence result in Example 2 to a triangular array. To proceed, we provide the following lemma.

**Lemma 53.** Let  $\lambda > 0$ , q > 4, and

$$X_i^N(\lambda) = \gamma_i^N + \lambda \alpha_i^N + \Delta_i^N(\lambda) + \beta, \text{ and } \hat{X}(\lambda) = \hat{\gamma} + \lambda \hat{\alpha} + \beta,$$

where

- $(\gamma^N, \alpha^N) = \{(\gamma^N_i, \alpha^N_i) : i \in \mathbb{N}\}$  is a sequence of 2-dimensional i.i.d. random variables with distribution identical to  $\mathcal{L}((\hat{\gamma}, \hat{\alpha}))$  with  $(\hat{\gamma}, \hat{\alpha}) \in L^q$  for some q > 4,
- $\beta \in L^q$  is independent to the random variables  $(\gamma_i^N, \alpha_i^N, \hat{\gamma}, \hat{\alpha})$ ,

• 
$$\max_{i=1,2,\ldots,N} \mathbb{E}\left[\sup_{\lambda \in [0,1]} |\Delta_i^N(\lambda)|^2\right] = O(N^{-1}).$$

Let  $\rho^N(X^N)$  be the empirical measure given by

$$\rho^N(X^N) = \frac{1}{N} \sum_{i=1}^N \delta_{X_i^N}.$$

Then, we have the following three results: For  $p \in [1, 2]$ ,

$$\mathbb{W}_p\left(\mathcal{L}\left(X_1^N(\lambda)\right), \mathcal{L}\left(\hat{X}(\lambda)\right)\right) = O\left(N^{-\frac{1}{2}}\right),\tag{4.3.7}$$

$$\sup_{\lambda \in [0,1]} \mathbb{E}\left[ \mathbb{W}_p^p\left(\rho^N\left(X^N(\lambda)\right), \mathcal{L}\left(\hat{X}(\lambda)\middle|\,\beta\right) \right) \right] = O\left(N^{-\frac{1}{2}}\right), \tag{4.3.8}$$

and

$$\mathbb{E}\left[\sup_{\lambda\in[0,1]}\mathbb{W}_{p}^{p}\left(\rho^{N}\left(X^{N}(\lambda)\right),\mathcal{L}\left(\hat{X}(\lambda)\middle|\,\beta\right)\right)\right] = \begin{cases} O\left(N^{-\frac{1}{2}}\ln(N)\right), & \text{if } p = 1, \\ O\left(N^{-\frac{1}{2}}\right), & \text{if } p > 1. \end{cases}$$
(4.3.9)

*Proof.* We will omit the dependence of  $\lambda$  if there is no confusion, for instance, we use X in lieu of  $X(\lambda)$ . Since  $\mathcal{L}(\hat{X}) = \mathcal{L}(X_1^N - \Delta_1^N)$ , the first result (4.3.7) directly follows from

$$\mathbb{W}_{p}^{p}\left(\mathcal{L}\left(X_{1}^{N}\right),\mathcal{L}\left(\hat{X}\right)\right) \leq \mathbb{E}\left[\left|\Delta_{1}^{N}\right|^{p}\right] \leq \left(\mathbb{E}\left[\left|\Delta_{1}^{N}\right|^{2}\right]\right)^{\frac{p}{2}} = O\left(N^{-\frac{p}{2}}\right)$$

Next, we set  $Y_i^N(\lambda) = \gamma_i^N + \lambda \alpha_i^N + \beta$ . By the definition of empirical measures, we have

$$\mathbb{W}_{p}^{p}\left(\rho^{N}\left(X^{N}\right),\rho^{N}\left(Y^{N}\right)\right) \leq \frac{1}{N}\sum_{i=1}^{N}\left|X_{i}^{N}-Y_{i}^{N}\right|^{p} = \frac{1}{N}\sum_{i=1}^{N}\left|\Delta_{i}^{N}\right|^{p}.$$
(4.3.10)

From the third condition on  $\Delta_i^N$ , we obtain

$$\mathbb{E}\left[\mathbb{W}_{p}^{p}\left(\rho^{N}\left(X^{N}\right),\rho^{N}\left(Y^{N}\right)\right)\right]=O\left(N^{-\frac{p}{2}}\right).$$

By Lemma 51, we also have

$$\mathbb{E}\left[\mathbb{W}_{p}^{p}\left(\rho^{N}\left(Y^{N}\right),\mathcal{L}\left(\hat{X}\middle|\beta\right)\right)\right]=O\left(N^{-\frac{1}{2}}\right).$$

In the end, (4.3.8) follows from the triangle inequality together with the fact that  $p \ge 1$ . Finally, for the proof of (4.3.9), we first use (4.3.10) to write

$$\mathbb{W}_{p}^{p}\left(\rho^{N}\left(X^{N}(\lambda)\right), \mathcal{L}\left(\hat{X}(\lambda)\middle|\,\beta\right)\right) \leq 2^{p-1}\left(\mathbb{W}_{p}^{p}\left(\rho^{N}\left(Y^{N}(\lambda)\right), \mathcal{L}\left(\hat{X}(\lambda)\middle|\,\beta\right)\right) + \frac{1}{N}\sum_{i=1}^{N}\left|\Delta_{i}^{N}(\lambda)\right|^{p}\right).$$
Applying Lemma 52 and the third condition on  $\Delta_i^N(\lambda)$ , we can conclude (4.3.9).

## 4.3.4 Proof of Theorem 44

For simplicity, let us introduce the following notations:

$$\mathcal{E}_t(a) = \exp\left\{\int_0^t a(s)ds\right\}, \quad \mathcal{E}_t(a,M) = \int_0^t \mathcal{E}_s(a)dM_s$$

for a deterministic function  $a(\cdot)$  and a martingale  $M = \{M_t\}_{t\geq 0}$ . With these notations, one can write the solution to the Ornstein–Uhlenbeck process

$$dX_t = -a_t X_t dt + dM_t$$

for a determinant function a in the form of

$$\mathcal{E}_t(a)X_t = X_0 + \mathcal{E}_t(a, M). \tag{4.3.11}$$

For MFG equilibrium, we define

$$\tilde{X}_t = \hat{X}_t - \hat{\mu}_t.$$

According to (4.3.1) in Proposition 45,  $\tilde{X}$  satisfies the following equation:

$$\tilde{X}_t = \tilde{X}_0 - \int_0^t 2a_s \tilde{X}_s ds + W_t.$$

Next, we express the solution of the above SDE in the form of

$$\tilde{Y}_t := \mathcal{E}_t(2a)\tilde{X}_t = \tilde{X}_0 + \mathcal{E}_t(2a, W).$$

Note that  $\tilde{Y}$  and  $\hat{\mu}$  are independent. Therefore,  $\hat{X}$  admits a decomposition of two independent processes as

$$\hat{X}_t = \tilde{X}_t + \hat{\mu}_t.$$

Furthermore, we have

$$\hat{Y}_t := \mathcal{E}_t(2a)\hat{X}_t = \tilde{X}_0 + \mathcal{E}_t(2a, W) + \mathcal{E}_t(2a)\left(\hat{\mu}_0 + \tilde{W}_t\right)$$

In the N-player game, we define the following quantities:

$$\bar{X}_t^{(N)} = \frac{1}{N} \sum_{i=1}^N \hat{X}_{it}^{(N)}, \quad \bar{W}_t^{(N)} = \frac{1}{N} \sum_{i=1}^N W_{it}^{(N)},$$

and

$$\tilde{X}_{it}^{(N)} = \hat{X}_{it}^{(N)} - \bar{X}_t^{(N)}.$$

It is worth noting that, by Proposition 46, we have

$$\hat{X}_{it}^{(N)} = \hat{X}_{i0}^{(N)} - \int_0^t 2\frac{N}{N-1} a^N(s) \left(\hat{X}_{is}^{(N)} - \frac{1}{N} \sum_{j=1}^N \hat{X}_{js}^{(N)}\right) ds + W_{it}^{(N)} + \tilde{W}_t$$

for all i = 1, 2, ..., N, then the mean-field term satisfies

$$\bar{X}_t^{(N)} = \bar{X}_0^{(N)} + \bar{W}_t^{(N)} + \tilde{W}_t$$

and the *i*-th player's path deviated from the mean-field path can be rewritten by

$$\tilde{X}_{it}^{(N)} = \tilde{X}_{i0}^{(N)} - \int_0^t 2\hat{a}^N(s)\tilde{X}_{is}^{(N)}ds + W_{it}^{(N)} - \bar{W}_t^{(N)},$$

where

$$\hat{a}^N = \frac{N}{N-1} a^N.$$

Next, we introduce

$$\hat{Y}_{it}^{(N)} = \mathcal{E}_t \left( 2\hat{a}^N \right) \hat{X}_{it}^{(N)}, \quad \tilde{Y}_{it}^{(N)} = \mathcal{E}_t \left( 2\hat{a}^N \right) \tilde{X}_{it}^{(N)}, \quad \bar{Y}_t^{(N)} = \mathcal{E}_t \left( 2\hat{a}^N \right) \bar{X}_t^{(N)}.$$

Consequently, we obtain the following relationships:

$$\tilde{Y}_{it}^{(N)} = \tilde{X}_{i0}^{(N)} + \mathcal{E}_t \left( 2\hat{a}^N, W_i^{(N)} - \bar{W}^{(N)} \right),$$
$$\bar{Y}_t^{(N)} = \mathcal{E}_t \left( 2\hat{a}^N \right) \left( \bar{W}_t^{(N)} + \tilde{W}_t + \bar{X}_0^{(N)} \right),$$
$$\hat{Y}_{it}^{(N)} = \bar{Y}_{it}^{(N)} + \tilde{Y}_{it}^{(N)}.$$

and

$$\mathbf{r}_{it} = \mathbf{r}_{it} + \mathbf{r}_{it}$$

To compare the process  $\hat{Y}_{it}^{(N)}$  with the target process

$$\hat{Y}_t = \tilde{X}_0 + \mathcal{E}_t (2a, W) + \mathcal{E}_t (2a) \left( \hat{\mu}_0 + \tilde{W}_t \right) 
= \tilde{X}_0 + \lambda_t Z_t + \mathcal{E}_t (2a) \left( \hat{\mu}_0 + \tilde{W}_t \right),$$
(4.3.12)

where

$$\lambda_t = \left(\int_0^t \mathcal{E}_s(4a)ds\right)^{1/2},$$

and

$$Z_t = \lambda_t^{-1} \mathcal{E}_t \left( 2a, W \right) \sim \mathcal{N}(0, 1),$$

we write  $\hat{Y}_{it}^{(N)}$  by

$$\hat{Y}_{it}^{(N)} = \tilde{X}_{i0}^{(N)} + \mathcal{E}_t \left( 2a, W_i^{(N)} \right) + \Delta_{it}^{(N)} + \mathcal{E}_t (2a) \left( \hat{\mu}_0 + \tilde{W}_t \right) 
= \tilde{X}_{i0}^{(N)} + \lambda_t Z_{it}^{(N)} + \Delta_{it}^{(N)} + \mathcal{E}_t (2a) \left( \hat{\mu}_0 + \tilde{W}_t \right),$$
(4.3.13)

where

$$Z_{it}^{(N)} = \lambda_t^{-1} \mathcal{E}_t \left( 2a, W_i^{(N)} \right) \sim \mathcal{N}(0, 1),$$

and

$$\Delta_{it}^{(N)} = \left( \mathcal{E}_t \left( 2\hat{a}^N, W_i^{(N)} \right) - \mathcal{E}_t \left( 2a, W_i^{(N)} \right) \right) - \mathcal{E}_t \left( 2\hat{a}^N, \bar{W}^{(N)} \right) + \left( \mathcal{E}_t \left( 2\hat{a}^N \right) - \mathcal{E}_t(2a) \right) \left( \hat{\mu}_0 + \tilde{W}_t \right) + \mathcal{E}_t \left( 2\hat{a}^N \right) \left( \bar{X}_0^{(N)} - \hat{\mu}_0 + \bar{W}_t^{(N)} \right) := I_{it}^{(N)} + II_t^{(N)} + III_t^{(N)} + IV_t^{(N)}.$$
(4.3.14)

To apply Lemma 53 to the processes of (4.3.13) and (4.3.12), we only need to show the second moment on  $\sup_{t \in [0,T]} \Delta_{it}^{(N)}$  is  $O(N^{-1})$  for each i = 1, 2, ..., N. In the following analysis, we will utilize the explicit solution of the ODE:

• Let c, d > 0 be two constants. The solution of

$$v'(t) - c^2 v^2(t) + d^2 = 0, \quad v(T) = 0$$

is

$$v(t) = \frac{d}{c} \cdot \frac{1 - e^{2dc(t-T)}}{1 + e^{2dc(t-T)}}.$$
(4.3.15)

We will employ this solution to derive the second-moment estimations of  $\sup_{t \in [0,T]} \Delta_{it}^{(N)}$ .

1. From (4.3.15), we have an estimation of

$$\left|a^{N}(t) - a(t)\right| = \frac{k|T - t|}{N} + o\left(N^{-1}\right).$$
(4.3.16)

Therefore, we have

$$\left|\mathcal{E}_{t}(2\hat{a}^{N}) - \mathcal{E}_{t}(2a)\right| = \frac{2t(T-t)}{N} + o\left(N^{-1}\right)$$
(4.3.17)

and thus by Burkholder-Davis-Gundy (BDG) inequality

$$\mathbb{E}\left[\sup_{t\in[0,T]} \left(I_{it}^{(N)}\right)^2\right] = \mathbb{E}\left[\sup_{t\in[0,T]} \left(\int_0^t \left(\mathcal{E}_s\left(2\hat{a}^N\right) - \mathcal{E}_s\left(2a\right)\right) dW_{is}^{(N)}\right)^2\right]$$
$$\leq C\mathbb{E}\left[\left(\int_0^T \left(\mathcal{E}_s\left(2\hat{a}^N\right) - \mathcal{E}_s\left(2a\right)\right) dW_{is}^{(N)}\right)^2\right] \text{ for some constant } C > 0$$
$$= C\int_0^T \left(\mathcal{E}_s\left(2\hat{a}^N\right) - \mathcal{E}_s\left(2a\right)\right)^2 ds$$
$$= O\left(N^{-2}\right).$$

2. Since  $\hat{a}^N$  is uniformly bounded by  $\sqrt{k/2}$ ,  $II_t^{(N)}$  is a martingale with its quadratic variance

$$[II^{(N)}]_T = \frac{1}{N} \int_0^T \mathcal{E}_s(4\hat{a}^N) ds = O\left(N^{-1}\right).$$

So, we have

$$\mathbb{E}\left[\sup_{t\in[0,T]}\left(II_t^{(N)}\right)^2\right] = O\left(N^{-1}\right).$$

3. From the estimation (4.3.17), we also have

$$\mathbb{E}\left[\sup_{t\in[0,T]}\left(III_{t}^{(N)}\right)^{2}\right]=O\left(N^{-2}\right).$$

4. By the assumption of i.i.d. initial states, we have

$$\mathbb{E}\left[\sup_{t\in[0,T]}\left(IV_t^{(N)}\right)^2\right] = \mathcal{E}_T\left(4\hat{a}^N\right)\left(Var\left(\bar{X}_0^{(N)}\right) + \mathbb{E}\left[\sup_{t\in[0,T]}\left(\bar{W}_t^{(N)}\right)^2\right]\right) = O\left(N^{-1}\right).$$

As a result, we have the following expression:

$$\mathbb{E}\left[\sup_{t\in[0,T]} \left(\Delta_{it}^{(N)}\right)^2\right] = O\left(N^{-1}\right), \quad \forall i = 1, 2, \dots, N.$$
(4.3.18)

By combining equations (4.3.12), (4.3.13), and (4.3.18), we can conclude Theorem 44 by applying Lemma 53.

## 4.4 Proposition 45: Derivation of the MFG path

This section is dedicated to proving Proposition 45, which provides insights into the MFG solution. To proceed, in Subsection 4.4.1, we begin by reformulating the MFG problem, assuming a Markovian structure for the equilibrium. Then, in Subsection 4.4.2, we solve the underlying control

problem and derive the corresponding Riccati system. Finally, in Subsection 4.4.3, we examine the fixed-point condition of the MFG problem, leading to the conclusion.

## 4.4.1 Reformulation

To determine the equilibrium measure, as defined in Definition 41, one needs to explore the infinitedimensional space of random measure flows  $m: (0,T] \times \Omega \to \mathcal{P}_2(\mathbb{R})$  until a measure flow satisfies the fixed-point condition  $m_t = \mathcal{L}(\hat{X}_t | \tilde{\mathcal{F}}_t)$  for all  $t \in (0,T]$ , as illustrated in Figure 4.1.

The first observation is that the cost function F in (4.2.8) is only dependent on the measure m through the first two moments with the quadratic cost structure, which is given by

$$F(x,m) = k(x^2 - 2x[m]_1 + [m]_2).$$

Consequently, the underlying stochastic control problem for MFG can be entirely determined by the input given by the  $\mathbb{R}^2$  valued random processes  $\mu_t = [m_t]_1$  and  $\nu_t = [m_t]_2$ , which implies that the fixed point condition can be effectively reduced to merely checking two conditions:

$$\mu_t = \mathbb{E}\left[\left.\hat{X}_t\right| \tilde{\mathcal{F}}_t\right], \ \nu_t = \mathbb{E}\left[\left.\hat{X}_t^2\right| \tilde{\mathcal{F}}_t\right].$$

This observation effectively reduces our search from the space of random measure-valued processes  $m: (0,T] \times \Omega \mapsto \mathcal{P}_2(\mathbb{R})$  to the space of  $\mathbb{R}^2$ -valued random processes  $(\mu,\nu): (0,T] \times \Omega \mapsto \mathbb{R}^2$ .

It is important to note that if the underlying MFG does not involve common noise, the aforementioned observation is adequate to transform the original infinite-dimensional MFG into a finitedimensional system. In this case, the moment processes  $(\mu, \nu)$  become deterministic mappings  $[0,T] \rightarrow \mathbb{R}^2$ . However, the following example demonstrates that this is not applicable to MFG with common noise, which presents a significant drawback in characterizing LQG-MFG using a finite-dimensional system.

**Example 3.** To illustrate this point, let's consider the following uncontrolled mean field dynamics: Let the mean field term  $\mu_t := \mathbb{E}[\hat{X}_t | \tilde{\mathcal{F}}_t]$ , where the underlying dynamic is given by

$$d\hat{X}_t = -\mu_t \tilde{W}_t dt + dW_t + d\tilde{W}_t, \quad \hat{X}_0 = X_0.$$

Here are two key observations:

•  $\mu_t$  is path dependent on entire path of  $\tilde{W}$ , i.e.,

$$\mu_t = \mu_0 e^{-\int_0^t \tilde{W}_s ds} + e^{-\int_0^t \tilde{W}_s ds} \int_0^t e^{\int_0^s \tilde{W}_r dr} d\tilde{W}_s.$$

This implies that the  $(t, \tilde{W}) \mapsto \mu_t$  is a function on an infinite dimensional domain.

•  $\mu_t$  is Markovian, i.e.,

$$d\mu_t = -\mu_t \tilde{W}_t dt + d\tilde{W}_t$$

It is possible to express the  $\mu_t$  via a SDE with finite-dimensional coefficient functions of  $(t, \mu_t)$ .

To make the previous idea more concrete, we propose the assumption of a Markovian structure for the first and second moments of the MFG equilibrium. In other words, we restrict our search for equilibrium to a smaller space  $\mathcal{M}$  of measure flows that capture the Markovian structure of the first and second moments.

**Definition 54.** The space  $\mathcal{M}$  is the collection of all  $\tilde{\mathcal{F}}_t$ -adapted measure flows  $m : [0, T] \times \Omega \mapsto \mathcal{P}_2(\mathbb{R})$ , whose first moment  $[m_t]_1 := \mu_t$  and second moment  $[m_t]_2 := \nu_t$  satisfy a system of SDE

$$\mu_t = \mu_0 + \int_0^t \left( w_1(s)\mu_s + w_2(s) \right) ds + \tilde{W}_t,$$

$$\nu_t = \nu_0 + \int_0^t \left( w_3(s)\mu_s + w_4(s)\nu_s + w_5(s)\mu_s^2 + w_6(s) \right) ds + 2\int_0^t \mu_s d\tilde{W}_s,$$
(4.4.1)

for some smooth deterministic functions  $(w_i : i = 1, 2, ..., 6)$  for all  $t \in [0, T]$ .



Figure 4.2: Equivalent MFG diagram 2.

The MFG problem originally given by Definition 40 can be recast as the following combination of stochastic control problem and fixed point condition:

• RLQG(Revised LQG):

Given smooth functions  $w = (w_i : i = 1, 2, ..., 6)$ , we want to find the value function  $\overline{V} = \overline{V}[w] : [0,T] \times \mathbb{R}^3 \to \mathbb{R}$  and optimal path  $(\hat{X}, \hat{\mu}, \hat{\nu})[w]$  from the following control problem:

$$\bar{V}(t,x,\bar{\mu},\bar{\nu}) = \inf_{\alpha \in \mathcal{A}} \mathbb{E}\left[\int_t^T \left(\frac{1}{2}\alpha_s^2 + \bar{F}(X_s,\mu_s,\nu_s)\right) ds \,\middle| \, X_t = x, \mu_t = \bar{\mu}, \nu_t = \bar{\nu}\right]$$

with the underlying process X of (4.2.1) and  $(\mu, \nu)$  of (4.4.1) and with the cost functions:  $\overline{F} : \mathbb{R}^3 \mapsto \mathbb{R}$  given by

$$\bar{F}(x,\bar{\mu},\bar{\nu}) = k(x^2 - 2x\bar{\mu} + \bar{\nu}),$$

where  $\bar{\mu}, \bar{\nu}$  are scalars, while  $\mu, \nu$  are used as processes.

• RFP(Revised fixed point condition):

Determine w satisfying the following fixed point condition:

$$\hat{\mu}_s = \mathbb{E}\left[\left.\hat{X}_s\right|\tilde{\mathcal{F}}_s\right] \text{ and } \hat{\nu}_s = \mathbb{E}\left[\left.\hat{X}_s^2\right|\tilde{\mathcal{F}}_s\right], \quad \forall s \in [0,T].$$
(4.4.2)

The equilibrium measure is then  $\mathcal{N}(\hat{\mu}_t, \hat{\nu}_t - \hat{\mu}_t^2)$ .

**Remark 55.** It is important to highlight that the Markovian structure for the first and second moments of the MFG equilibrium in this manuscript differs significantly from that presented in [50]. In [50], the processes  $\mu_t$  and  $\nu_t$  are pairs of processes with finite variation, while in our case, they are quadratic variation processes.

Specifically, in [50], the coefficient functions depend on the common noise Y, whereas in (4.4.1), the coefficient functions ( $w_i : i = 1, 2, ..., 6$ ) are independent of the common noise  $\tilde{W}$ . Instead, the first and second moments of the MFG equilibrium are only influenced by the common noise through an additive term.

## 4.4.2 The generic player's control with a given population measure

This section is devoted to the control problem RLQG parameterized by w.

### 4.4.2.1 HJB equation

To simplify the notation, let's denote each function  $w_i(t)$  as  $w_i$  for  $i \in \{1, 2, ..., 6\}$ . Assuming sufficient regularity conditions, and according to the dynamic programming principle (refer to [69] for more details), the value function  $\bar{V}$  defined in the RLQG problem can be obtained as a solution v of the following Hamilton-Jacobi-Bellman (HJB) equation

$$\begin{cases} \partial_{t}v + \inf_{a \in \mathbb{R}} \left( a \partial_{x}v + \frac{1}{2}a^{2} \right) + \left( w_{1}\bar{\mu} + w_{2} \right) \partial_{\bar{\mu}}v + \left( w_{3}\bar{\mu} + w_{4}\bar{\nu} + w_{5}\bar{\mu}^{2} + w_{6} \right) \partial_{\bar{\nu}}v + \partial_{xx}v + \frac{1}{2}\partial_{\bar{\mu}\bar{\mu}}v \\ + \partial_{x\bar{\mu}}v + 2\bar{\mu}^{2}\partial_{\bar{\nu}\bar{\nu}}v + 2\bar{\mu}\partial_{\bar{\mu}\bar{\nu}}v + 2\bar{\mu}\partial_{x\bar{\nu}}v + k(x^{2} - 2\bar{\mu}x + \bar{\nu}) = 0, \\ v(T, x, \mu_{T}, \nu_{T}) = 0. \end{cases}$$

Therefore, the optimal control has to admit the feedback form of

$$\hat{\alpha}(t) = -\partial_x v\left(t, \hat{X}_t, \mu_t, \nu_t\right), \qquad (4.4.3)$$

and then the HJB equation can be reduced to

$$\begin{cases} \partial_t v - \frac{1}{2} (\partial_x v)^2 + (w_1 \bar{\mu} + w_2) \, \partial_{\bar{\mu}} v + (w_3 \bar{\mu} + w_4 \bar{\nu} + w_5 \bar{\mu}^2 + w_6) \, \partial_{\bar{\nu}} v + \partial_{xx} v + \frac{1}{2} \partial_{\bar{\mu}\bar{\mu}} v \\ + \partial_{x\bar{\mu}} v + 2 \bar{\mu}^2 \partial_{\bar{\nu}\bar{\nu}} v + 2 \bar{\mu} \partial_{\bar{\mu}\bar{\nu}} v + 2 \bar{\mu} \partial_{x\bar{\nu}} v + k (x^2 - 2 \bar{\mu} x + \bar{\nu}) = 0, \end{cases}$$

$$(4.4.4)$$

$$v(T, x, \mu_T, \nu_T) = 0.$$

Next, we identify what conditions are needed for equating the control problem RLQG and the above HJB equation. Denote S to be the set of v such that  $v \in C^{\infty}$  satisfies

$$(1+|x|^2)^{-1} (|v|+|\partial_t v|) + (1+|x|+|\mu|)^{-1} (|\partial_x v|+|\partial_\mu v|) + (|\partial_{xx}v|+|\partial_x \mu v|+|\partial_{\mu\mu}v|+|\partial_\nu v|) < K$$

for all  $(t, x, \mu, \nu)$  for some positive constant K.

**Lemma 56.** Consider the control problem RLQG with some given smooth functions  $w = (w_i : i = 1, 2, ..., 6)$ .

- 1. (Verification theorem) Suppose there exists a solution  $v \in S$  of (4.4.4). Then  $v(t, x, \bar{\mu}, \bar{\nu}) = \bar{V}(t, x, \bar{\mu}, \bar{\nu})$ , and an optimal control is provided by (4.4.3).
- 2. Suppose that the value function  $\overline{V}$  belongs to S, and then  $\overline{V}(t, x, \overline{\mu}, \overline{\nu})$  solves HJB equation (4.4.4). Moreover,  $\hat{\alpha}$  of (4.4.3) is an optimal control.
- *Proof.* 1. First, we prove the verification theorem. Since  $v \in S$ , for any admissible  $\alpha \in \mathcal{H}^4_{\mathbb{F}}$ , the process  $X^{\alpha}$  is well defined and one can apply Itô's formula to obtain

$$\mathbb{E}\left[v(T, X_T, \mu_T, \nu_T)\right] = v(t, x, \bar{\mu}, \bar{\nu}) + \mathbb{E}\left[\int_t^T \mathcal{G}^{\alpha(s)} v(s, X_s, \mu_s, \nu_s) ds\right],$$

where

$$\begin{aligned} \mathcal{G}^a f(s, x, \bar{\mu}, \bar{\nu}) &= \left(\partial_t + a\partial_x + \partial_{xx} + (w_1\bar{\mu} + w_2)\partial_{\bar{\mu}} + \left(w_3\bar{\mu} + w_4\bar{\nu} + w_5\bar{\mu}^2 + w_6\right)\partial_{\bar{\nu}} \right. \\ &+ \frac{1}{2}\partial_{\bar{\mu}\bar{\mu}} + 2\bar{\mu}^2\partial_{\bar{\nu}\bar{\nu}} + \partial_{x\bar{\mu}} + 2\bar{\mu}\partial_{\bar{\mu}\bar{\nu}} + 2\bar{\mu}\partial_{x\bar{\nu}}\right) f(s, x, \bar{\mu}, \bar{\nu}). \end{aligned}$$

Note that the HJB equation actually implies that

$$\inf_{a} \left\{ \mathcal{G}^{a}v + \frac{1}{2}a^{2} \right\} = -\bar{F},$$

which again yields

$$-\mathcal{G}^a v \le \frac{1}{2}a^2 + \bar{F}.$$

Hence, we obtain that for all  $\alpha \in \mathcal{H}^4_{\mathbb{F}}$ ,

$$v(t, x, \bar{\mu}, \bar{\nu})$$

$$= \mathbb{E}\left[\int_{t}^{T} -\mathcal{G}^{\alpha(s)}v(s, X_{s}, \mu_{s}, \nu_{s})ds\right] + \mathbb{E}\left[v(T, X_{T}, \mu_{T}, \nu_{T})\right]$$

$$\leq \mathbb{E}\left[\int_{t}^{T} \left(\frac{1}{2}\alpha^{2}(s) + \bar{F}(X_{s}, \mu_{s}, \nu_{s})\right)ds\right]$$

$$= J(t, x, \alpha, \bar{\mu}, \bar{\nu}).$$

In the above, if  $\alpha$  is replaced by  $\hat{\alpha}$  given by the feedback form (4.4.3), then since  $\partial_x v$  is Lipschitz continuous in x, there exists corresponding optimal path  $\hat{X} \in \mathcal{H}^4_{\mathbb{F}}$ . Thus,  $\hat{\alpha}$  is also in  $\mathcal{H}^4_{\mathbb{F}}$ . One can repeat all the above steps by replacing X and  $\alpha$  by  $\hat{X}$  and  $\hat{\alpha}$ , and  $\leq$  sign by = sign to conclude that v is indeed the optimal value.

2. The opposite direction of the verification theorem follows by taking  $\theta \to t$  for the dynamic programming principle, for all stopping time  $\theta \in [t, T]$ ,

$$V(t, x, \bar{\mu}, \bar{\nu})$$

$$= \mathbb{E}\left[\int_{t}^{\theta} \left(\frac{1}{2}\alpha_{s}^{2} + \bar{F}(X_{s}, \mu_{s}, \nu_{s})\right) ds + \bar{V}(\theta, X_{\theta}, \mu_{\theta}, \nu_{\theta}) \middle| X_{t} = x, \mu_{t} = \bar{\mu}, \nu_{t} = \bar{\nu}\right],$$

which is valid under our regularity assumptions on all the partial derivatives.

## 4.4.2.2 LQG solution

It is worth noting that the costs  $\bar{F}$  of RLQG are quadratic functions in  $(x, \bar{\mu}, \bar{\nu})$ , while the drift function of the process  $\nu$  of (4.4.1) is not linear in  $(x, \bar{\mu}, \bar{\nu})$ . Therefore, the stochastic control problem RLQG does not fit into the typical LQG control structure. Nevertheless, similarly to the LQG solution, we guess the value function to be a quadratic function in the form of

$$v(t, x, \bar{\mu}, \bar{\nu}) = a(t)x^2 + b(t)\bar{\mu}^2 + c(t)\bar{\nu} + d(t) + e(t)x + f(t)\bar{\mu} + g(t)x\bar{\mu}.$$
(4.4.5)

Under the above setup for the value function v, for  $t \in [0, T]$ , the optimal control is given by

$$\hat{\alpha}_t = -\partial_x v(t, \hat{X}_t, \mu_t, \nu_t) = -2a(t)\hat{X}_t - e(t) - g(t)\mu_t, \qquad (4.4.6)$$

and the optimal path  $\hat{X}$  is

$$\begin{cases} d\hat{X}_t = \left(-2a(t)\hat{X}_t - e(t) - g(t)\mu_t\right)dt + dW_t + d\tilde{W}_t, \\ \hat{X}_0 = X_0. \end{cases}$$
(4.4.7)

To proceed, we introduce the following Riccati system of ODEs for  $t \in [0, T]$ ,

$$\begin{cases} a' - 2a^2 + k = 0, \\ b' - \frac{1}{2}g^2 + 2bw_1 + cw_5 = 0, \\ c' + cw_4 + k = 0, \\ d' - \frac{1}{2}e^2 + fw_2 + cw_6 + 2a + b + g = 0, \\ e' - 2ae + w_2g = 0, \\ f' - eg + w_1f + 2bw_2 + cw_3 = 0, \\ g' - 2ag + w_1g - 2k = 0, \end{cases}$$
(4.4.8)

with terminal conditions

$$a(T) = b(T) = c(T) = d(T) = e(T) = f(T) = g(T) = 0.$$
(4.4.9)

**Lemma 57.** Suppose there exists a unique solution (a, b, c, d, e, f, g) to the Riccati system of ODEs (4.4.8)-(4.4.9) on [0,T]. Then the value function of (RMFG) is given by

$$\bar{V}(t, x, \bar{\mu}, \bar{\nu}) = v(t, x, \bar{\mu}, \bar{\nu})$$
  
=  $a(t)x^2 + b(t)\bar{\mu}^2 + c(t)\bar{\nu} + d(t) + e(t)x + f(t)\bar{\mu} + g(t)x\bar{\mu}$  (4.4.10)

for  $t \in [0, T]$  and the optimal control and optimal path are given by (4.4.6) and (4.4.7), respectively. *Proof.* With the form of value function v given in (4.4.5) and the conditional first and second moment of  $\hat{X}_t$  under the  $\sigma$ -algebra  $\tilde{\mathcal{F}}_t$  given in (4.4.1), we have

$$\begin{aligned} \partial_t v &= a'(t)x^2 + e'(t)x + b'(t)\bar{\mu}^2 + f'(t)\bar{\mu} + g'(t)x\bar{\mu} + c'(t)\bar{\nu} + d'(t), \\ \partial_x v &= 2xa(t) + e(t) + g(t)\bar{\mu}, \\ \partial_{xx}v &= 2a(t), \\ \partial_{\bar{\mu}\bar{\nu}}v &= 2b(t)\bar{\mu} + f(t) + g(t)x, \\ \partial_{\bar{\nu}}v &= c(t), \\ \partial_{\bar{\mu}\bar{\mu}}v &= 2b(t), \\ \partial_{x\bar{\mu}}v &= g(t), \\ \partial_{\bar{\mu}\bar{\nu}}v &= \partial_{\bar{\nu}\bar{\nu}}v = \partial_{x\bar{\nu}}v = 0. \end{aligned}$$

Plugging them back to the HJB equation in (4.4.4), we get a system of ODEs in (4.4.8) by equating  $x, \bar{\mu}, \bar{\nu}$ -like terms in each equation with the terminal conditions given in (4.4.9).

Therefore, any solution (a, b, c, d, e, f, g) of a system of ODEs (4.4.8) leads to the solution of HJB (4.4.4) in the form of the quadratic function given by (4.4.10). Since the (a, b, c, d, e, f, g) are differentiable functions on the closed set [0, T], they are also bounded, and thus the regularity conditions needed for  $v \in S$  is valid. Finally, we invoke the verification theorem given by Lemma 56 to conclude the desired result.

#### 4.4.3 Fixed point condition and the proof of Proposition 45

Returning to the ODE system (4.4.8), there are 7 equations, whereas we need to determine a total of 13 deterministic functions of  $[0, T] \times \mathbb{R}$  to characterize MFG. These are

$$(a, b, c, d, e, f, g)$$
 and  $(w_i : i = 1, 2, \dots, 6).$ 

In this below, we identify the missing 6 equations by checking the fixed point condition of RFP. This leads to a complete characterization of the equilibrium for MFG in Definition 40.

**Lemma 58.** With the dynamic of the optimal path  $\hat{X}$  defined in (4.4.7), the fixed point condition (4.4.2) implies that the first moment  $\hat{\mu}_s := \mathbb{E}[\hat{X}_s|\tilde{\mathcal{F}}_s]$  and the second moment  $\hat{\nu}_s := \mathbb{E}[\hat{X}_s^2|\tilde{\mathcal{F}}_s]$  of the

optimal path conditioned on  $\tilde{\mathcal{F}}_t$  satisfy

$$\begin{cases} \hat{\mu}_s = \bar{\mu} + \int_t^s \left( \left( -2a(r) - g(r) \right) \hat{\mu}_r - e(r) \right) dr + \tilde{W}_s, \\ \hat{\nu}_s = \bar{\nu} + \int_t^s \left( 2 - 4a(r)\hat{\nu}_r - 2e(r)\hat{\mu}_r - 2g(r)\hat{\mu}_r^2 \right) dr + \int_t^s 2\hat{\mu}_r d\tilde{W}_r, \end{cases}$$
(4.4.11)

for  $s \ge t$ , and thus the coefficient functions  $w = (w_i : i = 1, 2, ..., 6)$  in (4.4.1) satisfy the following equations:

$$w_1 = -2a - g, \ w_2 = -e, \ w_3 = -2e, \ w_4 = -4a, \ w_5 = -2g, \ w_6 = 2, \ \forall t \in [0, T].$$
 (4.4.12)

*Proof.* With the dynamic of the optimal path  $\hat{X}$  given by (4.4.7), we have

$$\hat{X}_t = X_0 + \int_0^t \left( -2a(s)\hat{X}_s - e(s) - g(s)\hat{\mu}_s \right) ds + W_t + \tilde{W}_t,$$

and since the functions a, e, g are continuous on [0, T], then we can change of order of integration and expectation and it yields

$$\begin{aligned} \hat{\mu}_t &= \mathbb{E}\left[\left.\hat{X}_t\right|\left.\tilde{\mathcal{F}}_t\right] \\ &= \mathbb{E}\left[\left.X_0\right|\left.\tilde{\mathcal{F}}_t\right] + \int_0^t \left(-2a(s)\hat{\mu}_s - e(s) - g(s)\hat{\mu}_s\right)ds + \mathbb{E}\left[\left.W_t + \tilde{W}_t\right|\left.\tilde{\mathcal{F}}_t\right] \right] \\ &= \mathbb{E}\left[\left.X_0\right|\left.\tilde{\mathcal{F}}_t\right] + \int_0^t \left(-2a(s)\hat{\mu}_s - e(s) - g(s)\hat{\mu}_s\right)ds + \tilde{W}_t. \end{aligned}$$

Similarly, applying Itô's formula, we obtain

$$\hat{X}_{t}^{2} = X_{0}^{2} + \int_{0}^{t} \left( 2 - 4a(s)\hat{X}_{s}^{2} - 2e(s)\hat{X}_{s} - 2g(s)\hat{\mu}_{s}\hat{X}_{s} \right) ds + \int_{0}^{t} 2\hat{X}_{s}dW_{s} + \int_{0}^{t} 2\hat{X}_{s}d\tilde{W}_{s},$$

and it follows that

$$\begin{split} \hat{\nu}_{t} &= \mathbb{E}\left[\left.\hat{X}_{t}^{2}\right|\tilde{\mathcal{F}}_{t}\right] \\ &= \mathbb{E}\left[\left.X_{0}^{2}\right|\tilde{\mathcal{F}}_{t}\right] + \int_{0}^{t}\left(2 - 4a(s)\hat{\nu}_{s} - 2e(s)\hat{\mu}_{s} - 2g(s)\hat{\mu}_{s}^{2}\right)ds + \mathbb{E}\left[\left.\int_{0}^{t}2\hat{X}_{s}dW_{s} + \int_{0}^{t}2\hat{X}_{s}d\tilde{W}_{s}\right|\tilde{\mathcal{F}}_{t}\right] \\ &= \mathbb{E}\left[\left.X_{0}^{2}\right|\tilde{\mathcal{F}}_{t}\right] + \int_{0}^{t}\left(2 - 4a(s)\hat{\nu}_{s} - 2e(s)\hat{\mu}_{s} - 2g(s)\hat{\mu}_{s}^{2}\right)ds + \int_{0}^{t}2\hat{\mu}_{s}d\tilde{W}_{s}. \end{split}$$

Thus the desired result in (4.4.11) is obtained. Next, comparing the terms in (4.4.1) and (4.4.11), to satisfy the fixed point condition in MFG, we require another 6 equations in (4.4.12) for the coefficient functions  $w = (w_i : i = 1, 2, ..., 6)$ .

Using further algebraic structures, one can reduce the ODE system of 13 equations composed by (4.4.8) and (4.4.12) into a system of 4 equations.

**Proof of Proposition 45.** Let the smooth and bounded functions  $\{w_i : i = 1, 2, ..., 6\}$  be given, the functions (a, b, c, d, e, f, g) in (4.4.8) is a coupled linear system, and thus their existence, uniqueness and boundedness is shown by Theorem 12.1 in [2].

Plugging the 6 equations in (4.4.12) to the ODE system (4.4.8), we obtain

$$\begin{cases} a' - 2a^2 + k = 0, \\ b' - \frac{1}{2}g^2 - 4ab - 2bg - 2cg = 0, \\ c' - 4ac + k = 0, \\ d' - \frac{1}{2}e^2 - ef + 2c + 2a + b + g = 0, \\ e' - 2ae - eg = 0, \\ f' - eg - 2af - gf - 2be - 2ce = 0, \\ g' - 4ag - g^2 - 2k = 0, \end{cases}$$

with the terminal conditions

$$a(T) = b(T) = c(T) = d(T) = e(T) = f(T) = g(T) = 0.$$

Let l = 2a + g, and it is easy to obtain

$$l'(t) - l^2(t) = 0, \quad l(T) = 0,$$

which implies that l(t) = 2a(t) + g(t) = 0 for all  $t \in [0, T]$ . This gives the result that g = -2a and it yields e' = 0. Then with e(T) = 0, we have e(t) = 0 for all  $t \in [0, T]$  and thus one can obtain f' = 0, which indicates that f(t) = 0 for all  $t \in [0, T]$  as f(T) = 0. Therefore the ODE system (4.4.8) can be simplified to the following form about  $(a(t), b(t), c(t), d(t) : t \in [0, T])$ :

$$\begin{cases} a'(t) - 2a^{2}(t) + k = 0, \\ b'(t) - 2a^{2}(t) + 4a(t)c(t) = 0, \\ c'(t) - 4a(t)c(t) + k = 0, \\ d'(t) + b(t) + 2c(t) = 0, \end{cases}$$

$$(4.4.13)$$

with the terminal conditions

$$a(T) = b(T) = c(T) = d(T) = 0.$$
(4.4.14)

The unique solvability of the Riccati system (4.4.13)-(4.4.14) is proven in Lemma 61 in the Appendix. Note that the solution a of (4.3.2) is consistent with the solution of the Riccati system given by equations (4.4.13)-(4.4.14).

In this case, since 2a + g = 0 and e = 0 for all  $t \in [0, T]$ , it follows that  $\hat{\mu}_s = \bar{\mu} + \bar{W}_s$  for all  $s \in [t, T]$  from the fixed point result (4.4.11). Similarly,

$$\hat{\nu}_s = \bar{\nu} + \int_t^s \left( 2 + 4a(r)\hat{\mu}_r^2 - 4a(r)\hat{\nu}_r \right) \, dr + \int_t^s 2\hat{\mu}_r \, d\tilde{W}_r, \quad \forall s \in [t, T].$$

Plugging e = 0 and  $\hat{\mu}_s = \bar{\mu} + \tilde{W}_r$  back to (4.4.6), we obtain the optimal control by

$$\hat{\alpha}_s = 2a(s)(\bar{\mu} + \tilde{W}_s - \hat{X}_s).$$

Moreover, since e = f = 0 and g = -2a for  $s \in [t, T]$ , the value function can be simplified from (4.4.5) to

$$v(t, x, \bar{\mu}, \bar{\nu}) = a(t)x^2 - 2a(t)x\bar{\mu} + b(t)\bar{\mu}^2 + c(t)\bar{\nu} + d(t)$$

This concludes Proposition 45.

## 4.5 The *N*-Player Game

This section focuses on proving Proposition 46 regarding the corresponding N-player game. For simplicity, we can omit the superscript (N) when referring to the processes in the sample space  $\Omega^{(N)}$ .

To begin, we address the N-player game in Subsection 4.5.1, where we solve it and obtain a Riccati system containing  $O(N^3)$  equations. Subsequently, we reduce the relevant Riccati system to an ODE system in Subsection 4.5.2, which has a dimension independent of N. This simplified system forms the fundamental component of the convergence result.

## 4.5.1 Characterization of the *N*-player game by Riccati system

It is important to emphasize that based on the problem setting in Subsection 4.2.2 and the running cost for each player specified in (4.2.9), the *N*-player game can be classified as an *N*-coupled stochastic LQG problem. As a result, the value function and optimal control for each player can be determined by means of the following Riccati system:

For  $i = 1, 2, \ldots, N$ , consider

$$\begin{cases} A'_{i} - 2A_{i}^{\top} e_{i} e_{i}^{\top} A_{i} - 4 \sum_{j \neq i}^{N} A_{j}^{\top} e_{j} e_{j}^{\top} A_{i} + \frac{k}{N} \sum_{j \neq i}^{N} (e_{i} - e_{j}) (e_{i} - e_{j})^{\top} = 0, \\ B'_{i} - 2A_{i}^{\top} e_{i} e_{i}^{\top} B_{i} - 2 \sum_{j \neq i}^{N} \left( A_{i}^{\top} e_{j} e_{j}^{\top} B_{j} + A_{j}^{\top} e_{j} e_{j}^{\top} B_{i} \right) = 0, \\ C'_{i} - \frac{1}{2} B_{i}^{\top} e_{i} e_{i}^{\top} B_{i} - \sum_{j \neq i}^{N} B_{j}^{\top} e_{j} e_{j}^{\top} B_{i} + 2tr(A_{i}) = 0, \\ A_{i}(T) = B_{i}(T) = C_{i}(T) = 0, \end{cases}$$

$$(4.5.1)$$

where  $A_i$  is  $N \times N$  symmetric matrix,  $B_i$  is N-dimensional vector,  $C_i \in \mathbb{R}$  is a real constant, and  $e_i$  is the *i*-th natural basis in  $\mathbb{R}^N$  for each i = 1, 2, ..., N.

**Lemma 59.** Suppose  $(A_i, B_i, C_i : i = 1, 2, ..., N)$  is the solution of the Riccati system (4.5.1). Then, the value functions of the N-player game defined by (4.2.7) is

$$V_i(x^{(N)}) = (x^{(N)})^\top A_i(0)x^{(N)} + (x^{(N)})^\top B_i(0) + C_i(0), \quad i = 1, 2, \dots, N.$$

Moreover, the path and the control under the equilibrium are given by

$$d\hat{X}_{it}^{(N)} = \left(-2(A_i(t))_i^{\top}\hat{X}_t^{(N)} - (B_i(t))_i\right)dt + dW_{it}^{(N)} + d\tilde{W}_t,$$
(4.5.2)

and

$$\hat{\alpha}_{it}^{(N)} = -2(A_i(t))_i^{\top} \hat{X}_t^{(N)} - (B_i(t))_i$$

for each i = 1, 2, ..., N, where  $(A)_i$  denotes the *i*-th column of matrix A,  $(B)_i$  denotes the *i*-th entry of vector B and  $\hat{X}_t^{(N)} = [\hat{X}_{1t}^{(N)}, \hat{X}_{2t}^{(N)}, ..., \hat{X}_{Nt}^{(N)}]^\top$ .

*Proof.* From the dynamic programming principle, it is standard that, under enough regularities, the players' value function  $V(x^{(N)}) = (V_1, V_2, \ldots, V_N)(x^{(N)})$  can be lifted to the solution  $v_i(t, x^{(N)})$  of the following system of HJB equations, for  $i = 1, 2, \ldots, N$ ,

$$\begin{cases} \partial_t v_i + \inf_{a_{it} \in \mathbb{R}} \left( a_{it} \partial_{x_i} v_i + \frac{1}{2} a_{it}^2 \right) + \sum_{j \neq i}^N a_{jt} \partial_{x_j} v_i + \Delta v_i + \frac{k}{N} \sum_{j \neq i}^N \left( (e_i - e_j)^\top x^{(N)} \right)^2 = 0, \\ v_i \left( T, x^{(N)} \right) = 0. \end{cases}$$

Note that with  $a_{it} = -\partial_{x_i} v_i(t, x^{(N)})$  for each i = 1, 2, ..., N, the term in the infimum attains the optimal value and thus the HJB equation can be reduced to

$$\begin{cases} \partial_t v_i - \frac{1}{2} (\partial_{x_i} v_i)^2 - \sum_{j \neq i}^N \partial_{x_j} v_j \partial_{x_j} v_i + \Delta v_i + \frac{k}{N} \sum_{j \neq i}^N \left( (e_i - e_j)^\top x^{(N)} \right)^2 = 0, \\ v_i \left( T, x^{(N)} \right) = 0. \end{cases}$$
(4.5.3)

Then, the value functions V of the N-player game defined by (4.2.7) is  $V_i(x^{(N)}) = v_i(0, x^{(N)})$  for all i = 1, 2, ..., N. Moreover, the path and the control under the equilibrium are given by

$$d\hat{X}_{it}^{(N)} = -\partial_{x_i} v_i \left(t, \hat{X}_t^{(N)}\right) dt + dW_{it}^{(N)} + d\tilde{W}_t,$$

and

$$\hat{\alpha}_{it}^{(N)} = -\partial_{x_i} v_i \left( t, \hat{X}_t^{(N)} \right)$$

for i = 1, 2, ..., N. The proof is the application of Itô's formula and the details are omitted here.

Due to its LQG structure, the value function leads to a quadratic function of the form

$$v_i(t, x^{(N)}) = (x^{(N)})^\top A_i(t) x^{(N)} + (x^{(N)})^\top B_i(t) + C_i(t).$$

Plugging  $V_i$  into (4.5.3), and matching the coefficient of variables, we get the Riccati system of ODEs in (4.5.1) and the desired results are obtained.

## 4.5.2 Proof of Proposition 46: Reduced Riccati form for the equilibrium

At present, MFG and the corresponding N-player game can be characterized by Proposition 45 and Lemma 59, respectively. One of our primary objectives is to examine the convergence of the representative optimal path  $\hat{X}_{1t}^{(N)}$  generated by the N-player game defined in (4.5.1)-(4.5.2) to the optimal path  $\hat{X}_t$  of the MFG described in Proposition 45.

It should be noted that  $\hat{X}_t$  is solely dependent on the function a(t), as indicated in the ODE (4.3.2). In contrast,  $\hat{X}_{1t}^{(N)}$  depends on  $O(N^3)$  many functions derived from the solutions of a substantial Riccati system (4.5.1) involving matrices  $(A_{it}, B_{it} : i = 1, 2, ..., N)$ . Consequently, comparing these two processes meaningfully becomes an exceedingly challenging task without gaining further insight into the intricate structure of the Riccati system (4.5.1).

**Proof of Proposition 46.** Inspired from the setup in [50] and [46], we may seek a pattern for the matrix  $A_i$  in the following form:

$$(A_i)_{pq} = \begin{cases} a_1(t), & \text{if } p = q = i, \\ a_2(t), & \text{if } p = q \neq i, \\ a_3(t), & \text{if } p \neq q, p = i \text{ or } q = i, \\ a_4(t), & \text{otherwise.} \end{cases}$$
(4.5.4)

The next result justifies the above pattern: the  $N^2$  entries of the matrix  $A_i$  can be embedded to a 2-dimensional vector space no matter how big N is.

For the Riccati system (4.5.1), with the given of  $A_i$  and suppose each function in  $A_i$  is continuous on [0, T], it is obvious to see that  $B_i = 0$  for all  $t \in [0, T]$  and for all i = 1, 2, ..., N. Note that in this case, for i = 1, 2, ..., N, the optimal control is given by

$$\hat{\alpha}_{i} = -2\sum_{j=1}^{N} (A_{i})_{ij} \hat{X}_{jt}^{(N)} = -2 (A_{i})_{i}^{\top} \hat{X}_{t}^{(N)},$$

where  $(A)_i$  is the *i*-th column of matrix A.

Plugging the pattern (4.5.4) into the differential equation of  $A_i$ , we obtain the following system

of ODEs:

$$\begin{cases} a_1' - 2a_1^2 - 4(N-1)a_3^2 + \frac{N-1}{N}k = 0, \\ a_2' - 2a_3^2 - 4a_1a_2 - 4(N-2)a_3a_4 + \frac{k}{N} = 0, \\ a_3' - 2a_1a_3 - 4a_1a_3 - 4(N-2)a_3^2 - \frac{k}{N} = 0, \\ a_3' - 2a_1a_3 - 4a_2a_3 - 4(N-2)a_3a_4 - \frac{k}{N} = 0, \\ a_4' - 2a_3^2 - 4a_2a_3 - 4a_1a_4 - 4(N-3)a_3a_4 = 0. \end{cases}$$

with the terminal conditions

$$a_1(T) = a_2(T) = a_3(T) = a_4(T) = 0.$$

It is worth noting that there are two ODEs for  $a_3$ , and the two expressions should be equal, thus

$$a_1a_3 + (N-2)a_3^2 = a_2a_3 + (N-2)a_3a_4,$$

which implies that  $(a_1 + (N-2)a_3)' = (a_2 + (N-2)a_4)'$  or

$$2a_1^2 + 2(N-2)a_1a_3 + 4(N-1)a_3^2 + 4(N-2)a_2a_3 + 4(N-2)^2a_3a_4 - \frac{k}{N}$$
  
= 2(N-1)a\_3^2 + 4a\_1a\_2 + 4(N-2)(a\_2a\_3 + a\_3a\_4 + a\_1a\_4) + 4(N-2)(N-3)a\_3a\_4 - \frac{k}{N}.

After combining terms and substituting  $a_2 + (N-2)a_4$  with  $a_1 + (N-2)a_3$ , we get

$$a_1^2 + (N-2)a_1a_3 - (N-1)a_3^2 = 0,$$

which yields  $a_3 = a_1$  or  $a_3 = -\frac{1}{N-1}a_1$ . Note that, since  $a_1$  and  $a_3$  satisfies different differential equations, it follows that  $a_3 \neq a_1$ . Hence, we can conclude that  $a_3 = -\frac{1}{N-1}a_1$ . Next, from the equation  $a_1 + (N-2)a_3 = a_2 + (N-2)a_4$ , we have

$$a_4 = \frac{1}{N-2}a_1 + a_3 - \frac{1}{N-2}a_2$$

In conclusion, for i = 1, 2, ..., N,  $A_i$  has the following expressions:

$$(A_i)_{pq} = \begin{cases} a_1(t), & \text{if } p = q = i, \\ a_2(t), & \text{if } p = q \neq i, \\ -\frac{1}{N-1}a_1(t), & \text{if } p \neq q, p = i \text{ or } q = i, \\ \frac{1}{(N-1)(N-2)}a_1(t) - \frac{1}{N-2}a_2(t), & \text{otherwise}, \end{cases}$$

where  $a_1$  and  $a_2$  satisfies the system of ODEs (4.5.5)

$$\begin{cases} a_1' - \frac{2(N+1)}{N-1}a_1^2 + \frac{N-1}{N}k = 0, \\ a_2' + \frac{2}{(N-1)^2}a_1^2 - \frac{4N}{N-1}a_1a_2 + \frac{k}{N} = 0, \\ a_1(T) = a_2(T) = 0. \end{cases}$$
(4.5.5)

The existence and uniqueness of  $A_i$  in (4.5.1) are equivalent to the existence and uniqueness of (4.5.5). Firstly, the existence, uniqueness, and boundness of  $a_1$  in (4.5.5) is from the same argument for a in (4.4.13), which is shown as the proof of Lemma 61 in Appendix. The explicit solution of  $a_1$  is given by

$$a_1(t) = \sqrt{\frac{k}{2} \frac{(N-1)^2}{N(N+1)}} \frac{1 - e^{-2\sqrt{2}\sqrt{\frac{N+1}{N}k}(T-t)}}{1 + e^{-2\sqrt{2}\sqrt{\frac{N+1}{N}k}(T-t)}}$$

for all  $t \in [0, T]$ . Next, with the given of  $a_1$ , the existence, uniqueness, and boundness of  $a_2$  in (4.5.5) is guaranteed by Theorem 12.1 in [2]. Therefore, we can express the equilibrium paths and associated controls as the following:

$$d\hat{X}_{it}^{(N)} = -2a_1^N(t) \left( \hat{X}_{it}^{(N)} - \frac{1}{N-1} \sum_{j \neq i}^N \hat{X}_{jt}^{(N)} \right) dt + dW_{it}^{(N)} + d\tilde{W}_t,$$
(4.5.6)

and

$$\hat{\alpha}_{it}^{(N)} = -2a_1^N(t) \left( \hat{X}_{it}^{(N)} - \frac{1}{N-1} \sum_{j \neq i}^N \hat{X}_{jt}^{(N)} \right)$$

respectively for i = 1, 2, ..., N, where  $a_1^N$  is the solution to the ODE for  $a_1$  in (4.5.5). This concludes Proposition 46.

## 4.6 Further remark

We have now established Proposition 45 concerning the MFG in Section 4.4 and Proposition 46 regarding the N-player game in Section 4.5. With these propositions proven, we are now able to conclude the proof of Theorem 44, which was presented in Section 4.3.4.

## 4.7 Appendix

**Lemma 60.** Let  $\mathbb{W}_p$  be the *p*-Wasserstein metric. If X and Y are two real-valued random variables and c is a constant, then

$$\mathbb{W}_p(\mathcal{L}(X), \mathcal{L}(Y)) = \mathbb{W}_p(\mathcal{L}(X+c), \mathcal{L}(Y+c)).$$
(4.7.1)

Moreover, if  $\alpha = \{\alpha_i : i \in \mathbb{N}\}$  is a sequence of random variables, then

$$\mathbb{W}_p\left(\frac{1}{N}\sum_{i=1}^N \delta_{\alpha_i+c}, \mathcal{L}(Y+c)\right) = \mathbb{W}_p\left(\frac{1}{N}\sum_{i=1}^N \delta_{\alpha_i}, \mathcal{L}(Y)\right).$$
(4.7.2)

*Proof.* By definition of the *p*-Wasserstein metric, we have:

$$\mathbb{W}_p(\mathcal{L}(X), \mathcal{L}(Y)) = \left(\inf_{\pi \in \Pi(\mathcal{L}(X), \mathcal{L}(Y))} \int_{\mathbb{R}^2} |x - y|^p d\pi(x, y)\right)^{\frac{1}{p}},$$

where  $\Pi(\mathcal{L}(X), \mathcal{L}(Y))$  is the set of all joint probability measures with marginals  $\mathcal{L}(X)$  and  $\mathcal{L}(Y)$ . Similarly,

$$\mathbb{W}_p(\mathcal{L}(X+c), \mathcal{L}(Y+c)) = \left(\inf_{\pi \in \Pi(\mathcal{L}(X+c), \mathcal{L}(Y+c))} \int_{\mathbb{R}^2} |x-y|^p d\pi(x,y)\right)^{\frac{1}{p}},$$

where  $\Pi(\mathcal{L}(X+c), \mathcal{L}(Y+c))$  is the set of all joint probability measures with marginals  $\mathcal{L}(X+c)$ and  $\mathcal{L}(Y+c)$ .

Now, consider the mapping  $\Phi : \mathbb{R}^2 \to \mathbb{R}^2$  given by  $\Phi(x, y) = (x + c, y + c)$ . For any  $\pi \in \Pi(\mathcal{L}(X), \mathcal{L}(Y))$ , the pushforward measure of  $\pi$  under  $\Phi$  belongs to  $\Pi(\mathcal{L}(X + c), \mathcal{L}(Y + c))$ , i.e.,  $\pi' = \Phi_*\pi \in \Pi(\mathcal{L}(X + c), \mathcal{L}(Y + c))$ . Thus, we have

$$\Phi_*\Pi(\mathcal{L}(X),\mathcal{L}(Y)) \subset \Pi(\mathcal{L}(X+c),\mathcal{L}(Y+c)).$$

Moreover,  $\Phi$  is bijective and measure preserving, then

$$\int_{\mathbb{R}^2} |x - y|^p d\pi'(x, y) = \int_{\mathbb{R}^2} |(x + c) - (y + c)|^p d\pi(x, y) = \int_{\mathbb{R}^2} |x - y|^p d\pi(x, y).$$

Therefore, we know that

$$\begin{split} \mathbb{W}_p^p\left(\mathcal{L}(X), \mathcal{L}(Y)\right) &= \inf_{\pi \in \Pi(\mathcal{L}(X), \mathcal{L}(Y))} \int_{\mathbb{R}^2} |x - y|^p d\pi(x, y) \\ &= \inf_{\pi \in \Pi(\mathcal{L}(X), \mathcal{L}(Y))} \int_{\mathbb{R}^2} |x - y|^p d\Phi_* \pi(x, y) \\ &= \inf_{\pi' \in \Phi_* \Pi(\mathcal{L}(X), \mathcal{L}(Y))} \int_{\mathbb{R}^2} |x - y|^p d\pi'(x, y) \\ &\geq \mathbb{W}_p^p(\mathcal{L}(X + c), \mathcal{L}(Y + c)). \end{split}$$

by the definition of the *p*-Wasserstein metric. If we apply the above inequality to X' = X + c, Y' = Y + c, and c' = -c, the opposite inequality is provided. Thus, it completes the proof of (4.7.1).

Next, we note that

$$\frac{1}{N}\sum_{i=1}^{N}\delta_{\alpha_{i}+c} = \mathcal{L}(\alpha_{u}+c|\alpha),$$

where u be a uniform random variable on  $\{1, 2, ..., N\}$  independent to  $\alpha$ . Using (4.7.1), we conclude (4.7.2) from

$$\mathbb{W}_{p}\left(\frac{1}{N}\sum_{i=1}^{N}\delta_{\alpha_{i}+c},\mathcal{L}(Y+c)\right) = \mathbb{W}_{p}\left(\mathcal{L}(\alpha_{u}+c|\alpha),\mathcal{L}(Y+c)\right)$$
$$= \mathbb{W}_{p}\left(\mathcal{L}(\alpha_{u}|\alpha),\mathcal{L}(Y)\right)$$
$$= \mathbb{W}_{p}\left(\frac{1}{N}\sum_{i=1}^{N}\delta_{\alpha_{i}},\mathcal{L}(Y)\right).$$

**Lemma 61.** Under the Assumption 43, there exists a unique solution  $(a(t), b(t), c(t), d(t) : t \in [0, T])$ for the Riccati system of ODEs (4.4.13)-(4.4.14) and the solution can given explicitly by

$$\begin{cases} a(t) = \sqrt{\frac{k}{2}} \frac{1 - e^{-2\sqrt{2k}(T-t)}}{1 + e^{-2\sqrt{2k}(T-t)}}, \\ b(t) = \int_t^T \left(4a(s)c(s) - 2a^2(s)\right) ds, \\ c(t) = k \int_t^T e^{\int_t^s -4a(r)dr} ds, \\ d(t) = \int_t^T \left(b(s) + 2c(s)\right) ds. \end{cases}$$

*Proof.* Firstly, with the given of k > 0, we can solve the ODE

$$a'(t) - 2a^{2}(t) + k = 0, \quad a(T) = 0$$

explicitly by the method of separating variables. Note that with the differential form, we have

$$\frac{da}{\left(\sqrt{2}a - \sqrt{k}\right)\left(\sqrt{2}a + \sqrt{k}\right)} = \frac{1}{2\sqrt{k}}\left(\frac{1}{\sqrt{2}a - \sqrt{k}} - \frac{1}{\sqrt{2}a + \sqrt{k}}\right)da = dt.$$

It follows that

$$\ln\left(\left|\frac{\sqrt{2}a - \sqrt{k}}{\sqrt{2}a + \sqrt{k}}\right|\right) = 2\sqrt{2k}t + C_1$$

for some constant  $C_1$  by taking integration on both sides. Thus by calculation, we obtain

$$a(t) = \sqrt{\frac{k}{2}} \frac{1 - C_2 e^{2\sqrt{2kt}}}{1 + C_2 e^{2\sqrt{2kt}}}$$

for some constant  $C_2$  to be determined. Since a(T) = 0, it yields that  $C_2 = e^{-2\sqrt{2kT}}$  and thus

$$a(t) = \sqrt{\frac{k}{2}} \frac{1 - e^{-2\sqrt{2k}(T-t)}}{1 + e^{-2\sqrt{2k}(T-t)}}.$$

It is easy to verify that  $a(\cdot)$  is in  $C^{\infty}([0,T])$  and is bounded. With the given of a, the functions (b, c, d) in the Riccati system (4.4.13)-(4.4.14) is a coupled linear system, and thus their existence, uniqueness, and boundedness are given by Theorem 12.1 in [2].

## Chapter 5

# On the graphon mean field game equations: Individual agent affine dynamics and mean field dependent performance functions

## 5.1 Introduction

Mean Field Game (MFG) theory establishes Nash equilibirum conditions for large populations of asymptotically negligible non-cooperating agents via an analysis of the infinite limit population (Huang, Caines, and Malhame [40, 42, 45]; Lasry and Lions [55]). The resulting PDEs (Partial Differential Equations) consist of a backward Hamilton-Jacobi-Bellman (HJB) equation and a forward Fokker-Planck-Kolmogorov (FPK) equation for each generic agent. These equations are linked by the state distribution of a generic agent which is called the mean field of the system.

In this Chapter, our objective is to establish the unique solvability of the GMFG equation in an appropriate function space. The GMFG equations consist of a collection of parameterized Hamilton-Jacobi- Bellman equations,  $HJB(\alpha), \alpha \in [0, 1]$ , and a collection of parameterized Fokker-Planck-Kolmogorov equations,  $FPK(\alpha)$  with  $\alpha \in [0, 1]$ . The solution of a set of GMFG equations is a parameterized pair  $(v, \mu)$ , where  $v[\alpha] = v(t, \alpha, x)$  solves the  $HJB(\alpha)$  equation and  $\mu[\alpha] = \mu(t, \alpha, x)$ solves the  $FPK(\alpha)$  equation. The coupling of the system PDEs in this chapter has the following features (see [12] for a more general framework subject to different hypotheses):

- $FPK(\alpha)$  depends upon  $HJB(\alpha)$  through its first order coefficient  $\nabla v$ .
- $HJB(\alpha)$  depends upon  $FPK(\alpha')$  for all  $\alpha' \in [0, 1]$  through the graphon g acting on  $\mu[\alpha']$ ; this is the major difference from MFG.

The GMFG equations with a constant graphon reduce to the classical MFG system as a special case, and the original methods to establish solvability of the classical MFG equations are helpful in the present case. In [45] and [68], a Banach fixed point analysis is used depending on a contraction

argument; this is based on assumptions on the Lipschitz continuity of the functions appearing in the MFG equations and their derivatives, and yields uniqueness as well as existence. This approach is used in the parallel study [12] of the solvability of the GMFG equations. On the other hand, [13] and [72] carry out the existence analysis using the Schauder fixed point theorem based upon regularity assumptions and then obtain uniqueness via a monotonicity assumption on the running cost.

In this work, similar to the aforementioned analyses, we will establish the existence of solutions via the application of a fixed point theorem. Our existence proof adopts Schauder's argument on the fixed point theorem and is more closely relevant to [36], [13], and [72] in this sense. Unlike [36] on the solvability in Sobolev space, our solvability is to answer the existence in Hölder space along the lines of [13] and [72]. Nevertheless, different from all aforementioned papers, our proof on the continuity of the gradient of the value function with respect to the coefficient functions relies on probabilistic estimates rather than the theory of viscosity solutions. The main advantage of our approach is that we can conclude the local Lipschitz continuity of the solution map, which is stronger than continuity and beneficial to the subsequent analysis of the GMFG.

Having said that, the major difficulty generalizing existence from the MFG case to the GMFG case is to obtain the regularity of the solution with respect to the variable  $\alpha$ , which is essential for the existence result by the Schauder's fixed point theorem. To be more illustrative, for instance, to obtain a uniform first order estimate of  $|\nabla v(t, \alpha, x) - \nabla v(t, \alpha', x)|$  for the solution v of the HJB equation, one has to compare the solutions from two different HJBs parameterized by  $\alpha$  and  $\alpha'$ . This leads to a study of the sensitivity with respect to coefficient functions of corresponding PDEs. Therefore, the local Lipschitz continuity of the HJB solution map becomes essential for this procedure.

The chapter is organized as follows. Section 5.2 gives the problem set up. Section 5.3 presents the regularity of parabolic PDE and applies this to the FPK. Section 5.4 presents the existence result and Section 5.5 treats uniqueness. Section 5.6 presents a summary and extensions of the main result. For better clarity, all notations used in this chapter have been collected and explained in the Appendix Section 5.7.

## 5.2 Problem setup

Let  $\mathbb{T}^d$  be a d-torus.  $\mathcal{P}_1(\mathbb{T}^d)$  is the Wasserstein space of probability measures on  $\mathbb{T}^d$  satisfying

$$\int_{\mathbb{T}^d} |x| d\mu(x) < \infty$$

endowed with 1-Wasserstein metric  $\mathbb{W}_1(\cdot, \cdot)$  defined by

$$\mathbb{W}_1(\mu,\nu) = \inf_{\pi \in \Pi(\mu,\nu)} \int_{\mathbb{T}^d \times \mathbb{T}^d} |x-y| d\pi(x,y),$$

where  $\Pi(\mu, \nu)$  is the collection of all probability measures on  $\mathbb{T}^d \times \mathbb{T}^d$  with its marginals agreeing with  $\mu$  and  $\nu$ .

We consider the following large system of multi-agent problems. A generic agent can be identified by its state pair  $(\alpha, x) \in [0, 1] \times \mathbb{T}^d$ , where  $\alpha$  is the cluster index and x is a  $\mathbb{T}^d$  valued state. The weights of connections between clusters are given by a symmetric measurable function  $g: [0, 1]^2 \mapsto \mathbb{R}$ , which is commonly referred to a graphon [58]. The population density at the cluster  $\alpha$  at time t will be given by  $\mu(t, \alpha) \in \mathcal{P}_1(\mathbb{T}^d)$ .

*Example.* Two examples of graphons are given in the following discussion, while the reader is referred to [58] for the fundamental theory of this subject. A uniform graphon which corresponds to the limit of a sequence of Erdös-Rényi graphs with parameter  $p, 0 \le p \le 1$ , is given by

$$g(\alpha, \alpha') = p, \quad \forall \alpha, \alpha' \in [0, 1]$$
(5.2.1)

and the uniform attachment graph limit has the graphon

$$g(\alpha, \alpha') = 1 - \max\{\alpha, \alpha'\}, \quad \forall \alpha, \alpha' \in [0, 1].$$

$$(5.2.2)$$

A running cost incurred to the generic agent of  $(\alpha, x)$  with a feedback control exertion **a** :  $[0,T] \times [0,1] \times \mathbb{T}^d \mapsto \mathbb{R}^d$  at time t is given by

$$\ell(\mu, g, \mathbf{a}, t, \alpha, x) = \frac{1}{2} |\mathbf{a}(t, \alpha, x)|^2 + \ell_1(\mu, g, t, \alpha, x)$$
(5.2.3)

for some given function  $\ell_1(\cdot, \cdot, \cdot, \cdot, \cdot)$ . The following cost can be considered as an example for  $\ell_1$ 

$$\ell_1(\mu, g, t, \alpha, x) = \int_0^1 \int_{\mathbb{T}^d} \ell_2(x, y) \mu(t, \alpha', dy) g(\alpha, \alpha') \, d\alpha'$$
(5.2.4)

for some  $\ell_2 : \mathbb{T}^d \times \mathbb{T}^d \mapsto \mathbb{R}$ .

Let  $b: [0,T] \times [0,1] \times \mathbb{T}^d \mapsto \mathbb{R}$  and  $m_0: [0,1] \times \mathbb{T}^d \mapsto \mathbb{R}^+$  be two given smooth enough functions. By  $\nabla b$ , we denote the gradient of b on the domain  $\mathbb{T}^d$ , which is mapping  $[0,T] \times [0,1] \times \mathbb{T}^d \mapsto \mathbb{R}^d$ . Finding a solution of the GMFG equations consists of solving for the unknown triples  $(v, \mathbf{a}^*, \mu)$ :

- the value function  $v: [0,T] \times [0,1] \times \mathbb{T}^d \mapsto \mathbb{R}$ ,
- optimal control  $\mathbf{a}^* : [0, T] \times [0, 1] \times \mathbb{T}^d \mapsto \mathbb{R}^d$ ,
- and the density  $\mu: [0,T] \times [0,1] \times \mathbb{T}^d \mapsto \mathbb{R}^+$ ,

satisfying the  $\alpha$  parameterized family

$$\begin{cases} \partial_t v + (\nabla b + \mathbf{a}^*) \cdot \nabla v + \frac{1}{2} \Delta v + \ell(\mu, g, \mathbf{a}^*) = 0 \\ \mathbf{a}^*(t, \alpha, x) = \arg\min_{a \in \mathbb{R}^d} \left\{ a \cdot \nabla v(t, \alpha, x) + \frac{1}{2} |a|^2 \right\} \\ \partial_t \mu = -\operatorname{div}_x((\nabla b + \mathbf{a}^*)\mu) + \frac{1}{2} \Delta \mu \\ v(T, \alpha, x) = 0, \quad \mu(0, \alpha, x) = m_0(\alpha, x). \end{cases}$$
(5.2.5)

In the first and third equation of (5.2.5), each term is a function of  $(t, \alpha, x)$  without further specification. In particular, the  $\ell(\mu, g, \mathbf{a}^*)$  shall be understood as a mapping

$$(t,\alpha,x) \mapsto \ell(\mu,g,\mathbf{a}^*)(t,\alpha,x) := \ell(\mu,g,\mathbf{a}^*,t,\alpha,x)$$

Our goal in this chapter is to establish existence, uniqueness for the solution of (5.2.5) in an appropriate solution space. We close this section with a brief illustration of the probabilistic formulation on the GMFG for the motivational purpose. A generic player in GMFG is identified by a pair  $(\alpha, x) \in [0, 1] \times \mathbb{T}^d$ , where  $\alpha$  is geographical information and x is a state. The population density at index  $\alpha$  at time t is denoted by  $\mu(t, \alpha) \in \mathcal{P}_1(\mathbb{T}^d)$  and the relation between two generic players in  $\alpha$  and  $\alpha'$  is given by a graphon  $g(\alpha, \alpha')$ . Given a population density  $(t, \alpha) \mapsto \mu(t, \alpha)$  and a graphon  $(\alpha, \alpha') \mapsto g(\alpha, \alpha')$ , a generic player exerts its optimal strategy of the following stochastic control problem described below. State evolution of the generic player at  $\alpha$  follows a controlled stochastic differential equation:

$$X_t^{\alpha} = X_0^{\alpha} + \int_0^t \left(\nabla b\left(s, \alpha, X_s^{\alpha}\right) + \mathbf{a}\left(s, \alpha, X_s^{\alpha}\right)\right) ds + W_t^{\alpha},\tag{5.2.6}$$

where the drift is formed by a control process  $\mathbf{a}$  and a conservative vector field  $\nabla b$ ,  $W^{\alpha}$  is a Brownian motion in a filtered probability space independent to  $W^{\beta}$  for any  $\beta \neq \alpha$ , and  $X_0^{\alpha}$  is an initial random variable with a given distribution  $m_0(\alpha)$ . In the above, the left hand side is understood as the coset of  $\mathbb{Z}^d$  that contains the right hand side by a mapping  $\pi(x) = x + \mathbb{Z}^d$ . We use  $X^{\alpha}[\mathbf{a}]$  to denote the process with the dependence on  $\mathbf{a}$ . The objective of the generic player at  $\alpha$  with a given population density flow  $\mu$  is to minimize the total cost incurred during [0, T] of the form

$$J^{\alpha}(\mathbf{a},\mu) = \mathbb{E}\left[\int_{0}^{T} \ell\left(\mu, g, \mathbf{a}, t, \alpha, X_{t}^{\alpha}[\mathbf{a}]\right) dt\right]$$

over a reasonably rich enough control space of **a**. Note that the optimal strategy  $\mathbf{a}^*$  depends on  $\mu$ . Given an initial distribution  $m_0$ , the goal of the GMFG is to find the Nash equilibrium  $\mu^*$  and the corresponding  $\mathbf{a}^*$ , i.e. the pair ( $\mu^*, \mathbf{a}^*$ ) satisfies

$$J^{\alpha}(\mathbf{a}^*, \mu^*) \leq J^{\alpha}(\mathbf{a}, \mu^*), \ \forall \mathbf{a} \ \text{and} \ \mu^*(t, \alpha) \sim X_t^{\alpha}[\mathbf{a}^*], \ \forall (t, \alpha).$$

Indeed, the above formulation poses a class of mean field game problems indexed by  $\alpha \in [0, 1]$  and

couplings between mean field games are imposed by the running cost  $\ell$  via graphon g. For more detailed discussion and various applications are referred to [10, 11, 12].

## 5.3 Some regularity results

We are going to present sensitivity results of the parabolic PDE and FPK equations with respect to their coefficients separately, which eventually serve for the proof of fixed point theorem as key elements. Throughout the chapter, we will use  $\Psi(\cdot)$  in various places as a generic positive function increasing with respect to its variables. Moreover, all function spaces and relevant norms are sorted out in Section 5.7.

#### 5.3.1 Parabolic equations

Consider the equation

$$\begin{cases} \partial_t u = \frac{1}{2} \Delta u - cu + f, & \text{on } (0, T) \times \mathbb{T}^d \\ u(0, x) = 0, & \text{on } x \in \mathbb{T}^d. \end{cases}$$
(5.3.1)

We will denote the solution map by u = u[c, f] whenever it is necessary to emphasize its dependence on the coefficient functions.

#### 5.3.1.1 Preliminaries on solvability

If the coefficients c and f are Hölder in both variables (t, x), then there exists a unique classical solution. Recall that  $\Psi(\cdot)$  is a generic function mentioned in the first paragraph of Section 3.

**Lemma 62.** If  $c, f \in C^{\delta/2,\delta}([0,T] \times \mathbb{T}^d)$  holds for some  $\delta \in (0,1)$ , then there exists unique solution  $u \in C^{1+\delta/2,2+\delta}([0,T] \times \mathbb{T}^d)$  of (5.3.1) satisfying

$$|u|_{1+\delta/2,2+\delta} \le \Psi(|c|_{\delta/2,\delta}, |f|_{\delta/2,\delta}).$$

Moreover, v(t,x) := u(T-t,x) has a probabilistic representation v[c, f] of the form

$$v(t,x) = v[c,f](t,x) := \mathbb{E}\left[\int_{t}^{T} \exp\left\{-\int_{t}^{s} c\left(r, X^{t,x}(r)\right) dr\right\} f\left(s, X^{t,x}(s)\right) ds\right],$$
(5.3.2)

where

$$X^{t,x}(s) = x + W(s) - W(t)$$
(5.3.3)

for some Brownian motion W.

*Proof.* The solvability and its Hölder estimate is from Theorem 8.7.2 and Theorem 8.7.3 of [53], Theorem IV.5.1 of [54]. The probabilistic representation v[c, f] is from Feynman-Kac formula, see [31].

In the above, we remark that, (5.3.3) reads by  $X^{t,x}(s) = \pi(x + W(s) - W(t))$ , where  $\pi$  is the generic mapping  $\mathbb{R}^d \mapsto \mathbb{R}^d / \mathbb{Z}^d$ . Later we also need to use the following definition of weak solution, see [25].

**Definition 63.** A function  $u \in L^2([0,T], H^1(\mathbb{T}^d))$  is a weak solution of (5.3.1) if u satisfies

$$\begin{cases} \int_{\mathbb{T}^d} \phi(-\partial_t u - cu + f) dx = \frac{1}{2} \int_{\mathbb{T}^d} \nabla \phi \cdot \nabla u dx, \quad \forall \phi \in H^1(\mathbb{T}^d) \\ u(0, x) = 0, \quad on \ x \in \mathbb{T}^d. \end{cases}$$
(5.3.4)

We have the following uniqueness with the same assumptions as in Lemma 62.

**Lemma 64.** If  $c, f \in C^{\delta/2,\delta}([0,T] \times \mathbb{T}^d)$  holds for some  $\delta \in (0,1)$ , then there exists unique weak solution of (5.3.1) in  $L^2([0,T], H^1(\mathbb{T}^d))$ .

*Proof.* By Lemma 62, there exists a classical solution u. Together with the compactness of the domain, it yields  $u \in L^2([0,T], H^1(\mathbb{T}^d))$ . By Theorem 7.4 of [25], uniqueness in  $L^2([0,T], H^1(\mathbb{T}^d))$  holds if  $c \in L^\infty$  and  $f \in L^2$ , and this is valid, since all coefficients are continuous on the compact domain.

#### 5.3.1.2 First order regularity and sensitivity of the solution map

Although Lemma 62 has an estimation on  $|u|_{1,2}$ , it is controlled by an upper bound relevant to the Hölder norm of coefficients in the t variable, which is not desirable, see Section 5.4.5 for further remarks. Next, we will develop an upper bound independent of t-Hölder norm of the coefficients. To proceed, we define a linear operator

$$Lu = \partial_t u - \frac{1}{2}\Delta u. \tag{5.3.5}$$

The first result is on an estimate of  $|u|_0 = \sup_{[0,T] \times \mathbb{T}^d} |u(t,x)|$ .

**Lemma 65.** If  $c, f \in C^{\delta/2,\delta}([0,T] \times \mathbb{T}^d)$ , then u of (5.3.1) satisfies  $|u|_0 \leq e^{|c|_0 T} |f|_0 T$ .

*Proof.* If c = 0, then with  $u_1 = |f|_0 t$ ,

$$Lu_1 - f = |f|_0 - f \ge 0.$$

If  $c \neq 0$ , then with  $u_2 = \frac{|f|_0(e^{|c|_0 t} - 1)}{|c|_0}$ ,

$$(L+c)u_2 = |f|_0 e^{|c|_0 t} \left(1 + \frac{c}{|c|_0}\right) - \frac{c}{|c|_0} |f|_0$$
$$= |f|_0 (e^{|c|_0 t} - 1) \left(1 + \frac{c}{|c|_0}\right) + |f|_0$$
$$\ge f.$$

Note that both  $u_1$  and  $u_2$  are no greater than  $e^{|c|_0 T} |f|_0 T$ , and finally the comparison principle yields the result.

Next we will have the first order estimate independent to the Hölder norm in t of the coefficients. It also gives sensitivity of the solution map with respect to the coefficients.

**Lemma 66.** Let c, f be in  $C^{\delta,1}([0,T] \times \mathbb{T}^d)$  for some  $\delta \in (0,1)$ . Then the solution u of (5.3.1) belongs to  $C^{1,2}([0,T] \times \mathbb{T}^d)$  with

$$|u|_{0,1} \le \Psi(|c|_{0,1} + |f|_{0,1}).$$

Furthermore, the solution map u = u[c, f] satisfies

$$|u[c_1, f_1] - u[c_2, f_2]|_0 \le \Psi(K)(|c_1 - c_2|_0 + |f_1 - f_2|_0)$$

for  $K := |c_1|_0 + |c_2|_0 + |f_1|_0 + |f_2|_0$ .

*Proof.* u of (5.3.1) can be written by u(t, x) = v[c, f](T - t, x) with its probabilistic representation of (5.3.2). By setting  $X^i := X^{t,x_i}$  of (5.3.3), we have

$$X_s^1 - X_s^2 = x_1 - x_2, \ \forall s \ge t.$$

If we define

$$\Lambda^i_s = e^{-\int_t^s c(r, X^i(r))dr},$$

then

$$v[c, f](t, x_i) = \mathbb{E}\left[\int_t^T \Lambda_s^i f(s, X^i(s)) ds\right].$$

We first note that, by mean value theorem,

$$\left| \int_{t}^{s} c\left(r, X^{1}(r)\right) dr - \int_{t}^{s} c\left(r, X^{2}(r)\right) dr \right| \leq T |c|_{0,1} |x_{1} - x_{2}|.$$

Once again by mean value theorem and the fact of  $|-\int_t^s c(r, X^i(r))dr| \leq T|c|_0$ , we obtain

$$\begin{aligned} \left| \Lambda_s^1 - \Lambda_s^2 \right| &\le e^{T|c|_0} \left| \int_t^s c(r, X^1(r)) dr - \int_t^s c(r, X^2(r)) dr \right| \\ &\le \Psi(|c|_{0,1}) |x_1 - x_2| \end{aligned}$$

with probability one for  $\Psi = T|c|_{0,1}e^{T|c|_0}$ . Therefore, we have

$$\begin{aligned} |v[c,f](t,x_1) - v[c,f](t,x_2)| &\leq \mathbb{E}\left[\int_t^T \left|\Lambda_s^1 f\left(s, X^1(s)\right) - \Lambda_s^2 f\left(s, X^2(s)\right)\right| ds\right] \\ &\leq \mathbb{E}\left[\int_t^T \left(|\Lambda^1|_0 \left|f\left(s, X^1(s)\right) - f\left(s, X^2(s)\right)\right| + |f|_0 \left|\Lambda_s^1 - \Lambda_s^2\right|\right) ds\right], \end{aligned}$$

which gives

$$\begin{split} &|v[c,f](t,x_1) - v[c,f](t,x_2)| \\ &\leq \mathbb{E}\left[\int_t^T \Psi(|c|_0) |\nabla f|_0 \left| X^1(s) - X^2(s) \right| ds\right] + \mathbb{E}\left[\int_t^T |f|_0 \left| \Lambda_s^1 - \Lambda_s^2 \right| ds\right] \\ &\leq T \Psi(|c|_0) |\nabla f|_0 |x_1 - x_2| + T |f|_0 |\Psi(|c|_{0,1})|x_1 - x_2| \\ &\leq \Psi(|c|_{0,1} + |f|_{0,1}) |x_1 - x_2|. \end{split}$$

This implies that  $|\nabla v|_0 \leq \Psi(|c|_{0,1} + |f|_{0,1})$ . Together with Lemma 65, we conclude that

$$|u|_{0,1} \le \Psi(|c|_{0,1} + |f|_{0,1}).$$
(5.3.6)

Next, we estimate  $|u[c, f_1] - u[c, f_2]|_0$ . For any (t, x), we set  $\Lambda_s = e^{-\int_t^s c(r, X(r))dr}$ , and note that

$$\begin{split} |v[c,f_1](t,x) - v[c,f_2](t,x)| &\leq \mathbb{E}\left[\int_t^T |\Lambda_s f_1(s,X_s) - \Lambda_s f_2(s,X_s)|ds\right] \\ &\leq |f_1 - f_2|_0 \mathbb{E}\left[\int_t^T |\Lambda_s|ds\right] \\ &\leq Te^{T|c|_0}|f_1 - f_2|_0. \end{split}$$

This concludes that

$$|u[c, f_1] - u[c, f_2]|_0 \le \Psi(|c|_0)|f_1 - f_2|_0.$$
(5.3.7)

In the following, we estimate  $|u[c_1, f] - u[c_2, f]|_0$ . By setting  $\Lambda_s^i = e^{-\int_t^s c_i(r, X(r))dr}$ , we have

$$|\Lambda_s^1 - \Lambda_s^2| \le e^{T(|c_1|_0 + |c_2|_0)} \int_t^s |c_1(r, X_r) - c_2(r, X_r)| dr \le e^{T(|c_1|_0 + |c_2|_0)} |c_1 - c_2|_0 T dr$$

with probability one. Therefore,

$$|v[c_1, f](t, x) - v[c_2, f](t, x)| \le \mathbb{E} \left[ \int_t^T \left| \Lambda_s^1 f(s, X_s) - \Lambda_s^2 f(s, X_s) \right| ds \right]$$
  
$$\le |f|_0 \mathbb{E} \left[ \int_t^T \left| \Lambda_s^1 - \Lambda_s^2 \right| ds \right]$$
  
$$\le T^2 e^{T(|c_1|_0 + |c_2|_0)} |f|_0 |c_1 - c_2|_0.$$

This implies that

$$|u[c_1, f] - u[c_2, f]|_0 \le \Psi(|c_1|_0 + |c_2|_0 + |f|_0)|c_1 - c_2|_0.$$
(5.3.8)

The conclusion yields from (5.3.6), (5.3.7), (5.3.8).

## 5.3.1.3 Second order regularity and first order sensitivity

Next, we will see that under better regularity of c and f in x, we can improve regularity and sensitivity. Formally, if u of (5.3.1) is smooth enough, one can take derivatives of the equation to

conclude that  $\bar{u}_j = \partial_j u$  is the solution of the following equation depending on c, f and u of (5.3.1)

$$\begin{cases} \partial_t \bar{u}_j = \frac{1}{2} \Delta \bar{u}_j - c \bar{u}_j - u \partial_j c + \partial_j f, & \text{on } (0, T) \times \mathbb{T}^d \\ \bar{u}_j(0, x) = 0, & \text{on } x \in \mathbb{T}^d. \end{cases}$$
(5.3.9)

However, (5.3.9) is valid only if  $u \in C^{1,3}$  is given a priori.

**Lemma 67.** If  $c, f \in C^{\delta,2}([0,T] \times \mathbb{T}^d)$  for some  $\delta \in (0,1)$ , then the solution u of (5.3.1) is in  $C^{1,3}([0,T] \times \mathbb{T}^d)$  and  $\bar{u}_j = \partial_j u$  is the unique solution of (5.3.9).

*Proof.* By Lemma 64, u satisfies, for any  $\phi \in H^2(\mathbb{T}^d)$ ,

$$\int_{\mathbb{T}^d} \phi(-\partial_t u - cu + f) dx = \frac{1}{2} \int_{\mathbb{T}^d} \nabla \phi \cdot \nabla u dx.$$

Now, if we replace the test function  $\phi$  by  $\partial_i \phi$  in the above variational form, then we have

$$\int_{\mathbb{T}^d} \partial_i \phi(-\partial_t u - cu + f) dx = \frac{1}{2} \int_{\mathbb{T}^d} \nabla \partial_i \phi \cdot \nabla u dx$$

Using integration by parts, we can show that  $\bar{u}_j$  solves the variational form of (5.3.9) for any  $\phi \in H^2(\mathbb{T}^d)$ . Since  $H^2(\mathbb{T}^d)$  is a dense subset in  $H^1(\mathbb{T}^d)$ ,  $\bar{u}_j$  is indeed a unique weak solution of (5.3.9).

Lastly, since the  $\nabla c, \nabla f \in C^{\delta,1}([0,T] \times \mathbb{T}^d)$ , we conclude that  $\bar{u}_j$  is indeed a classical solution from Lemma 62. This also implies that  $u \in C^{1,3}([0,T] \times \mathbb{T}^d)$ .

**Lemma 68.** Let  $c, f \in C^{\delta,2}([0,T] \times \mathbb{T}^d)$ . Then the solution u of (5.3.1) belongs to  $C^{1,3}([0,T] \times \mathbb{T}^d)$  with

$$|u[c,f]|_{0,2} \le \Psi(|c|_{0,2} + |f|_{0,2}).$$

Furthermore, the solution map u = u[c, f] of (5.3.1) satisfies

$$|u[c_1, f_1] - u[c_2, f_2]|_{0,1} \le \Psi(K)(|c_1 - c_2|_{0,1} + |f_1 - f_2|_{0,1})$$

for

$$K := |c_1|_{0,1} + |c_2|_{0,1} + |f_1|_{0,1} + |f_2|_{0,1}.$$

*Proof.* By Lemma 67,  $\bar{u}_j = \partial_j u$  is the classical solution of (5.3.9), which satisfies

$$\bar{u}_j = u[c, \bar{f}],$$

where

$$\bar{f} = -u\partial_j c + \partial_j f.$$

Applying Lemma 66, we have  $|\bar{u}_j|_{0,1} < \Psi(|c|_{0,1} + |\bar{f}|_{0,1})$ . Note that,  $|\bar{f}|_{0,1}$  is controlled by  $|u|_{0,1} + |\partial_j c|_{0,1} + |\partial_j f|_{0,1}$ , which implies that  $|\bar{f}|_{0,1} \le \Psi(|c|_{0,2} + |f|_{0,2})$  due to Lemma 66. Hence, we conclude that  $|u[c, f]|_{0,2} \le \Psi(|c|_{0,2} + |f|_{0,2})$ .

At last, applying Lemma 66 on  $u[c, \bar{f}]$  again, we have

$$|u[c_1, \bar{f}_1] - u[c_2, \bar{f}_2]|_0 \le \Psi(K)(|c_1 - c_2|_0 + |\bar{f}_1 - \bar{f}_2|_0)$$

for  $K = |c_1|_0 + |\bar{f}_1|_0 + |c_2|_0 + |\bar{f}_2|_0$ , which similarly concludes the desired result.

#### 5.3.1.4 Summary on regularity and sensitivity

Now we may summarize and generalize the results above to a PDE with non-zero initial conditions. Consider equation

$$\begin{cases} \partial_t u = \frac{1}{2} \Delta u - cu + f, & \text{on } (0,T) \times \mathbb{T}^d \\ u(0,x) = \psi(x), & \text{on } x \in \mathbb{T}^d. \end{cases}$$
(5.3.10)

To proceed, we recall the following notations:

- $C_{0,n'}^{\delta,n}$  be the space of all functions  $f \in C^{\delta,n}([0,T] \times \mathbb{T}^d)$  with the topology induced by the norm  $|\cdot|_{0,n'}$ .
- $C_{0,1}^{1,3}([0,T] \times \mathbb{T}^d)$  is the space of all  $u \in C^{1,3}([0,T] \times \mathbb{T}^d)$  topologized by  $|\cdot|_{0,1}$ .

For more details, we refer it to Section 5.7.

**Theorem 69.** The solution map  $u : [c, f, \psi] \mapsto u[c, f, \psi]$  given by (5.3.10) is a locally Lipschitz continuous map

$$C_{0,1}^{\delta,2} \times C_{0,1}^{\delta,2} \times C_3^4 \mapsto C_{0,1}^{1,3}$$

*Proof.* It is enough to show that

$$|u[c_1, f_1, \psi_1] - u[c_2, f_2, \psi_2]|_{0,1} \le \Psi(K)(|c_1 - c_2|_{0,1} + |f_1 - f_2|_{0,1} + |\psi_1 - \psi_2|_3)$$

for  $K = |c_1|_{0,1} + |c_2|_{0,1} + |f_1|_{0,1} + |f_2|_{0,1} + |\psi_1|_3 + |\psi_2|_3$ . Indeed, setting  $\tilde{u}(t, x) = u(t, x) - \psi(x)$ , we have

$$\tilde{u} = u \left[ c, f + \frac{1}{2} \Delta \psi - c \psi, 0 \right]$$

for the solution map  $u[\cdot, \cdot, \cdot]$  defined via (5.3.10), and observe that the desired result is a consequence of Lemma 68.

Note that the local Lipschitz continuity of Theorem 69 automatically yields its local boundedness, i.e.,

$$|u[c, f, \psi]|_{0,1} \le \Psi(|c|_{0,1} + |f|_{0,1} + |\psi|_3)$$
(5.3.11)

for some positive increasing function  $\Psi$ . The following Harnack type inequality will be useful.

**Corollary 70.** If  $f \equiv 0$ ,  $\psi = e^b$  for some  $c, b \in C^{\delta,2}([0,T] \times \mathbb{T}^d)$ , then the solution u of (5.3.10) satisfies the inequality

$$e^{-(|b|_0+|c|_0T)} < u(t,x) < e^{|b|_0+|c|_0T}, \quad \forall (t,x) \in [0,T] \times \mathbb{T}^d$$

*Proof.* The inequalities follow from the representation for v(t, x) = u(T - t, x) in the form of

$$v(t,x) = \mathbb{E}\left[\exp\left\{-\int_{t}^{T} c\left(r, X^{t,x}(r)\right) dr\right\} \psi\left(X^{t,x}(T)\right)\right],$$

where X is given by (5.3.3).

## 5.3.2 The FPK equation

We study the weak solution of FPK equation on  $[0, T) \times \mathbb{T}^d$ :

$$\begin{cases} \partial_t \nu(t, x) = -\operatorname{div}_x(b(t, x)\nu(t, x)) + \frac{1}{2}\Delta\nu(t, x) \\ \nu(0, x) = m_0(x). \end{cases}$$
(5.3.12)

We adopt the conventional notation of

$$\langle m, \psi \rangle := \int_{\mathbb{T}^d} \psi(x) m(dx)$$

for any  $m \in \mathcal{P}_1(\mathbb{T}^d)$  and  $\psi : \mathbb{T}^d \mapsto \mathbb{R}$  whenever it is well defined.

**Definition 71.**  $\nu$  is said to be a weak solution of FPK (5.3.12), if it satisfies, for any  $\phi \in C_c^{\infty}([0,T] \times \mathbb{T}^d)$ 

$$\langle m_0, \phi(0, x) \rangle + \int_0^T \langle \nu_t, (\partial_t + \mathcal{L}) \phi \rangle dt = 0,$$

where

$$\mathcal{L} = b \cdot \nabla + \frac{1}{2}\Delta.$$

We denote the solution map of (5.3.12) by  $\nu = \nu[b, m_0]$ . We recall that  $C([0, T], \mathcal{P}_1(\mathbb{T}^d))$  is the space of all continuous mappings  $\nu : [0, T] \mapsto \mathcal{P}_1(\mathbb{T}^d)$  with a metric given by

$$dist(\nu_1,\nu_2) = \sup_t \mathbb{W}_1(\nu_1(t),\nu_2(t)),$$

where  $\mathbb{W}_1$  is 1-Wasserstein metric for  $\mathcal{P}_1$ .

**Theorem 72.** Let  $m_0 \in \mathcal{P}_1(\mathbb{T}^d)$ . Then the solution map  $b \mapsto \nu[b, m_0]$  of (5.3.12) is a locally Lipschitz continuous mapping from  $C([0, T] \times \mathbb{T}^d)$  to  $C([0, T], \mathcal{P}_1(\mathbb{T}^d))$ . In particular, if  $|b_1|_0 + |b_2|_0 < K$  then

$$\sup_{t} \mathbb{W}_1(\nu_1(t), \nu_2(t)) \le \Psi(K) |b_1 - b_2|_0.$$

Moreover,  $\nu = \nu[b, m_0]$  satisfies,

$$\mathbb{W}_1(\nu(t),\nu(s)) \le (1+\sqrt{T}|b|_0)|t-s|^{1/2},$$
(5.3.13)

$$\sup_{t} \int_{\mathbb{T}^d} |x| \nu(t, dx) \le \int_{\mathbb{T}^d} |x| m_0(dx) + |b|_0 T + \sqrt{T}.$$
(5.3.14)

г	-	-	
L			
L			
L			

*Proof.* If  $|b|_0 < \infty$  and  $m_0 \in \mathcal{P}_1$ , then

$$X(t) = X(0) + \int_0^t b(s, X_s) ds + W(t), \ X(0) \sim m_0$$

has a unique solution. An application of Itô's formula and the definition of the weak solution verifies that  $\nu(t) = Law(X(t))$  is the weak solution of (5.3.12), see [13]. (5.3.13) also follows from [13].

Next, (5.3.14) follows from

$$\sup_{t} \mathbb{E}\left[|X(t)|\right] \le \mathbb{E}\left[|X(0)|\right] + |b|_0 T + \sqrt{T}.$$

Let's assume  $|b_1|_0 + |b_2|_0 < K$  and  $\nu_1$  and  $\nu_2$  are corresponding solutions of (5.3.12). We denote by  $X_1$  and  $X_2$  the solutions of the SDE above. Note that

$$\mathbb{E}[|X_1(t) - X_2(t)|] \le \mathbb{E}\left[\int_0^t |b_1(s, X_1(s)) - b_2(s, X_2(s))|ds\right]$$
  
$$\le |b_1 - b_2|_0 T + K \int_0^t \mathbb{E}[|X_1(s) - X_2(s)|]ds.$$

So, we can use the Gronwall's inequality to have

$$\mathbb{E}[|X_1(t) - X_2(t)|] \le |b_1 - b_2|_0 T e^{KT}.$$

Therefore, we can have local Lipschitz of  $b \mapsto \nu[b, m_0]$  from

$$\mathbb{W}_1(\nu_1(t),\nu_2(t)) \le \mathbb{E}[|X_1(t) - X_2(t)|] \le |b_1 - b_2|_0 T e^{KT}.$$

к			

## 5.4 Existence

We now return to the GMFG scheme. First observe that, by using the cost of the form (5.2.3), the triple  $(v, \mathbf{a}^*, \mu)$  is the solution of (5.2.5) if and only if the pair  $(\tilde{v} := v - b, \mu)$  is the solution of HJB equation

$$\begin{cases} \partial_t \tilde{v} - \frac{1}{2} |\nabla \tilde{v}|^2 + \frac{1}{2} \Delta \tilde{v} + \tilde{\ell}_1(\mu, g) = 0\\ \tilde{v}(T, \alpha, x) = -b(T, \alpha, x) \end{cases}$$
(5.4.1)

coupled with FPK equation

$$\begin{cases} \partial_t \mu = \operatorname{div}_x(\mu \nabla \tilde{v}) + \frac{1}{2} \Delta \mu \\ \mu(0, \alpha, x) = m_0(\alpha, x), \end{cases}$$
(5.4.2)

where  $\tilde{\ell}_1$  is

$$\tilde{\ell}_1(t,\alpha,x) = \ell_1(t,\alpha,x) + \left(\partial_t b + \frac{1}{2}|\nabla b|^2 + \frac{1}{2}\Delta b\right)(t,\alpha,x).$$
(5.4.3)

Next, we outline our approach to the existence as follows. We define an operator

$$\nu = \Phi(\mu) = \Phi_2 \circ \Phi_1(\mu),$$

where

1.  $\nabla \tilde{v} = \Phi_1(\mu)$ , where  $\tilde{v}$  is the solution of (5.4.4) with a given  $\mu$ :

$$\begin{cases} \partial_t \tilde{v} - \frac{1}{2} |\nabla \tilde{v}|^2 + \frac{1}{2} \Delta \tilde{v} + \tilde{\ell}_1(\mu, g) = 0\\ \tilde{v}(T, \alpha, x) = -b(T, \alpha, x) \end{cases}$$
(5.4.4)

2.  $\nu = \Phi_2(\bar{v})$  be the function solving (5.4.5) with a given  $\bar{v}$ :

$$\begin{cases} \partial_t \nu = \operatorname{div}_x(\bar{\nu}\nu) + \frac{1}{2}\Delta\nu \\ \nu(0,\alpha,x) = m_0(\alpha,x). \end{cases}$$
(5.4.5)

The existence of the solution for the GMFG can be accomplished by the Schauder's fixed point theorem in an appropriate space to be mentioned below.

To proceed, we recall that  $d_1$  is the Wasserstein metric on  $\mathcal{P}_1(\mathbb{T}^d)$ . We define the space  $S^{1/2}$  as the collection of  $\mu : [0,T] \times [0,1] \mapsto \mathcal{P}_1(\mathbb{T}^d)$  such that

$$|\mu|_{1/2} = |\mu|_0 + [\mu]_{1/2} < \infty,$$

where

$$|\mu|_0 = \sup_{t,\alpha} \int_{\mathbb{T}^d} |x| \mu(t,\alpha,dx)$$

and

$$[\mu]_{1/2} = \sup_{t_1 \neq t_2, \alpha} \frac{d_1(\mu(t_1, \alpha), \mu(t_2, \alpha))}{|t_1 - t_2|^{1/2}}.$$

Note that,  $S^{1/2}$  is metrizable by

$$\rho(\mu_1, \mu_2) = \sup_{t, \alpha} \mathbb{W}_1(\mu_1(t, \alpha), \mu_2(t, \alpha)),$$
(5.4.6)

and we denote the space  $S^{1/2}$  by  $(S^{1/2}, \rho)$  whenever we need to emphasize its underlying metric. Note that  $B_r := \{\mu \in S^{1/2} : |\mu|_{1/2} \leq r\}$  is a closed convex compact subset of  $(S^{1/2}, \rho)$  by generalized version of Arzelà–Ascoli theorem, see P232 of [52]. It is often useful by the duality representation of Wasserstein metric to write

$$\rho(\mu_1, \mu_2) = \sup_{t, \alpha, Lip(f) \le 1} \int_{\mathbb{T}^d} f(x) d(\mu_1(t, \alpha) - \mu_2(t, \alpha))(x)$$
(5.4.7)

where Lip(f) is the Lipschitz constant of the function f. Similarly, if  $\mu \in B_r$  and  $f \in C^1$ , then

$$\int_{\mathbb{T}^d} f(y) d(\mu(t_1, \alpha) - \mu(t_2, \alpha))(y) \le |\nabla f|_0 \mathbb{W}_1(\mu(t_1, \alpha), \mu(t_2, \alpha)) \le r |\nabla f|_0 |t_1 - t_2|^{1/2}.$$
(5.4.8)

### 5.4.1 Assumptions

To proceed, we define a space  $C_{0,0,m'}^{\delta,0,m}$  as the collection of all functions in  $C^{\delta,0,m}([0,T] \times [0,1] \times \mathbb{T}^d, \mathbb{R})$  equipped with a  $C^{0,0,m'}([0,T] \times [0,1] \times \mathbb{T}^d, \mathbb{R})$  norm. For instance, if  $f \in C_{0,0,2}^{0.5,0,2}$ , then we write its norm as

$$|f|_{0,0,2}^{0.5,0,2} = |f|_{0,0,2} = |f|_0 + \sum_i |\partial_{x_i} f|_0 + \sum_{ij} |\partial_{x_i x_j} f|_0.$$

For more details, we refer to Section 5.7.

**Assumption 73.**  $b : [0,T] \times [0,1] \times \mathbb{T}^d \mapsto \mathbb{R}^d$ ,  $g : [0,1]^2 \mapsto \mathbb{R}$ , and  $m_0 : [0,1] \times \mathbb{T}^d \mapsto \mathbb{R}^d$  are infinitely smooth functions in all variables.

We pose the following assumptions for the cost function  $\ell_1$ . Throughout the chapter, since g will be a priori given function, we will suppress g by writing

$$\ell_1(\mu, g, t, \alpha, x) = \ell_1(\mu, t, \alpha, x)$$

if this does not cause any confusion. For convenience, we will write

$$\ell_1[\mu](t, \alpha, x) = \ell_1(\mu, t, \alpha, x) = \ell_1(\mu, g, t, \alpha, x).$$

Assumption 74. The mapping  $\mu \mapsto \ell_1[\mu]$  is a bounded and Lipschitz continuous mapping from  $S^{1/2}$  to  $C_{0,0,1}^{0.5,0,2}$ , that is, for any  $\mu \in S^{1/2}$ ,  $\ell_1[\mu]$  belongs to  $C^{0.5,0,2}$  and

$$|\ell_1[\mu]|_{0,0,1} < M, \quad |\ell_1[\mu_1] - \ell_1[\mu_2]|_{0,0,1} \le M\rho(\mu_1,\mu_2),$$

for some M > 0 independent to the choice of  $\mu$ .

We check that the assumptions are valid for a class of examples given in Lemma 75.

**Lemma 75.** Suppose  $\ell_2 \in C^{\infty}(\mathbb{T}^d \times \mathbb{T}^d, \mathbb{R})$  and g are given smooth enough. Then, the cost  $\ell_1$  of (5.2.4) satisfies Assumption 74.

*Proof.* Let d = 1 for the simplicity. For  $\mu \in S^{1/2}$ , we have

 $|\ell_1[\mu]|_0 \le |\ell_2|_0|g|_0,$ 

$$\begin{aligned} |\partial_x \ell_1[\mu]|_0 &\leq |\partial_x \ell_2|_0 |g|_0, \\ |\partial_{xx} \ell_1[\mu]|_0 &\leq |\partial_{xx} \ell_2|_0 |g|_0, \end{aligned}$$

and

$$\begin{aligned} |\ell_1(\mu, t_1, \alpha, x) - \ell_1(\mu, t_2, \alpha, x)| &\leq \int_0^1 \int_{\mathbb{T}^d} |\ell_2(x, y)(\mu(t_1, \alpha', dy) - \mu(t_2, \alpha', dy))g(\alpha, \alpha')|d\alpha' \\ &\leq \int_0^1 |\partial_y \ell_2|_0 d_1(\mu(t_1, \alpha'), \mu(t_2, \alpha'))g(\alpha, \alpha')d\alpha' \\ &\leq |\partial_y \ell_2|_0 |g|_0 |\mu|_{1/2} |t_1 - t_2|^{1/2}. \end{aligned}$$

This implies that  $\ell_1[\mu] \in C^{1/2,0,2}$  with estimation

$$|\ell_1[\mu]|_{1/2,0,2} \le |\ell_2|_{2,0}|g|_0(1+|\mu|_{1/2}).$$
(5.4.9)

Note that (5.4.9) does not give a uniform upper bound due to the  $\mu$ -dependence on the right hand side of the inequality. Nevertheless, we have a uniform upper bound for the weaker norm  $|\cdot|_{0,0,1}$ :

$$|\ell_1[\mu]|_{0,0,1} \le |\ell_2|_{1,0}|g|_0, \ \forall \mu \in S^{1/2}.$$

For  $\mu_1, \mu_2 \in S^{1/2}$ , we have

$$\ell_1(\mu_1, t, \alpha, x) - \ell_1(\mu_2, t, \alpha, x) = \int_0^1 \int_{\mathbb{T}^d} \ell_2(x, y)(\mu_1(t, \alpha', dy) - \mu_2(t, \alpha', dy))g(\alpha, \alpha')d\alpha'$$
  
$$\leq |\partial_y \ell_2|_0 d_1(\mu_1(t, \alpha), \mu_2(t, \alpha))|g|_0.$$

This implies that

$$|\ell_1[\mu_1] - \ell_1[\mu_2]|_0 \le |\partial_y \ell_2|_0 |g|_0 \rho(\mu_1, \mu_2)$$

Similarly, we obtain

$$|\partial_x \ell_1[\mu_1] - \partial_x \ell_1[\mu_2]|_0 \le |\partial_y \partial_x \ell_2|_0 |g|_0 \rho(\mu_1, \mu_2).$$

Therefore, we have Lipschitz continuity

$$|\ell_1[\mu_1] - \ell_1[\mu_2]|_{0,0,1} \le |\ell_2|_{1,1}|g|_0\rho(\mu_1,\mu_2),$$

and this implies Assumption 74 with  $M = |\ell_2|_{1,1}|g|_0$ .

## 5.4.2 Operator $\Phi_1$

Recall that  $\nabla \tilde{v} = \Phi_1(\mu)$ , where  $\tilde{v}$  is the solution of (5.4.4) with given  $\mu$ . By Hopf-Cole transform  $\tilde{v}$  is the solution of (5.4.4) if and only if

$$w = \exp\{-\tilde{v}\}\tag{5.4.10}$$

is the solution of

$$\begin{cases} \partial_t w + \frac{1}{2} \Delta w - w \tilde{\ell}_1[\mu] = 0 & \text{ on } (0, T) \times [0, 1] \times \mathbb{T}^d \\ w(T, \alpha, x) = e^{b(T, \alpha, x)} & \text{ on } [0, 1] \times \mathbb{T}^d. \end{cases}$$
(5.4.11)

In addition, we have the following relation by chain rule:

$$\nabla \tilde{v} = -\frac{\nabla w}{w}, \ \Delta \tilde{v} = \frac{-w\Delta w + |\nabla w|^2}{w^2}.$$

Since w-term appears in the denominator, Harnack type inequality in Corollary 70 ensures that  $\nabla \tilde{v}$  and  $\Delta \tilde{v}$  are well defined.

#### 5.4.2.1 Estimates of parameterized PDEs

We define

$$w = G(f) \tag{5.4.12}$$

by the solution of

$$\begin{cases} \partial_t w + \frac{1}{2} \Delta w - wf = 0 \quad \text{on } (0,T) \times [0,1] \times \mathbb{T}^d \\ w(T,\alpha,x) = e^{b(T,\alpha,x)} \quad \text{on } [0,1] \times \mathbb{T}^d. \end{cases}$$

$$(5.4.13)$$

Note that  $w = G(\tilde{\ell}_1[\mu])$  is the solution of (5.4.11).

**Lemma 76.** The mapping G is a locally Lipschitz continuous mapping from  $C_{0,0,1}^{0.5,0,2}$  to  $C_{0,0,1}^{1,0,2}$ .

Proof. Let  $f \in C^{0.5,0,2}$  and w = G(f). By Theorem 69, we have  $w(\alpha) \in C^{1,3}$ . If  $\alpha \to \alpha_0$ , then  $f(\alpha) \to f(\alpha_0)$  holds pointwisely. Together with Dominated Convergence Theorem on the probabilistic representation of w, one can conclude  $w(t, \alpha, x) \to w(t, \alpha_0, x)$  whenever  $\alpha \to \alpha_0$ . Therefore, w belongs to  $C^{1,0,3}$ .

Given  $f_1, f_2 \in C^{0.5,0,2}$  and  $w_i = G(f_i)$  with

$$K(\alpha) = |f_1(\alpha)|_{0,1} + |f_2(\alpha)|_{0,1} + |e^{b(T,\alpha)}|_3,$$

we can use local Lipschitz continuity of Theorem 69 to obtain local Lipschitz of G,

$$|w_{1} - w_{2}|_{0,0,1} = \sup_{\alpha} |w_{1}(\alpha) - w_{2}(\alpha)|_{0,1}$$
  
$$\leq \sup_{\alpha} \Psi(K(\alpha))|f_{1}(\alpha) - f_{2}(\alpha)|_{0,1}$$
  
$$\leq \Psi(\sup_{\alpha} K(\alpha))|f_{1} - f_{2}|_{0,0,1}.$$

In the above, we used the monotonicity of  $\Psi(\cdot)$  to switch  $\Psi$  and sup. Since  $\sup_{\alpha} K(\alpha) \leq \Psi(|f_1|_{0,0,1} + |f_2|_{0,0,1} + |b|_3)$ , we can rewrite the above estimations as

$$|w_1 - w_2|_{0,0,1} \le \Psi(|f_1|_{0,0,1} + |f_2|_{0,0,1} + |b|_3)|f_1 - f_2|_{0,0,1}.$$
#### **5.4.2.2** $\Phi_1$ estimate

**Lemma 77.**  $\Phi_1$  is a uniformly bounded and Lipschitz continuous mapping from  $(S^{1/2}, \rho)$  to  $C^0([0, T] \times [0, 1] \times \mathbb{T}^d, \mathbb{R}^d)$ .

*Proof.* If  $\mu \in S^{1/2}$ , then  $\ell_1[\mu] \in C^{0.5,0,2}$  with  $|\ell_1[\mu]|_{0,0,1} < M$  by Assumption 74. We recall that

$$\tilde{\ell}_1 = \ell_1 + \left(\partial_t b + \frac{1}{2}|\nabla b|^2 + \frac{1}{2}\Delta b\right).$$

Due to the smoothness of b and compactness of its domain, we still have  $\tilde{\ell}_1[\mu] \in C^{0.5,0,2}$  with  $|\tilde{\ell}_1[\mu]|_{0,0,1} < \Psi(M)$ . Together with local Lipschitz continuity of  $G(\cdot)$  in Lemma 76, it implies uniform boundedness of  $w = G(\tilde{\ell}_1[\mu])$ , i.e.

$$|w|_{0,0,1} < \Psi(M).$$

Moreover, Corollary 70 says that the reciprocal of  $w = G(\tilde{\ell}_1[\mu])$  is bounded in the sense  $|w^{-1}|_0 < \Psi(|\tilde{\ell}_1[\mu]|_0)$ . Therefore, we have

$$|w|_{0,0,1} + |w^{-1}|_0 < \Psi(M)$$

Next, we can prove that  $\Phi_1$  is uniformly bounded in  $C^0$ :

$$|\Phi_1(\mu)|_0 = |\nabla \tilde{v}|_0 = |w^{-1}\nabla w|_0 \le |w^{-1}|_0|\nabla w|_0 \le |w^{-1}|_0|w|_{0,0,1} \le \Psi(M).$$

Finally, we can show the global Lipschitz for  $\Phi_1$  by the following estimates:

$$\begin{split} |\Phi_{1}(\mu_{1}) - \Phi_{1}(\mu_{2})|_{0} &= \left| w_{1}^{-1} \nabla w_{1} - w_{2}^{-1} \nabla w_{2} \right|_{0} \\ &= \left| \frac{w_{2} \nabla w_{1} - w_{1} \nabla w_{2}}{w_{1} w_{2}} \right|_{0} \\ &\leq \Psi(M)(|w_{2}|_{0}|\nabla w_{1} - \nabla w_{2}|_{0} + |\nabla w_{2}|_{0}|w_{1} - w_{2}|_{0}) \\ &\leq \Psi(M)|w_{1} - w_{2}|_{0,0,1} \\ &\leq \Psi(M) \left| \tilde{\ell}_{1}[\mu_{1}] - \tilde{\ell}_{1}[\mu_{2}] \right|_{0,0,1} \\ &\leq \Psi(M)\rho(\mu_{1},\mu_{2}). \end{split}$$

In the last two steps, we used Lipschitz continuity obtained by Lemma 76 and Assumption 74.  $\Box$ 

#### 5.4.3 Operator $\Phi_2$

Next, we will show the properties associated to  $\Phi_2$  mapping from  $C^0([0,T] \times [0,1] \times \mathbb{T}^d, \mathbb{R}^d)$  to  $S^{1/2}$ .

**Lemma 78.**  $\Phi_2$  is a locally Lipschitz continuous mapping from  $C^0([0,T] \times [0,1] \times \mathbb{T}^d, \mathbb{R}^d)$  to  $(S^{1/2}, \rho)$ . Moreover,  $|\Phi_2(\bar{v})|_{1/2} \leq \Psi(|\bar{v}|_0)$  for all  $\bar{v} \in C^0([0,T] \times [0,1] \times \mathbb{T}^d, \mathbb{R}^d)$  for some monotonically increasing positive function  $\Psi$ . *Proof.* Given  $\bar{v} \in C^0([0,T] \times [0,1] \times \mathbb{T}^d, \mathbb{R}^d)$  and  $\nu = \Phi_2(\bar{v})$ , applying (5.3.14) of Theorem 72, it yields that

$$\begin{aligned} |\nu|_{0} &= \sup_{t,\alpha} \int_{\mathbb{T}^{d}} |x|\nu(t,\alpha,dx) = \sup_{\alpha} \sup_{t} \int_{\mathbb{T}^{d}} |x|\nu(t,\alpha,dx) \\ &\leq \sup_{\alpha} \left( \int_{\mathbb{T}^{d}} |x|m_{0}(\alpha,dx) + |\bar{v}(\alpha)|_{0}T + \sqrt{T} \right) \\ &\leq \Psi(|\bar{v}|_{0}). \end{aligned}$$

Next, we show the following equicontinuity property again by (5.3.13) of Theorem 72:

$$\sup_{t_1 \neq t_2, \alpha} \mathbb{W}_1(\nu(t_1, \alpha), \nu(t_2, \alpha)) \le \sup_{\alpha} \left( 1 + \sqrt{T} |\bar{v}(\alpha)|_0 \right) |t_1 - t_2|^{1/2} \le \Psi(|\bar{v}|_0) |t_1 - t_2|^{1/2}.$$

This proves  $\nu \in S^{1/2}$  with

 $|\nu|_{1/2} \le \Psi(|\bar{v}|_0).$ 

For the continuity of  $\Phi_2$ , given  $\bar{v}_1, \bar{v}_2 \in C^0([0,T] \times [0,1] \times \mathbb{T}^d, \mathbb{R}^d)$ , we set  $\nu_i = \Phi_2(\bar{v}_i)$  for i = 1, 2. Then, we use the local Lipschitz continuity in Theorem 72 to obtain local Lipschitz continuity of  $\Phi_2$  as follows:

$$\rho(\nu_{1},\nu_{2}) = \sup_{t,\alpha} \mathbb{W}_{1}(\nu_{1}(t,\alpha),\nu_{2}(t,\alpha))$$
  
= 
$$\sup_{\alpha} \sup_{t} \mathbb{W}_{1}(\nu_{1}(t,\alpha),\nu_{2}(t,\alpha))$$
  
= 
$$\sup_{\alpha} \Psi(|\bar{v}_{1}(\alpha)|_{0} + |\bar{v}_{2}(\alpha)|_{0})|\bar{v}_{1}(\alpha) - \bar{v}_{2}(\alpha)|_{0}$$
  
$$\leq \Psi(|\bar{v}_{1}|_{0} + |\bar{v}_{2}|_{0})|\bar{v}_{1} - \bar{v}_{2}|_{0}.$$

#### 5.4.4 Existence by the Schauder's fixed point theorem

**Theorem 79.** Suppose Assumptions 73 - 74 are valid. Then there exists a solution of (5.2.5) in the space  $C^{1,0,2}([0,T] \times [0,1] \times \mathbb{T}^d, \mathbb{R}) \times C([0,T] \times [0,1], \mathcal{P}_1(\mathbb{T}^d)).$ 

*Proof.* It is enough to show that  $\Phi_2 \circ \Phi_1$  has a fixed point in  $S^{1/2}$ . Recall that  $B_r$  is a convex closed and compact subset of  $S^{1/2}$ . For simplicity, we denote by  $\hat{B}_r$  the closed ball of radius r in  $C^0([0,T] \times [0,1] \times \mathbb{T}^d, \mathbb{R}^d)$ .

1. By Lemma 77, there exists some positive increasing function  $\Psi_1$  independent to r, such that the mapping

$$\Phi_1: B_r \mapsto B_{\Psi_1(M)}$$

is continuous.

2. By Lemma 78, there exists some positive increasing function  $\Psi_2$  such that the mapping

$$\Phi_2: B_{\Psi_1(M)} \mapsto B_{\Psi_2 \circ \Psi_1(M)}$$

is continuous.

Now we take

$$r = \Psi_2(\Psi_1(M))$$

and we have

$$\Phi_2 \circ \Phi_1 : B_r \mapsto B_r$$

is a continuous mapping and this yields the existence of a fixed point for  $\Phi$  by the Schauder's fixed point theorem.

In the above, we have indeed proved the existence in the space  $C^{1,0,2}([0,T] \times [0,1] \times \mathbb{T}^d, \mathbb{R}) \times S^{1/2}$ .

#### 5.4.5 Further remarks on the fixed point theorem

In connection with GMFG, we explain why Theorem 69 establishes locally Lipschitz continuity of the solution map  $u: [c, f, \psi] \mapsto u[c, f, \psi]$  of (5.3.10) in the sense of

$$C_{0,1}^{\delta,2} \times C_{0,1}^{\delta,2} \times C_3^4 \mapsto C_{0,1}^{1,3}$$
(5.4.14)

instead of

$$C^{\delta,2} \times C^{\delta,2} \times C^4 \mapsto C^{1,3}.$$
(5.4.15)

For the illustration purpose, if we freeze  $c, \psi$  of the solution map u, then local Lipschitz continuity in the sense of (5.4.14) implies local boundedness

$$|u|_{0,1} \le \Psi(|f|_{0,1}),$$

while local Lipschitz continuity in the sense of (5.4.15) implies local boundedness

$$|u|_{0,1} \le |u|_{1,3} \le \Psi(|f|_{\delta,2}).$$

The main difference of these two local boundedness properties is that, the first one controls u by f with 0-norm in t-variable while the second one does by f with  $\delta$ -norm in t-variable, which is not desirable. The main reason is that the running cost  $|\ell_1[\mu]|_{1/2,0,1} \leq \Psi(|\mu|_{1/2})$  of (5.4.9) does not have uniform bound in  $\mu$ , while  $|\ell_1[\mu]|_{0,0,1}$  does. For this reason, we included the regularity results for parabolic PDE solutions by dropping t-regularity while increasing x-regularity as a tradeoff.

Recall that, we have established the existence of a fixed point of a mapping  $\Phi = \Phi_2 \circ \Phi_1$  for  $\Phi_1 : \mu \mapsto \nabla \tilde{v}$  and  $\Phi_2 : \nabla \tilde{v} \mapsto \nu$ . Our approach is along the the Schuader's fixed point theorem with estimates

$$\Phi_1: B_r \mapsto \ddot{B}_{\Psi_1(M)}, \quad \Phi_2: \ddot{B}_{\Psi_1(M)} \mapsto B_{\Psi_2 \circ \Psi_1(M)}.$$

In the above, it is crucial that the  $\Phi_1$  is upper bounded by  $\Psi_1(M)$  independent to r, and this can be inferred from local boundedness of (5.4.14) together with uniform boundedness of  $|\ell_1[\mu]|_{0,0,1}$ .

In contrast, if we use local boundedness in the sense of (5.4.15), then we have estimations in the form of

$$\Phi_1: B_r \mapsto B_{\Psi_1(r)}, \quad \Phi_2: B_{\Psi_1(r)} \mapsto B_{\Psi_2 \circ \Psi_1(r)}.$$

Since the norm of the running cost  $|\ell_1[\mu]|_{1,0,3}$  depends on  $\mu$ ,  $\Phi_1$  can not be uniformly bounded. As a result, the choice of  $r = \Psi_1(r)$  is infeasible.

## 5.5 Uniqueness of GMFG

**Assumption 80.** There exists some  $\alpha \in [0, 1]$  satisfying

$$\int_{\mathbb{T}^d} (\ell_1(\mu_1, g, t, \alpha, x) - \ell_1(\mu_2, g, t, \alpha, x))(\mu_1 - \mu_2)(t, \alpha, dx) > 0,$$

for all  $\mu_1 \neq \mu_2 \in C([0,T] \times [0,1], \mathcal{P}_1(\mathbb{T}^d))$  and  $t \in [0,T]$ .

**Theorem 81.** ([13], [72]) Suppose Assumptions 73 - 74 and 80 are valid. Then, there exists a unique solution of (5.2.5) in the space  $C^{1,0,2}([0,T] \times [0,1] \times \mathbb{T}^d, \mathbb{R}) \times C([0,T] \times [0,1], \mathcal{P}_1(\mathbb{T}^d)).$ 

*Proof.* For i = 1, 2, let  $(v_i, \mu_i)$  be two different solution pairs, and set

$$\bar{v} = v_1 - v_2, \ \bar{\mu} = \mu_1 - \mu_2.$$

Note that  $\bar{v}(T, \alpha, x) = \bar{\mu}(0, \alpha, x) = 0$  for all  $(\alpha, x)$  by their given initial-terminal data. We also write  $\ell_1[\mu_i] = \ell_1[\mu_i, g]$  for short. Then  $\bar{v}$  satisfies

$$\partial_t \bar{v} + \nabla b \cdot \nabla \bar{v} + \frac{1}{2} \Delta \bar{v} - \frac{1}{2} |\nabla v_1|^2 + \frac{1}{2} |\nabla v_2|^2 + \ell_1[\mu_1] - \ell_1[\mu_2] = 0$$

and  $\bar{\mu}$  satisfies

$$-\partial_t \bar{\mu} - \operatorname{div}(\nabla b\bar{\mu}) + \frac{1}{2}\Delta\bar{\mu} + \operatorname{div}(\nabla v_1\mu_1) - \operatorname{div}(\nabla v_2\mu_2) = 0.$$

The above two equations can be rewritten as

$$\langle \partial_t \bar{v} + \nabla b \cdot \nabla \bar{v} + \frac{1}{2} \Delta \bar{v}, \bar{\mu} \rangle + \langle -\frac{1}{2} |\nabla v_1|^2 + \frac{1}{2} |\nabla v_2|^2 + \ell_1[\mu_1] - \ell_1[\mu_2], \bar{\mu} \rangle = 0$$

and

$$\langle \partial_t \bar{v} + \nabla b \cdot \nabla \bar{v} + \frac{1}{2} \Delta \bar{v}, \bar{\mu} \rangle + \langle \bar{v}, \operatorname{div}(\nabla v_1 \mu_1) - \operatorname{div}(\nabla v_2 \mu_2) \rangle = 0.$$

By subtracting above two equations, and utilizing the identities

$$\langle \operatorname{div}(\nabla v_1 \mu_1), \bar{v} \rangle = -\langle |\nabla v_1|^2, \mu_1 \rangle + \langle \nabla v_1 \cdot \nabla v_2, \mu_1 \rangle$$

and

$$\langle \operatorname{div}(\nabla v_2 \mu_2), \bar{v} \rangle = \langle |\nabla v_2|^2, \mu_2 \rangle - \langle \nabla v_1 \cdot \nabla v_2, \mu_2 \rangle,$$

we obtain

$$\langle \frac{1}{2}(\mu_1 + \mu_2), |\nabla \bar{v}|^2 \rangle + \langle \ell_1[\mu_1] - \ell_1[\mu_2], \bar{\mu} \rangle = 0.$$

The first term is non-negative and the second term is strictly positive for some  $\alpha \in [0, 1]$  by (A3), which implies a contradiction.

### 5.6 Concluding remarks

Our main result of Theorem 81 provides existence and uniqueness of the GMFG equation under some assumptions. One limitation of the current setting is that the running cost in the current setup allows to use Hopf-Cole transformation, which is essential to the subsequent analysis on regularities. To deal with the full generalization of the running cost, one must adopt different approaches and it will be in our future research direction. It is also desirable to generalize the result to the whole domain  $\mathbb{R}^d$  instead of compact domain  $\mathbb{T}^d$ . Another limitation is that, the current setting requires the continuity of the graphon. Note that some graphons are not necessarily continuous. Nevertheless, the continuity condition of the graphon can be relaxed in the following sense by similar arguments with additional complexity of notations, which is sketched below briefly.

To proceed, we define  $\hat{C}^0$  as the collection of bounded measurable functions  $f: [0,T] \times [0,1] \times \mathbb{T}^d \mapsto \mathbb{R}$ , and we denote its norm as

$$|f|_0 = \sup_{[0,T] \times [0,1] \times \mathbb{T}^d} |f(t, \alpha, x)|.$$

With  $\hat{C}^{\delta,0,2}$ , we denote the set of functions  $f \in \hat{C}^0$  with finite norm

$$|f|_{\delta,0,2} = |f|_0 + \sup_{t_1 < t_2, \alpha, x} \frac{|f(t_1, \alpha, x) - f(t_2, \alpha, x)|}{|t_1 - t_2|^{\delta}} + \sum_i |\partial_i f|_0 + \sum_{ij} |\partial_{ij} f|_0.$$

By the above definition  $\hat{C}^{\delta,0,2}$  allows the discontinuity in  $\alpha$ .

**Assumption 82.** 1. b and  $m_0$  are infinitely smooth in their domains.

2. The graphon g is bounded measurable on  $[0,1]^2$  with

$$|g|_0 = \sup_{[0,1]^2} |g(\alpha, \alpha')| < \infty.$$

We recall that  $B_r$  is defined in  $S^{1/2}$ . We use  $\hat{C}_{0,0,2}^{\delta,0,2}$  to denote the same set  $\hat{C}^{\delta,0,2}$  with the norm  $|\cdot|_{0,0,2}$ , i.e.

$$|f|_{0,0,2} = |f|_0 + \sum_i |\partial_i f|_0 + \sum_{ij} |\partial_{ij} f|_0.$$

**Assumption 83.** The mapping  $\mu \mapsto \ell_1[\mu]$  is a bounded and Lipschitz continuous mapping from  $S^{1/2}$  to  $\hat{C}^{0.5,0,2}_{0,0,1}$ , that is, for any  $\mu \in S^{1/2}$ ,  $\ell_1[\mu]$  belongs to  $\hat{C}^{0.5,0,2}$  and

$$|\ell_1[\mu]|_{0,0,1} < M, \quad |\ell_1[\mu_1] - \ell_1[\mu_2]|_{0,0,1} \le M\rho(\mu_1,\mu_2),$$

for some M > 0 independent to the choice of  $\mu$ .

We also define  $\hat{C}^{m,0,n}$  as the collection of  $f \in \hat{C}^0$  with continuous bounded *m*-th derivatives in t and *n*-th derivatives x. For instance, for  $f \in \hat{C}^{1,0,2}$ , we have finite norm

$$|f|_{1,0,2} = |f|_0 + |\partial_t f|_0 + \sum_i |\partial_i f|_0 + \sum_{ij} |\partial_{ij} f|_0$$

Now we present a result in parallel to Theorem 81. The proof is similar and so omitted.

**Corollary 84.** Suppose Assumptions 82 - 83 and 80 are valid. Then there exists a unique solution of (5.2.5) in the space  $\hat{C}^{1,0,2}([0,T] \times [0,1] \times \mathbb{T}^d, \mathbb{R}) \times \hat{C}([0,T] \times [0,1], \mathcal{P}_1(\mathbb{T}^d)).$ 

## 5.7 Appendix

In this appendix, we will summarize the notations of Hölder space used in this chapter. For this purpose, we will define the following functionals for a function u from a product normed space  $S = X \times Y$  to  $\mathbb{R}^d$  whenever it is well defined.

- $|u|_0 = \sup_S |u(x,y)|.$
- For nonnegative integers l, m, define

$$|u|_{l,m} = \sum_{i=0}^{l} \sum_{|\alpha|=i} |D_x^{\alpha} u|_0 + \sum_{i=0}^{m} \sum_{|\alpha|=i} |D_y^{\alpha} u|_0.$$

In the above,  $\alpha$  is a multi-index for differential operators. For instance,  $|\alpha| = \sum_{i=1}^{d_1} |\alpha_i|$  for a multi-index  $\alpha = (\alpha_i : i = 1, 2, ..., d_1)$ .

• For positive numbers  $l', m' \in (0, 1)$ , define

$$[u]_{l',m'} = [u]_{l',0} + [u]_{0,m'},$$

where

$$[u]_{l',0} = \sup_{x_1 \neq x_2, y} \frac{|u(x_1, y) - u(x_2, y)|}{|x_1 - x_2|^{l'}},$$

and

$$[u]_{0,m'} = \sup_{x,y_1 \neq y_2} \frac{|u(x,y_1) - u(x,y_2)|}{|y_1 - y_2|^{m'}}.$$

• For nonnegative integers l, m and positive number  $l' \in (0, 1)$ , define

$$|u|_{l+l',m} = |u|_{l,m} + \sum_{|\alpha|=l} [D_x^{\alpha}u]_{l',0}.$$

• For nonnegative integers l, m and positive numbers  $l', m' \in (0, 1)$ , define

$$|u|_{l+l',m+m'} = |u|_{l,m} + \sum_{|\alpha|=l} [D_x^{\alpha} u]_{l',m'} + \sum_{|\alpha|=m} [D_y^{\alpha} u]_{l',m'}.$$

One can check that the following spaces are Banach spaces:

- $C^{l,m}(X \times Y; \mathbb{R}^d) := \{u : |u|_{l,m} < \infty\},\$
- $C^{l+l',m}(X \times Y; \mathbb{R}^d) := \{u : |u|_{l+l',m} < \infty\},\$
- $C^{l+l',m+m'}(X \times Y; \mathbb{R}^d) := \{u : |u|_{l+l',m+m'} < \infty\}.$

In this chapter, we also involve the space  $C^{l',0,m}$  of functions with a domain  $S = X \times Y \times Z$ , whose norm is defined as

$$|u|_{l',0,m} = |u|_{0,0,m} + [D_z^m u]_{l',0,0}$$

where

$$|u|_{0,0,m} = \sum_{i=0}^{m} \sum_{|\alpha|=i} |D_z^{\alpha} u|_0, \text{ and } [u]_{l',0,0} = \sup_{x_1 \neq x_2, y, z} \frac{|u(x_1, y, z) - u(x_2, y, z)|}{|x_1 - x_2|^{l'}}.$$

In this chapter, our functions involve state domain taking values in *d*-torus  $\mathbb{T}^d = \mathbb{R}^d / \mathbb{Z}^d$ . For  $x \in \mathbb{R}^d$ , let  $\pi(x)$  be the coset of  $\mathbb{Z}^d$  that contains x, i.e.

$$\pi(x) = x + \mathbb{Z}^d.$$

A canonical metric on  $\mathbb{T}^d$  can be induced from the Euclidean metric by

$$|\pi(x) - \pi(y)|_{\mathbb{T}^d} = \inf\{|x - y - z| : z \in \mathbb{Z}^d\}.$$

For the illustration purpose, we provide a list of representative Hölder spaces used throughout the chapter:

•  $C^{\delta/2,\delta}([0,T] \times \mathbb{T}^d)$  is a space of functions u(t,x) with a norm defined by

$$|u|_{\delta/2,\delta} = |u|_0 + [u]_{\delta/2,\delta},$$

where  $[u]_{\delta/2,\delta}$  is a seminorm defined by

$$[u]_{\delta/2,\delta} = \sup_{t_1 \neq t_2, x} \frac{|u(t_1, x) - u(t_2, x)|}{|t_1 - t_2|^{\delta/2}} + \sup_{t, x_1 \neq x_2} \frac{|u(t, x_1) - u(t, x_2)|}{|x_1 - x_2|^{\delta}}$$

This definition may be slightly different from different resources. For instance, the definition given by [53] for the seminorm is

$$[u]_{\delta/2,\delta}' = \sup_{(t_1,x_1) \neq (t_2,x_2) \in [0,T] \times \mathbb{T}^d} \frac{|u(t_1,x_1) - u(t_2,x_2)|}{(|t_1 - t_2|^{1/2} + |x_1 - x_2|)^{\delta}}$$

Indeed, two norms induced by  $[u]_{\delta/2,\delta}$  and  $[u]'_{\delta/2,\delta}$  are equivalent, which can be seen from below:

$$[u]_{\delta/2,\delta} = [u]_{\delta/2,0} + [u]_{0,\delta} \le 2[u]'_{\delta/2,\delta},$$

and

$$\begin{split} [u]_{\delta/2,\delta}' &\leq \sup_{\substack{(t_1,x_1)\neq(t_2,x_2)\\ t_1\neq t_2}} \frac{|u(t_1,x_1)-u(t_2,x_1)|+|u(t_2,x_1)-u(t_2,x_2)|}{(|t_1-t_2|^{1/2}+|x_1-x_2|)^{\delta}} \\ &\leq \sup_{\substack{t_1\neq t_2}} \frac{|u(t_1,x_1)-u(t_2,x_1)|}{|t_1-t_2|^{\delta/2}} + \sup_{\substack{x_1\neq x_2}} \frac{|u(t_2,x_1)-u(t_2,x_2)|}{|x_1-x_2|^{\delta}} \\ &\leq [u]_{\delta/2,0} + [u]_{0,\delta} = [u]_{\delta/2,\delta}. \end{split}$$

•  $C^{0,1}([0,T] \times \mathbb{T}^d)$  is a space of functions u(t,x) with a norm

$$|u|_{0,1} = |u|_0 + \sum_{i=1\dots d} |\partial_{x_i} u|_0,$$

and  $C^{\delta,1}([0,T] \times \mathbb{T}^d)$  is a space of functions u(t,x) with a norm

$$|u|_{\delta,1} = |u|_{0,1} + \sum_{i=1\dots d} [\partial_{x_i} u]_{\delta,0}.$$

•  $C^{1,2}([0,T] \times \mathbb{T}^d)$  is a space of functions u(t,x) with a norm

$$|u|_{1,2} = |u|_0 + |\partial_t u|_0 + \sum_{i=1\dots d} |\partial_{x_i} u|_0 + \sum_{i,j=1\dots d} |\partial_{x_i x_j} u|_0.$$

•  $C^{1+\delta/2,2+\delta}([0,T]\times\mathbb{T}^d)$  is a space with a norm

$$|u|_{1+\delta/2,2+\delta} = |u|_{1,2} + [\partial_t u]_{\delta/2,\delta} + \sum_{i,j=1\dots d} [\partial_{x_i x_j} u]_{\delta/2,\delta}.$$

•  $C^{0,2}([0,T] \times \mathbb{T}^d)$  is a space with a norm

$$|u|_{0,2} = |u|_0 + \sum_{i=1\dots d} |\partial_{x_i} u|_0 + \sum_{i,j=1\dots d} |\partial_{x_i x_j} u|_0.$$

 $C^{\delta,2}([0,T] \times \mathbb{T}^d)$  is a space with a norm

$$|u|_{\delta,2} = |u|_{0,2} + \sum_{i,j=1...d} [\partial_{x_i x_j} u]_{\delta,0}.$$

We use  $C_{0,2}^{\delta,2}([0,T] \times \mathbb{T}^d)$  to denote the space of all functions in  $C^{\delta,2}([0,T] \times \mathbb{T}^d)$  topologized by the norm  $|\cdot|_{0,2}$ . Such a space  $C_{0,2}^{\delta,2}([0,T] \times \mathbb{T}^d)$  is not complete. However, every  $|\cdot|_{\delta,2}$ -norm bounded ball in  $C_{0,2}^{\delta,2}([0,T] \times \mathbb{T}^d)$  is precompact since  $C^{\delta,2}([0,T] \times \mathbb{T}^d)$  is compactly embedded into  $C^{0,2}([0,T] \times \mathbb{T}^d)$ . •  $C_{0,1}^{1,3}([0,T] \times \mathbb{T}^d)$  is the space of all  $u \in C^{1,3}([0,T] \times \mathbb{T}^d)$  topologized by  $|\cdot|_{0,1}$ , i.e.

$$|u|_{0,1} = |u|_0 + \sum_{i=1\dots d} |\partial_{x_i} u|_0$$

•  $C^{0,0,2}([0,T] \times [0,1] \times \mathbb{T}^d)$  is the space of all  $u(t,\alpha,x)$  having finite norm of

$$|u|_{0,0,2} = |u|_0 + \sum_{i=1\dots d} |\partial_{x_i} u|_0 + \sum_{i,j=1\dots d} |\partial_{x_i x_j} u|_0.$$

•  $C^{\delta,0,2}([0,T] \times [0,1] \times \mathbb{T}^d)$  is the space of all  $u(t,\alpha,x)$  having finite norm of

$$|u|_{\delta,0,2} = |u|_{0,0,2} + \sum_{i,j=1...d} [\partial_{x_i x_j} u]_{\delta,0,0}.$$

•  $C_{0,0,2}^{\delta,0,2}([0,T] \times [0,1] \times \mathbb{T}^d)$  is the space of all  $u(t,\alpha,x)$  having finite norm of  $|u|_{\delta,0,2}$ , but topologized by  $|\cdot|_{0,0,2}$ .

## Chapter 6

## Conclusion and future work

## 6.1 Conclusion

In this thesis, our focus centers on the convergence analysis of stochastic nonlinear systems and we explore the examination of their asymptotic behavior, a critical area for understanding the prolonged dynamics of these systems and those with substantial populations. This thesis emphasizes two key aspects: firstly, the examination of the turnpike property within the context of stochastic control problems, and secondly, the investigation of the convergence of N-player games towards their corresponding mean field games.

Firstly, our investigation delves into the asymptotic behavior of systems exhibiting long-term dynamics, focusing on convergence concerning the time horizon. In Chapter 2, we scrutinize the limiting behavior of a specific category of linear quadratic stochastic optimal control problems and their associated value functions as the time horizon extends to infinity. We provide a distinct approach to show the turnpike properties in stochastic control theory by using the cell problem within the realm of weak KAM theory in PDE and contributions can be summarized as follows.

- (1) Our first contribution lies in the formulation of a verification theorem connecting the cell problem to a specific class of infinite time horizon control problems, referred to as the probabilistic cell problem, see Lemma 5. Unlike the typical cell problem explored in the literature (e.g., [77]), the underlying cell problem in our context lacks uniqueness due to the non-compactness of the domain. The above verification theorem establishes a connection between the cell problem and the static optimization problem.
- (2) Our second contribution provides the connection between the cell problem (2.1.6) and the ergodic cost problem. This involves determining the constant

$$-c_* := \lim_{T \to \infty} \frac{1}{T} V_T(x) = \lim_{T \to \infty} \frac{1}{T} J_T(x; u_T^*).$$

This connection identifies not only the aforementioned turnpike property of (2.1.5) with respect to the control process and state process, but also unveils a new turnpike property in terms of the cost function:

$$\lim_{T \to \infty} \frac{1}{T} J_T(x; u^*) = -c_*,$$

where  $u^*$  is a control process obtained from the probabilistic cell problem independent to the length of the terminal time T, see Theorem 9.

Next, we examine the system complexity and consider the convergence behavior of stochastic nonlinear systems with large populations. More precisely, we concentrate on the fundamental question arising in MFG and delves into the examination of convergence rate from the N-player games to the associated MFG within different settings.

In Chapter 3, we investigate the convergence rate of the N-player game, governed by a continuoustime Markov chain as the common noise, towards its asymptotic MFG under LQG structure. The main results and contributions are summarized as follows.

- (1) Firstly, we introduce a Markovian structure using two auxiliary processes for the first and second moments of the MFG equilibrium and employ the fixed point condition in MFG. By doing so, we characterize the equilibrium measure in MFG with a finite-dimensional Riccati system of ODEs. Consequently, we obtain the equilibrium path, equilibrium control, and the value function in MFG.
- (2) Subsequently, we address the N-player game under the LQG structure, and we characterize its equilibrium path, equilibrium control, and the value function through a Riccati system of ODEs with a dimension of  $O(N^3)$ . Leveraging the N-invariant algebraic structure of this system of ODEs, we establish a dimension reduction result, facilitating a comparison between the equilibrium path in the N-player game and the equilibrium path in the MFG.
- (3) To demonstrate the convergence between the two equilibrium paths, we embed the equilibrium path in the N-player game to the probability space of the equilibrium path in the MFG using a distribution copy, leading to the achievement of the convergence result and the computation of the convergence rate. We obtain the convergence rate  $O(N^{-1/2})$  with respect to 2-Wasserstein distance.
- (4) Lastly, some numerical examples are presented to demonstrate the convergence result.

In Chapter 3, the number of states of the common noise is finite, thus next we consider the case when the number of states of the common noise is infinity. We investigate the convergence rate of the N-player game with Brownian motion as its common noise in the following chapter.

We focus on a class of one-dimensional LQG mean field games with Brownian motion as the common noise in Chapter 4. It is worth noting that the equilibrium path, equilibrium control, and the value function in MFG and the N-player game can be obtained by a similar methodology as Chapter 3. Our main contribution is the establishment of three different convergence rates from the N-player games to the corresponding mean field games:

• Firstly, we establish that the convergence rate of the *p*-Wasserstein metric for the distribution of the representative player in the *N*-player game to the distribution of the generic player in MFG is  $O(N^{-1/2})$  for  $p \in [1, 2]$ ;

- Secondly, it demonstrates that the convergence rate of the *p*-Wasserstein metric for the empirical measure of the equilibrium path in the *N*-player game to the equilibrium measure in MFG under the  $L^p$  sense is  $O(N^{-1/(2p)})$  for  $p \in [1, 2]$ ;
- Lastly, we show that the convergence rate of the uniform p-Wasserstein metric for the empirical measure of the equilibrium path in the N-player game to the equilibrium measure in MFG under the  $L^p$  sense is  $O(N^{-1/(2p)})$  for  $p \in (1, 2]$ , and  $O(N^{-1/2} \ln(N))$  for p = 1.

To achieve the above convergence rates, the methodology relies on a specific decomposition of the equilibrium path in the *N*-player game and the associated MFG framework. We establish the convergence results for the empirical measure of a non-i.i.d. sequence of random variables and generalize the result to triangular arrays, which provides a desired structure to complete the establishment of the convergence rates.

In Chapter 5, we investigate the large population games with asymmetric graph connections. We consider the solvability of a type of graphon mean field games. A new type of mean field games PDE system associated with the graphon mean field games system, see (5.2.5), is proposed in this project. The graphon mean field games system consists of a collection of parameterized HJB equations and a collection of parameterized Fokker-Planck-Kolmogorov equations. We establish the existence of solutions via the application of Schauder's fixed point theorem and obtain the uniqueness via the Lasry-Lions monotonicity assumption on the running cost. The main difficulty is to obtain the regularity of the solution and the sensitivity of the corresponding HJB equations and Fokker-Planck-Kolmogorov equations.

### 6.2 Future work

There are multiple short and long-term research directions that we would like to explore in the future based on the results in this thesis.

In Chapter 2, the diffusion term is a constant in our model setting and it is independent of the control term. We could try to involve the control in the diffusion term and reexamine the corresponding turnpike properties. Next, we only investigate the model under the LQG structure. The general model will be considered in the future. For example, we could formulate the problem within a general setting and give some regularity conditions to drift term, diffusion term and the running cost in the cost functional. Moreover, with the resurgent of interests of mean field games and mean field models, we could also consider the ergodic property of the mean field control problems.

For the convergence of the N-player game to the associated MFG investigated in Chapter 3 and Chapter 4, we propose the following future ideas. Firstly, we could consider the mean field game under more general settings about the dynamic processes, such as with time delays, Poisson jumps, etc. Next, except for considering the LQG structure, we could examine the convergence of mean field games with common noise under more general structures. It is important to note that the assumptions made in the aforementioned papers usually account for linear growth in the state and control elements for the running cost, or they only focus on the linear quadratic structure. The convergence of mean field games with common noise under a more general setting should obtain more attention. Furthermore, in our two previous works, we require positive values for all sensitivity parameters in the cost functional. We find that there is no global solution for MFG when the coefficient of the cost functional is negative, while there is a global solution when the coefficient is positive. So, it is also an interesting problem to investigate the explosion when some sensitivities take negative value.

One limitation of the current setting in Chapter 5 is that the running cost in the current setup allows to use Hopf-Cole transformation, which is essential to the subsequent analysis of regularities. To deal with the full generalization of the running cost, one must adopt different approaches and it will be in our future research direction. It is also desirable to generalize the result to the whole domain  $\mathbb{R}^d$  instead of the compact domain  $\mathbb{T}^d$ . In addition, the convergence of graphon mean field game is not addressed in our work. The establishment of the convergence rate from the *N*subpopulation game to the corresponding graphon mean field game will be considered in our future works.

# Bibliography

- [1] Saran Ahuja. Mean Field Games with Common Noise. Stanford University, 2015.
- [2] Panos J Antsaklis and Anthony N Michel. *Linear systems*. Springer Science & Business Media, 2006.
- [3] Ari Arapostathis, Vivek S Borkar, and Mrinal K Ghosh. Ergodic control of diffusion processes, volume 143. Cambridge University Press, 2012.
- [4] Mariko Arisawa and P-L Lions. On ergodic stochastic control. Communications in partial differential equations, 23(11-12):2187–2217, 1998.
- [5] Martino Bardi. Explicit solutions of some linear-quadratic mean field games. Networks & Heterogeneous Media, 7(2):243, 2012.
- [6] Martino Bardi and Fabio S Priuli. Lqg mean-field games with ergodic cost. In 52nd IEEE Conference on Decision and Control, pages 2493–2498. IEEE, 2013.
- [7] Guy Barles and Panagiotis E Souganidis. On the large time behavior of solutions of hamiltonjacobi equations. SIAM Journal on Mathematical Analysis, 31(4):925–939, 2000.
- [8] Tobias Breiten and Laurent Pfeiffer. On the turnpike property and the receding-horizon method for linear-quadratic optimal control problems. SIAM Journal on Control and Optimization, 58(2):1077–1102, 2020.
- [9] Peter E Caines, Daniel Ho, Minyi Huang, Jiamin Jian, and Qingshuo Song. On the graphon mean field game equations: Individual agent affine dynamics and mean field dependent performance functions. ESAIM: Control, Optimisation and Calculus of Variations, 28:24, 2022.
- [10] Peter E Caines and Minyi Huang. Graphon mean field games and the gmfg equations. In 2018 IEEE Conference on Decision and Control (CDC), pages 4129–4134. IEEE, 2018.

- [11] Peter E Caines and Minyi Huang. Graphon mean field games and the gmfg equations: ε-nash equilibria. In 2019 IEEE 58th conference on decision and control (CDC), pages 286–292. IEEE, 2019.
- [12] Peter E Caines and Minyi Huang. Graphon mean field games and their equations. SIAM Journal on Control and Optimization, 59(6):4373–4399, 2021.
- [13] Pierre Cardaliaguet. Notes on mean field games. Technical report, Technical report, 2010.
- [14] Pierre Cardaliaguet, François Delarue, Jean-Michel Lasry, and Pierre-Louis Lions. The Master Equation and the Convergence Problem in Mean Field Games:(AMS-201), volume 201.
   Princeton University Press, 2019.
- [15] Dean A Carlson, Alain B Haurie, and Arie Leizarowitz. Infinite horizon optimal control: deterministic and stochastic systems. Springer Science & Business Media, 2012.
- [16] René Carmona and François Delarue. Probabilistic theory of mean field games with applications.
   I, volume 83 of Probability Theory and Stochastic Modelling. Springer, Cham, 2018. Mean field FBSDEs, control, and games.
- [17] René Carmona and François Delarue. Probabilistic theory of mean field games with applications.
   II, volume 84 of Probability Theory and Stochastic Modelling. Springer, Cham, 2018. Mean field games with common noise and master equations.
- [18] René Carmona and François Delarue. Forward-backward stochastic differential equations and controlled mckean-vlasov dynamics. The Annals of Probability, 43(5):2647–2700, 2015.
- [19] René Carmona, François Delarue, and Aimé Lachapelle. Control of mckean–vlasov dynamics versus mean field games. *Mathematics and Financial Economics*, 7(2):131–166, 2013.
- [20] Giovanni Conforti. Coupling by reflection for controlled diffusion processes: turnpike property and large time behavior of hamilton jacobi bellman equations. *arXiv preprint arXiv:2206.04009*, 2022.
- [21] François Delarue, Daniel Lacker, and Kavita Ramanan. From the master equation to mean field game limit theory: Large deviations and concentration of measure. Annals of Probability, 48(1):211–263, 2020.

- [22] Robert Dorfman, Paul Anthony Samuelson, and Robert M Solow. Linear programming and economic analysis. Courier Corporation, 1987.
- [23] Richard Durrett. Probability. The Wadsworth & Brooks/Cole Statistics/Probability Series. Wadsworth & Brooks/Cole Advanced Books & Software, Pacific Grove, CA, 3rd edition, 2005. Theory and examples.
- [24] Carlos Esteve, Hicham Kouhkouh, Dario Pighin, and Enrique Zuazua. The turnpike property and the long-time behavior of the hamilton-jacobi-bellman equation. arXiv preprint arXiv:2006.10430, 2020.
- [25] Lawrence C. Evans. Partial differential equations, volume 19 of Graduate Studies in Mathematics. American Mathematical Society, Providence, RI, 1998.
- [26] Lawrence C Evans. A survey of partial differential equations methods in weak kam theory. Communications on pure and applied mathematics, 57(4):445–480, 2004.
- [27] Lawrence C Evans. Further pde methods for weak kam theory. Calculus of Variations and Partial Differential Equations, 35(4):435–462, 2009.
- [28] Albert Fathi. Weak KAM theorem in Lagrangian dynamics. Cambridge University Press, 2003.
- [29] Xinwei Feng, Jianhui Huang, and Zhenghong Qiu. Mixed social optima and nash equilibrium in linear-quadratic-gaussian mean-field system. *IEEE Transactions on Automatic Control*, 67(12):6858–6865, 2021.
- [30] Dena Firoozi, Sebastian Jaimungal, and Peter E Caines. Convex analysis for lqg systems with applications to major-minor lqg mean-field game systems. Systems & Control Letters, 142:104734, 2020.
- [31] Wendell H Fleming and Halil Mete Soner. Controlled Markov processes and viscosity solutions, volume 25. Springer Science & Business Media, 2006.
- [32] Nicolas Fournier and Arnaud Guillin. On the rate of convergence in wasserstein distance of the empirical measure. *Probability theory and related fields*, 162(3-4):707–738, 2015.
- [33] Wilfrid Gangbo, Alpár R Mészáros, Chenchen Mou, and Jianfeng Zhang. Mean field games master equations with nonseparable hamiltonians and displacement monotonicity. *The Annals* of Probability, 50(6):2178–2217, 2022.

- [34] Shuang Gao and Peter E Caines. Graphon linear quadratic regulation of large-scale networks of linear systems. In 2018 IEEE Conference on Decision and Control (CDC), pages 5892–5897. IEEE, 2018.
- [35] Shuang Gao, Peter E Caines, and Minyi Huang. Lqg graphon mean field games. arXiv preprint arXiv:2004.00679, 2020.
- [36] Diogo A Gomes, Edgard A Pimentel, and Vardan Voskanyan. Regularity theory for mean-field game systems. Springer, 2016.
- [37] Jianhui Huang and Minyi Huang. Mean field lqg games with model uncertainty. In 52nd IEEE Conference on Decision and Control, pages 3103–3108. IEEE, 2013.
- [38] Jianhui Huang, Xun Li, and Tianxiao Wang. Mean-field linear-quadratic-gaussian (lqg) games for stochastic integral systems. *IEEE Transactions on Automatic Control*, 61(9):2670–2675, 2015.
- [39] Minyi Huang. Large-population LQG games involving a major player: the Nash certainty equivalence principle. SIAM Journal on Control and Optimization, 48(5):3318–3353, 2010.
- [40] Minyi Huang, Peter E Caines, and Roland P Malhamé. Individual and mass behaviour in large population stochastic wireless power control problems: centralized and nash equilibrium solutions. In 42nd IEEE International Conference on Decision and Control (IEEE Cat. No. 03CH37475), volume 1, pages 98–103. IEEE, 2003.
- [41] Minyi Huang, Peter E Caines, and Roland P Malhamé. An invariance principle in large population stochastic dynamic games. *Journal of Systems Science and Complexity*, 20(2):162– 172, 2007.
- [42] Minyi Huang, Peter E Caines, and Roland P Malhamé. Large-population cost-coupled lqg problems with nonuniform agents: individual-mass behavior and decentralized ε-nash equilibria. *IEEE transactions on automatic control*, 52(9):1560–1571, 2007.
- [43] Minyi Huang, Peter E Caines, and Roland P Malhamé. The nash certainty equivalence principle and mckean-vlasov systems: an invariance principle and entry adaptation. In 2007 46th IEEE Conference on Decision and Control, pages 121–126. IEEE, 2007.

- [44] Minyi Huang, Peter E Caines, and Roland P Malhamé. Social optima in mean field lqg control: centralized and decentralized strategies. *IEEE Transactions on Automatic Control*, 57(7):1736–1751, 2012.
- [45] Minyi Huang, Roland P Malhamé, Peter E Caines, et al. Large population stochastic dynamic games: closed-loop mckean-vlasov systems and the nash certainty equivalence principle. *Communications in Information & Systems*, 6(3):221–252, 2006.
- [46] Minyi Huang and Xuwei Yang. Linear quadratic mean field games: decentralized O(1/N)-Nash equilibria. Journal of Systems Science & Complexity, 34(5):2003–2035, 2021.
- [47] Naoyuki Ichihara and Shuenn-Jyi Sheu. Large time behavior of solutions of hamilton-jacobibellman equations with quadratic nonlinearity in gradients. SIAM Journal on Mathematical Analysis, 45(1):279–306, 2013.
- [48] Joe Jackson and Ludovic Tangpi. Quantitative convergence for displacement monotone mean field games with controlled volatility. arXiv preprint arXiv:2304.04543, 2023.
- [49] Jiamin Jian, Sixian Jin, and Qingshuo Song. Long-time behavior of stochastic lq control problem. *submitted*, 2023.
- [50] Jiamin Jian, Peiyao Lai, Qingshuo Song, and Jiaxuan Ye. The convergence rate of the equilibrium measure for the hybrid lqg mean field game. Nonlinear Analysis: Hybrid Systems, 52:101454, 2024.
- [51] Jiamin Jian, Qingshuo Song, and Jiaxuan Ye. Convergence rate of lqg mean field games with common noise. arXiv preprint arXiv:2307.00695, 2023.
- [52] John L Kelley. *General topology*. Courier Dover Publications, 2017.
- [53] Nikolaĭ Vladimirovich Krylov. Lectures on elliptic and parabolic equations in Holder spaces. American Mathematical Soc., 1996.
- [54] Olga Aleksandrovna Ladyzhenskaia, Vsevolod Alekseevich Solonnikov, and Nina N Ural'tseva. Linear and quasi-linear equations of parabolic type, volume 23. American Mathematical Soc., 1988.
- [55] Jean-Michel Lasry and Pierre-Louis Lions. Mean field games. Japanese journal of mathematics, 2(1):229–260, 2007.

- [56] Pierre-Louis Lions, George Papanicolaou, and Srinivasa RS Varadhan. Homogenization of hamilton-jacobi equations. Unpublished preprint, 1987.
- [57] Hongwei Lou and Weihan Wang. Turnpike properties of optimal relaxed control problems. ESAIM: Control, Optimisation and Calculus of Variations, 25:74, 2019.
- [58] László Lovász. Large networks and graph limits, volume 60 of american mathematical society colloquium publications. American Mathematical Society, Providence, RI, 22, 2012.
- [59] Siyu Lv, Jie Xiong, and Xin Zhang. Linear quadratic leader-follower stochastic differential games for mean-field switching diffusions. *arXiv*, 2022.
- [60] Xuerong Mao and Chenggui Yuan. Stochastic differential equations with Markovian switching. Imperial college press, 2006.
- [61] John N Mather. Action minimizing invariant measures for positive definite lagrangian systems. Mathematische Zeitschrift, 207(1):169–207, 1991.
- [62] Lionel McKenzie. Turnpike theory. *Econometrica: Journal of the Econometric Society*, pages 841–865, 1976.
- [63] Gawtum Namah and Jean-Michel Roquejoffre. Remarks on the long time behaviour of the solutions of hamilton-jacobi equations. *Communications in partial differential equations*, 24(5-6):883–893, 1999.
- [64] John von Neumann. A model of general economic equilibrium. The Review of Economic Studies, 13(1):1–9, 1945.
- [65] Son L. Nguyen and Minyi Huang. Mean field lqg games with mass behavior responsive to a major player. In 2012 IEEE 51st IEEE Conference on Decision and Control (CDC), pages 5792–5797. IEEE, 2012.
- [66] Son L. Nguyen, Dung T Nguyen, and George Yin. A stochastic maximum principle for switching diffusions using conditional mean-fields with applications to control problems. ESAIM: Control, Optimisation and Calculus of Variations, 26:69, 2020.
- [67] Son L. Nguyen, George Yin, and Tuan A. Hoang. On laws of large numbers for systems with mean-field interactions and markovian switching. *Stochastic Processes and their Applications*, 130(1):262–296, 2020.

- [68] Mojtaba Nourian and Peter E. Caines. ε-Nash mean field game theory for nonlinear stochastic dynamical systems with major and minor agents. SIAM Journal on Control and Optimization, 51(4):3302–3331, 2013.
- [69] Huyên Pham. Continuous-time stochastic control and optimization with financial applications, volume 61. Springer Science & Business Media, 2009.
- [70] Alessio Porretta and Enrique Zuazua. Long time versus steady state optimal control. SIAM Journal on Control and Optimization, 51(6):4242–4273, 2013.
- [71] Alessio Porretta and Enrique Zuazua. Remarks on long time versus steady state optimal control. Mathematical paradigms of climate science, pages 67–89, 2016.
- [72] Lenya Ryzhik. Notes on mean field games. Technical report, Technical report, 2018.
- [73] Gilbert Strang. Introduction to linear algebra. Wellesley-Cambridge Press, Massachusetts, 2009.
- [74] Jingrui Sun, Hanxiao Wang, and Jiongmin Yong. Turnpike properties for stochastic linearquadratic optimal control problems. *Chinese Annals of Mathematics, Series B*, 43(6):999–1022, 2022.
- [75] Jingrui Sun and Jiongmin Yong. Turnpike properties for mean-field linear-quadratic optimal control problems. arXiv preprint arXiv:2209.11466, 2022.
- [76] Rinel Foguen Tchuendom. Uniqueness for linear-quadratic mean field games with common noise. Dynamic Games and Applications, 8(1):199–210, 2018.
- [77] Hung V Tran. Hamilton-Jacobi equations: theory and applications, volume 213. American Mathematical Soc., 2021.
- [78] Ky Tran, George Yin, and Le Yi Wang. A generalized goodwin business cycle model in random environment. Journal of Mathematical Analysis and Applications, 438(1):311–327, 2016.
- [79] Emmanuel Trélat and Enrique Zuazua. The turnpike property in finite-dimensional nonlinear optimal control. Journal of Differential Equations, 258(1):81–114, 2015.
- [80] George Yin and Chao Zhu. Hybrid switching diffusions, volume 63 of Stochastic Modelling and Applied Probability. Springer, New York, 2010. Properties and applications.

- [81] Alexander Zaslavski. Turnpike properties in the calculus of variations and optimal control, volume 80. Springer Science & Business Media, 2005.
- [82] Alexander Zaslavski. Turnpike properties of approximate solutions for discrete-time control systems. Commun Math Anal, 11:36–45, 2011.
- [83] Alexander Zaslavski. Turnpike conditions in infinite dimensional optimal control, volume 148. Springer, 2019.
- [84] Qing Zhang, George Yin, and El-Kébir Boukas. Controlled markov chains with weak and strong interactions: asymptotic optimality and applications to manufacturing. *Journal of* optimization theory and applications, 94:169–194, 1997.
- [85] Xun Yu Zhou and George Yin. Markowitz's mean-variance portfolio selection with regime switching: A continuous-time model. SIAM Journal on Control and Optimization, 42(4):1466– 1482, 2003.
- [86] Chao Zhu and George Yin. On competitive lotka-volterra model in random environments. Journal of Mathematical Analysis and Applications, 357(1):154–170, 2009.